Case 23-90088 Document 47 Filed in TXSR on 02/14/23 Page 1 of 237
Docket #0047 Date Filed: 02/14/2023

# IN THE UNITED STATES BANKRUPTCY COURT FOR THE SOUTHERN DISTRICT OF TEXAS HOUSTON DIVISION

		)
In re:		) Chapter 11
AVAYA INC., et al.,1		) Case No. 23-90088 (DRJ)
	Debtors.	) (Jointly Administered) (Emergency Hearing Requested)

DEBTORS' EMERGENCY MOTION FOR ENTRY
OF INTERIM AND FINAL ORDERS (I) AUTHORIZING
THE DEBTORS TO (A) OBTAIN POSTPETITION FINANCING
AND (B) UTILIZE CASH COLLATERAL, (II) GRANTING LIENS AND
SUPERPRIORITY ADMINISTRATIVE EXPENSE CLAIMS, (III) GRANTING
ADEQUATE PROTECTION TO THE PREPETITION SECURED PARTIES, (IV)
MODIFYING THE AUTOMATIC STAY, AND (V) GRANTING RELATED RELIEF

Emergency relief has been requested. Relief is requested not later than 9:00 a.m. (prevailing Central Time) on February 15, 2023.

If you object to the relief requested or you believe that emergency consideration is not warranted, you must appear at the hearing if one is set, or file a written response prior to the date that relief is requested in the preceding paragraph. Otherwise, the Court may treat the pleading as unopposed and grant the relief requested.

A hearing will be conducted on this matter on February 15, 2023, at 9:00 a.m. (prevailing Central Time) in Courtroom 400, 4th floor, 515 Rusk Street, Houston, Texas 77002. Participation at the hearing will only be permitted by an audio and video connection.

Audio communication will be by use of the Court's dial-in facility. You may access the facility at (832) 917-1510. Once connected, you will be asked to enter the conference room number. Judge Jones's conference room number is 205691. Video communication will be by use of the GoToMeeting platform. Connect via the free GoToMeeting application or click the link on Judge Jones's homepage. The meeting code is "Judge Jones". Click the settings icon in the upper right corner and enter your name under the personal information setting.

Hearing appearances must be made electronically in advance of both electronic and in-person hearings. To make your appearance, click the "Electronic Appearance" link on Judge Jones's homepage. Select the case name, complete the required fields and click "Submit" to complete your appearance.

A complete list of each of the Debtors in these chapter 11 cases may be obtained on the website of the Debtors' proposed claims and noticing agent at <a href="http://www.kccllc.net/avaya">http://www.kccllc.net/avaya</a>. The location of Debtor Avaya Inc.'s principal place of business and the Debtors' service address in these chapter 11 cases is 350 Mount Kemble Avenue, Morristown, New Jersey 07960.

The above-captioned debtors and debtors in possession (collectively, the "Debtors"), 2 state the following in support of this emergency motion (this "Motion").<sup>3</sup> In support of this Motion, the Debtors respectfully submit: (a) the Declaration of Eric Koza, Chief Restructuring Officer of Avaya Holdings Corp. and Certain of Its Affiliates and Subsidiaries, in Support of Debtors' Emergency Motion for Entry of Interim and Final Orders (I) Authorizing the Debtors to (A) Obtain Postpetition Financing and (B) Utilize Cash Collateral, (II) Granting Liens and Superpriority Administrative Expense Claims, (III) Granting Adequate Protection to the Prepetition Secured Modifying the Automatic Stay, and (V) Granting Related Relief Parties. (IV)(the "Koza Declaration"), filed contemporaneously herewith; (b) the Declaration of Roopesh Shah in Support of the Debtors' Emergency Motion for Entry of Interim and Final Orders (I) Authorizing the Debtors to (A) Obtain Postpetition Financing and (B) Utilize Cash Collateral, (II) Granting Liens and Superpriority Administrative Expense Claims, (III) Granting Adequate Protection to the Prepetition Secured Parties, (IV) Modifying the Automatic Stay, and (V) Granting Related Relief, filed contemporaneously herewith (the "Shah Declaration," together with the Koza Declaration, collectively, the "<u>DIP Declarations</u>"); and (c) the First Day Declaration. In further support of this Motion, the Debtors state the following:

<sup>&</sup>lt;sup>2</sup> Capitalized terms used but not otherwise defined herein shall have the meanings set forth in the First Day Declaration (as defined herein), the Plan, the DIP Term Loan Credit Agreement (as defined herein), or in the Interim Order (as defined herein), as applicable.

The Debtors, together with their non-Debtor affiliates (collectively, "<u>Avaya</u>" or the "<u>Company</u>"), are a leading provider of mission-critical, real-time communication applications. The facts and circumstances supporting this Motion are set forth in the *Declaration of Eric Koza, Chief Restructuring Officer of Avaya Holdings Corp. and Certain of Its Affiliates and Subsidiaries, in Support of Debtors' Chapter 11 Petitions and First Day Motions* (the "<u>First Day Declaration</u>"), filed contemporaneously on the date hereof and incorporated by reference herein.

#### **Preliminary Statement**

- 1. As described in the Koza Declaration, the Debtors enter these Chapter 11 Cases on the date hereof (the "Petition Date") with approximately \$45 million cash on hand. Koza Decl. ¶ 10. Absent immediate financing, the Debtors risk enterprise-wide operational disruption amidst ongoing efforts to optimize their business structure and enhance market offerings. *See* Koza Decl. ¶¶ 23-24. The Debtors require a certain minimum threshold liquidity in order to sustain operations in the ordinary course and complete an operational restructuring that will position the Company for long term success—the DIP Term Loan Facility achieves that end. *See* Koza Decl. ¶ 10.
- 2. Accordingly, prior to the filing of these Chapter 11 Cases, the Debtors and their advisors worked diligently to identify potential sources of postpetition financing. Through their proposed investment banker, Evercore Group L.L.C. ("Evercore"), the Debtors reached out to various stakeholders across the Debtors' capital structure and third-parties, including Holders of Prepetition ABL Claims, the PW Ad Hoc Group, and the Akin Ad Hoc Group (together with the PW Ad Hoc Group, the "Ad Hoc Groups"). Shah Decl. ¶ 13. This initial outreach revealed that members of the Ad Hoc Groups and Citibank, as the Prepetition ABL Agent, would be willing to provide postpetition financing to fund a comprehensive restructuring.
- 3. In connection with the filing of these Chapter 11 Cases, the Debtors have achieved extraordinary certainty regarding their exit from chapter 11, by obtaining (a) a \$500 million priming super priority senior secured debtor in possession facility (the "DIP Term Loan Facility") that converts into a exit term loan facility (the "Exit Facility") upon emergence, and (b) commitments for a debtor in possession asset-based revolving credit facility (the "DIP ABL Facility"), a U.S. only, new money, approximately \$128 million facility that will convert into a senior secured asset-based revolving credit (the "Exit ABL Facility"), upon emergence. The Debtors intend to return to the Bankruptcy Court in the very near term to seek approval of the DIP

ABL Facility, and upon entry of the Interim Order, are seeking limited relief to enter into certain commitment and fee letters to provide certainty regarding the availability and access of the DIP ABL Facility while the Debtors, the DIP ABL Agent, and the Consenting Stakeholders finalize the terms and documentation of such facility (such commitment letter, in the form substantially similar to that attached hereto as **Exhibit B**, (the "DIP-to-Exit ABL Commitment Letter") and fee letters (each, a "DIP-to-Exit ABL Fee Letter," and collectively, the "DIP-to-Exit ABL Fee Letter," and, together with the DIP ABL Commitment Letter, the "DIP-to-Exit ABL Commitment Papers").

- 4. As discussed in the Shah Declaration, concurrently with postpetition financing negotiations, the Debtors, their advisors, and the Ad Hoc Groups continued to work diligently and in good faith to negotiate the terms of the Debtors' restructuring. These negotiations successfully resulted in the RSA, which reflects massive consensus across the Debtors' capital structure. Moreover, pursuant to the Plan, the Debtors will emerge from chapter 11 with a right-sized capital structure, having reduced funded indebtedness by approximately 75%, from approximately \$3.4 billion to approximately \$800 million. The DIP Term Loan Facility, and, ultimately, the DIP ABL Facility (collectively, the "DIP Facilities") are a critical component of the overall RSA, ensuring the Debtors have sufficient capital to responsibly operate their businesses during these chapter 11 cases and emerge with the certainty of being well-capitalized. *See* Shah Decl. ¶ 19.
- 5. The DIP Term Facility is critical to, among other things, (a) continue satisfying obligations to the Debtors' contract counterparties; (b) provide the liquidity necessary to honor the obligations owed to ordinary course claimants and to continue favorable trade terms with such claimants; (c) fund the Initial Intercompany Transaction (as defined herein); (d) fund the Foreign Reserve Account (as defined herein); (e) reassure other stakeholders, including channel partners,

customers, and employees, that the Debtors have sufficient funds to continue operating in the ordinary course; (f) fund the Debtors' payroll obligations; and (g) fund the administrative costs of these Chapter 11 Cases, in each case in accordance with the Initial DIP Budget (as defined in the Interim Order) agreed upon by the Debtors and the Required DIP Lenders (as defined in the Interim Order), attached to the Interim Order as Exhibit B; and (h) cash collateralize letters of credit under the Cash Collateralized LC Facility (as defined herein). See Koza Decl. ¶¶ 20–22. On top of significant new money commitments, approximately \$56 million of the proceeds of the DIP Term Facility will be allocated to irrevocably refinancing the outstanding obligations under the Debtors' Prepetition ABL Facility (excluding amounts attributable to cash collateralizing letters of credit) (the "ABL Credit Facility Refinancing").

- 6. After careful consideration and extensive negotiation, the Debtors, in a prudent exercise of their business judgment, determined that these transactions are necessary to procure the proposed DIP Term Loan Facility, in the best interests of their estates, necessary to preserve the going-concern value of the enterprise, avoid unnecessary disputes and additional significant costs to the estates, and do not unduly prejudice any other creditors. Indeed, these transactions are critical elements of the Debtors' restructuring strategy and reflect the challenges posed by the unique structure of the Debtors' prepetition indebtedness, where valuable foreign non-Debtor affiliates (each, an "NFA," and collectively, the "NFAs") are guarantors of a portion of such indebtedness.
- 7. Foreign Subsidiary Funding. As set forth in the Koza Declaration, failure to obtain postpetition financing that provides for the Initial Intercompany Transaction would seriously jeopardize the Company's operations. Given that certain NFAs are incorporated in various foreign jurisdictions, filing such entities for chapter 11 and potentially ancillary

proceedings in multiple foreign jurisdictions would involve significant cross-border complexities that would be extremely disruptive to day-to-day operations and likely result in significantly increased professional fees. Additional entities filing for bankruptcy could also cause vendor and customer contraction, employee loss, and increase demands on the Debtors' management team. Moreover, many of the Debtors' contracts and supply chain agreements are intertwined between the Debtors and the NFAs and cessation of NFA operations could result in a material impact to the Debtors' revenue. By undertaking the Initial Intercompany Transaction, the Debtors will be able to maintain their ordinary course relationships with their NFAs and avoid value-destructive business interruptions to the Debtors' operations during the pendency of these Chapter 11 Cases. Accordingly, the Debtors seek the Court's authorization to use the proceeds of the DIP Term Loan Facility, in part, to consummate the Initial Intercompany Transaction and provide incremental, as-needed funding for postpetition intercompany transactions through the Foreign Reserve Account. Koza Decl. ¶¶ 10–18.

8. *Escrow Release*. In addition, the Debtors are seeking authority to execute and deliver an escrow release agreement, which provides that, once honored, the agreement governing that certain Escrow (described in greater detail in the First Day Declaration) shall be terminated, and, upon such termination, the escrow agent will release the cash therein to the Prepetition Term Loan Agent on account of the B-3 Escrow Claims (in essence, releasing the cash to the parties who first provided the cash that was funded into the Escrow). This Escrow Release is a critical component of the holistic RSA, returns funds the Debtors do not need to operate their businesses (and have never accessed since the funding of the Escrow) to the parties who provided the funding for such Escrow, and does not prejudice any party because it is subject to the Challenge Period and, if there is a successful Challenge, the Court may fashion an equitable remedy.

- 9. In addition to the requested relief regarding the DIP Term Loan Facility and the DIP-to-Exit ABL Commitment Papers, by this Motion, the Debtors seek the Court's authorization to use Cash Collateral. The nature of the Debtors' business requires the Debtors to have immediate use of Cash Collateral. Without it, the Debtors would be unable to operate their business and administer their estates, which would immediately and irreparably harm their stakeholders. With access to Cash Collateral, the Debtors will be able to continue their operations and preserve value for the benefit of their estates and stakeholders. The Prepetition Secured Parties have consented to the continued use of Cash Collateral under the terms included in the Interim Order (and as summarized in detail below). *See* Koza Decl. ¶ 22.
- 10. Overall, the DIP Term Loan Facility, in conjunction with the commitments for the DIP ABL Facility, are the culmination of extensive prepetition negotiations between the Debtors and the Ad Hoc Groups and are by far the best (and only) proposal that the Debtors received. With widespread support, no "roll-up" of First Lien Claims, and certainty at exit, there is no question that the DIP Term Loan Facility and the DIP-to-Exit ABL Commitment Papers are in the best interest of these estates.
- 11. As set forth in the Koza Declaration, the Debtors and their estates would suffer immediate and irreparable harm if the Debtors were denied the financing needed to sustain ongoing business operations during the critical first weeks of these Chapter 11 Cases. Koza Decl. ¶ 25. Without access to the DIP Term Loan Facility, the Debtors would be unable to fund the operational and liquidity needs for the NFAs, risking the commencement of additional cases for all their NFAs and their potential near-term liquidation—all to the serious detriment of their stakeholders due to vendor contraction and lack of confidence in the market. *See* Koza Decl. ¶ 14. Doing nothing and living with the risk of remedies being exercised against the NFAs is a highly untenable option.

The DIP Term Loan Facility ensures that the Debtors (a) have sufficient funding to consummate a value-maximizing restructuring transaction and (b) can continue to operate in the ordinary course in these Chapter 11 Cases. Further, as set forth in the Shah Declaration, the terms of the DIP Term Loan Facility are reasonable under the circumstances, and were the product of good faith, arm's-length negotiations. Shah Decl. ¶¶ 25, 32.

Debtors' operations and consummate a comprehensive restructuring transaction. For the reasons set forth in this Motion, the First Day Declaration, and the DIP Declarations, the Debtors firmly believe that approval of the DIP Term Loan Facility and the DIP-to-Exit ABL Commitment Papers will maximize value for all of the Debtors' stakeholders and is in the exercise of the Debtors' sound business judgment. Accordingly, the Debtors respectfully request that the Court approve the relief requested herein and enter an interim order (the "Interim Order") and a final order (the "Final Order") and, together with the Interim Order, the "DIP Orders").

## **Jurisdiction and Venue**

- 13. The United States Bankruptcy Court for the Southern District of Texas (the "Court") has jurisdiction over this matter pursuant to 28 U.S.C. § 1334. This is a core proceeding pursuant to 28 U.S.C. § 157 (b). The Debtors confirm their consent to the entry of a final order by the Court.
  - 14. Venue is proper pursuant to 28 U.S.C. §§ 1408 and 1409.
- 15. The bases for the relief requested herein are sections 105, 361, 362, 363, 364, 503, and 507 of title 11 of the United States Code (the "Bankruptcy Code"), rules 2002 and 4001 of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), and rules 4002-1 and 9013-1(b) of the Bankruptcy Local Rules for the Southern District of Texas (the "Bankruptcy Local Rules").

#### **Relief Requested**

- 16. The Debtors seek entry of the Interim Order, and the Final Order:<sup>4</sup>
  - a. authorizing Avaya Inc., in its capacity as borrower (the "DIP Borrower") to obtain senior secured postpetition financing on a superpriority basis, and for each of the other Debtors to guarantee unconditionally (the "Guarantors"), on a joint and several basis, the Borrower's obligation in connection therewith, consisting of the DIP Term Loan Facility (and all amounts extended under the DIP Term Loan Facility, pursuant to the terms and conditions of that certain Senior Secured Super-Priority Debtor-in-Possession Credit Agreement (as the same may be amended, restated, supplemented, or otherwise modified from time to time, the "DIP Term Loan Credit Agreement")), by and among Holdings, the Borrower, and Wilmington Savings Fund Society, FSB ("Wilmington") as administrative agent and collateral agent (the "DIP Term Loan Agent,") for and on behalf of itself and the other lenders party thereto (the "DIP Term Loan Lenders," and collectively with the DIP Term Loan Agent, the "DIP Secured Parties"), substantially in the form attached hereto as Exhibit A;
  - b. authorizing the Debtors to execute and deliver the DIP Term Loan Credit Agreement and any other agreements and documents related thereto (collectively with the DIP Term Loan Credit Agreement, the "DIP Term Loan Facility Documents,"), by and among the DIP Borrower, the Guarantors, the DIP Term Loan Agent, and to perform such other acts as may be necessary or desirable in connection with the DIP Term Loan Facility Documents;
  - c. authorizing the Debtors to pay certain fees required by the DIP Term Loan Agent in connection with the DIP Term Loan Facility;
  - d. approving entry into the DIP-to-Exit ABL Commitment Papers and authorizing the Debtors to take all actions necessary, appropriate, or required to comply with the Debtors' obligations under the DIP-to-Exit ABL Commitment Letter, including filing the DIP-to-Exit ABL Fee Letters under seal pursuant to the Fee Sealing Motion;<sup>5</sup>
  - e. authorizing the DIP Borrower to enter into the DIP Term Loan Facility Documents and related documents providing for a \$500 million DIP Term Loan Facility, to be drawn as \$400 million on an interim basis and an additional \$100 million on a final basis, that will:
    - i. upon entry of the Interim Order, be (a) used to fund the \$50 million required for an upfront postpetition intercompany transaction to certain NFAs (as

9

<sup>&</sup>lt;sup>4</sup> The Debtors will file the form of Final Order prior to the Final Hearing (as defined herein).

<sup>&</sup>lt;sup>5</sup> "<u>Fee Sealing Motion</u>" means the *Debtors' <u>Emergency</u> Motion Seeking Entry of an Order Authorizing the Debtors to File the Fee Letters Under Seal*, filed contemporaneously herewith.

further described below) (the "<u>Initial Intercompany Transaction</u>"), subject to the Intercompany Transfer Mechanic (as defined herein) and (b) fund \$40 million into a segregated account (as further described below) (the "<u>Foreign Reserve Account</u>"), subject to the Intercompany Transfer Mechanic and Foreign Reserve Protocol (as defined herein), to provide a liquidity backstop for the NFAs through intercompany borrowings on an as-needed basis;

- ii. upon entry of the Interim Order, cash collateralize outstanding letters of credit issued under the Prepetition ABL Facility, totaling approximately \$40 million, and cause all such letters of credit to be deemed to have been issued under a separate cash collateralized facility (the "Cash Collateralized LC Facility"); and
- iii. upon entry of the Interim Order, authorize the ABL Credit Facility Refinancing, pursuant to which the Debtors will repay all amounts outstanding or due and payable under the Prepetition ABL Facility, in cash in full, including accrued and unpaid interest and fees and any accrued but unpaid reimbursable costs and expenses;
- f. authorizing the Debtors to grant, in each case subject to the Carve-Out and the ABL Credit Facility Refinancing:
  - i. a first priority, perfected lien on substantially all unencumbered assets, subject to certain exceptions and exclusions; *provided* that such liens and claims will not attach to the proceeds of avoidance actions until entry of the Final Order;
  - ii. a perfected junior lien on all assets subject to other validly perfected liens as of the date hereof or liens that are perfected following the date hereof to the extent permitted by section 546(b) of the Bankruptcy Code;
  - iii. a first priority, perfected priming lien with respect to assets currently encumbered by the Debtors' prepetition funded debt; and
  - iv. superpriority claims with respect to obligations outstanding under the DIP Term Loan Facility;
- g. approving the Foreign Reserve Protocol, which, as defined and described more fully below, generally provides that any intercompany loans using funds from the Foreign Reserve Account (each, a "Foreign Reserve Account Withdrawal") shall

- be subject to certain notice and consent procedures with the Ad Hoc Groups providing the DIP Term Loan Facility;
- h. approving the Intercompany Transfer Mechanic, which, as defined and described more fully below, generally provides that:
  - any credit support or cash/cash equivalents transferred from any Debtor to any NFA shall (i) first be transferred to Sierra, (ii) second, transferred by Sierra to Avaya International Sales Ltd ("AISL" or the "Initial Intercompany Borrower"), and (iii) further transferred by the Initial Intercompany Borrower, in one or more steps, to the applicable NFA;
  - ii. any cash transferred from Sierra to any NFA shall be pursuant to an unsecured intercompany note or general ledger entry (the "NFA Notes/Receivables"), and notwithstanding the fact that the NFA Notes/Receivables shall be unsecured obligations of the applicable NFA, the NFA Notes/Receivables shall be deemed to be secured obligations of the NFAs for allocation of value in these Chapter 11 Cases; and
  - iii. all NFA Notes/Receivables shall be payable in full in cash upon maturity of or acceleration (except in the case of conversion of the DIP Loans to loans under the Exit Facility as contemplated by the RSA) the DIP Term Loan Facility.
- i. authorizing the Debtors to execute and deliver an escrow release agreement regarding the Escrow, as described above;
- j. authorizing the Debtors to use Cash Collateral within the meaning of sections 363(a) and 363(c) of the Bankruptcy Code;
- k. approval of the form and manner of adequate protection to be provided to the Prepetition Secured Parties<sup>6</sup> pursuant to sections 361 and 363 of the Bankruptcy Code, in each case subject to the Carve Out, including payment of professional fees and replacement liens and claims in favor of the Prepetition Secured Parties encumbering each of the Debtors' assets subject to the DIP Liens and Claims, such liens and claims granted priority in accordance with the Debtors' intercreditor agreements; and
- 1. vacating and modifying the automatic stay imposed by section 362 of the Bankruptcy Code to the extent necessary to implement and effectuate the terms and provisions of the DIP Term Loan Facility Documents and the DIP Orders, with

-

<sup>&</sup>lt;sup>6</sup> "<u>Prepetition Secured Parties</u>" means the agents, trustees, lenders and Holders under the Prepetition Term Loan Facility, the Legacy Notes, and the Secured Exchangeable Notes.

- actions against the Collateral requiring at least five calendar days' notice to the Debtors, the U.S. Trustee, and any official committee;
- m. waiving any applicable stay, including (to the extent applicable) under Bankruptcy Rule 6004, to provide for immediate effectiveness of the Interim Order;
- n. scheduling a date for a hearing on the Motion to consider entry of the Final Order (the "Final Hearing"); and
- o. granting related relief.

# Concise Statement of the Material Terms of the Interim Order Pursuant to Bankruptcy Rule 4001 and the Procedures for Complex Cases in the Southern District of Texas

17. The following chart contains a summary of the material terms of the proposed Interim Order, together with references to the applicable sections of the Interim Order and other relevant source documents, including the DIP Term Loan Credit Agreement, as required by Bankruptcy Rules 4001(b)(1)(B) and 4001(c)(1)(B) and the *Procedures for Complex Chapter 11 Cases in the Southern District of Texas* (the "Complex Case Procedures").<sup>7</sup>

SUMMARY OF MATERIAL TERMS OF PROPOSED DIP TERM LOAN FINANCING		
Bankruptcy Rule	DIP Term Loan Facility	
Parties to the DIP Term Loan Facility Bankruptcy Rule 4001(c)(1)(B)	Borrower: Avaya Inc. a Delaware corporation and a Debtor and Debtor in Possession under chapter 11 of the Bankruptcy Code (the "Borrower")  Guarantors: (a) Holdings, (b) each Domestic Subsidiary (other than an Excluded Subsidiary) that provides the Guarantee on the Closing Date, which in any event will include all Subsidiaries that are Debtors on the Closing Date, or becomes a party to the Guarantee after the Closing Date pursuant to the DIP Term Loan Credit Agreement or otherwise (including becoming a Debtor in the Chapter 11 Cases unless	
	waived), and (c) the Borrower (other than with respect to its own Obligations).  DIP Lenders: The lending institutions from time to time party to the DIP Term Loan Credit Agreement.  Administrative Agent and Collateral Agent: Wilmington Savings Fund Society, FSB.	

<sup>7</sup> The summaries contained in this Motion are qualified in their entirety by the provisions of the documents referenced. To the extent anything in this Motion is inconsistent with such documents, the terms of the applicable documents shall control. Capitalized terms used in this summary chart but not otherwise defined have the meanings ascribed to them in the DIP Term Loan Facility Documents or the Interim Order, as applicable.

SUMMARY OF	MATERIAL TERMS OF PROPOSED DIP TERM LOAN FINANCING	
Bankruptcy Rule	DIP Term Loan Facility	
	See DIP Term Loan Credit Agreement, Preamble.	
Term Bankruptcy Rule 4001(b)(1)(B)(iii) 4001(c)(1)(B)	The maturity date with respect to the DIP Facility (the "Maturity Date") shall be the earliest of:  (a) six (6) months after the Closing Date (or if such day shall not be a Business Day, the next succeeding Business Day) (the "Scheduled Maturity Date");	
	(b) 45 days after the Petition Date if the Final Order has not been entered prior to the expiration of such 45-day period, unless otherwise extended by the Required Lenders;	
	(c) the Consummation Date;	
	(d) the acceleration of the Loans and the termination of the Commitments with respect to the Term Facility;	
	(e) the consummation of a sale of all or substantially all of the assets of the Borrower (or the Borrower and the Guarantors) pursuant to section 363 of the Bankruptcy Code; and	
	(f) the termination of the Restructuring Support Agreement.	
	See DIP Term Loan Credit Agreement, § 1.1, "Defined Terms."	
Commitments Bankruptcy Rule 4001(c)(1)(B)	The DIP Facility commitments total \$500 million (the " <u>Term Facility</u> ").  See DIP Term Loan Credit Agreement, "Recitals."	
Conditions of Borrowing Bankruptcy Rule 4001(c)(1)(B)	The Order and the DIP Term Loan Credit Agreement include conditions to closing that are customary and appropriate for similar debtor-in-possession financings of this type.	
	See DIP Term Loan Credit Agreement § 6.2, "Collateral."	
Interest Rates Bankruptcy Rule	The DIP Term Loans will bear interest at a rate equal to SOFR plus 8.00%.	
4001(c)(1)(B)(ii)	See DIP Term Loan Credit Agreement, § 2.8, "Interest."	
Use of Proceeds and Cash Collateral Bankruptcy Rule 4001(b)(l)(B)(ii)	The Borrower will use the proceeds from the Loans (i) to consummate the Prepetition ABL Repayment, (ii) for working capital and general corporate purposes of the Debtors; (iii) to pay obligations arising from or related to the Carve Out; (iv) to pay professional fees in connection with the Chapter 11 Cases; (v) to make adequate protection payments; (vi) to pay fees and expenses incurred in connection with the transactions contemplated hereby and in the Restructuring Support Agreement; (vii) to fund an intercompany loan in an amount not to exceed \$50,000,000 (the "Initial Intercompany Transaction") by Sierra to Avaya International Sales Ltd (the "Initial Intercompany Borrower") to be used for working capital and general corporate purposes of foreign Subsidiaries of the Borrower, including, but not limited to, to fund certain cash pooling accounts of certain foreign Subsidiaries of the Borrower and opened at Citibank, N.A. and its affiliates; and (viii) to fund a deposit of up to \$40,000,000 to a segregated account held by the Borrower (the "Foreign Reserve Account") for purposes of backstopping the liquidity of certain foreign non-Debtor affiliates to the extent necessary to preserve the value of the Debtors' international	

SUMMARY OF	MATERIAL TERMS OF PROPOSED DIP TERM LOAN FINANCING
Bankruptcy Rule	DIP Term Loan Facility
	business; provided that the use of funds in the Foreign Reserve Account during the Chapter 11 Cases shall be subject to the Foreign Reserve Protocol.
	See DIP Term Loan Credit Agreement, § 9.13, "Use of Proceeds."
Parties with an Interest	The Prepetition ABL Secured Parties and the Prepetition Secured Parties
in Cash Collateral Bankruptcy Rule 4001(b)(1)(B)(i)	See Interim Order ¶ 4.
Fees Bankruptcy Rule	The Borrower has agreed to pay to the Agents and the Lenders:
4001(c)(1)(B)	(a) All fees in the amounts and at the times specified in the Fee Letter due and payable to the Agents;
	(b) to the Administrative Agent, for the account of each Lender, an upfront fee (the "Term Upfront Fee") (which, at the discretion of such Lender, may take the form of original issue discount) equal to 4.00% (the "Term Upfront Fee Percentage") of the aggregate principal amount of the Loans funded hereunder in respect of each Borrowing referred to in Section 2.01(a), which Term Upfront Fee shall be earned, due and payable on the date of such Borrowing and calculated by multiplying the Term Upfront Fee Percentage by the aggregate principal amount of Loans funded by such Lender on the date of such Borrowing; provided that notwithstanding the foregoing, for the avoidance of doubt no applicable original issue discount shall reduce the amount of Obligations required to be prepaid or repaid with respect to the Loans owing hereunder;
	(c) a fee (the "Put Option Premium") equal to 6.0% of the aggregate principal amount of the Commitments as of the date of the Commitment Letter without giving effect to any termination or reduction, (x) to the extent the Plan shall have been consummated on or prior to the Maturity Date, to any Lender thereof or any Person designated by such Lender (pro rata in accordance with the outstanding principal amount of the Commitments and Loans held by such Lender on the date that the Put Option Premium is payable), in shares of New Equity Interests at a 37.5% discount to an implied equity value of \$538,812,500, or (y) otherwise, to the Administrative Agent, for the ratable account of each Lender thereof (pro rata in accordance with the outstanding principal amount of the Commitments and Loans held by such Lender on the date that the Put Option Premium is payable), in cash;
	(d) an exit fee (the " <u>Exit Fee</u> ") in cash in an amount equal to 1.0% of the sum of the principal amount of such Loans repaid or prepaid or accelerated or Commitments terminated, which Exit Fee shall be earned on the Closing Date;
	(e) a fronting fee equal to 0.375% of the aggregate Commitments as of the Closing Date; and
	(f) all fees required to be paid on the Closing Date pursuant to the Fee Letter and reasonable and documented out-of-pocket expenses required to be paid on the Closing Date pursuant to the Commitment Letter and the DIP Term Loan Credit Agreement.
	No other fees and expenses are due and payable under the terms of the DIP Term Loan Facility.

SUMMARY OF MATERIAL TERMS OF PROPOSED DIP TERM LOAN FINANCING		
Bankruptcy Rule	DIP Term Loan Facility	
	See DIP Term Loan Credit Agreement, § 4.1, "Fees;" § 6.6, "Fees."	
Budget Bankruptcy Rule 4001 (c)(1)(B)	The Debtors are permitted to use the proceeds of the DIP Term Loan Facility and the Cash Collateral in accordance with the Initial DIP Budget, substantially in the form attached to the Interim Order as Exhibit B.  See Interim Order 23, Exhibit B.	
Milestones Bankruptcy Rule 4001(c)(1)(B)	The Borrower shall comply with the following Milestones (collectively, the "Milestones"):	
4001(c)(1)(b)	(a) The Interim Order shall have been entered by no later than three (3) days after the Petition Date;	
	(b) The Escrow Payment shall occur no later than the date when the Interim DIP Order is entered by the Bankruptcy Court;	
	(c) The Final Order shall have been entered by no later than forty-five (45) days after the Petition Date;	
	(d) The Rights Offering (as defined in the Restructuring Support Agreement) shall have been commenced by no later than ten (10) days after the Petition Date;	
	(e) The 2023 PBGC Settlement shall have been approved by the Court no later than confirmation of the Plan;	
	(f) The order provisionally approving the adequacy of the Disclosure Statement, in form and substance acceptable to the Debtors and the Required Lenders, shall have been entered no later than three (3) days after the Petition Date;	
	(g) The Plan and Disclosure Statement shall have been filed no later than one (1) day after the Petition Date;	
	(h) The order approving the adequacy of the Disclosure Statement shall have been entered no later than sixty (60) days after the Petition Date;	
	(i) The order confirming the Plan shall have been entered by no later than sixty (60) days after the Petition Date;	
	(j) The substantial consummation of the Plan shall have occurred no later than ninety (90) days after the Petition Date.	
	See DIP Term Loan Credit Agreement, § 9.16, "Milestones."	

SUMMARY OF	MATERIAL TERMS OF PROPOSED DIP TERM LOAN FINANCING
Bankruptcy Rule	DIP Term Loan Facility
Liens and Priority Bankruptcy Rule 4001(c)(1)(B)(i)	As security for the DIP Obligations, subject and subordinate in all respects to the Carve Out, effective and perfected upon the entry of the Interim Order and without the necessity of the execution, recordation or filing by the Loan Parties, the DIP Agent or any DIP Lender of mortgages, security agreements, control agreements, pledge agreements, financing statements or other similar documents, or the possession or control by the DIP Agent of, or over, any DIP Collateral, the following security interests and liens are hereby granted to the DIP Agent for their own respective benefit and the benefit of the DIP Lenders, subject only to the Carve Out and the Permitted Liens (all such liens and security interests granted to the DIP Agent, for their respective benefit and for the benefit of the respective DIP Lenders, pursuant to the Interim Order and the DIP Documents, the "DIP Liens"):
	(a) pursuant to Bankruptcy Code section 364(c)(2), and subject and subordinate in all respects to the Carve Out, valid, binding, continuing, enforceable, fully-perfected first priority senior security interests in and liens upon all DIP Collateral, to the extent such DIP Collateral is not subject to valid, perfected and non-avoidable liens as of the Petition Date or valid and non-avoidable liens in favor of third parties that were in existence immediately prior to the Petition Date that are perfected as permitted by Bankruptcy Code section 546(b) ("Unencumbered Property");
	(b) pursuant to Bankruptcy Code section 364(c)(3), and subject and subordinate in all respects to the Carve Out, valid, binding, continuing, enforceable, fully-perfected junior security interests in and liens on the DIP Collateral, to the extent such DIP Collateral is subject to (i) valid, perfected and non-avoidable liens as of the Petition Date or (ii) valid and unavoidable liens in favor of third parties that were in existence immediately prior to the Petition Date that are perfected as permitted by Bankruptcy Code section 546(b), in each case other than the Primed Liens, with the relative priorities among the DIP Liens as set forth on Exhibit A attached to the Interim Order; and
	(c) Priming Liens. Pursuant to Bankruptcy Code section 364(d)(1), and subject and subordinate in all respects to the Carve Out, a valid, binding, continuing, enforceable, fully-perfected priming senior security interests in and liens upon the Prepetition Collateral.
	The DIP Liens securing the DIP Obligations are valid, automatically perfected, non-avoidable, senior in priority, and superior to any security, mortgage, collateral interest, lien, or claim to any of the DIP Collateral, except that the DIP Liens shall be subject to the Carve Out in all respects and shall otherwise be junior only to the Permitted Liens.
	See Interim Order ¶¶ 10–11.
Carve-Out Bankruptcy Rule	The Order provide a "Carve-Out" of certain statutory fees and allowed professional fees of the Debtors pursuant to section 1103 of the Bankruptcy Code.
4001(c)(1)(B)	See Interim Order ¶ 14.

SUMMARY OF MATERIAL TERMS OF PROPOSED DIP TERM LOAN FINANCING		
Bankruptcy Rule	DIP Term Loan Facility	
Challenge Period Bankruptcy Rule 4001(c)(l)(B)	The Interim Order provides for a challenge period no later than the date that is the earlier of (x) the commencement of a hearing to consider confirmation of the Plan and (y) sixty (60) days after the Petition Date (the "Challenge Period"); provided that so long as the Tolling Conditions are met, the Challenge Period shall be tolled with respect to the Creditors' Committee and, upon the failure of the Tolling Conditions, the Creditors' Committee will have sixty (60) days from the date of such failure to bring any Challenge, and (b) there is entered a final, non-appealable order in favor of the plaintiff in any such timely filed Challenge sustaining such Challenge.  See Interim Order ¶ 31.	
Adequate Protection Bankruptcy Rule 4001(b)(1)(B)(iv),	The adequate protection provided to the Prepetition Secured Parties shall be in accordance with the Interim Order.	
4001(c)(1)(B)(ii)	See Interim Order ¶ 20.	
Events of Default Bankruptcy Rule 4001(c)(1)(B)	The DIP Term Loan Facility contains events of default that are usual and customary for debtor-in-possession financings, including without limitation, the failure to timely comply with any of the Case Milestones.	
	See DIP Term Loan Credit Agreement, § 11, "Events of Default."	
Waiver/Modification of the Automatic Stay Bankruptcy Rule 4001(c)(1)(B)(iv)	Pursuant to the Interim Order, the automatic stay provisions of Bankruptcy Code section 362(a) of the Bankruptcy Code shall be modified to the extent necessary to implement and effectuate the terms of the Interim Order.  See Interim Order ¶ 22.	
Indemnification Bankruptcy Rule 4001(c)(1)(B)(ix)	The DIP Facility contains indemnification provisions ordinary and customary for financings of this type.  See DIP Term Loan Credit Agreement § 12.7, "Indemnification."	
Non-Debtor Funding 4001(c)(1)(B)(i)-(ii)	The Borrower will establish and fund a segregated account held by Avaya Inc. with proceeds from the DIP Term Loan Facility in an amount equal to \$40,000,000 for purposes of backstopping the liquidity of certain foreign non-Debtor affiliates to the extent necessary to preserve the value of the Debtors' international business operations.	
Foreign Reserve Protocol 4001(c)(1)(B)(i)–(ii)	See Interim Order, "Recitals."  Any intercompany loans using funds from the Foreign Reserve Account (each, a "Foreign Reserve Account Withdrawal") in accordance with the DIP Documents shall be subject to procedures (the "Foreign Reserve Protocol") in addition to the Intercompany Transfer Mechanic:	
	(a) the Debtors shall submit in writing to the DIP Agent, with a copy to the advisors to the Akin Ad Hoc Group and the PW Ad Hoc Group a withdrawal request (the "Foreign Reserve Account Withdrawal Notice"), which Foreign Reserve Account Withdrawal Notice shall specify (i) the proposed foreign non-Debtor recipient of the transfer; (ii) the proposed transfer amount; and (iii) the proposed use of funds; and	
	(b) the DIP Agent shall, within three (3) business days from receipt of the Foreign Reserve Account Withdrawal Notice, advise the Debtors if the Required DIP Lenders consent to any such Foreign Reserve Account Withdrawal, which consent shall not be unreasonably withheld. All Foreign Reserve Account	

SUMMARY OF MATERIAL TERMS OF PROPOSED DIP TERM LOAN FINANCING		
Bankruptcy Rule	DIP Term Loan Facility	
	Withdrawals must be made in accordance with the Intercompany Transfer Mechanic.  See Interim Order, ¶ 35.	
Intercompany Transfer	No DIP Proceeds, DIP Collateral or any Cash Collateral may be transferred or	
Mechanic	otherwise provided by any Debtor, directly or indirectly, to or for the benefit of any	
4001(c)(1)(B)(i)–(ii)	non-Debtor subsidiary or subsequently by any non-Debtor subsidiary to any other non-Debtor subsidiary (either in the form of restricted payments, investments, intercompany advances, guarantee of obligations or otherwise (except for repayment	
	of account payables for goods delivered or services rendered postpetition in accordance with the order approving the relief sought in the Cash Management Motion, including pursuant to shared services or other intercompany service agreements, in the ordinary course of business, consistent with past practice and subject to the DIP Budget)), other than the Initial Intercompany Transaction and, subject to the Foreign Reserve Protocol, any Foreign Reserve Account Withdrawal ("Non Debtor Subsidiary Transfer"), which, in each case, shall be, (i) first, lent to Sierra Communications; (ii) second lent by Sierra Communications to the Initial Intercompany Borrower, evidenced by an intercompany unsecured note or intercompany ledger entry, in each case as is reasonably satisfactory to the Required DIP Lenders and the Debtors and (iii) further lent by the Initial Intercompany Borrower, in one or more steps, to the applicable non-Debtor subsidiary; each step of such transfer set forth in clause (iii) shall be evidenced by an intercompany unsecured note or intercompany ledger entry in each case as is reasonably satisfactory to the Required DIP Lenders and the Debtors (the transfers, intercompany notes or intercompany ledger entry referenced in clauses (ii) and (iii), collectively, the "Non-Debtor Subsidiary Notes/Receivable"). All Non-Debtor Subsidiary Notes/Receivables held by Sierra Communications shall accrue interest at the same rate as the DIP Term Loan Obligations and be pledged to secure the DIP Obligations.  See Interim Order, ¶ 36.	
Section 506(c) and 552(b) Waiver Bankruptcy Rule 4001(c)(l)(B)(x)	Subject to and effective upon the entry of the Final Order, the Prepetition Secured Parties and DIP Secured Parties are subject to a waiver of (a) any "equities of the case" exception under section 552(b) of the Bankruptcy Code and (b) 506(c) of the Bankruptcy Code.	
	See Interim Order, ¶ 16–17.	

## **Statement Regarding Significant Provisions**

18. The DIP Orders contain certain of the provisions (the "Significant Provisions")<sup>8</sup> identified in Section C, paragraph 8 of the Complex Case Procedures. The Significant Provisions included in the DIP Orders are as follows:

Significant Provisions refer to those provisions that: (a) grant cross-collateralization protection (other than replacement liens or other adequate protection) to prepetition secured creditors; (b) deem prepetition secured debt to be postpetition debt or that use postpetition loans from a prepetition secured creditor to pay part or all of that

- a. **Plan and Disclosure Statement-Related Milestones**. The DIP Term Loan Credit Agreement provides milestones for (i) filing a plan of reorganization and a disclosure statement of no later than one (1) day after the petition date and (ii) obtaining confirmation of the Plan no later than sixty (60) days after the Petition Date (see DIP Term Loan Credit Agreement § 9.16);
- b. *No-Cross Collateralization*. The DIP Orders do not authorize, and the DIP Term Loan Facility does not provide for, any cross-collateralization of the Debtors' funded-debt;
- c. Requirement that Postpetition Loans Be Used to Repay Prepetition Debt.

  The Interim Order authorizes the Debtors to refinance the Prepetition ABL Facility pursuant to the Prepetition ABL Repayment (see Interim Order ¶ 6);
- d. *Liens on Proceeds of Avoidance Actions*. The Final Order, but not the Interim Order, grants the DIP Secured Parties liens on the proceeds of claims and causes of action arising under chapter 5 of the Bankruptcy Code (the "Avoidance Action Proceeds"), excluding the actual claims or causes of action arising under chapter 5 of the Bankruptcy Code (collectively, "Avoidance Actions") (see Interim Order ¶ 13);
- e. Default Provisions and Remedies.
  - The Debtors shall promptly provide notice to the DIP Agent and the i. Prepetition First Lien Agents (with a copy to counsel of the Akin Ad Hoc Group and counsel to the PW Ad Hoc Group) of the occurrence of any DIP Termination Date. Upon the occurrence and during the continuation of any DIP Termination Date, the DIP Agent (at the direction of the Required DIP Lenders in accordance with the applicable DIP Credit Agreement) shall be required to provide five (5) business days' written notice (such period, the "Remedies Notice Period" and such notice, a "Termination Declaration") to the Debtors, counsel to any Creditors' Committee (if appointed), counsel to the Prepetition First Lien Agents, counsel of the Akin Ad Hoc Group, counsel to the PW Ad Hoc Group and the U.S. Trustee of the applicable DIP Agent's intent to exercise its rights and remedies, prior to the exercise of any of the following rights: (1) declaring all DIP Obligations, including any and all accrued interest, premiums, fees and expenses constituting the DIP Obligations

secured creditor's prepetition debt, other than as provided in section 552(b) of the Bankruptcy Code; (c) bind the bankruptcy estates or any parties in interest with respect to the validity, perfection, or amount of the secured creditor's prepetition lien or debt or the waiver of claims against the secured creditor; (d) waive or limit the estate's rights under section 506(c) of the Bankruptcy Code; (e) grant prepetition secured creditors liens on the debtor's claims and causes of action arising under chapter 5 of the Bankruptcy Code; (f) impose deadlines for the filing of a plan or disclosure statement; and (g) grant an administrative claim.

owing under the DIP Documents, to be immediately due and payable; (2) declaring the commitment of each DIP Lender to make DIP Loans to be terminated, whereupon such commitments and obligation shall be terminated to the extent any such commitment remains under the DIP Facility; (3) the termination of the DIP Facility and the DIP Documents as to any future liability or obligation of the DIP Secured Parties, but without affecting any of the DIP Liens or the DIP Obligations; (4) termination and/or revocation of the Debtors' right, if any, under the Interim Order and the DIP Documents to use any Cash Collateral of the DIP Secured Parties; (5) charging of interest at the default rate under the DIP Facility; (6) freezing of monies or balances in the DIP Proceeds Account; (7) enforcing any and all rights against the DIP Collateral in possession of the applicable DIP Agent, including, without limitation, disposition of the DIP Collateral, solely for the application towards the Carve Out and the DIP Obligations in accordance with their respective priorities; and (8) taking any other actions or exercise any other rights or remedies permitted under the Interim Order, the DIP Documents, or applicable law; provided that the DIP Lenders shall not be obligated to make any DIP Loans or advances under the DIP Facility during any Remedies Notice Period. The DIP Agent may provide a Termination Declaration, notwithstanding the provisions of Bankruptcy Code section 362, without any application, motion or notice to, hearing before, or order from the Court (see Interim Order ¶ 25);

ii. Upon delivery of a Termination Declaration, each of the DIP Agent, the DIP Lenders, the Debtors, the Creditors' Committee, and the applicable Prepetition Secured Parties consents to a hearing on an expedited basis to consider (a) whether a DIP Termination Date has occurred and (b) any appropriate relief (including, without limitation, the Debtors' non-consensual use of Cash Collateral). During the Remedies Notice Period, notwithstanding anything to the contrary set forth in paragraph 25 of the Interim Order, the Debtors shall continue to have the right to use Cash Collateral in accordance with the terms of the Interim Order, solely to pay necessary expenses set forth in the DIP Budget to avoid immediate and irreparable harm to the Debtors' estates. At the end of the Remedies Notice Period, unless the Court has entered an order to the contrary or otherwise fashioned an appropriate remedy, including whether DIP

<sup>&</sup>quot;DIP Term Loan Proceeds Account" has the meaning ascribed to it in the Debtors' Emergency Motion for Entry of Interim and Final Orders (I) Authorizing the Debtors to (A) Continue Using the Cash Management System, (B) Maintain Existing Bank Accounts and Business Forms and Books and Records, and (C) Continue Using the Investment Account and the Investment Policy, (II) Authorizing Continued Intercompany Transactions, (III) Granting Administrative Expenses Status to Postpetition Intercompany Transactions, and (IV) Granting Related Relief (the "Cash Management Motion") filed contemporaneously herewith.

Termination Date has occurred, the Debtors' right to use Cash Collateral shall immediately cease, unless otherwise provided herein, and the DIP Agent and DIP Lenders shall have the rights set forth immediately below (*see* Interim Order ¶ 26); and

iii. During the Remedies Notice Period, prior to the exercise or enforcement of any rights against the DIP Collateral (other than as set forth in paragraph 26 of the Interim Order), the DIP Agent (at the direction of the Required DIP Lenders in accordance with the DIP Documents) shall be required to file an emergency motion with the Court or file the appropriate written notice in accordance with the applicable Court procedures on five (5) Business Days' notice (the "Stay Relief Hearing") to determine whether a DIP Termination Event has occurred (and the Loan Parties and the Creditors' Committee, if any, shall not object to the shortened notice with respect to such Stay Relief Hearing). In the event the Court determines during a Stay Relief Hearing that a DIP Termination Date has occurred, the Court may fashion an appropriate remedy, which may include, inter alia, the exercise of any and all rights or remedies available to the DIP Secured Parties under the Interim Order, the DIP Documents or applicable law against the DIP Collateral; provided that the rights of the Debtors to contest such relief are expressly preserved; provided, further, that in the event that a party challenges the applicable DIP Agent's assertion that a DIP Termination Event has occurred or has occurred and is continuing and the Court is unavailable for a hearing during the Remedies Notice Period, the automatic stay pursuant to Bankruptcy Code section 362 shall remain in effect as to all actions other than those expressly identified in paragraph 25 of the Interim Order until the Court has an opportunity to rule on such challenge (see Interim Order  $\P$  27);

## f. Section 506(c) and Section 552 Waivers.

i. Except to the extent of the Carve Out, no costs or expenses of administration of the Chapter 11 Cases or any future proceeding that may result therefrom, including liquidation in bankruptcy or other proceedings under the Bankruptcy Code, shall be charged against or recovered from the DIP Collateral pursuant to Bankruptcy Code section 506(c) or any similar principle of law, as such pertains to the DIP Secured Parties or the DIP Obligations without the prior written consent of the DIP Agent and no such consent shall be implied from any other action, inaction, or acquiescence by the DIP Agent or the DIP Lenders, and nothing contained in the Interim Order shall be deemed to be a consent by the DIP Agent or the DIP Lenders or the Prepetition Secured Parties to any charge, lien, assessment or claim against the DIP Collateral under Bankruptcy Code section 506(c) or

otherwise. Subject only to and effective upon entry of the Final Order, except to the extent of the Carve Out, no costs or expenses of administration of the Chapter 11 Cases or any future proceeding that may result therefrom, including liquidation in bankruptcy or other proceedings under the Bankruptcy Code, shall be charged against or recovered from the Prepetition Collateral (including Cash Collateral) pursuant to Bankruptcy Code section 506(c) or any similar principle of law, without the prior written consent of the Prepetition First Lien Agents, as the case may be, and no such consent shall be implied from any other action, inaction, or acquiescence by the Prepetition Secured Parties, and nothing contained in the Interim Order shall be deemed to be a consent by the Prepetition Secured Parties to any charge, lien, assessment or claim against the Prepetition Collateral under Bankruptcy Code section 506(c) or otherwise (see Interim Order ¶17); and

- ii. Subject only to and effective upon the entry of the Final Order, none of the DIP Collateral, the DIP Lenders, the DIP Agent, the Prepetition Collateral, the Prepetition First Lien Adequate Protection Liens or the Prepetition Secured Parties shall be subject to the equitable doctrine of "marshaling" or any other similar doctrine, and all proceeds thereof shall be received and used in accordance with the Interim Order. Further, subject only to and effective upon entry of the Final Order, in no event shall the "equities of the case" exception in Bankruptcy Code section 552(b) apply to the secured claims of the Prepetition Secured Parties ¶ 16);
- g. Limitations on the Use of Cash Collateral Other than General "Carve-Outs" to Pay Approved Fees and Expenses of Advisors to Official Committees or Future Trustees.
  - i. Notwithstanding anything herein or in any other order of this Court to the contrary, none of the DIP Facility, the Prepetition Collateral, any Cash Collateral or the Carve Out (other than the Investigation Budget (as defined herein)) may be used to (a) investigate, analyze, commence, prosecute, threaten, litigate, object to, contest, or challenge in any manner or raise any defenses to, the validity, perfection, priority, extent or enforceability of any amount due under the DIP Documents, the Existing Agreements or the liens or claims granted under the Interim Order, the DIP Documents or the Existing Agreements, including the Primed Liens, the Cash Collateral Liens and the DIP Liens, or any mortgage, security interest, or lien with respect thereto, or any other rights or interests or replacement liens with respect thereto or any other rights or interests of any of the DIP Agent, the other DIP Secured Parties, the Prepetition Term Loan Agent, the Prepetition ABL Secured Parties, or the Prepetition Secured Parties, (b) assert any Claims and

Defenses, including any Avoidance Actions, or any other causes of action against the DIP Agent, the DIP Lenders, the Prepetition Secured Parties or their respective agents, affiliates, subsidiaries, directors, officers, representatives, attorneys or advisors, (c) prevent, hinder, or otherwise delay the DIP Agent's or the Prepetition First Lien Agents' assertion, enforcement, or realization on the Prepetition Collateral or the DIP Collateral, in accordance with the DIP Documents, the Existing Agreements or the Interim Order, the exercise of rights by the DIP Agent or the Prepetition Secured Parties once an Event of Default has occurred and is continuing, or any other rights or interest of any of the DIP Agent, the DIP Lenders, the Prepetition ABL Secured Parties or the Prepetition Secured Parties following the occurrence of a DIP Termination Date and after the Remedies Notice Period, (d) seek to subordinate (other than to the Carve-Out or as set forth in the Interim Order) or recharacterize the DIP Obligations or any of the Prepetition First Lien Obligations, or to disallow or avoid any claim, mortgage, security interest, lien, or replacement lien or payment thereunder, (e) seek to modify any of the rights granted to the DIP Agent, the DIP Lenders, or any of the Prepetition First Lien Agents hereunder or under the DIP Documents or the Existing Agreements, in the case of each of the foregoing clauses (a) through (d), without such party's prior written consent, (e) pay any amount on account of any claims arising prior to the Petition Date unless such payments are approved by an order of this Court or otherwise permitted under the DIP Documents, (f) file any motion seeking approval of a sale of any DIP Collateral without the consent of the Required DIP Lenders, other than a sale that indefeasibly satisfies the DIP Obligations in full in cash, (g) challenge the Escrow Release or the Escrow Payment or seek to transfer any portion of the Escrow Cash to the Debtors' estates, or (h) pay Allowed Professional Fees, disbursements, costs or expenses incurred by any person, including, without limitation, the Creditors' Committee (if any), in connection with any of the foregoing; *provided*, that this paragraph (including, for the avoidance of doubt, the Investigation Budget) shall not limit (or be deemed to limit) the Loan Parties' rights to challenge the claims of any creditor who received payment pursuant to such repurchase, redemption or other satisfaction by any Debtor entity of the HoldCo Convertible Notes with the proceeds of the Tranche B-3 Term Loans or otherwise prior to the Petition Date. "Investigation Budget" means a cap of \$125,000 with respect to Allowed Professional Fees to be incurred by the Creditors' Committee under the investigation budget (see Interim Order ¶ 34);

h. **Priming Liens**. Pursuant to Bankruptcy Code section 364(d)(1), and subject and subordinate in all respects to the Carve Out, a valid, binding, continuing, enforceable, fully-perfected priming senior security interest in

and liens upon the Prepetition Collateral, which security interests and liens shall prime the Primed Liens in accordance with the priorities shown on Exhibit A attached to the Interim Order ¶ 10(c); and

- i. No Limitation on the Ability of Estate Fiduciaries to Fulfill their Duties.
  - i. The DIP Orders bind the Debtors, and subject to certain challenge rights, all parties in interest, with respect to the validity, perfection, and amount of the liens and debt of the DIP Secured Parties and Prepetition Secured Parties, and the waiver of the Debtors' claims against the DIP Secured Parties and Prepetition Secured Parties (*see* Interim Order, ¶ 31); and
  - ii. The DIP Orders impose limitations on the use of Cash Collateral to, among other things, investigate or finance the prosecution of claims and causes of action against the DIP Secured Parties, the Prepetition Secured Parties (each in their capacities as such), and other related parties related to the DIP Obligations, the Prepetition Obligations, the DIP Liens, or the Prepetition Liens, *provided*, *however*, that such limitations shall not apply to investigations of the Creditors' Committee (if any), in an aggregate amount not to exceed \$125,000 (see Interim Order ¶ 34(b)).
- 19. As required by Section C, paragraph 8 of the Complex Case Procedures, the inclusion of each of the applicable Significant Provisions in the DIP Term Loan Facility is appropriate and necessary to permit the Debtors' access to the DIP Term Loan Facility. The terms and conditions of each of the Significant Provisions included in the DIP Orders are the result of arm's-length and hard-fought negotiations between the Debtors and the DIP Lenders. The DIP Lenders are unwilling to provide the DIP Term Loan Facility absent the inclusion of these provisions and no other existing stakeholder or third party has presented a lower cost or otherwise better postpetition financing proposal. As set forth herein and in the DIP Declarations, granting the relief requested pursuant to the Interim Order is critical to the continued operation of the Debtors' businesses and will permit the Debtors to smoothly transition into these Chapter 11 Cases, preserving the value of the Debtors' estates for all parties in interest. The DIP Term Loan Facility is critical to the Debtors' continuing operations. In light of the foregoing, the Debtors submit that

the Significant Provisions in the DIP Orders are appropriate and necessary under the facts and circumstances of these Chapter 11 Cases and should be approved.

## **The Debtors' Prepetition Capital Structure**

20. Avaya's prepetition capital structure includes approximately \$3.4 billion<sup>10</sup> in funded debt as of the Petition Date, consisting of: (a) the Prepetition ABL Facility; (b) the Term Loan Credit Facility; (c) the Legacy Notes; (d) the Secured Exchangeable Notes; and (e) the HoldCo Convertible Notes. Avaya's prepetition indebtedness is also subject to two intercreditor agreements, generally referred to as (i) the ABL Intercreditor Agreement<sup>11</sup> and (ii) the First Lien Pari Intercreditor Agreement.<sup>12</sup> The ABL Intercreditor Agreement governs the relative contractual rights of lenders under the ABL Credit Facility, on the one hand, and the Term Loan Credit Facility, on the other hand and, pursuant to certain joinders, the relative contractual rights of Holders of Legacy Notes and Secured Exchangeable Notes. The First Lien Pari Intercreditor Agreement, in turn, governs the relative contractual rights of holders under the Legacy Notes and Secured Exchangeable Notes with respect to lenders under the Term Loan Credit Facility.

Not including approximately \$40 million of letters of credit outstanding under the ABL Credit Facility as of the Petition Date.

The "ABL Intercreditor Agreement" means that certain intercreditor agreement dated as of December 15, 2017 (as amended, restated, modified, and supplemented from time to time) by and among Avaya Inc., as borrower, HoldCo, Citibank, and Goldman Sachs.

The "<u>First Lien Pari Intercreditor Agreement</u>" means that certain intercreditor agreement dated as of September 25, 2020 (as amended, restated, modified, and supplemented from time to time) by and among Avaya Inc., HoldCo, Goldman Sachs, and Wilmington Trust, National Association.

21. The following table summarizes the Debtors prepetition funded indebtedness:

Indebtedness	Balance Outstanding (\$ millions) <sup>13</sup>
Prepetition ABL Facility <sup>14</sup>	\$56
B-1 Term Loans	\$800
B-2 Term Loans	\$743
B-3 Term Loans	\$350
Legacy Notes	\$1,000
Senior Exchangeable Secured Notes	\$250
HoldCo Convertible Notes	\$221
Total	\$3,420
Legacy Liabilities	Balance Outstanding (\$ millions) <sup>15</sup>
US Pension (underfunded liability)	\$111
International Pension (underfunded liability)	\$319
OPEB (underfunded liability)	\$115
Total	\$545

These obligations are discussed below:

# I. Prepetition ABL Facility.

- 22. Avaya Inc., as the parent borrower, HoldCo, as holdings, foreign borrowers, and Citibank, N.A., Inc., as administrative agent (in such capacity, the "<u>Prepetition ABL Agent</u>"), entered into that certain credit agreement, dated as of December 15, 2017, and amended and restated as of September 25, 2020 (as amended, modified, or supplemented and in effect immediately prior to the Petition Date, the "<u>Prepetition ABL Credit Agreement</u>" and the credit facilities thereunder, the "<u>Prepetition ABL Facility</u>").
- 23. Obligations under the Prepetition ABL Facility are secured by substantially all assets of the U.S. borrower and guarantors under the Prepetition ABL Credit Agreement, and solely with respect to the foreign tranche, by certain foreign subsidiaries, subject to certain

Denotes balance outstanding, and not including accrued interest amounts, as of February 12, 2023.

<sup>&</sup>lt;sup>14</sup> Excludes approximately \$40 million of LCs issued under the Prepetition ABL Facility.

Denotes balance outstanding as of September 30, 2022.

limitations and exclusions. Pursuant to the ABL Intercreditor Agreement, the Prepetition ABL Facility holds a first priority lien versus the Debtors' other debt obligations with respect to ABL Priority Collateral (as defined in the ABL Intercreditor Agreement) which includes, among other things, cash, accounts receivable, and inventory. The Prepetition ABL Facility holds a second priority position versus the Debtors' obligations with respect to term loan collateral (the "Prepetition Term Loan Priority Collateral"), generally consisting of assets, such as equipment, fixtures, and stock pledges, except to the extent constituting ABL Priority Collateral. In addition, the foreign tranche of the Prepetition ABL Facility is secured by certain assets of foreign borrowers and guarantors.

24. Under the Prepetition ABL Credit Agreement, consisting of a \$150 million U.S. tranche and a \$50 million foreign tranche, each borrower's ability to incur loans and letters of credit under each tranche is capped at the lesser of aggregate revolving credit commitments and the applicable borrowing base(s), which is calculated by applying advance rate to the eligible inventory, accounts receivable and cash and subtracting certain reserves. In order to be included in the borrowing base, accounts receivable and inventory must satisfy strict eligibility criteria. The Prepetition ABL Agent has substantial rights with respect to monitoring and auditing of the borrowing base, including by means of period field examinations, inventory appraisals, and is entitled to receipt of frequent borrowing base reports.

## II. Prepetition Term Loans.

25. Avaya Inc., as borrower, HoldCo, as holdings, Goldman Sachs Bank USA, as administrative agent and collateral agent (in such capacity, the "Prepetition Term Loan Agent"),

For the avoidance of doubt, the Prepetition ABL Facility is senior to the loans under the Prepetition Term Loan Credit Agreement.

and the lenders that are party thereto from time to time (the "<u>Prepetition Term Loan Lenders</u>") are parties to that certain first lien term loan credit agreement (as amended, modified, or supplemented and in effect immediately prior to the Petition Date, the "<u>Prepetition Term Loan Credit Agreement</u>," or the "<u>Prepetition Term Loan Credit Facility</u>").

- 26. The Prepetition Term Loan Credit Agreement provides three tranches of the Prepetition Term Loan Credit Facility. An aggregate principal amount of \$1.9 billion is outstanding as of the Petition Date under the Prepetition Term Loan Credit Agreement, consisting of approximately: (a) \$800 million outstanding in term B-1 loans maturing December 15, 2027; (b) \$743 million outstanding in term B-2 loans maturing December 15, 2027; and (c) as discussed in greater detail below, \$350 million outstanding in term B-3 loans maturing December 15, 2027 (the "B-3 Term Loans," and collectively, the "Prepetition Term Loan Credit Facility"). Obligations under the Prepetition Term Loan Credit Facility are secured by a first priority lien on substantially all the U.S. borrowers' and guarantors' assets, subject to certain limitations and exclusions. As noted above, obligations outstanding under the Prepetition Term Loan Credit Facility are senior in priority versus the Prepetition ABL Facility with respect to Prepetition Term Loan Priority Collateral and are junior in priority versus the Prepetition ABL Facility with respect to ABL Priority Collateral.
- 27. On July 12, 2022, Avaya Inc. entered into that certain Amendment No. 4 to the Prepetition Term Loan Credit Agreement pursuant to which Avaya incurred the B-3 Term Loans in an aggregate principal amount of \$350 million (the "Prepetition Term Loan Amendment"). Avaya used a portion of the proceeds from the B-3 Term Loans to make a distribution to HoldCo, which used the dividend proceeds to repurchase certain of the HoldCo Convertible Notes, with the remaining funds received from the B-3 Term Loans placed in an account maintained by Goldman

Sachs Bank USA (the "<u>Escrow Account</u>"), resulting in approximately \$221 million being held in the Escrow Account (the "<u>Escrow Cash</u>"). As of the Petition Date, the Escrow Cash remains in the Escrow Account.

## III. Legacy Notes.

28. Avaya Inc., as issuer, HoldCo and the Subsidiary Guarantors, as guarantors, and Wilmington Trust, National Association, as indenture trustee and collateral agent (the "Legacy Notes Trustee"), issued senior first lien notes (the "Legacy Notes") pursuant to that certain indenture dated September 25, 2020 (the "Legacy Notes Indenture"). The Legacy Notes mature on September 15, 2028, and approximately \$1 billion in principal amount remains outstanding as of the Petition Date. Obligations under the Legacy Notes are secured by substantially all assets of each Debtor other than Sierra, KnoahSoft, Inc., and CTIntegrations, LLC, subject to certain limitations and exclusions.

## IV. Secured Exchangeable Notes.

29. Avaya Inc., as issuer, HoldCo and the Subsidiary Guarantors, as guarantors, and Wilmington Trust, National Association, as trustee (the "Secured Exchangeable Notes Trustee"), issued exchangeable secured notes (the "Secured Exchangeable Notes") pursuant to that certain indenture dated July 12, 2022 (the "Secured Exchangeable Notes Indenture"). The Secured Exchangeable Notes mature on December 15, 2027, and approximately \$250 million in principal amount remains outstanding as of the Petition Date. The indenture and the Secured Exchangeable Notes provide, among other things, that the Secured Exchangeable Notes are Avaya Inc.'s senior secured obligations. Obligations under the Secured Exchangeable Notes are secured by substantially all assets of each Debtor other than Sierra, KnoahSoft, Inc., and CTIntegrations, LLC, subject to certain limitations and exclusions.

## V. HoldCo Convertible Notes.

30. HoldCo, as issuer, and The Bank of New York Mellon Trust Company N.A., as trustee (the "HoldCo Convertible Notes Trustee"), issued HoldCo convertible notes (the "HoldCo Convertible Notes") pursuant to that certain indenture dated June 11, 2018 (the "HoldCo Convertible Notes Indenture"). The HoldCo Convertible Notes mature on June 15, 2023, and approximately \$221 million in principal amount remains outstanding as of the Petition Date. Approximately \$129 million of HoldCo Convertible Notes were repurchased by HoldCo in July 2022. The indenture and the HoldCo Convertible Notes provide, among other things, that the HoldCo Convertible Notes are unsecured obligations of HoldCo only.

## VI. The Debtors' Legacy Liabilities.

#### A. U.S. Pension Liabilities.

31. The Debtors maintain one qualified defined benefit pension plan under the Internal Revenue Code (the "IRC") and the Employee Retirement Income Security Act of 1974 ("ERISA") for their Employees and Retirees. The Avaya Inc. Pension Plan (the "Qualified Pension Plan") currently provides pension benefits to approximately 325 active employees and approximately 4,100 retirees and other separated participants beneficiaries (collectively, and the "Pension Recipients"). Under the IRC and ERISA, the Debtors make minimum funding contributions to the Qualified Pension Plan based on annual actuarial calculations. In addition, the Qualified Pension Plan makes annual premium payments to the Pension Benefit Guaranty Corporation. The Pension Recipients receive approximately \$20 million per quarter under the Qualified Pension Plan (collectively, the "Pension Benefit Payments"), consisting of payments per-participant of approximately \$4,800. The Pension Benefit Payments are paid out of the corpus of trusts established for the plan (collectively, the "Pension Plan Trusts").

- 32. As part of its 2017 restructuring, Avaya Inc. entered into a settlement with the Pension Benefit Guaranty Corporation (the "PBGC") whereby Avaya Inc. and the PBGC agreed to, among other key terms: (a) the assumption of the hourly pension plan; (b) the termination of the pension plan for salaried employees; (c) certain excess contributions that would be triggered by the occurrence of future, material, transactions; and (d) the plan recovery for the PBGC (collectively, the "2017 PBGC Settlement").<sup>17</sup>
- 33. Over the past few months, the Company has re-engaged in discussions with the PBGC regarding the terms of the hourly pension plan. Prior to the Petition Date, after a series of negotiations and discussions, Avaya and the PBGC entered into an agreement (called the "2023 PBGC Settlement" in the Plan), effective upon the Debtors' emergence from chapter 11, that provides for the assumption of the hourly pension plan and the consensual termination of the 2017 PBGC Settlement, including the excess contribution obligations thereunder.

## VII. Intercreditor Agreements.

34. The ABL Intercreditor Agreement addresses the rights of the ABL Agent under the Prepetition ABL Credit Agreement and "Senior Secured Parties" vis-à-vis the Prepetition Term Loan Agent, the Legacy Notes Trustee, and the Secured Exchangeable Notes Trustee under the Term Loan Credit Agreement, Legacy Notes Indenture, and Secured Exchangeable Notes Indenture, respectively, for itself and on behalf of the lenders under the Prepetition Term Loan Credit Agreement and the Holders of Legacy Notes and Secured Exchangeable Notes, by virtue of the ABL Intercreditor Agreement, the Legacy Notes Indenture, and the Secured Exchangeable Notes Indenture, as noted above. Pursuant to Section 6.01 of the ABL Intercreditor Agreement,

See In re Avaya Inc., et al., Case No. 17-10089 (SMB) (Bankr. S.D.N.Y. Nov. 27, 2017) [Docket Nos. 1305, 1567].

the Term Priority Representative, <sup>18</sup> for itself and on behalf of the Term Loan Lenders, agreed not to object to a proposed debtor-in-possession financing or the use of cash collateral, and not to request or accept any adequate protection except: (1) to the extent permitted in Section 6.03, which provides for junior replacement liens and claims on additional collateral to the extent such liens and claims are granted to Senior Secured Intercreditor Parties; <sup>19</sup> and (2) in connection with the assertion of certain rights pursuant to section 2.03, which are inapplicable here. <sup>20</sup>

35. The First Lien Pari Intercreditor Agreement, in turn, addresses the Prepetition Term Loan Agent, as the authorized representative of the Prepetition Term Loan Lenders, the Legacy Notes Trustee, as the authorized representative for Holders of Legacy Notes, and the Secured Exchangeable Notes Trustee, as the authorized representative for holders of Secured Exchangeable Notes, on account of among other things, debtor in possession financing and the Debtors' use of the Cash Collateral. In particular, the First Lien Pari Intercreditor Agreement prohibits any "Pari Secured Party" from objecting to such financing encumbering their shared collateral, or the use of the Cash Collateral where such financing is not objected to by the administrative agent under the Term Loan Facility so long as holders of obligations under the Legacy Notes, the Secured Exchangeable Notes, and the Prepetition Term Loan Credit Agreement:

(a) will retain their liens on shared collateral, (b) are granted liens on any additional collateral

By definition, "<u>Term Loan Priority Representative</u>" includes Goldman Sachs and any additional trustee, administrative agent, collateral agent, security agent, or similar agent under the Term Loan Facility in an applicable joinder agreement.

By definition, "Senior Secured Intercreditor Parties" includes certain secured parties under the Prepetition Term Loan Credit Facility and the Prepetition ABL Facility.

Further, such Holders of Legacy Notes and Secured Exchangeable Notes agreed to subordinate their liens where, as here, the liens securing certain prepetition secured debt obligations are subordinated or *pari passu* with the liens securing debtor-in-possession borrowing.

By definition, "<u>Pari Secured Party</u>" includes Holders of Legacy Notes and Secured Exchangeable Notes and lenders under the Prepetition Term Loan Credit Agreement.

pledged in connection with postpetition financing, and (c) any periodic payments are applied in accordance with the First Lien Pari Intercreditor Agreement. The Debtors respectfully submit that such conditions are satisfied by the terms of the Debtors' proposed adequate protection package here.

## **The DIP Term Loan Facility**

# VIII. The Debtors' Need for Immediate Access to the DIP Term Loan Facility and Cash Collateral.

- 36. As described in the First Day Declaration, beginning in early fiscal 2022, Avaya's revenue began to decrease. The decrease was driven by the continued decline in the capex business and the delayed revenue generation from the cloud investments, which was compounded in the second half of 2022 by a deceleration in the growth of the Company's subscription revenue. These challenges, coupled with customers' related concerns regarding Avaya's financial health, have had a detrimental impact on Avaya's revenue and cash flow.
- 37. In the face of these headwinds, the Company's management commenced an extensive restructuring of their operations to reduce costs and improve the cash generation of the business. Additionally, with their advisors, the Company also assessed the need for contingency planning, and engaged in efforts to prepare the Debtors for a value-maximizing deleveraging transaction contemplated in these Chapter 11 Cases, pursuant to the RSA and the Plan. A key part of these efforts was to evaluate the Debtors' liquidity position and the need for postpetition financing, to enable the Debtors to effectuate the Company's operational restructuring and to fund this transaction through these Chapter 11 Cases.
- 38. The Debtors, with the assistance of their advisors, analyzed the quantum of capital needed and the potential forms that a financing and/or restructuring could take and determined that it would not be possible to administer these Chapter 11 Cases, pursue the transaction pursuant to

the RSA and the Plan, operate the Debtors' business in the ordinary course, and pay administrative costs during these cases solely through the use of Cash Collateral. Without immediate access to the DIP Term Loan Facility, the Debtors will be unable to meet their near-term working capital needs, stabilize their operations, fund the costs of administering these cases, and to continue to effectuate a value-maximizing operational restructuring commenced prepetition.

- 39. The Debtors have an urgent need for significant and immediate liquidity. The Debtors enter these Chapter 11 Cases with approximately \$45 million in cash on hand, which is gravely insufficient to meet the Debtors' liquidity needs both in the near-term and throughout these Chapter 11 Cases. Without immediate financing, the Debtors project that they will be unable to pay essential costs required to continue operating as a going concern, resulting in immediate and irreparable harm to the Debtors' business on a global scale. The Debtors require cash to, among other things, satisfy payroll obligations, honor obligations under their customer contracts, maintain insurance coverage, pay taxes, bolster the liquidity position of the NFAs, make other payments integral to the continued management, operation, and preservation of the Debtors' business, and provide an avenue to exit these Chapter 11 Cases expeditiously. Koza Decl. ¶ 10.
- 40. The Debtors' liquidity needs are particularly pressing in relation to ensuring the NFAs maintain sufficient and necessary liquidity to avoid value-destructive business interruptions. The NFAs operate more than 150 bank accounts and must maintain a certain level of global liquidity to sustain their international operations. Koza Decl. ¶ 11.
- 41. The NFAs also have immediate liquidity needs which need to be addressed. Prior to February 28, more than \$15 million in payroll obligations and \$7 million in pension obligations will need to be paid out of the Notional Cash Pool for the European entities. This would leave the cash pool with insufficient cash to support ongoing operational needs, outside of funding coming

from the Debtors to supplement their liquidity. The Initial Intercompany Transaction will provide additional liquidity for the NFAs to fund their operations.

- 42. The Debtors intend to use \$50 million of the DIP Term Loan Facility proceeds to fund the Initial Intercompany Transaction. Koza Decl. ¶ 3. These funds will be used to convert the existing international cash pooling system from a notional to a physical pool by eliminating certain negative balances; to satisfy certain payroll obligations; to fund certain pension and retirement obligations; and to otherwise fund the working capital needs of the NFAs. Koza Decl. ¶ 13.
- 43. There is an immediate and exigent need for the Debtors to use the proceeds of the DIP Term Loan Facility to ensure the viability of the Company's international business by funding the Initial Intercompany Transaction. Funding the NFAs immediately is both practical and beneficial to the administration of the Debtors' estates and the preservation of value. It is administratively efficient to fund the NFAs at the onset of these Chapter 11 Cases, when the DIP Lenders can be apprised of the uses for and can consent to a one-time outlay, rather than forcing the Debtors to make piecemeal requests for funding and eliminating their flexibility to respond to financial and market conditions as they arise. Further, the Initial Intercompany Transaction will allow the Company's international operations to run smoothly and bolster confidence in the market, benefiting the Debtors' estate by minimizing vendor contraction and ensuring that the Company continues business in the ordinary course. Koza Decl. ¶ 13.
- 44. In addition, a substantial number of NFAs are parties to the Cash Pooling Agreement,<sup>22</sup> which memorializes a cash management arrangement among the Cash Pooling Bank

<sup>&</sup>quot;Cash Pooling Agreement" means that certain cash pooling agreement between Citibank N.A., as cash pooling bank (the "Cash Pooling Bank"), AISL, as agent, and various Avaya entities as signatories to the Cash Pooling Agreement via separate Accession Agreement (as defined therein) (such entities, collectively with AISL, the "Group Companies"), dated March 20, 2002. The Group Companies include: (a) Avaya UK; (b) Avaya

and the Group Companies (which are comprised entirely of non-Debtor entities), whereby debit and credit balances of the Group Companies' cash pool accounts of the same currency are consolidated and notionally treated by the Cash Pooling Bank as an aggregate amount on deposit, allowing individual Group Companies to maintain positive and negative individual balances at various times (the "Cash Pooling Arrangement"). Among other benefits, the Cash Pooling Arrangement allows the Group Companies to control their receipts and disbursements and to manage liquidity on an intercompany basis without having to create intercompany notes or otherwise pay receivables.

45. On March 31, 2023, the Cash Pooling Arrangement will transition from notional balances to physical balances, necessitating the actual transfer of funds among the Group Companies to reconcile certain negative balances. Under the physical cash pool structure, each NFA bank account must maintain a positive balance, unlike the current notional cash pool system where only a positive balance must be maintained in the aggregate among all accounts. Further, the NFAs are required to maintain a certain minimum liquidity to comply with applicable jurisdictional requirements and to provide comfort to the management of the NFAs. This creates the potential for sudden and uncertain shifts in liquidity need as the cash pool structure conversion commences. Thus, rebalancing these accounts at the outset of these Chapter 11 Cases to ensure each account is properly funded and in compliance with foreign requirements is in the best interests of the Debtors and their estates insofar as minimizing the risk of any NFA entering into a foreign

Switzerland GMBH; (c) Avaya Communicaiones Espana S.L.U.; (d) Avaya Singapore Pte Ltd.; (e) Avaya Nederland BV; (f) Kontfel A.B.; (g) Avaya Japan Ltd.; (h) Avaya Italia SpA; (i) Avaya Hong Kong Ltd.; (j) Avaya Germany GmbH; (k) Avaya France S.A.S.; (l) Avaya ECS Ltd.; (m) Avaya Canada Corporation; (n) Avaya Belgium SPRL BVBA; (o) Avaya Austria GmbH; (p) Avaya Australia Pty. Ltd.; (q) Avaya International Enterprises Ltd.; and (r) Avaya Cyprus Ltd.

insolvency proceeding that would harm the enterprise as a whole and destroy value that would otherwise inure to the Debtors' estates. Koza Decl. ¶ 11, 13.

- 46. Further, there is an immediate need to cover certain NFAs' payroll and related benefit costs as well as other obligations arising from the enterprise-wide cost-saving initiative aimed at optimizing the Company's operational and cost structure. Ensuring that sufficient cash is available to make such payments, including severance, when they come due would minimize possible angst in the workforce, including the reduced portion thereof. Timely payment will also avoid any penalties or other issues arising from commitments to local works councils, that could result in a disruption in the Company's cost optimization initiatives. This cash infusion will mitigate the potential disruption to the Company's global operations while optimizing the overall allocation of foreign liquidity and ensuring continued ordinary-course operations. Koza Decl. ¶ 16.
- 47. The Foreign Reserve Account is also an integral component of stabilizing the Company's global operations. To provide further assurance that the NFAs will have sufficient liquidity during the Debtors' Chapter 11 Cases, the Debtors intend to fund a segregated bank account with \$40 million that would remain with the Debtors during the Chapter 11 Cases and only be accessed upon the consent of certain DIP Lenders (pursuant to the Foreign Reserve Protocol), if necessary, to provide additional liquidity to stabilize the operations of the NFAs. Koza Decl. ¶ 18. While the Debtors do not anticipate needing to draw from the Foreign Reserve Account, its existence allows the Company to operate its global business in an uninterrupted fashion, maximizing the Debtors' value for the benefit of their stakeholders. Koza Decl. ¶ 18.
- 48. In light of these circumstances, the Debtors submit that the Initial Intercompany Transaction and the funding of the Foreign Reserve Account is justified and essential for the

Debtors to seamlessly transition into chapter 11, continue ordinary course operations uninterrupted, and preserve their market share, the reputation of their businesses, and the loyalty and goodwill of their employees.

- 49. The size of the DIP Term Loan Facility and the amount requested on an interim basis has been determined based on a thorough analysis conducted by the Debtors' management team and its advisors. The amount was derived from a cash-flow projection developed from an analysis of the Debtors' projected receipts and disbursements during these cases, reflected in the Initial DIP Budget, and discussions with the Debtors' management team. The Debtors submit that the Budget presents a reasonable estimate of the Debtors' cash sources and needs during these Chapter 11 Cases and, given these estimates, that the DIP Term Loan Facility will provide the Debtors sufficient liquidity to stabilize their operations and fund the administration of these Chapter 11 Cases. Koza Decl. ¶ 19.
- 50. Specifically, the DIP Term Loan Facility will allow the Debtors to: (a) continue satisfying obligations to the Debtors' contract counterparties; (b) provide the liquidity necessary to honor amounts owed to ordinary course claimants and to continue favorable trade terms with such claimants; (c) fund the Initial Intercompany Transaction and the Foreign Reserve Account to bolster the liquidity of the NFAs and ensure continued stability of the Debtors' global business; (d) reassure other stakeholders, including channel partners, landlords, customers, and employees, that the Debtors have sufficient funds to continue operating in the ordinary course; (e) fund the Debtors' payroll obligations; (f) fund the administrative costs of these Chapter 11 Cases; (g) cash collateralize existing letters of credit and refinance the Prepetition ABL Facility; and (h) solidify access to liquidity upon emergence from these Chapter 11 Cases in the form of exit financing. Koza Decl. ¶ 20–21.

- 51. The Prepetition Secured Parties have also agreed to provide the Debtors with immediate access to the use of Cash Collateral on a consensual basis, subject to the terms and conditions of the DIP Term Loan Facility Documents and the DIP Orders. Immediate access to Cash Collateral will: (a) ensure that the Debtors have sufficient working capital to, among other things, pay their employees, vendors, landlords, and service providers; (b) enable the Debtors to honor their prepetition obligations under and in accordance with the proposed "first-day" relief if approved by the Bankruptcy Court; and (c) satisfy the administrative expenses of these Chapter 11 Cases. The ability to immediately use Cash Collateral also ensures that the Debtors avoid unnecessary, value-destructive operational disruptions that would otherwise be costly and potentially damaging to the business. The Debtors submit that utilizing Cash Collateral is fundamental to the preservation and maintenance of the Debtors' going-concern value during these Chapter 11 Cases, is critical for the Debtors' successful reorganization, and is in the best interests of the Debtors and their estates. Koza Decl. ¶ 22.
- 52. Absent funds available from the DIP Term Loan Facility and access to Cash Collateral, the Debtors could face a value-destructive interruption to their business and lose support from important stakeholders on which the Debtors' business depends, including channel partners, employees, customers, vendors, and other contract counterparties. This, in turn, would hinder the Debtors' ability to maximize the value of their estates, and the Debtors would be forced to curtail their operations significantly, if not entirely, to the detriment of all stakeholders. Koza Decl. ¶ 23.
- 53. Further, without access to the DIP Term Loan Facility, the Debtors will have extremely limited cash on hand, and based on the Debtors' liquidity forecast, would not be able to generate sufficient levels of operating cash to cover their working capital needs and the costs of

these Chapter 11 Cases. As a result, the Debtors submit that the DIP Term Loan Facility is critical to the Debtors' ability to administer these Chapter 11 Cases and provide the Debtors with sufficient liquidity to continue operations in the ordinary course and pursue the restructuring contemplated by the Plan and the RSA. Koza Decl. ¶ 9.

54. In light of the Debtors' immediate liquidity needs, the Debtors submit that they will materially benefit from the availability of the DIP Term Loan Facility. This financing, together with the RSA and the Plan, will provide the necessary liquidity for a successful reorganization of the Debtors' estates, which in turn will maximize value for the benefit of their creditors, employees, vendors, and other stakeholders. In addition, the DIP Term Loan Facility, and the assurance it provides to the Debtors' workforce and operations, will help stabilize the Debtors' operations at the outset of these Chapter 11 Cases and will provide the clearest path to an expeditious exit from chapter 11. Accordingly, the relief requested is necessary and appropriate to avoid immediate and irreparable harm to the Debtors' estates.

## IX. The Debtors' Proposed Adequate Protection Is Fair and Reasonable.

55. The adequate protection provided in the Interim Order (the "Adequate Protection") is fair and reasonable under the circumstances. The Adequate Protection package proposed for the Prepetition Secured Parties includes (a) Prepetition First Lien 507(b) Claims, (b) Prepetition First Lien Adequate Protection Liens, (c) delivery of certain reports and notices pursuant to certain sections of DIP Term Loan Credit Agreement, and (d) Adequate Protection Fees and Expenses, each as defined in the Interim Order and, critically, does not include current cash pay interest or payment of accrued but unpaid prepetition interest. Taken together, the cumulative effect of all of these provisions is to provide reasonable and appropriate adequate protection for the Prepetition Secured Parties. The proposed Adequate Protection is sufficient to protect the Prepetition First Lien Secured Parties from any diminution in value to the Prepetition Collateral, including Cash

Collateral. Moreover, the Required DIP Lenders have consented to the proposed Adequate Protection. In light of the foregoing, the proposed Adequate Protection to be provided for the benefit of the Prepetition Secured Parties is appropriate.

56. Access to the Prepetition Secured Parties' Cash Collateral will provide liquidity that is vital for the Debtors to maintain operations, pay employees, and fund these Chapter 11 Cases. Thus, the Debtors' provision of the Adequate Protection is not only necessary to protect against any diminution in value but is fair and appropriate under the circumstances of these Chapter 11 Cases to ensure the Debtors are able to continue using Cash Collateral for the benefit of all parties in interest and their estates.

#### III. Alternative Sources of Financing Are Not Available on Better Terms.

- 57. To evaluate alternatives to the proposed DIP Term Loan Facility, Evercore initiated a marketing process for postpetition financing. As part of this process, Evercore solicited interest from more than twenty sources of financing outside of the Debtors' capital structure. The potential third-party lenders Evercore contacted included various institutions that routinely provide postpetition financing, including both well-known commercial banks and specialty lenders. From that group of potential third-party lenders, nine executed confidentiality agreements and received access to non-public information. After evaluating the options, the Company eventually determined that all ABL proposals were either not actionable or inferior to the current DIP Financing, in part due to their higher contemplated cost of capital. Shah Decl. ¶ 15.
- 58. As a result, it became clear to the Debtors that their best path to financing their Chapter 11 Cases was through financing facilities from their existing lenders. As discussed above, Evercore solicited a DIP ABL proposal from Citibank and a new money term loan proposal from the Ad Hoc Groups. *See* Shah Decl. ¶ 16–18. Over the course of multiple weeks, the Debtors and their advisors engaged in various conversations and extensive negotiations with Citibank and the

Ad Hoc Groups to achieve the best possible terms for the DIP Term Loan Facility and DIP ABL Facility. *See* Shah Decl. ¶ 16–18.

## IV. The DIP Term Loan Facility Is Necessary to Preserve the Value of the Debtors' Estates.

Debtors will likely need to curtail their operations significantly. Koza Decl. ¶ 23. Even if the Debtors had the ability to access Cash Collateral, cash on hand and expected inflows would not be sufficient to fund their operations as a going concern in the near term. *See* Koza Decl. ¶ 10. Without a new source of liquidity, the Debtors' businesses may be irreparably harmed by their inability to maintain the minimum liquidity necessary to fund certain operating and business expenses of both the Debtors and certain foreign non-Debtor affiliates and provide a backstop for intercompany liquidity throughout the duration of the Debtors' Chapter 11 Cases. Consequently, the DIP Term Loan Facility is necessary to preserve the value of the Debtors' estates.

## **Basis for Relief**

- I. The Debtors Should Be Authorized to Obtain Postpetition Financing Through the DIP Term Loan Facility Documents.
  - A. Entry into the DIP Term Loan Facility Documents Is an Exercise of the Debtors' Sound Business Judgment.
- 60. The Court should authorize the Debtors, as an exercise of their sound business judgment, to enter into the DIP Term Loan Facility Documents, obtain access to the DIP Term Loan Facility, and continue using Cash Collateral. Section 364 of the Bankruptcy Code authorizes a debtor to obtain secured or superpriority financing under certain circumstances discussed in detail below. Courts grant a debtor in possession considerable deference in acting in accordance with its business judgment in obtaining postpetition secured credit, so long as the agreement to obtain such credit does not run afoul of the provisions of, and policies underlying, the Bankruptcy

Code. *See, e.g., In re N. Bay Gen. Hosp., Inc.*, No. 08-20368 (Bankr. S.D. Tex. July 11, 2008) (order approving postpetition financing on an interim basis as exercise of debtors' business judgment); *In re Trans World Airlines, Inc.*, 163 B.R. 964, 974 (Bankr. D. Del. 1994) (approving a postpetition loan and receivables facility because such facility "reflect[ed] sound and prudent business judgment"); *In re L.A. Dodgers LLC*, 457 B.R. 308, 313 (Bankr. D. Del. 2011) ("[C]ourts will almost always defer to the business judgment of a debtor in the selection of the lender."); *In re Ames Dep't Stores, Inc.*, 115 B.R. 34, 40 (Bankr. S.D.N.Y. 1990) ("[C]ases consistently reflect that the court's discretion under section 364 is to be utilized on grounds that permit reasonable business judgment to be exercised so long as the financing agreement does not contain terms that leverage the bankruptcy process and powers or its purpose is not so much to benefit the estate as it is to benefit a party in interest.").

- 61. To determine whether the business judgment standard is met, a court need only "examine whether a reasonable business person would make a similar decision under similar circumstances." *In re Exide Techs.*, 340 B.R. 222, 239 (Bankr. D. Del. 2006); *see also In re Curlew Valley Assocs.*, 14 B.R. 506, 513–14 (Bankr. D. Utah 1981) (noting that courts should not second guess a debtor's business decision when that decision involves "a business judgment made in good faith, upon a reasonable basis, and within the scope of the debtor's authority under the [Bankruptcy] Code").
- 62. In considering whether the terms of postpetition financing are fair and reasonable, courts consider the terms in light of the relative circumstances of both the debtor and the potential lender. *See In re Farmland Indus., Inc.*, 294 B.R. 855, 886 (Bankr. W.D. Mo. 2003) (while many of the terms favored the DIP lenders, "taken in context, and considering the relative circumstances of the parties," the court found them to be reasonable); *see also Unsecured Creditors' Comm.*

Mobil Oil Corp. v. First Nat'l Bank & Trust Co. (In re Elingsen McLean Oil Co., Inc.), 65 B.R. 358, 365 n.7 (W.D. Mich. 1986) (recognizing a debtor may have to enter into "hard bargains" to acquire funds for its reorganization).

- 63. The Debtors' determination to move forward with the DIP Term Loan Facility is an exercise of their sound business judgment following an arm's-length process and careful evaluation of available alternatives. Specifically, the Debtors and their advisors determined that the Debtors would require significant postpetition financing to support their operational and chapter 11 activities. The Debtors and their advisors determined that the DIP Term Loan Facility provides financing on more favorable economic terms than other alternatives. The Debtors and their advisors also determined that the DIP Term Loan Facility not only provides certainty with respect to the capital necessary for the administration of these Chapter 11 Cases and the Debtors' ongoing business operations but also to successfully emerge from these Chapter 11 Cases through the access to the Exit Facility, which will provide liquidity upon emergence.
- 64. The DIP Term Loan Facility will allow the Debtors to fund payroll obligations discussed above and administrative cost of these Chapter 11 Cases and provide a path to emergence by allowing the Debtors to implement the restructuring contemplated by the RSA and the Plan. The Debtors negotiated the DIP Term Loan Facility and other DIP Term Loan Facility Documents with the DIP Lenders in good faith, at arm's length, and with the assistance of their respective advisors, and the Debtors believe that they have obtained the best financing available under the circumstances. Accordingly, the Court should authorize the Debtors' entry into the DIP Term Loan Credit Agreement, as it is a reasonable exercise of the Debtors' business judgment.

- B. The Prepetition ABL Repayment Contemplated in the Interim Order Is Appropriate Under the Circumstances.
- 65. The proposed Prepetition ABL Facility Refinancing upon entry of the Interim Order is also an exercise of the Debtors' sound business judgment. Section 363(b) of the Bankruptcy Code permits a debtor to use, sell, or lease property, other than in the ordinary course of business, with court approval. Courts in the Fifth Circuit have recognized that it is appropriate to authorize the payment of prepetition obligations where necessary to protect and preserve the estate, including an operating business's going-concern value. *See, e.g., In re CoServ, L.L.C.*, 273 B.R. 487, 497 (Bankr. N.D. Tex. 2002) (authorizing payment of certain prepetition claims pursuant to "doctrine of necessity"); *In re Equalnet Commc'ns Corp.*, 258 B.R. 368, 369–70 (Bankr. S.D. Tex. 2000) (business transactions critical to the survival of the business of the debtor are exceptions to the general rule of nonpayment of prepetition claims prior to plan confirmation). The business judgment rule shields a debtor's management from judicial second-guessing. *In re Johns-Manville Corp.*, 60 B.R. 612, 615–16 (Bankr. S.D.N.Y. 1986) ("[T]he [Bankruptcy] Code favors the continued operation of a business by a debtor and a presumption of reasonableness attaches to a debtor's management decisions.").
- 66. Upon entry of the Interim Order, the Prepetition ABL Facility will be repaid in full with proceeds of the DIP Term Loan Facility and approximately \$40 million of letters of credit issued under the Prepetition ABL Facility will be cash collateralized. Such refinancing is necessary to allow the Debtors to take full advantage of the favorable economic terms offered by the DIP Term Loan Facility while also de-risking the DIP ABL Facility, thereby maximizing value for all stakeholders.
- 67. Notably, the Debtors do not have the ability to borrow additional amounts under the Prepetition ABL Credit Facility following the Petition Date. Further, as described in more

detail in the First Day Declaration, the outstanding obligations under the Prepetition ABL Credit Facility are oversecured based on the value of the ABL Priority Collateral, as reflected in the significant borrowing base cushion from which the Prepetition ABL Facility benefits. Shah Decl. ¶ 30. Where, as here, a prepetition secured creditor is oversecured by their collateral and therefore stands to receive payment in full, repaying such creditor with the proceeds of postpetition loans will not harm the Debtors' estates and other creditors. Indeed, Prepetition ABL Repayment will merely accelerate the satisfaction of the prepetition obligations thereunder without affecting recovery to other creditors because these prepetition obligations are fully secured by first-priority liens on prepetition collateral with a value in excess of the outstanding obligations, and therefore would have received a full recovery upon emergence. Absent the ABL Credit Facility Refinancing, interest and fees would accrue on the drawn amount and outstanding letters of credit at the default rate, and such interest would have to be paid in connection of the ultimate paydown of the Prepetition ABL Credit Facility. Shah Decl. ¶ 30. Accordingly, Prepetition ABL Repayment is appropriate and does not prejudice any creditors.

- 68. Moreover, without the Prepetition ABL Repayment, the Debtors would not be able to use their cash, or the cash generated by the collection of prepetition accounts receivables and sale of prepetition inventory, without the consent of the Prepetition ABL Lenders or a showing of adequate protection. The Debtors do not have sufficient unencumbered assets to assure that they can adequately protect the Prepetition ABL Facility for the use of their cash collateral. Accordingly, by refinancing the Prepetition ABL Facility now, the Debtors are assured of the use of cash collateral.
- 69. Finally, Prepetition ABL Repayment is a critical component of the structure of the DIP Term Loan Facility negotiated with the DIP Lenders, who specifically negotiated for such

refinancing in the context of their commitment or willingness to provide the DIP Term Loan Facility. The Debtors and the DIP Lenders engaged in arm's-length negotiations and ultimately agreed on Prepetition ABL Repayment in order to ensure access to the liquidity generated by the collateral securing the Prepetition ABL Credit Facility without the use of cash collateral being contested. Shah Decl. ¶ 31. Specifically, Prepetition ABL Repayment will provide continuity with counterparties holding the letters and allow the Company to secure new letters of credit, if required. Additionally, letters of credit can be issued at a lower cost using the DIP Term Loan Facility than if they were cash collateralized under a new facility. Koza Decl. ¶ 21. Indeed, the refinancing of the Prepetition ABL Credit Facility is a condition of the DIP Term Loan Facility and was critical to reaching an agreement with the Prepetition Secured Parties regarding the consensual use of Cash Collateral and the terms of the global restructuring transaction embodied in the RSA and Plan. Shah Decl. ¶ 31. Absent the DIP Term Loan Facility, the Debtors' ability to continue operating as a going concern will be jeopardized to the detriment of all parties in interest.

70. For the foregoing reasons, Prepetition ABL Repayment is a sound exercise of the Debtors' business judgment, reasonable, and should be approved.

## C. Funding the NFAs Is in the Estates' Best Interests.

71. The Debtors request approval to fund certain NFAs in the amount of \$50 million through the Initial Intercompany Transaction and to fund \$40 million of the proceeds of the DIP Term Loan Facility into the Foreign Reserve Account for purposes of providing and backstopping the liquidity of certain of the Debtors' NFAs to the extent necessary to preserve the value of the Debtors' international business and sustain operations critical to the Debtors' success on a goforward basis. The Debtors also seek authority to make disbursements from the

Foreign Reserve Account in their discretion, subject only to the Foreign Reserve Protocol and Intercompany Transfer Mechanic.

- 72. The DIP Financing provides for the funding of the Initial Intercompany Transaction with proceeds of the DIP Term Loan Facility upon the entry of the Interim Order. The Initial Intercompany Transaction will provide necessary upfront liquidity for certain NFAs to fund obligations vital to maintaining their operations and businesses. Such obligations include foreign payroll and other general business expenses. Additionally, the Debtors anticipate that the infusion of liquidity provided by the Initial Intercompany Transaction will be necessary to sustain certain NFAs throughout the duration of these Chapter 11 Cases, mitigating against the liquidity of such NFAs decreasing below the minimum liquidity thresholds required to operate in certain foreign jurisdictions. By undertaking the Initial Intercompany Transaction, the Debtors will be able to maintain their ordinary course relationships with their NFAs and avoid significant disruption to the Debtors' operations during the pendency of these Chapter 11 Cases.
- 73. Further, the DIP Financing provides for the Foreign Reserve Account to be funded with proceeds of the DIP Term Loan Facility. Funds in the Foreign Reserve Account will be available to the Debtors for disbursement to NFAs, on an as-needed basis (subject to the Foreign Reserve Protocol and Intercompany Transfer Mechanic, as described below) following a written request by the requesting NFAs which indicates the amount and purpose of the funds.
- 74. The Intercompany Transfer Mechanic provides that, except where prohibited by local law, any credit support, guarantee, cash or cash equivalents transferred by any Debtor to or for the benefit of an NFA (except for repayments of intercompany liabilities incurred in the ordinary course of business), shall be evidenced by an intercompany unsecured note. Such notes

or ledgers, as applicable, are pledged as security for the intercompany loan from Avaya Inc. to Sierra, which, in turn, is pledged as security for the DIP Term Loan Facility.

- As more fully described above and in the First Day Declaration, the Debtors operate as a single global enterprise with over 100 NFAs. The operations of the NFAs are critical to the overall value of the Debtors' enterprise. In order for these NFAs to avoid their own insolvency filings while continuing to generate revenues that increase the value of the Debtors' estates, it is imperative that these entities and their directors are assured that they will have access to capital in the event of a funding shortfall or grave need for liquidity. Due to the prospect of insolvency-related civil or criminal liability in certain jurisdictions, certain of the NFAs could be likely to initiate insolvency proceedings without such assurances. In that regard, the Initial Intercompany Transaction and the Foreign Reserve Account provide vital stability and flexibility to manage the liquidity of the NFAs, in the Debtors' business judgment.
- Recognizing the global nature of many debtors' businesses, courts in this district and elsewhere often grant relief designed to preserve the value of a debtor's foreign affiliates in order to benefit the debtor. *See, e.g., In re Cineworld Grp. Plc*, No. 22-90168 (MI) (Bankr. S.D. Tex. Sept. 8, 2022) [Docket No. 173]; *In re Talen Energy Supply, LLC*, No. 22-90054 (MI) (Bankr. S.D. Tex. May 11, 2022) [Docket No. 127]; *see also, e.g., In re Lumileds Holding B.V.*, No. 22-11155 (LGB) (Bankr. S.D.N.Y. Aug. 31, 2022) [Docket No. 73]; *In re Grupo Aeroméxico, S.A.B. de C.V.*, No. 20-11563 (SCC) (Bankr. S.D.N.Y. Aug. 21, 2020) [Dkt. No. 318]; *In re Aegean Marine Petroleum Network Inc.*, No. 18-13374 (MEW) (Bankr. S.D.N.Y. Nov. 9, 2018) [Docket No. 51].
- 77. Absent funding the Initial Intercompany Transaction and the Foreign Reserve Account contemplated under the DIP Term Loan Facility, the Debtors' ability to

continue operating as a going concern and maintain necessary liquidity available for certain NFAs will be jeopardized to the detriment of all parties in interest. Thus, the Debtors respectfully submit that the use of estate property to backstop NFAs through the Initial Intercompany Transaction and the Foreign Reserve Account is within the Debtors' business judgment, reasonable, and should be approved.

## D. The Debtors Should Be Authorized to Grant Liens and Superpriority Claims to the DIP Lenders.

- 78. The Debtors propose to obtain financing under the DIP Term Loan Facility by providing security interests and liens as set forth in the DIP Term Loan Facility Documents pursuant to section 364(c) of the Bankruptcy Code. Specifically, the Debtors propose to provide to the DIP Lenders postpetition security interest in and liens on the DIP Collateral (as defined in the Interim Order) and Prepetition Collateral that are valid, perfected, allowed, enforceable, non-avoidable and not subject to challenge, dispute or subordination immediately upon entry of the Interim Order.
- 79. The statutory requirement for obtaining postpetition credit under section 364(c) of the Bankruptcy Code is a finding, made after notice and hearing, that a debtor is "unable to obtain unsecured credit allowable under Section 503(b)(1) of [the Bankruptcy Code]." 11 U.S.C. § 364(c). See In re Crouse Grp., Inc., 71 B.R. 544, 549 (Bankr. E.D. Pa. 1987) (secured credit under section 364(c) of the Bankruptcy Code is authorized, after notice and hearing, upon showing that unsecured credit cannot be obtained). Courts have articulated a three-part test to determine whether a debtor is entitled to financing under section 364(c) of the Bankruptcy Code. Specifically, courts look to whether:
  - a. the debtor is unable to obtain unsecured credit under section 364(b) of the Bankruptcy Code, *i.e.*, by allowing a lender only an administrative claim;

- b. the credit transaction is necessary to preserve the assets of the estate; and
- c. the terms of the transaction are fair, reasonable, and adequate, given the circumstances of the debtor-borrower and proposed lenders.

See In re Ames Dep't Stores, 115 B.R. 34, 37–40 (Bankr. S.D.N.Y. 1990); see also In re St. Mary Hosp., 86 B.R. 393, 401–02 (Bankr. E.D. Pa. 1988); Crouse Grp., 71 B.R. at 549.

- 80. The Debtors meet each part of this test. As described above, no lenders were willing to provide sufficient postpetition financing on a junior lien or an unsecured or administrative priority basis. *See* Shah Decl. ¶ 16. The DIP Lenders will not fund the DIP Term Loan Facility on any other terms, and no other existing stakeholder or third party has presented a higher or otherwise better debtor-in-possession financing proposal. Absent the DIP Term Loan Facility, which will provide certainty that the Debtors will have sufficient liquidity to administer these Chapter 11 Cases, and comfort to their employees and vendor constituencies that business will continue in the ordinary course, the value of the Debtors' estates would be impaired to the detriment of all stakeholders. Given the Debtors' circumstances, the Debtors believe that the terms of the DIP Term Loan Facility, as set forth in the DIP Term Loan Facility Documents are fair, reasonable, and meet the standard for obtaining postpetition financing.
- 81. In the event that a debtor is unable to obtain unsecured credit allowable as an administrative expense under section 503(b)(1) of the Bankruptcy Code, section 364(c) of the Bankruptcy Code provides that a court "may authorize the obtaining of credit or the incurring of debt (a) with priority over any or all administrative expenses of the kind specified in section 503(b) or 507(b) of [the Bankruptcy Code]; (b) secured by a lien on property of the estate that is not otherwise subject to a lien; or (c) secured by a junior lien on property of the estate that is subject to a lien." 11 U.S.C. § 364(c). As described above, the Debtors are unable to obtain unsecured

credit. Therefore, approving (a) a superpriority claim in favor of the DIP Lenders, (b) liens in favor of the DIP Lenders on unencumbered property of the estate, and (c) junior liens in favor of the DIP Lenders on encumbered property of the estate is reasonable and appropriate.

- 82. Section 364(d) of the Bankruptcy Code provides that a debtor may obtain credit secured by a senior or equal lien on property of the estate already subject to a lien, after notice and a hearing, where the debtor is "unable to obtain such credit otherwise" and "there is adequate protection of the interest of the holder of the lien on the property of the estate on which such senior or equal lien is proposed to be granted." 11 U.S.C. § 364(d)(1). The Debtors may incur "priming" liens under the DIP Term Loan Facility if either (a) the Prepetition Secured Parties have consented or (b) the Prepetition Secured Parties' interest in collateral are adequately protected. *See Anchor Savs. Bank FSB v. Sky Valley, Inc.*, 99 B.R. 117, 122 (N.D. Ga. 1989) ("[B]y tacitly consenting to the superpriority lien, those [undersecured] creditors relieved the debtor of having to demonstrate that they were adequately protected.").
- 83. Further, section 364(d) provides that a debtor may obtain credit secured by a senior or equal lien on property of the estate already subject to a lien where the debtor is "unable to obtain such credit otherwise" and "there is adequate protection of the interest of the holder of the lien on the property of the estate on which such senior or equal lien is proposed to be granted." 11 U.S.C. § 364(d)(1). Accordingly, the Debtors may incur "priming" liens under the Facility if they are unable to obtain unsecured or junior secured credit and either (a) the Prepetition Secured Parties have consented or (b) the Prepetition Secured Parties' interests in collateral are adequately protected.
- 84. Here, the Ad Hoc Groups, representing over 90% of the First Lien Claims, have affirmatively consented to the DIP Term Loan Facility and Adequate Protection and actively

participated in facilitating the proposed DIP Term Loan Facility. As set forth more fully herein and in the Interim Order, the Debtors propose to provide a variety of adequate protection to protect the interests of the Prepetition Secured Parties. Therefore, the relief requested pursuant to section 364(d)(1) of the Bankruptcy Code is appropriate.

- E. No Comparable Alternative to the DIP Term Loan Facility Is Reasonably Available on More Favorable Overall Terms.
- 85. A debtor need only demonstrate "by a good faith effort that credit was not available without" the protections afforded to potential lenders by sections 364(c) of the Bankruptcy Code. *In re Snowshoe Co., Inc.*, 789 F.2d 1085, 1088 (4th Cir. 1986); *see also In re Plabell Rubber Prods., Inc.*, 137 B.R. 897, 900 (Bankr. N.D. Ohio 1992). In circumstances where only a few lenders likely can or will extend the necessary credit to a debtor, "it would be unrealistic and unnecessary to require [the debtor] to conduct such an exhaustive search for financing." *In re Sky Valley, Inc.*, 100 B.R. 107, 113 (Bankr. N.D. Ga. 1988), *aff'd sub nom. Anchor Sav. Bank FSB v. Sky Valley, Inc.*, 99 B.R. 117, 120 n.4 (N.D. Ga. 1989); *see also In re Snowshoe Co.*, 789 F.2d 1085, 1088 (4th Cir. 1986) (demonstrating that credit was unavailable absent the senior lien by establishment of unsuccessful contact with other financial institutions in the geographic area); *In re Stanley Hotel, Inc.*, 15 B.R. 660, 663 (D. Colo. 1981) (finding that two national banks refused to grant unsecured loans was sufficient to support conclusion that section 364 requirement was met); *In re Ames Dep't Stores*, 115 B.R. at 37–39 (explaining that a debtor must show that it made reasonable efforts to seek other sources of financing under section 364(a) and (b)).
- 86. As noted above, the Debtors do not believe that more favorable alternative postpetition financing is reasonably available given the realities imposed by the Debtors' existing capital structure and the Debtors' solicitation of alternative financing proposals. Additionally, the Debtors' overall restructuring is closely tied to the successful transaction described in the RSA.

Furthermore, as set forth in the Shah Declaration, the Debtors engaged with certain stakeholders and third parties regarding potential financing for a chapter 11 process and entered into non-disclosure agreements with respect to potential financing for a chapter 11 process, but ultimately the Debtors did not receive any offer or combination of offers superior to the DIP Term Loan Facility. Shah Decl. ¶ 15. Simply put, the DIP Term Loan Facility provides the Debtors with the liquidity they need at the lowest cost available while simultaneously placing the Debtors on an optimal path for a successful restructuring. Thus, the Debtors have determined that the DIP Term Loan Facility provides the most favorable terms because it offers the most efficient transaction costs while reducing execution risks and is an integral piece of the debt reduction included in the RSA and is provided on reasonable terms under the circumstances. Therefore, the requirement of section 364 of the Bankruptcy Code that alternative credit on more favorable terms be unavailable to the Debtors is satisfied.

#### II. The Debtors Should Be Authorized to Use Cash Collateral.

87. Section 363 of the Bankruptcy Code generally governs the use of estate property. Section 363(c)(2)(A) of the Bankruptcy Code permits a debtor in possession to use Cash Collateral with the consent of the secured party. Here, the DIP Lenders and the Prepetition Secured Parties consent or are deemed to consent to the Debtors' use of the Cash Collateral, subject to the terms and limitations set forth in the Interim Order. As described above and in the Koza Declaration, access to Cash Collateral on an interim basis is essential to the continued operation of the Debtors' businesses and smooth entry into the Chapter 11 Cases. Use of Cash Collateral is in the best interests of the Debtors' estates and all of their stakeholders, including the Prepetition Secured Parties, and that the Interim Order should be approved.

### III. Adequate Protection Provided to the Prepetition Secured Parties Is Appropriate.

- 88. Section 363(e) of the Bankruptcy Code provides for adequate protection of interests in property when a debtor uses Cash Collateral. Further, section 362(d)(1) of the Bankruptcy Code provides for adequate protection of interests in property due to the imposition of the automatic stay. See In re Cont'l Airlines, 91 F.3d 553, 556 (3d Cir. 1996) (en banc). While section 361 of the Bankruptcy Code provides examples of forms of adequate protection, such as granting replacement liens and administrative claims, courts decide what constitutes sufficient adequate protection on a case-by-case basis. See, e.g., In re Swedeland Dev. Grp., Inc., 16 F.3d 552, 564 (3d Cir. 1994) (explaining that the "determination of whether there is adequate protection is made on a case by case basis"); In re Satcon Tech. Corp., No. 12-12869 (KG), 2012 WL 6091160, at \*6 (Bankr. D. Del. Dec. 7, 2012) (same); In re N.J. Affordable Homes Corp., No. 05-60442 (DHS), 2006 WL 2128624, at \*14 (Bankr. D.N.J. June 29, 2006) ("the circumstances of the case will dictate the necessary relief to be given"); In re Columbia Gas Sys., Inc., Nos. 91-803, 91-804, 1992 WL 79323, at \*2 (Bankr. D. Del. Feb. 18, 1992) ("what interest is entitled to adequate protection and what constitutes adequate protection must be decided on a case-by-case basis"); see also In re Dynaco Corp., 162 B.R. 389, 394 (Bankr. D.N.H. 1993) (citing 2 Collier on Bankruptcy ¶ 361.01[1] at 361–66 (15th ed. 1993) (explaining that adequate protection can take many forms and "must be determined based upon equitable considerations arising from the particular facts of each proceeding")).
- 89. As described more fully herein, and as set forth in the Interim Order, the Debtors propose to provide the Prepetition Secured Parties with a variety of adequate protection to protect against the postpetition diminution in value of the Cash Collateral resulting from the use, sale, or lease of the Cash Collateral by the Debtors and the imposition of the automatic stay. The Adequate Protection package includes (a) Prepetition First Lien 507(b) Claims, (b) Prepetition

First Lien Adequate Protection Liens, (c) delivery of certain reports and notices pursuant to certain sections of DIP Term Loan Credit Agreement, and (d) Adequate Protection Fees and Expenses (collectively, the "Adequate Protection Liens"). In light of the foregoing, the proposed Adequate Protection Liens to be provided for the benefit of the Prepetition Secured Parties are appropriate.<sup>23</sup> The Debtors' provision of the Adequate Protection Liens is not only necessary to protect against any diminution in value but is fair and appropriate under the circumstances of these Chapter 11 Cases to ensure the Debtors are able to continue using the Cash Collateral, subject to the terms and limitations set forth in the Interim Order, for the benefit of all parties in interest and their estates.

### IV. The Scope of the Carve Out Is Appropriate.

Orders. Without the Carve Out, the Debtors and other parties in interest may be deprived of certain rights and powers because the services for which professionals may be paid in these Chapter 11 Cases would be restricted. *See In re Ames Dep't Stores*, 115 B.R. at 40 (observing that courts insist on carve outs for professionals representing parties in interest because "[a]bsent such protection, the collective rights and expectations of all parties-in-interest are sorely prejudiced"). The Carve Out does not directly or indirectly deprive the Debtors' estates or other parties in interest of possible rights and powers. Additionally, the Carve Out protects against administrative insolvency during the course of these Chapter 11 Cases by ensuring that assets remain for the payment of the Clerk of the Court, U.S. Trustee fees, and professional fees of the Debtors and any statutory committee appointed under section 1102 of the Bankruptcy Code in these Chapter 11 Cases.

Pursuant to the DIP Orders, the Prepetition Secured Parties are permitted to seek additional adequate protection in accordance with the terms thereof.

- V. The Debtors Should Be Authorized to Pay the Interest and Fees Required by the DIP Term Loan Agent and the DIP Term Loan Lenders Under the DIP Term Loan Facility Documents and Enter Into the DIP-to-Exit ABL Commitment Papers and Pay Fees and Expenses Thereunder.
- 91. In connection with negotiating the DIP Term Loan Facility, the Debtors have agreed, subject to Court approval, to pay certain interest, fees, and premiums to the DIP Term Loan Agent and the DIP Lenders. In particular, as noted above, the Debtors have agreed to pay:<sup>24</sup>
  - a. *Interest Rate*. Term SOFR plus 8.00% with respect to the DIP Term Loan Facility, subject to a 1.00% SOFR floor, paid in cash at the end of each applicable interest period;
  - b. **Put Option Premium**. a put option premium of the principal amount of the aggregate commitments under the DIP Term Loan Facility, earned on the date of the Commitment Letter (as defined in the DIP Term Loan Credit Agreement) and payable upon any repayment of principal amount of the DIP Term Loan Facility, whether through optional prepayments, mandatory prepayments, at maturity, upon acceleration or otherwise, to any person that is a DIP Term Loan Lender on the date thereof (*pro rata* in accordance with the outstanding principal amount of the commitments and loans under the DIP Term Loan Facility held by such DIP Term Loan Lender on the date thereof), (x) to the extent the Plan shall have been consummated on or prior to the DIP Term Loan Facility Termination Date, in shares of New Equity Interests (as defined in the RSA) at a 37.5% discount to an implied equity value of \$538,821,500, or (y) otherwise, in cash;
  - c. *OID/Upfront*. an upfront fee of 4.00% upon each borrowing under the DIP Term Loan Facility, which amount may be netted from such borrowing; and
  - d. *Exit Fee*. an exit fee of 1.00% of the principal amount of the aggregate commitments under the DIP Term Loan Facility.
- 92. As set forth in the Shah Declaration, the interest and fees to be paid under the DIP Term Loan Facility are reasonable and appropriate, particularly in light of the circumstances of these Chapter 11 Cases and the marketing process undertaken, and represent the only viable option

The summaries contained in this Motion are qualified in their entirety by the provisions of the documents referenced. To the extent anything in this Motion is inconsistent with such documents, the terms of the applicable documents shall control. Capitalized terms used in this summary but not otherwise defined have the meanings ascribed to them in the DIP Term Loan Facility Documents.

presently available to the Debtors. The fees and rates to be paid under the proposed DIP Term Loan Facility (a) were the subject of rigorous arm's-length negotiation between the Debtors and the DIP Lenders, and (b) are an integral component of the overall terms of the proposed DIP Term Loan Facility. *See* Shah Decl. ¶ 28.

93. Further, the fees payable in equity of the reorganized entity (if the Plan is confirmed), preserves liquidity that provides additional runway for the Debtors to consummate the Plan contemplated in the RSA. See Shah Decl. ¶ 27. DIP fees payable in reorganized equity are not without precedent in this and other jurisdictions. See, e.g., In re Party City Holdco Inc., No. 23-90005 (DRJ) (Bankr. S.D. Tex. Jan. 18, 2023) [Docket No. 112] (interim DIP order approving backstop commitment fee payable subject to confirmation of a chapter 11 plan); In re Neiman Marcus Grp. Ltd. LLC, No. 20-32519 (DRJ) (Bankr. S.D. Tex. May 8, 2020) [Docket No. 254] (interim DIP order approving exit and backstop commitment fees payable in reorganized equity, subject to confirmation of a chapter 11 plan); In re Am. Com. Lines Inc., No. 20-30982 (MI) (Bankr. S.D. Tex. Mar. 5, 2020) [Docket No. 202] (interim DIP order approving backstop premium and conversion discount payable in reorganized equity); see also, e.g., In re Lumileds Holding B.V., No. 22-11155 (LGB) (Bankr. S.D.N.Y. Aug. 31, 2022) [Docket No. 73] (interim DIP order approving backstop, participation, and exit commitment fees payable in 57.7% of reorganized equity); In re Grupo Aeroméxico, S.A.B. de C.V., No. 20-11563 (SCC) (Bankr. S.D.N.Y. Aug. 21, 2020) [Dkt. No. 318] (interim DIP Order approving exit fee payable in equity at DIP lenders' option); In re Pyxus Int'l, Inc., No. 20-11570 (LSS) (Bankr. D. Del. June 17, 2020) [Docket No. 84] (interim DIP order approving backstop fee payable in reorganized equity); In re Chinos Holdings, Inc., No. 20-32181 (KLP) (Bankr. E.D. Va. May 5, 2020) [Docket No. 84] (interim DIP order approving of backstop premium payable in reorganized equity).

- 94. The Debtors considered the fees described above when determining in their sound business judgment that the DIP Term Loan Facility is reasonable and constitutes the best terms on which the Debtors can obtain the postpetition financing necessary to continue their operations, prosecute their cases, and benefit the Debtors' estates. Accordingly, the Court should authorize the Debtors to pay the interest and fees provided under the DIP Term Loan Facility Documents in connection with the DIP Term Loan Facility.
- 95. Moreover, as set forth in the Shah Declaration, the Debtors' entry into the DIP-to-Exit ABL Commitment Papers and the payment of fees and expenses thereunder is also reasonable and appropriate. The DIP ABL Facility will provide the Debtors with access to liquidity and letters of credit to support their businesses during the course of these Chapter 11 Cases. And the commitment of the DIP ABL Facility lenders to convert their commitments under the DIP ABL Facility to exit financing provides a concrete, actionable path toward the Debtors' emergence from chapter 11 in an efficient and value-maximizing manner. While the Debtors, the DIP ABL Agent, the DIP ABL Lenders, the Consenting Stakeholders, and their advisors continue to finalize the terms and documentation of the underlying facilities, the Debtors have the opportunity to "lock-in" commitments of essential incremental liquidity to fund the Debtors' operations through and the administration of these Chapter 11 Cases and the wherewithal to exit therefrom for the benefit of their stakeholders. *See* Shah Decl. ¶ 18. Accordingly, the Court should authorize the Debtors to enter into the DIP-to-Exit ABL Commitment Papers.

#### VI. The DIP Lenders Should Be Deemed Good-Faith Lenders Under Section 364(e).

96. Section 364(e) of the Bankruptcy Code protects a good-faith lender's right to collect on loans extended to a debtor, and its right in any lien securing those loans, even if the authority of the debtor to obtain such loans or grant such liens is later reversed or modified on appeal. Section 364(e) of the Bankruptcy Code provides that:

The reversal or modification on appeal of an authorization under this section [364 of the Bankruptcy Code] to obtain credit or incur debt, or of a grant under this section of a priority or a lien, does not affect the validity of any debt so incurred, or any priority or lien so granted, to an entity that extended such credit in good faith, whether or not such entity knew of the pendency of the appeal, unless such authorization and the incurring of such debt, or the granting of such priority or lien, were stayed pending appeal.

97. As explained herein, the DIP Term Loan Facility Documents are the result of: (a) the Debtors' reasonable and informed determination that the DIP Lenders provided the best postpetition financing alternative available under the circumstances and (b) extended arm's-length, good-faith negotiations between the Debtors and the DIP Lenders. The terms and conditions of the DIP Term Loan Facility Documents are reasonable under the circumstances, and the proceeds of the DIP Term Loan Facility will be used only for purposes that are permissible under the Bankruptcy Code. Further, the Debtors market-tested the terms of the DIP Term Loan Facility before determining that the DIP Term Loan Facility provided the best terms available. Accordingly, the Court should find that the DIP Lenders are "good faith" lenders within the meaning of section 364(e) of the Bankruptcy Code and are entitled to all of the protections afforded by that section.

## VII. The Automatic Stay Should Be Modified on a Limited Basis.

98. The proposed Interim Order provides that the automatic stay provisions of section 362 of the Bankruptcy Code will be modified to allow the DIP Lenders to file any financing statements, security agreements, notices of liens, and other similar instruments and documents in order to validate and perfect the liens and security interests granted to them under the Interim Order. The proposed Interim Order further provides that the automatic stay is modified as necessary to permit the Debtors to grant liens to the DIP Lenders and the Prepetition Secured Parties and to incur all liabilities and obligations set forth in the Interim Order.

The automatic stay should be modified to allow the Debtors and the DIP Lenders to effectuate the terms of the Interim Order.

# VIII. Failure to Obtain Immediate Interim Access to the DIP Term Loan Facility and Cash Collateral Would Cause Immediate and Irreparable Harm.

- 99. Bankruptcy Rules 4001(b) and 4001(c) provide that a final hearing on a motion to obtain credit pursuant to section 364 of the Bankruptcy Code or to use Cash Collateral pursuant to section 363 of the Bankruptcy Code may not be commenced earlier than fourteen (14) days after the service of such motion. Upon request, however, the Court may conduct a preliminary, expedited hearing on the motion and authorize the obtaining of credit and use of cash collateral to the extent necessary to avoid immediate and irreparable harm to a debtor's estate. See Bankruptcy Rule 4001(b)(2). Section 363(c)(3) of the Bankruptcy Code authorizes the court to conduct a preliminary hearing and to authorize the use of cash collateral "if there is a reasonable likelihood that the [debtor] will prevail at the final hearing under [section 363(e) of the Bankruptcy Code]." 11 U.S.C. § 363(c)(3). Furthermore, the Complex Case Procedures provide that "on motion by the debtors, a hearing will routinely be conducted as a first day hearing to consider either cash collateral and/or interim debtor-in-possession financing." Complex Case Procedures ¶ 23.
- 100. As set forth in the Koza Declaration, the Debtors have an immediate postpetition need to use Cash Collateral, and access the liquidity provided by the DIP Term Loan Facility. The Debtors cannot maintain the value of their estates during the pendency of these Chapter 11 Cases without access to cash. The Debtors entered these Chapter 11 Cases with extremely limited cash on hand. Koza Decl. ¶ 10. Without immediate financing, the Debtors project that they will be unable to pay essential costs required to continue operating as a going concern, resulting in immediate and irreparable harm to the Debtors' business. Koza Decl. ¶ 10. In particular, the

Debtors require immediate liquidity to satisfy their obligations to, among others, employees, vendors, landlords, and service providers, all of whom are critical to ensuring that the Debtors can continue operating as a going concern. Koza Decl. ¶ 10. Absent funds available from the DIP Term Loan Facility and access to cash collateral, the Debtors could face value-destructive interruptions to their business, proceedings under foreign insolvency law, and lose support from important stakeholders on which the Debtors' business depends, including channel partners, employees, customers, vendors, and other contract counterparties. *See* Koza Decl. ¶¶ 15, 23.

- 101. The Debtors will therefore be unable to operate their business or otherwise fund these Chapter 11 Cases without access to Cash Collateral and the liquidity provided by the DIP Term Loan Facility, and will suffer immediate and irreparable harm to the detriment of all creditors and other parties in interest. Koza Decl. ¶¶ 23, 25. In short, the Debtors' ability to administer these Chapter 11 Cases through the use of Cash Collateral and the liquidity provided by the DIP Term Loan Facility is vital to preserve and maximize the value of the Debtors' estates.
- 102. The Debtors request that the Court hold and conduct a hearing to consider entry of the Interim Order authorizing the Debtors, from and after entry of the Interim Order until the Final Hearing, to receive initial funding under the DIP Term Loan Facility and to utilize Cash Collateral. The Debtors require the initial funding under the DIP Term Loan Facility prior to the Final Hearing and entry of the Final Order to continue operating, pay their administrative expenses, and to implement the relief requested in the Debtors' other "first day" motions. This relief will enable the Debtors to preserve and maximize value and, therefore, avoid immediate and irreparable harm and prejudice to their estates and all parties in interest, pending the Final Hearing.

#### **Emergency Consideration**

103. The Debtors request emergency consideration of this Motion pursuant to Bankruptcy Rule 6003, which empowers a court to grant relief within the first twenty-one days

after the commencement of a chapter 11 case "to the extent that relief is necessary to avoid immediate and irreparable harm." An immediate and orderly transition into chapter 11 is critical to the viability of the Debtors' operations. Failure to receive the requested relief during the first twenty-one days of these Chapter 11 Cases would imperil the Debtors' restructuring and cause irreparable harm. The Debtors have satisfied the "immediate and irreparable harm" standard of Bankruptcy Rule 6003 and request that the Court approve the relief requested in this Motion on an emergency basis.

### Waiver of Bankruptcy Rule 6004(a) and 6004(h)

104. The Debtors request that the Court enter an order providing that notice of the relief requested herein satisfies Bankruptcy Rule 6004(a) and that the Debtors have established cause to exclude such relief from the fourteen-day stay period under Bankruptcy Rule 6004(h).

## **Reservation of Rights**

105. Nothing contained herein or any actions taken pursuant to such relief requested is intended or shall be construed as: (a) an admission as to the amount of, basis for, or validity of any claim against a Debtor entity under the Bankruptcy Code or other applicable nonbankruptcy law; (b) a waiver of the Debtors' or any other party in interest's right to dispute any claim on any grounds; (c) a promise or requirement to pay any claim; (d) an implication or admission that any particular claim is of a type specified or defined in this Motion or any order granting the relief requested by this Motion or a finding that any particular claim is an administrative expense claim or other priority claim; (e) a request or authorization to assume, adopt, or reject any agreement, contract, or lease pursuant to section 365 of the Bankruptcy Code; (f) an admission as to the validity, priority, enforceability, or perfection of any lien on, security interest in, or other encumbrance on property of the Debtors' estates; (g) a waiver or limitation of the Debtors' or any other party in interest's rights under the Bankruptcy Code or any other applicable law; or

(h) a concession by the Debtors that any liens (contractual, common law, statutory, or otherwise) that may be satisfied pursuant to the relief requested in this Motion are valid, and the rights of all parties in interest are expressly reserved to contest the extent, validity, or perfection or seek avoidance of all such liens. If the Court grants the relief sought herein, any payment made pursuant to the Court's order is not intended, and should not be construed as, an admission as to the validity of any particular claim or a waiver of the Debtors' or any other party in interest's rights to subsequently dispute such claim.

#### **Notice**

106. The Debtors have provided notice of this Motion to the following parties or their respective counsel: (a) the U.S. Trustee; (b) the Holders of the thirty largest unsecured claims against the Debtors (on a consolidated basis); (c) counsel to the Akin Ad Hoc Group; (d) counsel to the PW Ad Hoc Group; (e) the ABL Agent and counsel thereto; (f) the Term Loan Agent and counsel thereto; (g) the Legacy Notes Trustee and counsel thereto; (h) the Secured Exchangeable Secured Notes Trustee and counsel thereto; (i) the HoldCo Convertible Notes Trustee and counsel thereto; (j) the DIP Term Loan Agent and counsel thereto; (k) the Office of the United States Attorney for the Southern District of Texas; (l) the state attorneys general for states in which the Debtors conduct business; (m) the Internal Revenue Service; (n) the Securities and Exchange Commission; (o) the Environmental Protection Agency; (p) other governmental agencies having a regulatory or statutory interest in these cases; and (q) any party that has requested notice pursuant to Bankruptcy Rule 2002. In light of the nature of the relief requested, no other or further notice is required.

The Debtors request that the Court enter the Interim Order and the Final Order granting the relief requested herein and such other relief as the Court deems appropriate under the circumstances.

Houston, Texas

Dated: February 14, 2023

#### /s/ Matthew D. Cavenaugh

#### JACKSON WALKER LLP

Matthew D. Cavenaugh (TX Bar No. 24062656) Genevieve M. Graham (TX Bar No. 24085340) Rebecca Blake Chaikin (TX Bar No. 24133055) Emily Meraia (TX Bar No. 24129307) 1401 McKinney Street, Suite 1900

Houston, TX 77010

Telephone: (713) 752-4200 Facsimile: (713) 752-4221

Email: mcavenaugh@jw.com

rchaikin@jw.com ggraham@jw.com emeraia@jw.com

### KIRKLAND & ELLIS LLP KIRKLAND & ELLIS INTERNATIONAL LLP

Joshua A. Sussberg, P.C. (*pro hac vice* pending) Aparna Yenamandra, P.C. (*pro hac vice* pending) Rachael M. Bentley (*pro hac vice* pending) Andrew Townsell (*pro hac vice* pending) 601 Lexington Avenue

New York, New York 10022 Telephone: (212) 446-4800 Facsimile: (212) 446-4900

Email: joshua.sussberg@kirkland.com

aparna.yenamandra@kirkland.com rachael.bentley@kirkland.com andrew.townsell@kirkland.com

-and-

Patrick J. Nash, Jr., P.C. (pro hac vice pending)

300 North LaSalle Street Chicago, Illinois 60654

Telephone: (312) 862-2000 Facsimile: (312) 862-2200

Email: patrick.nash@kirkland.com

Proposed Co-Counsel to the Debtors and Debtors in Possession

Proposed Co-Counsel to the Debtors and Debtors in Possession

## **Certificate of Accuracy**

I certify that the foregoing statements are true and accurate to the best of my knowledge. This statement is being made pursuant to Bankruptcy Local Rule 9013-1(i).

/s/ Matthew D. Cavenaugh

Matthew D. Cavenaugh

## **Certificate of Service**

I certify that on February 14, 2023, I caused a copy of the foregoing document to be served by the Electronic Case Filing System for the United States Bankruptcy Court for the Southern District of Texas.

/s/ Matthew D. Cavenaugh

Matthew D. Cavenaugh

## Exhibit A

**DIP Term Loan Credit Agreement** 

#### SUPERPRIORITY SECURED DEBTOR IN POSSESSION CREDIT AGREEMENT

Dated as of February 15, 2023,

among

## AVAYA HOLDINGS CORP.,

a Debtor and Debtor in Possession under chapter 11 of the Bankruptcy Code, as Holdings,

#### **AVAYA INC.,**

a Debtor and Debtor in Possession under chapter 11 of the Bankruptcy Code, as the Borrower,

The Several Lenders from Time to Time Parties Hereto,

## WILMINGTON SAVINGS FUND SOCIETY, FSB,

as Administrative Agent and Collateral Agent,

## TABLE OF CONTENTS

		Page
SECTION 1	DEFINITIONS	1
1.1	Defined Terms	
1.2	Other Interpretive Provisions	
1.3	Accounting Terms	
1.4	Divisions	
1.5	References to Agreements, Laws, Etc	
1.6	Times of Day	
1.7	Timing of Payment or Performance	
1.8	Currency Equivalents Generally	
1.9	Classification of Loans and Borrowings	
1.10	Permitted Liens.	
SECTION 2	AMOUNT AND TERMS OF CREDIT	54
2.1	The Loans	
2.2	Maximum Number of Borrowings	
2.3	Notice of Borrowing	
2.4	Disbursement of Funds	
2.5	Repayment of Term Loans; Evidence of Debt	
2.6	SOFR/Base Rate Conversions and Continuations	
2.7	Benchmark Replacement Setting	
2.8	Interest	
2.9	Interest Periods	
2.10	Increased Costs, Illegality, Etc	
2.11	Compensation	
2.12	Change of Lending Office	
2.13	Notice of Certain Costs	
2.14	[Reserved]	
2.15	Conversion to Exit Term Loans	
2.16	Defaulting Lenders	
2.17	Priority and Liens	
2.18	No Discharge; Survival of Claims	
2.19	Payment of Obligations	
SECTION 3	[RESERVED]	66
SECTION 4	FEES; COMMITMENTS	66
4.1	Fees.	
4.2	Mandatory Termination or Reduction of Commitments	
SECTION 5	PAYMENTS	68
5.1	Voluntary Prepayments	
5.2	Mandatory Prepayments	
5.3	Method and Place of Payment	

5.4	Net Payments	71
5.5	Computations of Interest and Fees.	75
5.6	Limit on Rate of Interest	75
SECTION	CONDITIONS DESCEDENT TO THE SLOSING DATE	70
SECTION 6	CONDITIONS PRECEDENT TO THE CLOSING DATE	
6.1	Credit Documents	
6.2	Collateral	
6.3	Legal Opinions	
6.4	Closing Certificates	
6.5	Secretary's Certificate; Authorization of Proceedings of Each Credit Party	
6.6	Fees	
6.7	Representations and Warranties	
6.8	Material Adverse Effect	
6.9	Approved Budget; DIP Projection	
6.10	Prepetition ABL Payoff Letter	
6.11	Prepetition ABL Payoff	
6.12	Governmental Approvals	78
6.13	Patriot Act	78
6.14	The Chapter 11 Cases	78
6.15	Escrow Payment	79
6.16	No Default	79
6.17	No Litigation	79
6.18	Restructuring Support Agreement	
SECTION 7	CONDITIONS PRECEDENT TO THE SUBSEQUENT DRAW	70
7.1	Closing Date	
7.1	Interim Order	
7.2		
	First Day Orders	
7.4	Representations and Warranties	
7.5	No Default or Event of Default	
7.6	Request for Borrowing	
7.7	No Violation of Law	
7.8	Payments of Fees and Expenses	
7.9	Material Adverse Effect	
7.10	The Chapter 11 Cases	
7.11	Milestones	
7.12	Approved Budget	
7.13	No Litigation	82
SECTION 8	REPRESENTATIONS AND WARRANTIES	82
8.1	Corporate Status; Compliance with Laws	
8.2	Corporate Power and Authority	
8.3		
8.4	NO VIOIATION	
	No Violation	
_	Litigation	83
8.5	LitigationMargin Regulations	83 83
_	Litigation	83 83

8.8	True and Complete Disclosure	84
8.9	[Reserved]	
8.10	Tax Matters	84
8.11	Compliance with ERISA	85
8.12	Subsidiaries	85
8.13	Intellectual Property	85
8.14	Environmental Laws	85
8.15	Properties	86
8.16	No Default	86
8.17	Security Interests	86
8.18	Labor Matters	87
8.19	Sanctioned Persons; Anti-Corruption Laws; Patriot Act	87
8.20	Use of Proceeds	87
8.21	Orders	87
8.22	Status of Obligations; Perfection and Priority of Security Interests	88
SECTION 9	AFFIRMATIVE COVENANTS	88
9.1	Information Covenants	88
9.2	Books, Records and Inspections	90
9.3	Maintenance of Insurance	91
9.4	Payment of Taxes	91
9.5	Consolidated Corporate Franchises	91
9.6	Compliance with Statutes, Regulations, Etc	92
9.7	Lender Calls	92
9.8	Maintenance of Properties	92
9.9	Cash Management Order	92
9.10	End of Fiscal Years	92
9.11	Additional Guarantors and Grantors	92
9.12	Further Assurances	93
9.13	Use of Proceeds	94
9.14	Maintenance of Ratings	95
9.15	Changes in Business	95
9.16	Milestones	95
9.17	Bankruptcy Matters	96
9.18	DIP Proceeds Account	97
SECTION 1	0 NEGATIVE COVENANTS	97
10.1	Limitation on Indebtedness	98
10.2	Limitation on Liens	101
10.3	Limitation on Fundamental Changes	
10.4	Limitation on Disposition	
10.5	Limitation on Investments	
10.6	Limitation on Restricted Payments	
10.7	Limitations on Debt Prepayments and Amendments	
10.8	Limitation on Subsidiary Distributions	
10.9	Amendment of Organizational Documents	
	Permitted Activities and Transfer to Parent Companies	

10.11	Transactions with Affiliates	112
10.12	Permitted Variance	113
10.13	Intercompany Limitations	113
10.14	Bankruptcy Matters	114
	1 EVENTS OF DEFAULT	
11.1	Payments	
11.2	Representations, Etc	
11.3	Covenants	
11.4	Default Under Other Agreements	
11.5	Bankruptcy	
11.6	ERISA	
11.7	Guarantee	
11.8	Security Agreement	
11.9	Judgments	
11.10	$\mathcal{E}$	
11.11	,	
11.12	1 1 2	
11.13	J	
11.14	,	
11.15	1 2	
11.16	L J	
11.17	$\mathcal{C}$	
11.18		
11.19		
11.20		
11.21		
11.22		
11.23	Application of Proceeds	121
SECTION 1	2 THE AGENTS	122
12.1	Appointment	
12.1	11	
12.2	Delegation of Duties  Exculpatory Provisions	
12.3	Reliance by Agents	
12.4	Notice of Default	
12.5	Non-Reliance on Administrative Agent, Collateral Agent and Other	120
12.0	LendersLenders	126
12.7	Indemnification	
12.7	Agents in their Individual Capacities	
12.9 12.10	Successor Agents	
	Administrative Agent May File Proofs of Claim	
12.11	· · · · · · · · · · · · · · · · · · ·	
12.12 12.13		130
12.13	and Guarantee	121
12.17		1211

<b>SECTION 13</b>	MISCELLANEOUS	132
13.1	Amendments, Waivers and Releases	132
13.2	Notices	135
13.3	No Waiver; Cumulative Remedies	136
13.4	Survival of Representations and Warranties	136
13.5	Payment of Expenses; Indemnification	136
13.6	Successors and Assigns; Participations and Assignments	138
13.7	Replacements of Lenders under Certain Circumstances	143
13.8	Adjustments; Set-off	
13.9	Counterparts; Electronic Execution	
13.10	Severability	
13.11	INTEGRATION; ORDERS CONTROL	145
13.12	GOVERNING LAW	146
13.13	Submission to Jurisdiction; Waivers	146
13.14	Acknowledgments	146
	WAIVERS OF JURY TRIAL	
	Confidentiality	
	Direct Website Communications	
	USA PATRIOT Act	
13.19	Payments Set Aside	
13.20	Judgment Currency	151
13.21	Acknowledgement and Consent to Bail-In of EEA Financial Institutions	152

**SCHEDULES** 

Schedule 1.1(a) Commitments of Lenders

Schedule 1.1(b) Re-allocated Commitments of Lenders

Schedule 8.4 Litigation
Schedule 8.12 Subsidiaries

Schedule 8.14 Environmental Matters

Schedule 8.15 Property Matters

Schedule 10.4 Scheduled Dispositions

Schedule 10.11 Closing Date Affiliate Transactions

Schedule 13.2 Notice Addresses

**EXHIBITS** 

Exhibit A Form of Notice of Borrowing/Notice of Conversion or Continuation

Exhibit B Form of Promissory Note

Exhibit C [Reserved]
Exhibit D [Reserved]
Exhibit E [Reserved]
Exhibit F [Reserved]
Exhibit G [Reserved]
Exhibit H [Reserved]

Exhibit I Form of Assignment and Assumption
Exhibit J 1-4 Form of Non-U.S. Lender Certification

Exhibit K Form of Approved Budget

Exhibit L Interim Order

Exhibit M Form of Variance Report

SUPERPRIORITY SECURED DEBTOR IN POSSESSION CREDIT AGREEMENT (as amended, restated, supplemented or otherwise modified from time to time, the "Agreement"), dated as of February 15, 2023, among AVAYA HOLDINGS CORP., a Delaware corporation and a Debtor and Debtor in Possession under chapter 11 of the Bankruptcy Code ("Holdings"), in its capacity as Holdings, AVAYA INC., a Delaware corporation and a Debtor and Debtor in Possession under chapter 11 of the Bankruptcy Code (the "Borrower"), the lending institutions from time to time parties hereto (each a "Lender" and, collectively, the "Lenders") and WILMINGTON SAVINGS FUND SOCIETY, FSB, as Administrative Agent and Collateral Agent.

#### **RECITALS:**

WHEREAS, capitalized terms used and not defined in the preamble and these recitals shall have the respective meanings set forth for such terms in Section 1.1 hereof;

WHEREAS, on February 14, 2023 (the "Petition Date"), Holdings, the Borrower and certain of the Borrower's Domestic Subsidiaries (collectively, the "Debtors") filed voluntary petitions for relief under chapter 11 in the United States Bankruptcy Court for the Southern District of Texas (such court, together with any other court having exclusive jurisdiction over the Case from time to time and any Federal appellate court thereof, the "Bankruptcy Court") and commenced cases, jointly administered under Case No. 23-90088 (collectively, the "Chapter 11 Cases"), and have continued in the possession and operation of their assets and in the management of their businesses pursuant to sections 1107 and 1108 of the Bankruptcy Code;

WHEREAS, the Lenders agree, upon the satisfaction (or waiver) of certain conditions precedent set forth in Section 6, to extend credit to the Borrower in the form of term loans in an aggregate principal amount of \$500,000,000 (the "Term Facility");

NOW, THEREFORE, in consideration of the premises and the covenants and agreements contained herein, the parties hereto hereby agree as follows:

#### **SECTION 1 Definitions**

### 1.1 Defined Terms

As used herein, the following terms shall have the meanings specified in this Section 1.1 unless the context otherwise requires:

"ABR" shall mean for any day a fluctuating rate per annum equal to the greatest of (a) the Federal Funds Effective Rate *plus* 1/2 of 1%, (b) the rate of interest in effect for such day as publicly announced from time to time by the Wall Street Journal as the "U.S. prime rate" and (c) greater of (x) Adjusted Term SOFR for a one-month tenor in effect on such day plus 1.00% and (y) 2.00%. If the Administrative Agent is unable to ascertain the Federal Funds Effective Rate due to its inability to obtain sufficient quotations in accordance with the definition thereof, after notice is provided to the Borrower, the ABR shall be determined without regard to clause (a) above until the circumstances giving rise to such inability no longer exist. Any change in the ABR due to a change in such rate publicly announced by the Wall Street Journal or in the Federal Funds Effective Rate or the Adjusted Term SOFR shall take effect at the opening of business on the day

specified in the public announcement of such change or on the effective date of such change in the Federal Funds Effective Rate or the Adjusted Term SOFR.

- "ABR Loan" shall mean each Term Loan bearing interest based on the ABR.
- "ABR Term SOFR Determination Day" shall have the meaning specified in the definition of "Term SOFR".
- "Acceptable Reinvestment Commitment" shall mean a binding commitment of the Borrower or any Subsidiary entered into at any time prior to the end of the Reinvestment Period to reinvest the proceeds of a Recovery Prepayment Event.
- "Adjusted Term SOFR" shall mean, for purposes of any calculation, the rate per annum equal to (a) Term SOFR for such calculation plus (b) the Term SOFR Adjustment; provided that if Adjusted Term SOFR as so determined shall ever be less than the Floor, then Adjusted Term SOFR shall be deemed to be the Floor.
- "Administrative Agent" shall mean Wilmington Savings Fund Society, FSB, as the administrative agent for the Lenders under this Agreement and the other Credit Documents, or any successor administrative agent pursuant to Section 12.9.
- "Administrative Agent's Office" shall mean the Administrative Agent's address and, as appropriate, account as set forth on Schedule 13.2, or such other address or account as the Administrative Agent may from time to time notify to the Borrower and the Lenders.
- "Administrative Questionnaire" shall mean an administrative questionnaire in a form supplied by the Administrative Agent.
- "Affected Financial Institution" means (a) any EEA Financial Institution or (b) any UK Financial Institution.
- "Affiliate" shall mean, with respect to any Person, any other Person directly or indirectly controlling, controlled by, or under direct or indirect common control with such Person. A Person shall be deemed to control another Person if such Person possesses, directly or indirectly, the power to direct or cause the direction of the management and policies of such other Person, whether through the ownership of voting securities or by contract. The terms "controlling" and "controlled" shall have meanings correlative thereto.
  - "Agent Parties" shall have the meaning provided in Section 13.17(d).
  - "Agents" shall mean the Administrative Agent and the Collateral Agent.
- "Aggregate Quarterly Subscription Contract ARR Revenue" means, with respect to each fiscal quarter of the Borrower, (a) the arithmetic average of the sum of (i) the Aggregate Subscription Contract ARR with respect to all Subscription Contracts in effect on the last day of the previous fiscal quarter and (ii) the Aggregate Subscription Contract ARR with respect to all Subscription Contracts in effect on the last day of such fiscal quarter divided by (b) four (4).

- "Aggregate Quarterly Subscription Contract GAAP Revenue" means, with respect to each fiscal quarter of the Borrower, the revenue generated by all Subscription Contracts during such fiscal quarter as determined pursuant GAAP, including the provisions of ASC 606, Contracts with Customers.
- "Aggregate Subscription Contract ARR" means, at any time, the aggregate Subscription Contract ARR for all Subscription Contracts then in effect at such time.
- "Agreement" shall have the meaning provided in the introductory paragraph hereto.
  - "Agreement Currency" shall have the meaning provided in Section 13.20.
  - "Akin" means Akin Gump Strauss Hauer & Feld LLP.
- "Akin Ad Hoc Group" has the meaning set forth in the Restructuring Support Agreement.
  - "Anti-Corruption Laws" shall have the meaning provided in Section 8.19.
- "Applicable ABR Margin" shall mean at any date, with respect to each ABR Loan, 7.00% per annum.
- "Applicable Laws" shall mean, as to any Person, any law (including common law), statute, regulation, ordinance, rule, order, decree, judgment, consent decree, writ, injunction, settlement agreement or governmental requirement enacted, promulgated or imposed or entered into or agreed by any Governmental Authority, in each case applicable to or binding on such Person or any of its property or assets is subject.
- "Applicable SOFR Margin" shall mean, with respect to each SOFR Loan, 8.00% per annum.
- "Approved Budget" shall mean a projected statement of sources and uses of cash for the Borrower and the other Debtors on a weekly basis for the following 13 calendar weeks including the anticipated uses of the Term Facility for each week during such period, in substantially the form of Exhibit K hereto. As used herein, "Approved Budget" shall initially refer to the initial Approved Budget delivered in accordance with Section 6.9 and thereafter shall refer to the most recent Approved Budget delivered by the Borrower in accordance with Section 9.1(b), to the extent such Approved Budget is accepted by the Required Lenders.
- "Approved Fund" shall mean any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender, or (c) an entity or an Affiliate of an entity that administers or manages a Lender.
- "Asset Sale Prepayment Event" shall mean any Disposition under and pursuant to Section 10.4(b).

"Assignment and Assumption" shall mean an assignment and assumption substantially in the form of Exhibit I, or such other form as may be approved by the Administrative Agent and the Borrower.

"Authorized Officer" shall mean the President, the Chief Executive Officer, the Chief Financial Officer, the Chief Operating Officer, the Chief Restructuring Officer, the Treasurer, any Assistant Treasurer, the Controller, any Vice President, with respect to certain limited liability companies or partnerships that do not have officers, any manager, managing member or general partner thereof, any other senior officer of Holdings, the Borrower or any other Credit Party designated as such in writing to the Administrative Agent by Holdings, the Borrower or such other Credit Party, as applicable from time to time, and, with respect to any document delivered on the Closing Date, the Secretary or any Assistant Secretary of any Credit Party. Any document delivered hereunder that is signed by an Authorized Officer shall be conclusively presumed to have been authorized by all necessary corporate, limited liability company, partnership and/or other action on the part of Holdings, the Borrower or any other Credit Party and such Authorized Officer shall be conclusively presumed to have acted on behalf of such Person.

"Available Tenor" shall mean, with respect to any Term Loans, as of any date of determination and with respect to the then-current Benchmark, as applicable, (x) if such Benchmark is a term rate, any tenor for such Benchmark (or component thereof) that is or may be used for determining the length of an interest period pursuant to this Agreement or (y) otherwise, any payment period for interest calculated with reference to such Benchmark (or component thereof) that is or may be used for determining any frequency of making payments of interest calculated with reference to such Benchmark pursuant to this Agreement, in each case, as of such date and not including, for the avoidance of doubt, any tenor for such Benchmark that is thenremoved from the definition of "Interest Period" pursuant to Section 2.7(d).

"Avoidance Action" shall have the meaning specified in the Interim Order or the Final Order, as applicable.

"B-3 Escrow Claims" shall have the meaning specified in the Restructuring Support Agreement.

"B-3 Re-allocation" shall mean the adjustments to the Term Commitments of the Lenders that shall occur effective upon the earlier of the occurrence of the Escrow Payment and entry of the Final Order pursuant to the terms of the Commitment Letter to the extent that the B-3 Escrow Claims are repaid following the Petition Date and prior to the entry of, or will be repaid pursuant to the terms of the Final Order (the "B-3 Re-allocation Triggering Event"), based on the methodology as illustrated by the adjustments to the "Annex I Commitments" as reflected in the "Annex II Commitments," in each case, set forth in the Commitment Letter; *provided* that the Borrower shall provide prompt written notice of the B-3 Re-allocation Triggering Event, which notice may be made by Kirkland & Ellis LLP, counsel to the Debtors, and the Administrative Agent shall be deemed to have no knowledge of such B-3 Re-allocation Triggering Event until receipt of such notice.

"Bail-In Action" shall mean the exercise of any Write-Down and Conversion Powers by the applicable Resolution Authority in respect of any liability of an Affected Financial Institution.

"Bail-In Legislation" shall mean, (a) with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule, and (b) with respect to the United Kingdom, Part I of the United Kingdom Banking Act 2009 (as amended from time to time) and any other law, regulation or rule applicable in the United Kingdom relating to the resolution of unsound or failing banks, investment firms or other financial institutions or their affiliates (other than through liquidation, administration or other insolvency proceedings).

"Bankruptcy Code" shall mean title 11 of the United States Code, as heretofore and hereafter amended, and codified in 11 U.S.C. section 101 et seq.

"Bankruptcy Court" shall have the meaning provided in the preamble to this Agreement.

"Benchmark" shall mean, initially, the Term SOFR Reference Rate; provided that if a Benchmark Transition Event has occurred with respect to the Term SOFR Reference Rate or the then-current Benchmark, then "Benchmark" means the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to Section 2.7.

"Benchmark Replacement" shall mean, with respect to any Benchmark Transition Event, the sum of: (a) the alternate benchmark rate that has been selected by the Administrative Agent and the Borrower giving due consideration to any evolving or then-prevailing market convention for determining a benchmark rate as a replacement to the then-current Benchmark for Dollar-denominated syndicated credit facilities at such time and (b) the related Benchmark Replacement Adjustment; *provided* that, if such Benchmark Replacement as so determined would be less than the Floor, such Benchmark Replacement will be deemed to be the Floor for the purposes of this Agreement and the other Credit Documents.

"Benchmark Replacement Adjustment" shall mean, with respect to any replacement of the then-current Benchmark with an Unadjusted Benchmark Replacement, the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by the Administrative Agent and the Borrower giving due consideration to any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement for Dollar-denominated syndicated credit facilities at such time.

"Benchmark Replacement Date" shall mean the earliest to occur of the following events with respect to the then-current Benchmark:

(a) in the case of clause (a) or (b) of the definition of "Benchmark Transition Event," the later of (i) the date of the public statement or publication of information

- referenced therein and (ii) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide all Available Tenors of such Benchmark (or such component thereof); or
- (b) in the case of clause (c) of the definition of "Benchmark Transition Event," the first date on which such Benchmark (or the published component used in the calculation thereof) has been determined and announced by the regulatory supervisor for the administrator of such Benchmark (or such component thereof) to be non-representative; *provided* that such non-representativeness will be determined by reference to the most recent statement or publication referenced in such clause (c) and even if any Available Tenor of such Benchmark (or such component thereof) continues to be provided on such date.

For the avoidance of doubt, the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (a) or (b) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Event" shall mean the occurrence of one or more of the following events with respect to the then-current Benchmark:

- (a) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely; *provided* that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof);
- (b) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the Federal Reserve Board, the Federal Reserve Bank of New York, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component), a resolution authority with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component), which states that the administrator of such Benchmark (or such component) has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely; *provided* that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof); or
- (c) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that all Available Tenors of such Benchmark (or

such component thereof) are not, or as of a specified future date will not be, representative.

For the avoidance of doubt, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Start Date" shall mean, in the case of a Benchmark Transition Event, the earlier of (a) the applicable Benchmark Replacement Date and (b) if such Benchmark Transition Event is a public statement or publication of information of a prospective event, the 90th day prior to the expected date of such event as of such public statement or publication of information (or if the expected date of such prospective event is fewer than 90 days after such statement or publication, the date of such statement or publication).

"Benchmark Unavailability Period" shall mean, the period (if any) (a) beginning at the time that a Benchmark Replacement Date has occurred if, at such time, no Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Credit Document in accordance with Section 2.7 and (b) ending at the time that a Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Credit Document in accordance with Section 2.7.

"Benefited Lender" shall have the meaning provided in Section 13.8(a).

"Board" shall mean the Board of Governors of the Federal Reserve System of the United States (or any successor).

"Borrower" shall have the meaning provided in the preamble to this Agreement.

"Borrowing" shall mean the incurrence of one Type of Term Loan on a given date (or resulting from conversions on a given date), and in the case of SOFR Loans, having the same Interest Period.

"Broker-Dealer Subsidiary" shall mean any Subsidiary that is registered as a broker-dealer under the Exchange Act or any other Applicable Law requiring similar registration.

"Business Day" shall mean any day excluding Saturday, Sunday and any other day on which banking institutions in New York City are authorized by law or other governmental actions to close.

"Capital Expenditures" shall mean, for any period, the aggregate of all expenditures (whether paid in cash or accrued as liabilities and including in all events all amounts expended or capitalized under Capital Leases) by the Borrower and the Subsidiaries during such period that, in conformity with GAAP, are or are required to be included as capital expenditures on a consolidated statement of cash flows of the Borrower or the Subsidiary.

"Capital Lease" shall mean, as applied to the Borrower and the Subsidiaries, any lease of any property (whether real, personal or mixed) by the Borrower or any Subsidiary as lessee

that, in conformity with GAAP, is, or is required to be, accounted for as a capital lease on the balance sheet of the Borrower; *provided*, *however*, that notwithstanding anything to the contrary in this Agreement or in any other Credit Document, any leases that were not capital leases when entered into but are recharacterized as capital leases due to a change in accounting rules that becomes effective after December 15, 2017 shall for all purposes of this agreement not be treated as Capital Leases.

"Capitalized Lease Obligations" shall mean, as applied to the Borrower and the Subsidiaries at the time any determination is to be made, the amount of the liability in respect of a Capital Lease that would at such time be required to be capitalized and reflected as a liability on the balance sheet (excluding the footnotes thereto) of the Borrower or the Subsidiary in accordance with GAAP, and the Stated Maturity thereof shall be the date of the last payment of rent or any other amount due under such Capital Lease prior to the first date upon which such Capital Lease may be prepaid by the lessee without payment of a penalty; *provided*, *however*, that notwithstanding anything to the contrary in this Agreement or in any other Credit Document, any obligations that were not required to be included on the balance sheet of the Borrower or the Subsidiary as capital lease obligations when incurred but are recharacterized as capital lease obligations due to a change in accounting rules that becomes effective after December 15, 2017 shall for all purposes of this Agreement not be treated as Capitalized Lease Obligations.

"Capitalized Software Expenditures" shall mean, for any period, the aggregate of all expenditures (whether paid in cash or accrued as liabilities) by the Borrower and the Subsidiaries during such period in respect of purchased software or internally developed software and software enhancements that, in conformity with GAAP are or are required to be reflected as capitalized costs on the consolidated balance sheet of the Borrower.

"Captive Insurance Subsidiary" shall mean a Subsidiary of the Borrower established for the purpose of, and to be engaged solely in the business of, insuring the businesses or facilities owned or operated by the Borrower or any of its Subsidiaries or joint ventures or to insure related or unrelated businesses.

"Carve Out" shall have the meaning specified in the Interim Order or the Final Order, as applicable.

"Carve Out Reserve" shall have the meaning specified in the Interim Order or the Final Order, as applicable.

"Case" shall have the meaning provided in the preamble to this Agreement.

"Cash Collateral" has the meaning specified in the Interim Order or the Final Order, as applicable.

# "Cash Equivalent" shall mean:

(a) Dollars and cash in such foreign currencies held by the Borrower or any Subsidiary from time to time in the ordinary course of business;

- (b) securities issued or unconditionally guaranteed by the United States government or any agency or instrumentality thereof, in each case having maturities and/or reset dates of not more than 24 months from the date of acquisition thereof;
- (c) securities issued by any state of the United States of America or any political subdivision of any such state or any public instrumentality thereof or any political subdivision of any such state or any public instrumentality thereof having maturities of not more than 24 months from the date of acquisition thereof and, at the time of acquisition, having an investment grade rating generally obtainable from either S&P or Moody's (or, if at any time neither S&P nor Moody's shall be rating such obligations, then from another nationally recognized rating service);
- (d) commercial paper or variable or fixed rate notes maturing no more than 12 months after the date of creation thereof and, at the time of acquisition, having a rating of at least A-3 or P-3 from either S&P or Moody's (or, if at any time neither S&P nor Moody's shall be rating such obligations, an equivalent rating from another nationally recognized rating service);
- (e) time deposits with, or domestic and LIBOR certificates of deposit or bankers' acceptances maturing no more than two years after the date of acquisition thereof issued by, the Administrative Agent (or any Affiliate thereof), any Lender or any other bank having combined capital and surplus of not less than \$500,000,000 in the case of domestic banks and \$100,000,000 (or the dollar equivalent thereof) in the case of foreign banks;
- (f) repurchase agreements with a term of not more than 90 days for underlying securities of the type described in clauses (b), (c) and (e) above entered into with any bank meeting the qualifications specified in clause (e) above or securities dealers of recognized national standing;
- (g) marketable short-term money market and similar funds (x) either having assets in excess of \$500,000,000 or (y) having a rating of at least A-3 or P-3 from either S&P or Moody's (or, if at any time neither S&P nor Moody's shall be rating such obligations, an equivalent rating from another nationally recognized rating service);
- (h) shares of investment companies that are registered under the Investment Company Act of 1940 and substantially all the investments of which are one or more of the types of securities described in clauses (a) through (g) above; and
- (i) in the case of Investments by any Foreign Subsidiary or Investments made in a country outside the United States of America, other customarily utilized high-quality Investments in the country where such Foreign Subsidiary is located or in which such Investment is made.

"Cash Management Order" means an order entered by the Bankruptcy Court regarding the Credit Parties' cash management system, bank accounts, cash collections and disbursements, intercompany transactions, bank fees, business forms, corporate cards and related

matters, in form and substance satisfactory to the Required Lenders, as the same may be amended, modified or supplemented from time to time.

"Certificated Securities" shall have the meaning provided in Section 8.17.

"CFC" shall mean a Subsidiary of the Borrower that is a "controlled foreign corporation" within the meaning of Section 957 of the Code.

"CFC Holding Company" shall mean a Subsidiary of the Borrower that has no material assets other than (a) the equity interests (including, for this purpose, any debt or other instrument treated as equity for U.S. federal income tax purposes) in (x) one or more Foreign Subsidiaries that are CFCs or (y) one or more other CFC Holding Companies and (b) cash and Cash Equivalents and other assets being held on a temporary basis incidental to the holding of assets described in clause (a) of this definition.

"Change in Law" shall mean (a) the adoption of any Applicable Law after the Closing Date, (b) any change in any Applicable Law or in the interpretation or application thereof by any Governmental Authority after the Closing Date or (c) compliance by any party with any guideline, request, directive or order issued or made after the Closing Date by any central bank or other governmental or quasi-governmental authority (whether or not having the force of law); provided that notwithstanding anything herein to the contrary, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (y) all requests, rules, guidelines or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the U.S. or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Law", regardless of the date enacted, adopted or issued.

"Change of Control" shall mean and be deemed to have occurred if (a) any "person" (as such term is used in Sections 13(d) and 14(d) of the Exchange Act as in effect on the Closing Date), other than one or more Permitted Holders, becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 of the Exchange Act as in effect on the Closing Date) of more than 50% of the total voting power of the Voting Stock of Holdings; provided that (x) so long as Holdings is a Subsidiary of any Parent Entity, no Person shall be deemed to be or become a beneficial owner of more than 50% of the total voting power of the Voting Stock of Holdings unless such Person shall be or become a beneficial owner of more than 50% of the total voting power of the Voting Stock of such Parent Entity (other than a Parent Entity that is a Subsidiary of another Parent Entity) and (y) any Voting Stock of which any Permitted Holder is the beneficial owner shall not in any case be included in any Voting Stock of which any such Person is the beneficial owner; or (b) Holdings shall not own, directly or indirectly, beneficial ownership of 100% of the Stock and Stock Equivalents of the Borrower.

Notwithstanding the preceding or any provision of Section 13d-3 of the Exchange Act, (i) a Person or group shall not be deemed to beneficially own Voting Stock subject to a stock or asset purchase agreement, merger agreement, option agreement, warrant agreement or similar agreement (or voting or option or similar agreement related thereto) until the consummation of the acquisition of the Voting Stock in connection with the transactions contemplated by such

agreement, (ii) if any group includes one or more Permitted Holders, the issued and outstanding Voting Stock of Holdings owned, directly or indirectly, by any Permitted Holders that are part of such group shall not be treated as being beneficially owned by such group or any other member of such group for purposes of determining whether a Change of Control has occurred, (iii) a Person or group will not be deemed to beneficially own the Voting Stock of another Person as a result of its ownership of Voting Stock or other securities of such other Person's parent entity (or related contractual rights) unless it owns 50% or more of the total voting power of the Voting Stock entitled to vote for the election of directors of such parent entity having a majority of the aggregate votes on the board of directors (or similar body) of such parent entity and (iv) the right to acquire Voting Stock (so long as such Person does not have the right to direct the voting of the Voting Stock subject to such right) or any veto power in connection with the acquisition or disposition of Voting Stock will not cause a party to be a beneficial owner.

"Claim" shall have the meaning provided in the definition of "Environmental Claims".

"Closing Date" shall mean February 15, 2023, on which the conditions set forth in Section 6 are first satisfied.

"Code" shall mean the Internal Revenue Code of 1986, as amended from time to time. Section references to the Code are to the Code, as in effect on the Closing Date, and any subsequent provisions of the Code, amendatory thereof, supplemental thereto or substituted therefore.

"Collateral" shall mean all property pledged, mortgaged or purported to be pledged or mortgaged pursuant to the Security Documents (including the Orders) including Owned Real Property owned by the Credit Parties as of the Closing Date and all Owned Real Property acquired by the Credit Parties after the Closing Date, but excluding, for the avoidance of doubt, all Excluded Collateral.

"Collateral Agent" shall mean Wilmington Savings Fund Society, FSB, in its capacity as collateral agent for the Secured Parties under this Agreement and the Security Documents, or any successor collateral agent appointed pursuant hereto.

"Commitment Letter" shall mean that certain Debtor in Possession Term Loan Facility Commitment Letter, dated as of February 14, 2023, by and among the Commitment Parties and the Borrower.

"Commitment Party" shall mean any "Commitment Party" party to the Commitment Letter.

"Commitments" shall mean, with respect to each Lender (to the extent applicable), such Lender's Term Commitments.

"Commodity Exchange Act" shall mean the Commodity Exchange Act (7 U.S.C. §1 et seq.), as amended from time to time, and any successor statute.

"Communications" shall have the meaning provided in Section 13.17(a).

"Confidential Information" shall mean with respect to information provided pursuant to a customary "private side" virtual data room, all information received from Holdings, the Borrower or any of their respective Subsidiaries relating to Holdings, the Borrower or any of their respective Subsidiaries or any of their respective businesses in connection with this Agreement or the transactions contemplated hereby, other than any such information that is available to any Agent or any Lender on a nonconfidential basis prior to disclosure by Holdings, the Borrower or any of their respective Subsidiaries or any such information that is clearly marked as non-confidential or provided pursuant to a customary "public side" virtual data room.

"Conforming Changes" shall mean, in each case solely to the extent relating to the Term Loans, with respect to either the use or administration of Term SOFR or the use, administration, adoption or implementation of any Benchmark Replacement, any technical, administrative or operational changes (including changes to, in each case solely to the extent relating to the Term Loans, the definition of "ABR," the definition of "Business Day," the definition of "U.S. Government Securities Business Day," the definition of "Interest Period" or any similar or analogous definition (or the addition of a concept of "interest period"), timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, the applicability and length of lookback periods, the applicability of Section 2.11 and other technical, administrative or operational matters) that the Administrative Agent decides may be appropriate to reflect the adoption and implementation of any such rate or to permit the use and administration thereof by the Administrative Agent in a manner substantially consistent with market practice (or, if the Administrative Agent decides that adoption of any portion of such market practice is not administratively feasible or if the Administrative Agent determines that no market practice for the administration of any such rate exists, in such other manner of administration as the Administrative Agent decides is reasonably necessary in connection with the administration of this Agreement and the other Credit Documents).

"Consolidated Depreciation and Amortization Expense" shall mean, with respect to the Borrower and the Subsidiaries for any period, the total amount of depreciation and amortization expense, including the amortization of deferred financing fees or costs, debt issuance costs, commissions, fees and expenses, capitalized expenditures, Capitalized Software Expenditures, amortization of expenditures relating to software, license and intellectual property payments, amortization of any lease related assets recorded in purchase accounting, customer acquisition costs, unrecognized prior service costs and actuarial gains and losses related to pensions and other post-employment benefits, amortization of original issue discount resulting from the issuance of Indebtedness at less than par and incentive payments, conversion costs, and contract acquisition costs of the Borrower and the Subsidiaries for such period on a consolidated basis and otherwise determined in accordance with GAAP.

"Consolidated EBITDA" shall mean, for any period, Consolidated Net Income for such period, *plus*:

(a) without duplication and (except in the case of the add-backs set forth in clauses (vii) and (xi) below) to the extent deducted (and not added back) in arriving at such Consolidated Net Income, the sum of the following amounts for the Borrower and the Subsidiaries for such period:

- (i) Fixed Charges (including (x) net losses on Hedging Obligations or other derivative instruments entered into for the purpose of hedging interest rate risk and (y) costs of surety bonds in connection with financing activities in each case to the extent included in Consolidated Interest Expense, together with items excluded from Consolidated Interest Expense pursuant to clause (1)(o) (z) of the definition thereof),
- (ii) provision for taxes based on income or profits or capital gains, including federal, foreign, state, franchise, excise, value-added and similar taxes and foreign withholding taxes (including penalties and interest related to such taxes or arising from tax examinations) paid or accrued during such period, including any penalties and interest related to such taxes or arising from any tax examination, to the extent the same were deducted (and not added back) in computing such Consolidated Net Income and the net tax expense associated with any adjustments made pursuant to clauses (a) through (t) of the definition of "Consolidated Net Income",
  - (iii) Consolidated Depreciation and Amortization Expense for such period,
- (iv) the amount of any restructuring cost, charge or reserve (including any costs incurred in connection with acquisitions after the Closing Date and costs related to the closure and/or consolidation of facilities) and any one time expense relating to enhanced accounting function or other transaction costs, public company costs, costs and expenses in connection with the implementation of fresh start accounting, and costs related to the implementation of operational and reporting systems and technology initiatives (*provided* that such costs related to the implementation of operation and reporting systems and technology initiatives shall not exceed \$50,000,000 for any such period),
- (v) any other non-cash charges, expenses or losses, including any non-cash asset retirement costs, non-cash increase in expenses resulting from the revaluation of inventory (including any impact of changes to inventory valuation policy methods including changes in capitalization of variances) or other inventory adjustments or due to purchase accounting, or any other acquisition, non-cash compensation charges, non-cash expense relating to the vesting of warrants, write-offs or write-downs for such period (*provided* that if any such non-cash charges represent an accrual or reserve for potential cash items in any future period, the cash payment in respect thereof in such future period shall be subtracted from Consolidated EBITDA to such extent, and excluding amortization of a prepaid cash item that was paid in a prior period),
- (vi) the amount of any minority interest expense consisting of Subsidiary income attributable to minority equity interests of third parties in any non-Wholly Owned Subsidiary,
- (vii) an amount (which, for the avoidance of doubt, if positive, increases Consolidated EBITDA or, if negative, reduces Consolidated EBITDA) equal to (x) the Aggregate Quarterly Subscription Contract ARR Revenue for such period minus (y) the Aggregate Quarterly Subscription Contract GAAP Revenue for such period,

# (viii) [reserved],

- (ix) contract termination costs and any costs, charges or expenses incurred pursuant to any management equity plan or stock option plan or any other management or employee benefit plan or agreement or any stock subscription or shareholder agreement or other equity-based compensation, to the extent that such costs or expenses are funded with cash proceeds contributed to the capital of the Borrower or Net Cash Proceeds of an issuance of Stock or Stock Equivalents (other than Disqualified Stock) of the Borrower (or any direct or indirect parent thereof),
  - (x) [reserved],
  - (xi) the proceeds of any business interruption insurance,
- (xii) extraordinary, unusual or non-recurring charges, expenses or losses (including unusual or non-recurring expenses), transaction fees and expenses and consulting and advisory fees, indemnities and expenses, severance, integration costs, costs of strategic initiatives, relocation costs, consolidation and closing costs, facility opening and pre-opening costs, business optimization expenses or costs, transition costs, restructuring costs, signing, retention, recruiting, relocation, signing, stay or completion bonuses and expenses (including payments made to employees who are subject to non-compete agreements),
- (xiii) any impairment charge or asset write-off or write-down including impairment charges or asset write-offs or write-downs related to intangible assets, long-lived assets and Investments in debt and equity securities, in each case pursuant to GAAP, and the amortization of intangibles arising pursuant to GAAP,
- (xiv) cash receipts (or any netting arrangements resulting in increased cash receipts) not added in arriving at Consolidated EBITDA or Consolidated Net Income in any period to the extent the non-cash gains relating to such receipts were deducted in the calculation of Consolidated EBITDA pursuant to paragraph (b) below for any previous period and not added, *less*
- (b) without duplication and to the extent included in arriving at such Consolidated Net Income for the Borrower and the Subsidiaries, the sum of the following amounts for such period:
- (i) non-cash gains increasing Consolidated Net Income for such period (excluding any non-cash gain to the extent it represents the reversal of an accrual or reserve for a potential cash item that reduced Consolidated Net Income or Consolidated EBITDA in any prior period),
  - (ii) extraordinary, unusual or non-recurring gains,
- (iii) cash expenditures (or any netting arrangements resulting in increased cash expenditures) not deducted in arriving at Consolidated EBITDA or Consolidated Net Income in any period to the extent non-cash losses relating to such expenditures were added in the calculation of Consolidated EBITDA pursuant to paragraph (a) above for any previous period and not deducted, and

(iv) the amount of any minority interest income consisting of Subsidiary losses attributable to minority equity interests of third parties in any non-Wholly Owned Subsidiary,

in each case, as determined on a consolidated basis for the Borrower and the Subsidiaries in accordance with GAAP.

"Consolidated Interest Expense" shall mean, with respect to any period, without duplication, the sum of:

- consolidated interest expense of the Borrower and the Subsidiaries for such period, (1) to the extent such expense was deducted (and not added back) in computing Consolidated Net Income (including (a) amortization of original issue discount resulting from the issuance of Indebtedness at less than par, (b) all commissions, discounts and other fees and charges owed with respect to letters of credit, bankers' acceptances or collateral posting facilities, (c) non-cash interest payments (but excluding any non-cash interest expense attributable to the movement in the mark to market valuation of Hedging Obligations or other derivative instruments pursuant to GAAP), (d) the interest component of Capitalized Lease Obligations and (e) net payments, if any, pursuant to interest rate Hedging Obligations with respect to Indebtedness, and excluding (o) annual agency fees paid to the administrative agents and collateral agents under this Agreement and the DIP ABL Facility, if any, (p) additional interest with respect to failure to comply with any registration rights agreement owing to holders of any securities, (q) costs associated with obtaining Hedging Obligations, (r) accretion of asset retirement obligations and accretion or accrual of discounted liabilities not constituting Indebtedness, (s) any expense resulting from the discounting of any Indebtedness in connection with the application of fresh start accounting or purchase accounting, (t) penalties and interest relating to taxes (u) amortization of reacquired Indebtedness, deferred financing fees, debt issuance costs, commissions, fees and expenses, (v) any expensing of bridge, commitment and other financing fees, (w) [reserved], (x) any prepayment premium or penalty, (y) any interest expense attributable to a parent entity resulting from push-down accounting and (z) any lease, rental or other expenses from operating leases); plus
- (2) consolidated capitalized interest of the Borrower and the Subsidiaries, in each case for such period, whether paid or accrued; *less*
- (3) interest income for such period.

For purposes of this definition, interest on a Capitalized Lease Obligation shall be deemed to accrue at an interest rate reasonably determined by such Person to be the rate of interest implicit in such Capitalized Lease Obligation in accordance with GAAP.

"Consolidated Net Income" shall mean, for any period, the net income (loss) of the Borrower and the Subsidiaries for such period determined on a consolidated basis in accordance with GAAP, excluding, without duplication, the net after-tax effect of,

- (a) any extraordinary, unusual or nonrecurring losses, gains, fees, costs, charges or expenses for such period,
- (b) Transaction Expenses,
- (c) the cumulative effect of a change in accounting principles and changes as a result of adoption or modification of accounting policies during such period,
- (d) any income (or loss) from disposed, abandoned or discontinued operations and any gains or losses on disposal of disposed, abandoned, transferred, closed or discontinued operations,
- (e) any gains or losses (less all fees and expenses relating thereto) attributable to asset dispositions or abandonments other than in the ordinary course of business, as determined in good faith by the Borrower,
- (f) [reserved],
- (g) [reserved],
- (h) all adjustments (including the effects of such adjustments pushed down to the Borrower and the Subsidiaries) in the Borrower's consolidated financial statements pursuant to GAAP, resulting from the application of purchase accounting in relation to the Transactions or any consummated acquisition, in each case, including the amortization, write-off or write-down of any assets, any deferred revenue and any other amounts and other similar adjustments and, whether consummated before or after the Closing Date,
- (i) any income (or loss) for such period attributable to the early extinguishment of Indebtedness (other than Hedging Obligations, but including, for the avoidance of doubt, debt exchange transactions and the extinguishment of pre-petition indebtedness in connection with the Transactions),
- (j) any unrealized income (or loss) for such period attributable to Hedging Obligations or other derivative instruments,
- (k) any impairment charge or asset write-off or write-down including impairment charges or asset write-offs or write-downs related to intangible assets, long-lived assets and investments in debt and equity securities or as a result of a change in law or regulation, in each case pursuant to GAAP,
- (l) any non-cash compensation expense recorded from grants of stock appreciation or similar rights, stock options, restricted stock or other rights, and any cash charges associated with the rollover, acceleration or payout of Stock or Stock Equivalents by management of the Borrower or any of its direct or indirect parent companies in connection with the Transactions,

- (m) accruals and reserves established or adjusted within twelve months after the Closing Date that are so required to be established as a result of the Transactions in accordance with GAAP or changes as a result of adoption of or modification of accounting policies during such period,
- (n) any accruals, payments, fees, expenses or charges (including rationalization, legal, tax, structuring, and other costs and expenses, but excluding depreciation or amortization expense) related to, or incurred in connection with, the Transactions (including letter of credit fees), the Plan, any offering of Stock or Stock Equivalents (including any equity offering), Investment, acquisition, Disposition, Restricted Payment, recapitalization or the issuance or incurrence of Indebtedness permitted to be incurred by the Borrower and the Subsidiaries pursuant hereto (including any refinancing transaction or amendment, waiver, or other modification of any debt instrument), in each case whether or not consummated, including (A) such fees, expenses or charges related to the negotiation, execution and delivery and other transactions contemplated by this Agreement, the other Credit Documents, (B) any amendment or other modification of this Agreement and the other Credit Documents, (C) any such transaction consummated prior to the Closing Date and any such transaction undertaken but not completed, (D) any charges or nonrecurring merger costs as a result of any such transaction, and (E) earnout obligations paid or accrued during such period with respect to any acquisition or other Investment,
- (o) [reserved],
- (p) restructuring-related or other similar charges, fees, costs, commissions and expenses or other charges incurred during such period in connection with this Agreement, the other Credit Documents, the DIP ABL Facility, the Chapter 11 Cases, any reorganization plan in connection with the Case, and any and all transactions contemplated by the foregoing, including the write-off of any receivables, the termination or settlement of executory contracts, professional and accounting costs fees and expenses, management incentive, employee retention or similar plans (in each case to the extent such plan is approved by the Bankruptcy Court to the extent required), litigation costs and settlements, asset write-downs, income and gains recorded in connection with the corporate reorganization of the Debtors;
- (q) any expenses, charges or losses that are covered by indemnification or other reimbursement provisions in connection with any Investment or any sale, conveyance, transfer or other disposition of assets permitted under this Agreement, to the extent actually reimbursed, or, so long as the Borrower has made a determination that a reasonable basis exists for indemnification or reimbursement and only to the extent that such amount is in fact indemnified or reimbursed within 365 days of such determination (with a deduction in the applicable future period for any amount so added back to the extent not so indemnified or reimbursed within such 365 days),

- (r) to the extent covered by insurance and actually reimbursed, or, so long as the Borrower has made a determination that there exists reasonable evidence that such amount will in fact be reimbursed by the insurer and only to the extent that such amount is in fact reimbursed within 365 days of the date of such determination (with a deduction in the applicable future period for any amount so added back to the extent not so reimbursed within such 365 days), expenses, charges or losses with respect to liability or casualty events or business interruption,
- (s) any net unrealized gain or loss (after any offset) resulting from currency translation gains or losses relating to currency remeasurements of Indebtedness (including any gain or loss resulting from obligations under any Hedging Obligation for currency exchange risk) and any foreign currency translation gains or losses, and
- (t) to the extent non-cash and deducted in calculating net income (or loss), any net pension, post-employment benefit or long-term disability costs, including interest cost, service cost, actuarial expected return on plan assets, amortization of unrecognized prior service costs, actuarial losses, including amortization of such amounts arising in prior periods, amortization of unrecognized net obligations (and loss or cost) existing at the date of initial application of FASB Standard 87, 106 and 112 (or their equivalents under the ASC), and any other items of a similar nature and any gain or loss attributable to mark-to-market adjustments in the valuation of pension liabilities, including actuarial gain or loss on pension and post-retirement plans, curtailments and settlements and prior service cost adjustment.

"Consummation Date" means the date of the substantial consummation (as defined in section 1101 of the Bankruptcy Code) of a Reorganization Plan that is confirmed pursuant to an order of the Bankruptcy Court; provided, that for purposes hereof the Consummation Date of the Reorganization Plan shall be no later than the "effective date" thereof.

"Contingent Obligation" shall mean indemnification Obligations and other similar contingent Obligations for which no claim has been made in writing.

"Contractual Requirement" shall have the meaning provided in Section 8.3.

"Conversion Date" means the date upon which the Plan is consummated and all conditions precedent to the effectiveness of the Exit Term Loan Facility shall have been satisfied or waived.

"Credit Documents" shall mean this Agreement, the Guarantee, the Security Documents, the Fee Letter, any promissory notes issued by the Borrower hereunder, the Commitment Letter and any other document jointly identified by the Borrower and the Administrative Agent as a "Credit Document."

"Credit Facility" shall mean any category of Commitments and/or Term Loans and other extensions of credit thereunder.

"Credit Party" shall mean each of Holdings, the Borrower and each of the Subsidiary Guarantors.

- "Creditors' Committee" shall have the meaning specified in the Interim Order or the Final Order, as applicable.
- "Debt Incurrence Prepayment Event" shall mean any issuance or incurrence by the Borrower or any of the Subsidiaries of any Indebtedness (other than as permitted to be issued or incurred under Section 10.1).
  - "Debtors" shall have the meaning provided in the Recitals to this Agreement.
  - "Declined Proceeds" shall have the meaning provided in Section 5.2(f).
- "**Default**" shall mean any event, act or condition that with notice or lapse of time hereunder, or both, would constitute an Event of Default.
  - "**Default Rate**" shall have the meaning provided in Section 2.8(d).
- "Defaulting Lender" shall mean any Lender with respect to which a Lender Default is in effect.
- "Deferred Net Cash Proceeds" shall have the meaning provided such term in the definition of "Net Cash Proceeds".
- "Deferred Net Cash Proceeds Payment Date" shall have the meaning provided such term in the definition of "Net Cash Proceeds".
- "DIP ABL Facility" shall mean an asset-based revolving facility to be incurred by the Borrower and the other Credit Parties on or after the Closing Date in an aggregate commitment amount up to \$200,000,000 pursuant to terms reasonably satisfactory to the Required Lenders (it being understood and agreed that the terms set forth in the term sheet dated as of February 13 (the "DIP ABL Term Sheet") shall be deemed to be satisfactory to the Lenders, including with respect to the lien priority exhibit attached thereto, which for the avoidance of doubt is identical to the lien priority exhibit attached to the Commitment Letter).
- "DIP ABL Intercreditor Agreement" shall mean an intercreditor agreement providing for, among other things, priority of the DIP ABL Facility (and any cash management obligations and hedging obligations secured in connection therewith) over the Specified Collateral and subordination of the DIP ABL Facility with respect to other Collateral, in form and substance pursuant to the terms set forth in the Commitment Letter (and the term sheet exhibited thereto) or otherwise reasonably satisfactory to the Agents and the Required Lenders.
- "DIP ABL Term Sheet" shall have the meaning set forth in the definition of "DIP ABL Facility".
- "DIP Motion" shall mean that certain Debtors' Emergency Motion for Entry of Interim and Final Orders (I) Authorizing the Debtors to (A) Obtain Postpetition Financing and (B) Utilize Cash Collateral, (II) Granting Liens and Superpriority Administrative Expense Claims, (III) Granting Adequate Protection to the Prepetition Secured Parties, (IV) Modifying the

Automatic Stay, and (V) Granting Related Relief, filed in the Chapter 11 Cases on the Petition Date [Docket No. []].

"DIP Proceeds Account" shall have the meaning set forth in Section 9.18.

"DIP Projection" shall mean the sensitized latest financial outlook projections of the Borrower and its Subsidiaries (prepared on a monthly basis) through the end of the fiscal year ended September 30, 2023, in a form customary for "DIP budgets" in form and substance satisfactory to the Required Lenders in their sole discretion.

"DIP Term Priority Collateral" means all Collateral other than Specified Collateral, but including, for the avoidance of doubt, the Foreign Reserve Account, the DIP Proceeds Account, and any amounts deposited therein, and proceeds of Avoidance Action (after the entry of the Final Order).

"Disclosure Statement" shall have the meaning set forth in Section 9.16.

"Disposition" or "Dispose" shall mean (i) the convey, sale, lease, assignment, transfer or other disposition of any of property, business or assets (including receivables and leasehold interests), whether owned on the Closing Date or hereafter acquired or (ii) the sale to any Person (other than to the Borrower or a Subsidiary Guarantor) any shares owned by it of any Subsidiary's Stock and Stock Equivalents.

"Disqualified Stock" shall mean, with respect to any Person, any Stock or Stock Equivalents of such Person which, by its terms, or by the terms of any security into which it is convertible or for which it is putable or exchangeable, or upon the happening of any event, matures or is mandatorily redeemable (other than solely for Stock or Stock Equivalents that is not Disqualified Stock), other than as a result of a change of control, asset sale or similar event so long as any rights of the holders thereof upon the occurrence of such change of control, asset sale or similar event shall be subject to the prior repayment in full of the Term Loans and all other Obligations (other than Contingent Obligations) and the termination of all Commitments, pursuant to a sinking fund obligation or otherwise, or is redeemable at the option of the holder thereof (other than as a result of a change of control, asset sale or similar event so long as any rights of the holders thereof upon the occurrence of such change of control, asset sale or similar event shall be subject to the prior repayment in full of the Term Loans and all other Obligations (other than Contingent Obligations) and the termination of all Commitments), in whole or in part, in each case prior to the date that is ninety-one (91) days after the Scheduled Maturity Date as determined at the time of the issuance.

"Dollars" and "\$" shall mean dollars in lawful currency of the United States of America.

"**Domestic Subsidiary**" shall mean each Subsidiary of the Borrower that is organized under the laws of the United States or any state thereof, or the District of Columbia.

"EEA Financial Institution" shall mean (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" shall mean any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"EEA Resolution Authority" shall mean any public administrative authority or any person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution.

"Employee Benefit Plan" shall mean an employee benefit plan (as defined in Section 3(3) of ERISA), other than a Foreign Plan, that is maintained or contributed to by Holdings, Borrower or any Subsidiary (or, with respect to an employee benefit plan subject to Title IV of ERISA, any ERISA Affiliate).

"Environmental Claims" shall mean any and all actions, suits, proceedings, orders, decrees, demands, demand letters, claims, liens, notices of noncompliance, violation or potential responsibility or investigation (other than reports prepared by or on behalf of Holdings, the Borrower or any other Subsidiary of Holdings (a) in the ordinary course of such Person's business or (b) as required in connection with a financing transaction or an acquisition or disposition of Real Estate) or proceedings in each case relating in any way to any applicable Environmental Law or any permit issued, or any approval given, under any applicable Environmental Law (hereinafter, "Claims"), including (i) any and all Claims by Governmental Authorities for enforcement, cleanup, removal, response, remedial or other actions or damages pursuant to any applicable Environmental Law and (ii) any and all Claims by any third party seeking damages, contribution, indemnification, cost recovery, compensation or injunctive relief relating to the presence, release or threatened release into the environment of Hazardous Materials or arising from alleged injury or threat of injury to human health or safety (to the extent relating to human exposure to Hazardous Materials), or to the environment, including ambient air, indoor air, surface water, groundwater, land surface and subsurface strata and natural resources such as wetlands.

"Environmental Law" shall mean any applicable Federal, state, foreign or local statute, law, rule, regulation, ordinance, code and rule of common law now or, with respect to any post-Closing Date requirements of the Credit Documents, hereafter in effect, and in each case as amended, and any legally binding judicial or administrative interpretation thereof, including any legally binding judicial or administrative order, consent decree or judgment, relating to the protection of the environment, including ambient air, indoor air, surface water, groundwater, land surface and subsurface strata and natural resources such as wetlands, or to human health or safety (to the extent relating to human exposure to Hazardous Materials), or Hazardous Materials.

**"ERISA**" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time. Section references to ERISA are to ERISA as in effect on the Closing Date and any subsequent provisions of ERISA amendatory thereof, supplemental thereto or substituted therefor.

"ERISA Affiliate" shall mean each person (as defined in Section 3(9) of ERISA) that together with the Borrower or any Subsidiary of the Borrower would be deemed to be a "single employer" within the meaning of Section 414(b) or (c) of the Code or, solely for purposes of Section 302 of ERISA and Section 412 of the Code, is treated as a single employer under Section 414 of the Code.

"ERISA Event" shall mean (i) the failure of any Employee Benefit Plan to comply with any provisions of ERISA and/or the Code or with the terms of such Employee Benefit Plan; (ii) any Reportable Event; (iii) the existence with respect to any Employee Benefit Plan of a nonexempt Prohibited Transaction; (iv) any failure by any Pension Plan to satisfy the minimum funding standards (within the meaning of Section 412 of the Code or Section 302 of ERISA) applicable to such Pension Plan, whether or not waived; (v) the filing pursuant to Section 412(c) of the Code or Section 302(c) of ERISA of an application for a waiver of the minimum funding standard with respect to any Pension Plan; (vi) the occurrence of any event or condition which would reasonably be expected to constitute grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan or the incurrence by any Credit Party or any of its ERISA Affiliates of any liability under Title IV of ERISA with respect to the termination of any Pension Plan, including but not limited to the imposition of any Lien in favor of the PBGC or any Pension Plan; (vii) the receipt by any Credit Party or any of its ERISA Affiliates from the PBGC or a plan administrator of any written notice to terminate any Pension Plan under Section 4042(a) of ERISA or to appoint a trustee to administer any Pension Plan under Section 4042(b)(1) of ERISA; (viii) the incurrence by any Credit Party or any of its ERISA Affiliates of any liability with respect to the withdrawal or partial withdrawal from any Pension Plan (or a cessation of operations that is treated as such a withdrawal under Section 4062(e) of ERISA) or Multiemployer Plan; (ix) the receipt by any Credit Party or any of its ERISA Affiliates of any notice concerning the imposition on it of Withdrawal Liability or a determination that a Multiemployer Plan is, or is expected to be, Insolvent or in "endangered" or "critical" status (within the meaning of Section 432 of the Code or Section 305 of ERISA), or terminated (within the meaning of Section 4041A of ERISA), (x) a determination that any Pension Plan is or is expected to be in "at risk" status (within the meaning of Section 430 of the Code or Section 303 of ERISA); or (xi) any other event or condition with respect to a Pension Plan or Multiemployer Plan that could result in liability to the Borrower or any Subsidiary.

"Escrow Payment" shall have the meaning provided in the Restructuring Support Agreement,

"EU Bail-In Legislation Schedule" shall mean the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor person), as in effect from time to time.

"Event of Default" shall have the meaning provided in Section 11.

**"Exchange Act"** shall mean the Securities Exchange Act of 1934, as amended, and rules and regulations promulgated thereunder.

"Excluded Collateral" shall mean (i) any "intent to use" trademark application filed and accepted in the United States Patent and Trademark Office unless and until an amendment

to allege use or a statement of use has been filed and accepted by the United States Patent and Trademark Office to the extent, if any, that, and solely during the period, if any, in which the grant of security interest therein could impair the validity or enforceability of such "intent to use" trademark application under federal law; (ii) any charter, permit, franchise, authorization, lease, license or agreement, in each case, only to the extent and for so long as the grant of a security interest therein (or the assets subject thereto) by the applicable Credit Party (x) would violate invalidate such charter, permit, franchise, authorization, lease, license, or agreement or (y) would give any party (other than a Credit Party) to any such charter, permit, franchise, authorization, lease, license or agreement the right to terminate its obligations thereunder or (z) is permitted under such charter, permit, franchise, lease, license or agreement only with consent of the parties thereto (other than consent of a Credit Party) and such necessary consents to such grant of a security interest have not been obtained (it being understood and agreed that no Credit Party or Subsidiary has any obligation to obtain such consents) other than, in each case referred to in clauses (x) and (y) and (z), as would be rendered ineffective pursuant to Sections 9-406, 9-407, 9-408 or 9-409 of the Uniform Commercial Code of any relevant jurisdiction, in each case excluding the proceeds and receivables thereof which are not otherwise Excluded Collateral; (iii) any Excluded Stock and Stock Equivalents; (iv) any margin stock; (v) all Leased Real Property, provided, that proceeds of Leased Real Property shall not be Excluded Collateral; (vi) any assets with respect to which granting a security interest in such assets is prohibited by or would violate law, treaty, rule, or regulation or determination of an arbitrator or a court or other Governmental Authority or which would require obtaining the consent, approval, license or authorization of any Governmental Authority (unless such consent, approval, license or authorization has been received; provided that there shall be no obligation to obtain such consent) or create a right of termination in favor of any governmental or regulatory third party, in each case after giving effect to Sections 9-406, 9-407, 9-408 or 9-409 of the Uniform Commercial Code of any relevant jurisdiction or other Applicable Law, (vii) any payroll, tax and trust account and any account holding cash and Cash Equivalents collateralizing the Existing Letters of Credit, Existing Cash Management Obligations and Existing Secured Hedging Obligations and (viii) Avoidance Actions and, prior to entry of the Final Order, the proceeds of Avoidance Actions (it being understood that notwithstanding such exclusion of Avoidance Actions, upon entry of the Final Order, to the extent approved by the Bankruptcy Court, such liens shall attach to the proceeds of Avoidance Actions), excluding, in each case of clauses (i) through (vii), the proceeds and receivables thereof (to the extent not otherwise constituting Excluded Collateral; provided that with respect to clauses (ii) and (vi), such property shall be Excluded Collateral only to the extent and for so long as such prohibition, violation, invalidation or consent right, as applicable, is in effect and in the case of any such agreement or consent, was not created in contemplation thereof or of the creation of a security interest therein.

"Excluded Stock and Stock Equivalents" shall mean (i) any Stock or Stock Equivalents to the extent the pledge thereof would violate any Applicable Law or any Contractual Requirement (including any legally effective requirement to obtain the consent or approval of, or a license from, any Governmental Authority or any other regulatory third party unless such consent, approval or license has been obtained (it being understood that the foregoing shall not be deemed to obligate the Borrower or any Subsidiary of the Borrower to obtain any such consent, approval or license)) and (ii) any Stock or Stock Equivalents that are margin stock; *provided* that Excluded Stock and Stock Equivalents shall not include proceeds of the foregoing property to the extent otherwise constituting Collateral as in effect on the Closing Date or any date of such acquisition of such Stock or Stock Equivalents.

"Excluded Subsidiary" shall mean (a) each Domestic Subsidiary that is (i) prohibited by any applicable (x) Contractual Requirement, (y) Applicable Law (including without limitation as a result of applicable financial assistance, directors' duties or corporate benefit requirements) or (z) Organizational Document (in the case of clauses (x) and (z), in effect on the Closing Date or any date of acquisition of such Subsidiary (to the extent such prohibition was not entered into in contemplation of the Guarantee)) from guaranteeing or granting Liens to secure the Obligations at the time such Subsidiary becomes a Subsidiary (and for so long as such restriction or any replacement or renewal thereof is in effect), or (ii) required to obtain consent, approval, license or authorization of a Governmental Authority for such guarantee or grant (unless such consent, approval, license or authorization has already been received); provided that there shall be no obligation to obtain such consent, (b) any Foreign Subsidiary, (c) any Captive Insurance Subsidiary, (d) any non-profit Subsidiary or (e) any Broker-Dealer Subsidiary; provided that Excluded Subsidiary shall not include any Debtor.

"Excluded Taxes" shall mean, any of the following Taxes imposed on or with respect to any Agent or any Lender or required to be deducted or withheld from a payment to any Agent or Lender, (a) net income Taxes and franchise and excise Taxes (imposed in lieu of net income Taxes) and any branch profits Taxes imposed on such Agent or Lender imposed as a result of such Agent or Lender being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable lending office located in the jurisdiction imposing such Tax (or any political subdivision thereof), (b) any Taxes imposed on any Agent or any Lender as a result of any current or former connection between such Agent or Lender and the jurisdiction of the Governmental Authority imposing such Tax or any political subdivision or taxing authority thereof or therein (other than any such connection arising solely from such Agent or Lender having executed, delivered or performed its obligations or received a payment under, or having been a party to or having enforced, this Agreement or any other Credit Document), (c) any U.S. federal withholding Tax that is imposed on amounts payable to or for the account of any Agent or Lender under the law in effect at the time such Agent or Lender becomes a party to this Agreement (or designates a new lending office other than a new lending office designated at the request of the Borrower pursuant to Section 13.7(a)); provided that this clause (c) shall not apply to the extent that the indemnity payments or additional amounts any Lender would be entitled to receive (without regard to this clause (c)) do not exceed the indemnity payment or additional amounts that the person making the assignment, participation or transfer to such Lender (or designation of a new lending office by such Lender) would have been entitled to receive pursuant to Section 5.4 immediately before such assignment, participation, transfer or change in lending office in the absence of such assignment, participation, transfer or change in lending office (it being understood and agreed, for the avoidance of doubt, that any withholding Tax imposed on a Lender as a result of a Change in Law occurring after the time such Lender became a party to this Agreement (or designates a new lending office) shall not be an Excluded Tax under this clause (c)), (d) any Tax to the extent attributable to such Agent's or Lender's failure to comply with Sections 5.4(e), (f) (in the case of any Non-U.S. Lender) or Section 5.4(i) (in the case of a U.S. Lender) or Section 5.4(j) and (e) any Taxes imposed by FATCA.

**"Existing Cash Management Obligations"** means any and all obligations in respect of "Cash Management Agreements" identified in the Prepetition ABL Payoff Letter.

"Existing Hedging Obligations" means any and all obligations in respect of "Secured Hedging Agreements" identified in the Prepetition ABL Payoff Letter.

"Existing Letters of Credit" means those letters of credit identified in the Prepetition ABL Payoff Letter.

"Exit Fee" shall have the meaning assigned to such term in Section 4.1(d).

**"Exit Term Loan Facility"** means the credit agreement to be entered into on the Conversion Date as contemplated by the terms specified in the Exit Term Loan Term Sheet.

**"Exit Term Loan Term Sheet"** means the "Exit Term Loan Term Sheet" as defined in the Restructuring Support Agreement.

"Exit Term Loans" means term loans incurred under the Exit Term Loan Facility.

"Extraordinary Receipts" shall mean any cash receipts received by any Credit Party or any Subsidiary thereof not in the ordinary course of business in respect of any Tax refunds or Employee Benefit Plan reversions received by any Credit Party.

**"Extraordinary Receipts Prepayment Event"** shall mean the receipt of Net Cash Proceeds with respect to any Extraordinary Receipts.

"FATCA" shall mean Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future Treasury regulations or official interpretations thereof, any agreements entered into pursuant to Section 1471(b)(1) of the Code, and intergovernmental agreement (together with any Applicable Law implementing such agreement) entered into in connection with any of the foregoing.

"Federal Funds Effective Rate" shall mean, for any day, the weighted average of the *per annum* rates on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers on such day, as published on the next succeeding Business Day by the Federal Reserve Bank of New York; *provided* that (a) if such day is not a Business Day, the Federal Funds Effective Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Effective Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of 1/100 of 1%) charged on such day on such transactions as determined by the Administrative Agent; *provided* that is the Federal Funds Effective Rate for any day is less than zero, the Federal Funds Effective Rate for such day will be deemed to be zero.

"Fee Letter" shall mean the Agent Fee Letter, dated as of February 15, 2023, between Wilmington Savings Fund Society, FSB and Avaya Inc.

"Fees" shall mean all amounts payable pursuant to, or referred to in, Section 4.1.

"Final Order" has the meaning specified in Section 7.2.

"Final Order Entry Date" means the date on which the Final Order is entered by the Bankruptcy Court.

"First Day Orders" means all orders entered by the Bankruptcy Court on or, within five days of, the Petition Date, or based on motions filed on or about the Petition Date.

"Fiscal Year" shall have the meaning provided in Section 9.10.

"Fixed Charges" shall mean, the sum of, without duplication:

- (1) Consolidated Interest Expense; *plus*
- (2) all cash dividends or cash distributions (other than return of capital) paid (excluding items eliminated in consolidation) on any series of preferred stock during such period; *plus*
- (3) all cash dividends or cash distributions (other than return of capital) paid (excluding items eliminated in consolidation) on any series of Disqualified Stock during such period.

"Flood Laws" means (i) the National Flood Insurance Act of 1968 as now or hereafter in effect or any successor statute thereto, (ii) the Flood Disaster Protection Act of 1973 as now or hereafter in effect or any successor statute thereto, (iii) the National Flood Insurance Reform Act of 1994 (amending 42 USC 4001, et seq.) and (iv) the Flood Insurance Reform Act of 2004, and any regulations promulgated thereunder.

"Floor" shall mean a rate of interest equal to 1.00%.

"Foreign Asset Sale" shall have the meaning provided in Section 5.2(g).

"Foreign Intercompany Note" shall mean the Foreign Intercompany Note, dated as of the date hereof, evidencing the Non-Debtor Subsidiary Notes/Receivables, by and among the payees and payors party thereto.

"Foreign Plan" shall mean any employee benefit plan, program, policy, arrangement or agreement maintained or contributed to by the Borrower or any of its Subsidiaries with respect to employees employed outside the United States.

"Foreign Recovery Event" shall have the meaning provided in Section 5.2(g).

"Foreign Reserve Account" shall have the meaning set forth in Section 9.13(a).

"Foreign Reserve Account Withdrawal" shall have the meaning set forth in Section 9.13(b).

"Foreign Reserve Account Withdrawal Notice" shall have the meaning set forth in Section 9.13(b).

"Foreign Reserve Protocol" shall have the meaning set forth in Section 9.13(b).

"Foreign Subsidiary" shall mean each Subsidiary of the Borrower that is not a Domestic Subsidiary.

"Fund" shall mean any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course.

"GAAP" shall mean generally accepted accounting principles in the United States of America, as in effect from time to time; provided, however, that if the Borrower notifies the Administrative Agent that the Borrower requests an amendment to any provision hereof to eliminate the effect of any change occurring after the Closing Date in GAAP or in the application thereof on the operation of such provision (or if the Administrative Agent notifies the Borrower that the Required Lenders request an amendment to any provision hereof for such purpose), regardless of whether any such notice is given before or after such change in GAAP or in the application thereof, then such provision shall be interpreted on the basis of GAAP as in effect and applied immediately before such change shall have become effective until such notice shall have been withdrawn or such provision amended in accordance herewith.

"Governmental Authority" shall mean any nation, sovereign or government, any state, province, territory or other political subdivision thereof, and any entity or authority exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, including a central bank or stock exchange.

"Granting Lender" shall have the meaning provided in Section 13.6(f).

"Guarantee" shall mean the Guarantee, dated as of the Closing Date, made by each Guarantor in favor of the Administrative Agent for the benefit of the Secured Parties.

"Guarantee Obligations" shall mean, as to any Person, any obligation of such Person guaranteeing or intended to guarantee any Indebtedness of any other Person (the "primary obligor") in any manner, whether directly or indirectly, including any obligation of such Person, whether or not contingent, (a) to purchase any such Indebtedness or any property constituting direct or indirect security therefor, (b) to advance or supply funds (i) for the purchase or payment of any such Indebtedness or (ii) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (c) to purchase property, securities or services primarily for the purpose of assuring the owner of any such Indebtedness of the ability of the primary obligor to make payment of such Indebtedness or (d) otherwise to assure or hold harmless the owner of such Indebtedness against loss in respect thereof; provided, however, that the term "Guarantee Obligations" shall not include endorsements of instruments for deposit or collection in the ordinary course of business or customary and reasonable indemnity obligations in effect on the Closing Date or entered into in connection with any acquisition or Disposition of assets permitted under this Agreement (other than such obligations with respect to Indebtedness). The amount of any Guarantee Obligation shall be deemed to be an amount equal to the stated or determinable amount of the Indebtedness in respect of which such Guarantee Obligation is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof (assuming such Person is required to perform thereunder) as determined by such Person in good faith.

"Guarantors" shall mean (a) Holdings, (b) each Domestic Subsidiary (other than an Excluded Subsidiary) that provides the Guarantee on the Closing Date, which in any event will include all Subsidiaries that are Debtors on the Closing Date, or becomes a party to the Guarantee after the Closing Date pursuant to Section 9.11 or otherwise (including becoming a Debtor in the Chapter 11 Cases unless waived) and (c) the Borrower (other than with respect to its own Obligations).

"Hazardous Materials" shall mean (a) any petroleum or petroleum products spilled or released into the environment, radioactive materials, friable asbestos, urea formaldehyde foam insulation, polychlorinated biphenyls, and radon gas; (b) any chemicals, materials or substances defined as or included in the definition of "hazardous substances", "hazardous waste", "hazardous materials", "extremely hazardous waste", "restricted hazardous waste", "toxic substances", "toxic pollutants", "contaminants", or "pollutants", or words of similar import, under any applicable Environmental Law; and (c) any other chemical, material or substance, for which a release into the environment is prohibited, limited or regulated by any Environmental Law.

"Hedging Agreements" shall mean (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement or any other master agreement (any such master agreement, together with any related schedules, a "Master Agreement"), including any such obligations or liabilities under any Master Agreement.

"Hedging Obligations" shall mean, with respect to any Person, the obligations of such Person under Hedging Agreements.

"Holdings" shall have the meaning in the introductory paragraph hereto.

"Indebtedness" of any Person shall mean (a) all indebtedness of such Person for borrowed money, (b) all obligations of such Person evidenced by bonds, debentures, notes, loan agreements or other similar instruments, (c) the deferred purchase price of assets or services that in accordance with GAAP would be included as a liability on the balance sheet of such Person, (d) the face amount of all letters of credit issued for the account of such Person and, without duplication, all drafts drawn thereunder, (e) all Indebtedness of any other Person secured by any Lien on any property owned by such Person, whether or not such Indebtedness has been assumed by such Person, (f) the principal component of all Capitalized Lease Obligations of such Person, (g) the Swap Termination Value of Hedging Obligations of such Person, (h) without duplication, all Guarantee Obligations of such Person, (i) Disqualified Stock of such Person and (j) Receivables Indebtedness of such Person; provided that Indebtedness shall not include (i) trade and other

ordinary course payables and accrued expenses arising in the ordinary course of business, (ii) deferred or prepaid revenue, (iii) purchase price holdbacks in respect of a portion of the purchase price of an asset to satisfy warranty or other unperformed obligations of the respective seller, (iv) any Indebtedness defeased by such Person or by any Subsidiary of such Person, (v) contingent obligations incurred in the ordinary course of business and (vi) earnouts or similar obligation until earned, due and payable and not paid for a period of thirty (30) days.

For all purposes hereof, (a) the Indebtedness of any Person shall include the Indebtedness of any partnership or joint venture (other than a joint venture that is itself a corporation or limited liability company) in which such Person is a general partner or a joint venture, except to the extent such Person's liability for such Indebtedness is otherwise limited and only to the extent such Indebtedness constitutes Indebtedness for borrowed money, obligations in respect of Capitalized Lease Obligations and obligations evidenced by bonds, debentures, notes, loan agreement or other similar instruments and (b) the amount of Indebtedness of any Person for purposes of clause (e) shall be deemed to be equal to the lesser of (i) the aggregate unpaid principal amount of such Indebtedness and (ii) the fair market value of the property encumbered thereby as determined by such Person in good faith.

"indemnified liabilities" shall have the meaning provided in Section 13.5.

"Indemnified Taxes" shall mean (a) all Taxes imposed on or with respect to any payment made on account of any obligation of any Credit Party under any Credit Document other than Excluded Taxes and (b) to the extent not otherwise described in (a), Other Taxes.

"Initial Draw" shall have the meaning set forth in Section 2.1.

"Initial Intercompany Borrower" shall have the meaning set forth in Section 9.13.

"Initial Intercompany Transaction" shall have the meaning set forth in Section 9.13.

"Insolvent" shall mean, with respect to any Multiemployer Plan, the condition that such Multiemployer Plan is insolvent within the meaning of Section 4245 of ERISA.

"Intercompany Transfer Mechanic" shall have the meaning set forth in Section 10.13(a).

"Interest Period" shall mean, with respect to any Term Loan, the interest period applicable thereto, as determined pursuant to Section 2.9.

"Interim Order" means an order of the Bankruptcy Court (as the same may be amended, supplemented, or modified from time to time after entry thereof in accordance with the terms hereof) approving the Credit Documents on an interim basis and in the form attached hereto as Exhibit L, with any changes to such form as are satisfactory to the Borrower and the Required Lenders. For the avoidance of doubt, on the Closing Date, all references to the Interim Order shall be to that certain *Interim Order (I) Authorizing Debtors (A) to Obtain Postpetition Financing Pursuant to 11 U.S.C. §§ 105, 361, 362, 363(b), 364(c)(1), 364(c)(2), 364(c)(3), 364(d)(1) and* 

364(e), and (B) to Utilize Cash Collateral Pursuant to 11 U.S.C. § 363, (II) Granting Adequate Protection to Prepetition Secured Parties Pursuant to 11 U.S.C. §§ 361, 362, 363, 364 and 507(b), (III) Scheduling Final Hearing Pursuant to Bankruptcy Rules 4001(b) and (c) and (IV) Granting Related Relief [Docket No. []].

"Interim Order Entry Date" means the date on which the Interim Order is entered by the Bankruptcy Court, which date was February 15, 2023.

"Investment" shall mean, for any Person: (a) the acquisition (whether for cash, property, services or securities or otherwise) of Stock, Stock Equivalents, bonds, notes, debentures, partnership, limited liability company membership or other ownership interests or other securities of any other Person (including any "short sale" or any sale of any securities at a time when such securities are not owned by the Person entering into such sale); (b) the making of any deposit with, or advance, loan or other extension of credit to, any other Person (including the purchase of property from another Person subject to an understanding or agreement, contingent or otherwise, to resell such property to such other Person) (including any partnership or joint venture); (c) the entering into of any Guarantee Obligation with respect to Indebtedness; or (d) the purchase or other acquisition (in one transaction or a series of transactions) of all or substantially all of the property and assets or business of another Person or assets constituting a business unit, line of business or division of such Person; provided that, in the event that any Investment is made by the Borrower or any Subsidiary in any Person through substantially concurrent interim transfers of any amount through one or more other Subsidiaries, then such other substantially concurrent interim transfers shall be disregarded for purposes of Section 10.5 (excluding, in the case of the Borrower and the Subsidiaries, intercompany loans, advances and Indebtedness having a term not exceeding 364 days (inclusive of any roll-over or extensions of terms) and made in the ordinary course of business). The amount of any Investment outstanding at any time shall be the original cost of such Investment reduced by any Returns of the Borrower or a Subsidiary in respect of such Investment (provided that, with respect to amounts received other than in the form of cash or Cash Equivalents, such amount shall be equal to the fair market value of such consideration).

"Judgment Currency" shall have the meaning provided in Section 13.20.

"Junior Indebtedness" shall have the meaning provided in Section 10.7(a).

"Leased Real Property" means any Real Estate that is leased or otherwise rented and not owned in fee simple by any Credit Party.

"Lender" shall have the meaning provided in the preamble to this Agreement.

"Lender Default" shall mean (a) the refusal or failure (which has not been cured) of a Lender to make available its portion of any Borrowing that it is required to make hereunder, (b) a Lender having notified the Administrative Agent and/or the Borrower that it does not intend to comply with its funding obligations under this Agreement or has made a public statement to that effect with respect to its funding obligations under this Agreement, (c) a Lender has failed to confirm in a manner reasonably satisfactory to the Administrative Agent, the Borrower that it will comply with its funding obligations under this Agreement, (d) a Lender being deemed insolvent or becoming the subject of a bankruptcy or insolvency proceeding or has admitted in writing that

it is insolvent, *provided* that a Lender Default shall not be in effect with respect to a Lender solely by virtue of the ownership or acquisition of any Stock or Stock Equivalents in that Lender or any direct or indirect parent company thereof by a Governmental Authority so long as such ownership interest does not result in or provide such Lender with immunity from the jurisdiction of courts within the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such governmental authority or instrumentality) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lender or (e) a Lender that has, or has a direct or indirect parent company that has, become the subject of a Bail-In Action.

"Lien" shall mean any mortgage, pledge, security interest, hypothecation, collateral assignment, lien (statutory or other) or similar encumbrance (including any conditional sale or other title retention agreement or any Capital Lease).

"Loan" shall mean a Term Loan.

"Management Stockholders" means the members of management of Holdings (or any Parent Entity) or its Subsidiaries who are holders of Stock and Stock Equivalents of Holdings or of any Parent Entity on the Closing Date.

"Master Agreement" shall have the meaning provided in the definition of the term "Hedging Agreement".

"Material Adverse Effect" shall mean any circumstance or condition that would individually or in the aggregate, have a material adverse effect on (i) the business, assets, liabilities (actual or contingent), operations, properties or condition (financial or otherwise) of the Borrower and its Subsidiaries taken as a whole (other than as a result of events leading up to and following the commencement of the Chapter 11 Cases and the continuation and prosecution thereof), (ii) the ability of the Credit Parties (taken as a whole) to perform or consummate their respective payment obligations under any Credit Document (other than as a result of events leading up to and resulting from the commencement of the Chapter 11 Cases and the continuation and prosecution thereof), or (iii) the rights, benefits and remedies of the Lenders or the Administrative Agent under any Credit Document, including the ability of the Administrative Agent and Lenders to enforce the Credit Documents; provided that nothing disclosed in any filings by the Borrower on Form 8-K or 12b-25 made through February 10, 2023, shall, in any case, in and of itself and based solely on facts as disclosed therein (without giving effect to any developments not disclosed therein) constitute a Material Adverse Effect.

"Material Real Property" shall mean (i) the Owned Real Property set forth on Schedule 8.15 and (ii) any Owned Real Property owned by any Credit Party with a fair market value in excess of \$5,000,000.

"Maturity Date" shall mean the earliest to occur of (a) the date that is six (6) months after the Closing Date (or if such day shall not be a Business Day, the next succeeding Business Day) (the "Scheduled Maturity Date"); (b) the date that is 45 days after the Petition Date if the Final Order has not been entered prior to the expiration of such 45-day period, unless otherwise extended by the Required Lenders; (c) the Consummation Date; (d) the acceleration of the Loans and the termination of the Commitments with respect to the Term Facility in accordance

with this Agreement; (e) the consummation of a sale of all or substantially all of the assets of the Borrower (or the Borrower and the Guarantors) pursuant to section 363 of the Bankruptcy Code; and (f) the termination of the Restructuring Support Agreement.

"Milestones" shall have the meaning set forth in Section 9.16.

"Minimum Borrowing Amount" shall mean (a) with respect to a Borrowing of SOFR Loans, \$5,000,000 (or, if less, the entire remaining Commitments of any applicable Credit Facility at the time of such Borrowing), and (b) with respect to a Borrowing of ABR Loans, \$1,000,000 (or, if less, the entire remaining Commitments of any applicable Credit Facility at the time of such Borrowing).

"Monthly Booking Value" means, with respect to a Subscription Contract, the total contract value divided by the term of the contract (in number of months).

"Moody's" shall mean Moody's Investors Service, Inc. or any successor by merger or consolidation to its business.

"Mortgage" shall mean a mortgage or a deed of trust, deed to secure debt, trust deed or other security document entered into by the owner of a Material Real Property and the Collateral Agent for the benefit of the Secured Parties in respect of that Owned Real Property that constitutes Material Real Property, in a form to be mutually agreed with the Administrative Agent.

"Multiemployer Plan" shall mean a plan that is a multiemployer plan as defined in Section 4001(a)(3) of ERISA (i) to which any of the Borrower, any Subsidiary of the Borrower or any ERISA Affiliate is then making or has an obligation to make contributions or (ii) with respect to which the Borrower, any Subsidiary of the Borrower or any ERISA Affiliate could incur liability pursuant to Title IV of ERISA.

### "Net Cash Proceeds" shall mean,

- (1) with respect to any Asset Sale Prepayment Event or any Recovery Prepayment Event, (a) the gross cash proceeds (including payments from time to time in respect of installment obligations, if applicable, but only as and when received) received by or on behalf of the Borrower or any Subsidiary in connection therewith, as the case may be, less (b) the sum of:
  - (i) the amount, if any, of all taxes (including in connection with any repatriation of funds) paid or estimated by the Borrower in good faith to be payable by the Borrower or any Subsidiary in connection with such Prepayment Event,
  - (ii) the amount of any reasonable reserve established in accordance with GAAP against any liabilities (other than any taxes deducted pursuant to clause (i) above) (x) associated with the assets that are the subject of such Prepayment Event and (y) retained by the Borrower or any Subsidiary (including any pension and other post-employment benefit liabilities and liabilities related to environmental matters or against any indemnification obligations associated with such transaction); provided that the amount of any subsequent reduction of such reserve (other than in connection with a payment in

respect of any such liability) shall be deemed to be Net Cash Proceeds of such Prepayment Event occurring on the date of such reduction,

- (iii) the amount of any Indebtedness (other than Indebtedness hereunder and any other Indebtedness secured by a Lien that ranks *pari passu* with or is subordinated to the Liens securing the Obligations) secured by a Lien on the assets that are the subject of such Prepayment Event, to the extent that the instrument creating or evidencing such Indebtedness requires that such Indebtedness be repaid upon consummation of such Prepayment Event,
- (iv) the amount of any proceeds of any Recovery Prepayment Event that the Borrower or any Subsidiary has reinvested (or intends to reinvest within the Reinvestment Period, has entered into an Acceptable Reinvestment Commitment prior to the last day of the Reinvestment Period to reinvest or, with respect to any Recovery Prepayment Event, provided an Acceptable Reinvestment Commitment or a Restoration Certification prior to the last day of the Reinvestment Period) in the business of the Borrower or any Subsidiary (subject to Section 9.15), including for the repair, restoration or replacement of an asset or assets subject to such Recovery Prepayment Event; provided that any portion of such proceeds that has not been so reinvested within such Reinvestment Period (with respect to such Recovery Prepayment Event, the "Deferred Net Cash Proceeds") shall, unless the Borrower or any Subsidiary has entered into an Acceptable Reinvestment Commitment or provided a Restoration Certification prior to the last day of such Reinvestment Period to reinvest such proceeds, (x) be deemed to be Net Cash Proceeds of such Prepayment Event occurring on the last day of such Reinvestment Period or, if later, 180 days after the date the Borrower or such Subsidiary has entered into such Acceptable Reinvestment Commitment or provided such Restoration Certification, as applicable (such last day or 180th day, as applicable, the "Deferred Net Cash Proceeds Payment Date"), and (y) be applied to the repayment of Term Loans in accordance with Section 5.2(a)(i),
- (v) in the case of any Asset Sale Prepayment Event, any funded escrow established pursuant to the documents evidencing any such sale or Disposition to secure any indemnification obligations or adjustments to the purchase price associated with any such sale or Disposition; *provided* that the amount of any subsequent reduction of such escrow (other than in connection with a payment in respect of any such liability) shall be deemed to be Net Cash Proceeds of such a Prepayment Event occurring on the date of such reduction solely to the extent that the Borrower and/or any Subsidiaries receives cash in an amount equal to the amount of such reduction,
- (vi) in the case of any Asset Sale Prepayment Event or Recovery Prepayment Event by a non-Wholly Owned Subsidiary, the *pro rata* portion of the Net Cash Proceeds thereof (calculated without regard to this clause (vi)) attributable to minority interests and not available for distribution to or for the account of the Borrower or a Wholly Owned Subsidiary as a result thereof, and
- (vii) reasonable and customary fees, commissions, expenses (including attorney's fees, investment banking fees, survey costs, title insurance premiums and recording charges, transfer taxes, deed or mortgage recording taxes and other customary

expenses and brokerage, consultant and other customary fees), issuance costs, premiums, discounts and other costs paid by the Borrower or any Subsidiary, as applicable, in connection with such Prepayment Event, in each case only to the extent not already deducted in arriving at the amount referred to in clause (a) above;

- (2) with respect to the incurrence or issuance of any Indebtedness or the issuance of any Stock or Stock Equivalent or capital contribution, the excess, if any, of (a) the sum of cash and Cash Equivalents received in connection with such incurrence or issuance over (b) reasonable and customary fees, commissions, expenses (including attorney's fees, investment banking fees, survey costs, title insurance premiums and recording charges, transfer taxes, deed or mortgage recording taxes and other customary expenses and brokerage, consultant and other customary fees), issuance costs, premiums, discounts and other costs paid by the Borrower or any Subsidiary in connection with such incurrence or issuance; and
- (3) with respect to any Extraordinary Receipts, the gross cash proceeds received by or on behalf of the Borrower or any Subsidiary in connection therewith, as the case may be, less any reasonable costs incurred in connection with such Extraordinary Receipt, including all reasonable out-of-pocket costs and expenses incurred in connection with the collection of such proceeds; *provided* that if such amount shall be less than \$5,000,000, the Net Cash Proceeds shall be deemed to be zero.

"New Equity Interests" shall have the meaning provided in the Restructuring Support Agreement.

"Non-Consenting Lender" shall have the meaning provided in Section 13.7(b).

"Non-Debtor Subsidiary Notes/Receivables" shall have the meaning provided in Section 10.13(a).

"Non-Debtor Subsidiary Transfer" shall have the meaning provided in Section 10.13(a).

"Non-Defaulting Lender" shall mean and include each Lender other than a Defaulting Lender.

"Non-U.S. Lender" shall mean any Agent or Lender that is not, for U.S. federal income tax purposes, (a) an individual who is a citizen or resident of the U.S., (b) a corporation, partnership or entity treated as a corporation or partnership created or organized in or under the laws of the U.S., or any political subdivision thereof, (c) an estate whose income is subject to U.S. federal income taxation regardless of its source or (d) a trust if a court within the U.S. is able to exercise primary supervision over the administration of such trust and one or more U.S. persons have the authority to control all substantial decisions of such trust or a trust that has a valid election in effect under applicable U.S. Treasury regulations to be treated as a U.S. person.

"Notice of Borrowing" shall mean a request of the Borrower in accordance with the terms of Section 2.3 and substantially in the form of Exhibit A or such other form as shall be approved by the Administrative Agent (acting reasonably).

"Notice of Conversion or Continuation" shall have the meaning provided in Section 2.6(a).

"Obligations" shall mean all advances to, and debts, liabilities, obligations, covenants and duties of, any Credit Party arising under any Credit Document or otherwise with respect to any Term Loan entered into with Holdings, the Borrower or any Subsidiary, whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees (including the Put Option Premium and the Exit Fee) that accrue after the commencement by or against any Credit Party of any proceeding under any bankruptcy or insolvency law naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding. Without limiting the generality of the foregoing, the Obligations of the Credit Parties under the Credit Documents (and any of their Subsidiaries to the extent they have obligations under the Credit Documents) include the obligation (including guarantee obligations) to pay principal, interest, charges, expenses, fees, attorney costs, indemnities and other amounts payable by any Credit Party under any Credit Document.

"Orders" shall mean, collectively, the Interim Order and the Final Order.

"Organizational Documents" shall mean, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction), (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement and (c) with respect to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization and, if applicable, any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization with the applicable Governmental Authority in the jurisdiction of its formation or organization and, if applicable, any certificate or articles of formation or organization of such entity.

"Other Taxes" shall mean any and all present or future stamp, registration, documentary or other similar Taxes arising from any payment made or required to be made under this Agreement or any other Credit Document or from the execution or delivery of, registration or enforcement of, consummation or administration of, or otherwise with respect to, this Agreement or any other Credit Document except any such Taxes that are any Taxes imposed on any Agent or any Lender as a result of any current or former connection between such Agent or Lender and the jurisdiction of the Governmental Authority imposing such Tax or any political subdivision or taxing authority thereof or therein (other than any such connection arising solely from such Agent or Lender having executed, delivered or performed its obligations or received a payment under, or having been a party to or having enforced, this Agreement or any other Credit Document) imposed with respect to an assignment (other than an assignment made pursuant to Section 13.7 or Section 2.12).

"Owned Real Property" shall mean Real Estate owned in fee simple by any Credit Party.

"Overnight Rate" shall mean, for any day, the greater of (a) the Federal Funds Effective Rate and (b) an overnight rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation.

"Parent Entity" shall mean any direct or indirect parent of Avaya Holdings.

"Participant" shall have the meaning provided in Section 13.6(c)(i).

"Participant Register" shall have the meaning provided in Section 13.6(c)(iii).

"Patriot Act" shall have the meaning provided in Section 13.18.

"Paul, Weiss" means Paul, Weiss, Rifkind, Wharton & Garrison LLP.

"Payment in Full" or "Paid in Full" shall mean the time at which no Agent or Lender shall have any Commitments, any Loan or other Obligations unpaid, unsatisfied or outstanding (other than in respect of contingent obligations, indemnities and expenses related thereto that are not then payable or in existence) as a result of all such Loans and other Obligations having been paid in full in cash.

"PBGC" shall mean the Pension Benefit Guaranty Corporation established pursuant to Section 4002 of ERISA, or any successor thereto.

"Pension Plan" shall mean any employee pension benefit plan (as defined in Section 3(2) of ERISA, but excluding any Multiemployer Plan) which is covered by Title IV of ERISA or subject to the minimum funding standards under Section 412 of the Code or Section 302 of ERISA and is maintained or contributed to by the Borrower, any Subsidiary or ERISA Affiliate or with respect to which the Borrower, any Subsidiary or any ERISA Affiliate could incur liability pursuant to Title IV of ERISA.

"Perfection Certificate" shall mean a perfection certificate of the Borrower in any form approved by the Administrative Agent (acting reasonably).

"Periodic Term SOFR Determination Day" shall have the meaning specified in the definition of "Term SOFR".

## "Permitted Encumbrances" shall mean:

- (a) Liens for taxes, assessments or governmental charges or claims (including Liens imposed by the PBGC or similar Liens) not yet delinquent or that are being contested in good faith and by appropriate proceedings for which appropriate reserves have been established to the extent required by and in accordance with GAAP or that are not required to be paid pursuant to Section 9.4 (including as a result of nonpayment of taxes permitted or required under the Bankruptcy Code);
- (b) Liens in respect of property or assets of the Borrower or any Subsidiary imposed by Applicable Law, such as carriers', landlords', construction contractors', warehousemen's and mechanics' Liens and other similar Liens, arising in the

- ordinary course of business, in respect of amounts not more than 60 days overdue or being contested in good faith;
- (c) Liens arising from judgments or decrees in circumstances not constituting an Event of Default under Section 11.9;
- (d) Liens incurred or deposits of cash and Cash Equivalents (i) made in connection with workers' compensation, unemployment insurance, employee benefit and pension liability and other types of social security or similar legislation, or to secure the performance of tenders, statutory obligations, trade contracts (other than for payment of Indebtedness), leases, statutory obligations, surety, stay, customs and appeal bonds, bids, leases, government contracts, surety, performance and return-of-money bonds and other similar obligations, in each case incurred in the ordinary course of business or otherwise constituting Investments permitted by Section 10.5 or (ii) to provide "adequate assurances of payment" as that term is used in section 366 of the Bankruptcy Code;
- (e) ground leases or subleases, licenses or sublicenses in respect of Real Estate on which facilities owned or leased by the Borrower or any of the Subsidiaries are located;
- (f) easements, rights-of-way, licenses, reservations, servitudes, permits, conditions, covenants, rights of others, restrictions (including zoning restrictions), royalty interests and leases, minor defects, exceptions or irregularities in title or survey, encroachments, protrusions and other similar charges or encumbrances (including those to secure health, safety and environmental obligations), which do not interfere in any material respect with the business of the Borrower and the Subsidiaries, taken as a whole;
- (g) with respect to any Owned Real Property that constitutes Material Real Property, any exception on the title policy issued and matters shown on the Survey delivered which do not in the aggregate materially adversely affect the value of said property or materially impair its use in the operation of the business of the Borrower or any of the Subsidiaries;
- (h) any interest or title of a lessor, sublessor, licensor, sublicensor or grantor of an easement or secured by a lessor's, sublessor's, licensor's, sublicensor's interest or grantor of an easement under any lease, sublease, license, sublicense or easement to be entered into by the Borrower or any Subsidiary as lessee, sublessee, licensee, grantee or sublicensee to the extent permitted or not prohibited by this Agreement;
- (i) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods;
- (j) leases, licenses, subleases or sublicenses granted to others not interfering in any material respect with the business of the Borrower and the Subsidiaries, taken as a whole or constituting Disposition permitted under Section 10.4;

- (k) Liens arising from precautionary Uniform Commercial Code financing statement or similar filings made in respect of operating leases entered into by the Borrower or any Subsidiary;
- (l) any zoning, land use, environmental or similar law or right reserved to or vested in any Governmental Authority to control or regulate the use of any Real Estate that does not materially interfere with the ordinary conduct of the business of the Borrower and the Subsidiaries, taken as a whole;
- (m) any Lien arising by reason of deposits with or giving of any form of security to any Governmental Authority for any purpose at any time as required by Applicable Law as a condition to the transaction of any business or the exercise of any privilege or license, or to enable the Borrower or any Subsidiary to maintain self-insurance or to participate in any fund for liability on any insurance risks;
- (n) rights reserved to or vested in any Governmental Authority by the terms of any right, power, franchise, grant, license or permit, or by any provision of Applicable Law, to terminate or modify such right, power, franchise, grant, license or permit or to purchase or recapture or to designate a purchaser of any of the property of such person;
- (o) Liens arising under any obligations or duties affecting any of the property, the Borrower or any Subsidiary to any Governmental Authority with respect to any franchise, grant, license or permit which do not materially impair the use of such property for the purposes for which it is held;
- (p) rights reserved to or vested in any Governmental Authority to use, control or regulate any property of such Person, which do not materially impair the use of such property for the purposes for which it is held;
- (q) any obligations or duties, affecting the property of the Borrower or any Subsidiary, to any Governmental Authority with respect to any franchise, grant, license or permit;
- (r) a set-off or netting rights granted by the Borrower or any Subsidiary pursuant to any Hedging Agreements solely in respect of amounts owing under such agreements;
- (s) [reserved];
- (t) [reserved];
- (u) [reserved];
- (v) with respect to any Foreign Subsidiary, other Liens and privileges arising mandatorily by Applicable Laws;
- (w) [reserved];

- (x) Liens (i) of a collecting bank arising under Section 4-210 of the Uniform Commercial Code on items in the course of collection and (ii) in favor of a banking institution arising as a matter of law encumbering deposits (including the right of set-off) or attaching to commodity trading accounts or other commodity brokerage accounts incurred in the ordinary course of business, and (iii) in favor of banking or other financial institutions or other electronic payment service providers arising as a matter of law or customary contract encumbering deposits, including deposits in "pooled deposit" or "sweep" accounts (including the right of set-off) and which are within the general parameters customary in the banking or finance industry;
- (y) Liens arising out of conditional sale, title retention, consignment or similar arrangements for sale or purchase of goods entered into by the Borrower or any Subsidiary in the ordinary course of business permitted or not prohibited by this Agreement;
- (z) [reserved];
- (aa) [reserved];
- (bb) Liens that are contractual rights of set-off (i) relating to the establishment of depository relations with banks not given in connection with the issuance of Indebtedness, (ii) relating to pooled deposit or sweep accounts of the Borrower or any Subsidiary to permit satisfaction of overdraft or similar obligations incurred in the ordinary course of business of the Borrower and the Subsidiaries or (iii) relating to purchase orders and other agreements entered into with customers of the Borrower or any Subsidiary in the ordinary course of business;
- (cc) [reserved];
- (dd) Liens on insurance policies and the proceeds thereof securing the financing of the premiums with respect thereto;
- (ee) Liens on specific items of inventory or other goods and the proceeds thereof securing such Person's obligations in respect of documentary letters of credit or banker's acceptances issued or created for the account of such Person to facilitate the purchase, shipment or storage of such inventory or goods in the ordinary course of business or consistent with past practice;
- (ff) any restrictions on any Stock or Stock Equivalents or other joint venture interests of the Borrower or any Subsidiary providing for a breach, termination or default under any owners, participation, shared facility, joint venture, stockholder, membership, limited liability company or partnership agreement between such Person and one or more other holders of such Stock or Stock Equivalents or interest of such Person, if a security interest or other Lien is created on such Stock or Stock Equivalents or interest as a result thereof and other similar Liens; and

(gg) Liens securing Indebtedness or other obligations (i) of the Borrower or any Subsidiary in favor of a Credit Party and (ii) of any other Subsidiary that is not a Credit Party in favor of any other Subsidiary that is not a Credit Party.

"Permitted Holders" means, collectively, (i) the Management Stockholders (including any Management Stockholders holding Stock and Stock Equivalents through an equityholding vehicle), (ii) any Person who is acting solely as an underwriter in connection with a public or private offering of Stock and Stock Equivalents of any Parent Entity or Holdings, acting in such capacity, (iii) any group (within the meaning of Section 13(d)(3) or Section 14(d)(2) of the Exchange Act or any successor provision) of which any of the foregoing, any Permitted Plan or any Person or group that becomes a Permitted Holder specified in the last sentence of this definition are members and any member of such group; *provided* that, in the case of such group and without giving effect to the existence of such group or any other group, Persons referred to in subclauses (i) through (ii), collectively, have beneficial ownership of more than 50% of the total voting power of the Voting Stock of Holdings or any Parent Entity held by such group, and (iv) any Permitted Plan.

"Permitted Plan" means any employee benefits plan of Holdings or any of its Affiliates and any Person acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan.

"Permitted Variance" means the amount of aggregate total receipts and operating disbursements made or accrued during any applicable Variance Test Period that (i) with respect to total receipts, is not less than 85% of projected total receipts for such Variance Test Period and (ii) with respect to operating disbursements, is not greater than 115% of project operating disbursements for such Variance Test Period.

"**Person**" shall mean any individual, partnership, joint venture, firm, corporation, limited liability company, association, trust or other enterprise or any Governmental Authority.

"Petition Date" shall have the meaning specified in the Recitals to this Agreement.

"Plan" shall have the meaning provided in the Restructuring Support Agreement.

"Platform" shall have the meaning provided in Section 13.17(c).

"Prepayment Event" shall mean any Asset Sale Prepayment Event, Recovery Prepayment Event, Debt Incurrence Prepayment Event or Extraordinary Receipts Prepayment Event.

"Prepetition ABL Credit Agreement" shall mean that certain ABL Credit Agreement, dated as of December 15, 2017, by and among the Borrower, Holdings, Avaya Canada Corp., Avaya UK, Avaya International Sales Limited, Avaya Deutschland GMBH, Avaya GMBH & Co. KG, the lenders party thereto, the lending institutions party thereto as L/C Issuers and Swing Line Lenders and Citibank, N.A., as Administrative Agent and Collateral Agent.

"Prepetition ABL Facility" shall mean the credit facility provided for pursuant to the Prepetition ABL Credit Agreement.

- "Prepetition ABL Payoff" shall mean (a)(i)the indefeasible repayment in full of all obligations under the Prepetition ABL Facility, including interest and fees through the date of repayment (at the non-default contract rate) and (ii) the cash collateralization of all Existing Letters of Credit, Existing Cash Management Obligations and Existing Secured Hedging Obligations, in each case, with the proceeds of the Initial Draw and pursuant to the terms of the Interim DIP Order and (b) the termination of all commitments thereunder and the release of all guarantees and lien in respect thereof.
- "Prepetition ABL Payoff Letter" shall mean a payoff letter evidencing the Prepetition ABL Payoff in form and substance reasonably acceptable to the Required Lenders.
- "Prepetition Debt" shall mean, collectively, the Indebtedness of each Debtor outstanding and unpaid on the date on which such Person becomes a Debtor.
- "Prepetition Exchangeable Notes" shall mean the 8.00% exchangeable senior secured notes issued pursuant to that certain Indenture, dated as of July 12, 2022, by and among the Borrower, the guarantors party thereto and Wilmington Trust, National Association. as Trustee and as Notes Collateral Agent.
- "Prepetition First Lien Debt" means all obligations in respect of (i) the Prepetition ABL Facility, (ii) the Prepetition Term Loan Credit Facility, (iii) the Prepetition First Lien Notes and (iv) the Prepetition Exchangeable Notes.
- "Prepetition First Lien Notes" shall mean the 6.125% senior secured first lien notes issued pursuant to that certain Indenture, dated as of September 25, 2020, by and among the Borrower, the guarantors party thereto and Wilmington Trust, National Association. as Trustee and as Notes Collateral Agent.
- "Prepetition Payment" shall mean a payment (by way of adequate protection or otherwise) of principal or interest or otherwise on account of any (i) Prepetition Debt, (ii) "critical or foreign vendor payments" or (iii) trade payables (including, without limitation, in respect of reclamation claims), or other pre-petition claims against any Credit Party.
- "Prepetition Term Loan Credit Agreement" shall mean that certain Term Loan Credit Agreement, dated as of December 15, 2017, as amended, restated, amended and restated, supplemented or otherwise modified from time to time prior to the date hereof, by and among the Borrower, Holdings, certain guarantors party thereto, the lenders party thereto and Goldman Sachs Bank USA, as Administrative Agent and Collateral Agent.
- "Prepetition Secured Parties" shall mean the lender parties under the Prepetition ABL Facility and Prepetition Term Loan Facility and the beneficial holders of the Prepetition First Lien Notes and Prepetition Exchangeable Notes.
- **"Prepetition Term Loan Facility"** shall mean the credit facilities provided for pursuant to the Prepetition Term Loan Credit Agreement.
- "**Prohibited Transaction**" shall have the meaning assigned to such term in Section 406 of ERISA or Section 4975(c) of the Code.

"Public Reporting Entity" shall mean an entity that (i) complies with the reporting obligations under U.S. securities laws, (ii) is designated by the Borrower as a "Public Reporting Entity" and (iii) whose consolidated financial results include the financial results of the Borrower and its consolidated subsidiaries and customary reconciliations to eliminate the financial results of entities other than the Borrower and its consolidated subsidiaries.

"Put Option Premium" shall have the meaning set forth in Section 4.1(c).

"PW Ad Hoc Group" has the meaning set forth in the Restructuring Support Agreement.

"Real Estate" shall mean any interest in land, buildings and improvements owned, leased or otherwise held by any Credit Party, but excluding all operating fixtures and equipment.

"Receivables Indebtedness" shall mean, at any time, with respect to any receivables, securitization or similar facility (excluding any account receivable factoring facility entered into incurred in the ordinary course of business), the aggregate principal, or stated amount, of the "indebtedness", fractional undivided interests (which stated amount may be described as a "net investment" or similar term reflecting the amount invested in such undivided interest) or other securities incurred or issued pursuant to such receivables, securitization or similar facility, at such time, in each case outstanding at such time.

"Recovery Event" shall mean (a) any damage to, destruction of or other casualty or loss involving any property or asset or (b) any seizure, condemnation, confiscation or taking (or transfer under threat of condemnation) under the power of eminent domain of, or any requisition of title or use of or relating to, or any similar event in respect of, any property or asset.

"Recovery Prepayment Event" shall mean the receipt of Net Cash Proceeds with respect to any settlement or payment in connection with any Recovery Event in respect of any property or asset of the Borrower or any Subsidiary; *provided* that the term "Recovery Prepayment Event" shall not include any Asset Sale Prepayment Event.

"Register" shall have the meaning provided in Section 13.6(b)(iii).

"Regulation T" shall mean Regulation T of the Board as from time to time in effect and any successor to all or a portion thereof establishing margin requirements.

"Regulation U" shall mean Regulation U of the Board as from time to time in effect and any successor to all or a portion thereof establishing margin requirements.

"Regulation X" shall mean Regulation X of the Board as from time to time in effect and any successor to all or a portion thereof establishing margin requirements.

"Reinvestment Period" shall mean 6 months following the date of receipt of Net Cash Proceeds of a Recovery Prepayment Event.

"Rejection Notice" shall have the meaning provided in Section 5.2(f).

"Related Parties" shall mean, with respect to any specified Person, such Person's Affiliates (or, for purposes of clauses (A) and (B) of the last proviso of Section 13.5 and the penultimate paragraph of Section 13.5, such Person's controlled Affiliates) and the directors, officers, employees, agents, trustees and advisors of such Person and any Person that possesses, directly or indirectly, the power to direct or cause the direction of the management or policies of such Person, whether through the ability to exercise voting power, by contract or otherwise.

"Reorganization Plan" shall mean a plan of reorganization in the Chapter 11 Cases.

"Reportable Event" shall mean an event described in Section 4043 of ERISA and the regulations thereunder, other than any event as to which the thirty day notice period has been waived.

"Required Akin Ad Hoc Group Lenders" shall mean, at any date, Lenders who are members of the Akin Ad Hoc Group having or holding a majority of the sum of (a) the outstanding principal amount of the Term Loans in the aggregate that are held by members of the Akin Ad Hoc Group at such date and (b) the outstanding amount of the aggregate unfunded Commitments in respect of the Term Loans that are held by members of the Akin Ad Hoc Group at such date.

"Required Lenders" shall mean, at any date, two or more unaffiliated (if there are two or more such unaffiliated Lenders) Lenders who are Non-Defaulting Lenders having or holding a majority of the sum of (a) the outstanding principal amount of the Term Loans in the aggregate at such date and (b) the outstanding amount of the unfunded Commitments in the aggregate in respect of the Term Loans at such date.

"Required PW Ad Hoc Group Lenders" shall mean, at any date, Lenders who are members of the PW Ad Hoc Group having or holding a majority of the sum of (a) the outstanding amount of the Term Loans in the aggregate that are held by members of the PW Ad Hoc Group at such date and (b) the outstanding principal amount of the aggregate unfunded Commitments in respect of the Term Loans that are held by members of the PW Ad Hoc Group at such date.

"Resolution Authority" shall mean an EEA Resolution Authority or, with respect to any UK Financial Institution, a UK Resolution Authority.

"Restoration Certification" shall mean, with respect to any Recovery Prepayment Event, a certification made by an Authorized Officer of the Borrower or any Subsidiary, as applicable, to the Administrative Agent prior to the end of the Reinvestment Period certifying that (a) the Borrower or such Subsidiary intends to use the proceeds received in connection with such Recovery Prepayment Event (x) to repair, restore, refurbish or replace the property or assets in respect of which such Recovery Prepayment Event occurred or (y) or to invest in assets used or useful in a Similar Business, (b) the approximate costs of completion of such repair, restoration, refurbishment or replacement and (c) that such repair, restoration or replacement will be completed within the later of (x) fifteen months after the date on which cash proceeds with respect to such

Recovery Prepayment Event were received and (y) 180 days after delivery of such Restoration Certification.

"Restricted Payment" shall mean, with respect to the Borrower or any Subsidiary, any dividend or return any capital to its stockholders or any other distribution, payment or delivery of property or cash to its stockholders on account of such Stock and Stock Equivalents, or redemption, retirement, purchase or other acquisition, directly or indirectly, for consideration, any shares of any class of its Stock or Stock Equivalents or set aside any funds for any of the foregoing purposes, other than dividends payable solely in its Stock or Stock Equivalents (other than Disqualified Stock).

"Restructuring Support Agreement" shall mean that certain Restructuring Support Agreement dated as of February 14, 2023 among Holdings, the Borrower, certain Subsidiaries of the Borrower, the Consenting Stakeholders (as defined therein).

"Retained Declined Proceeds" shall have the meaning provided in Section 5.2(f).

"Returns" shall mean, with respect to any Investment, any dividend, distribution, interest, fees, premium, return of capital, repayment of principal, income, profits (from a Disposition or otherwise) and other amounts received or realized in respect of such Investment.

"S&P" shall mean Standard & Poor's Financial Services LLC or any successor by merger or consolidation to its business.

"Sanctions" shall have the meaning provided in Section 8.19.

"Sanctions Laws" shall have the meaning provided in Section 8.19.

"SEC" shall mean the Securities and Exchange Commission or any successor thereto.

"Secured Parties" shall mean the Administrative Agent, the Collateral Agent, each Lender and each sub-agent pursuant to Section 12 appointed by the Administrative Agent with respect to matters relating to the Credit Facilities or appointed by the Collateral Agent with respect to matters relating to any Security Document.

"Securities Act" shall mean the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Security Agreement" shall mean the Security Agreement, dated as of the Closing Date (as the same may be amended, restated, amended and restated, supplemented or otherwise modified or replaced from time to time), entered into by the Borrower, the other grantors party thereto and the Collateral Agent for the benefit of the Secured Parties.

"Security Documents" shall mean, collectively, (a) the Security Agreement, (b) the Orders, (c) the Mortgages (to the extent requested by the Required Lenders), (d) each intellectual property security agreement, (e) the DIP ABL Intercreditor Agreement (if entered into) and (f) each other security agreement or other instrument or document executed and delivered

pursuant to Section 9.11 or 9.12 or pursuant to any other such Security Documents. The Security Documents (other than the Orders) shall supplement, and shall not limit, the grant of a Lien on and security interest in the Collateral pursuant to the Orders.

"Sierra" shall mean Sierra Communications International LLC.

"Similar Business" shall mean any business conducted or proposed to be conducted by the Borrower and the Subsidiaries, taken as a whole, on the Closing Date or any other business activities which are reasonable extensions thereof or otherwise similar, incidental, corollary, complementary, synergistic, reasonably related, or ancillary to any of the foregoing, in each case as determined by the Borrower in good faith.

"SOFR" shall mean a rate equal to the secured overnight financing rate as administered by the SOFR Administrator.

"SOFR Administrator" shall mean the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate).

"SOFR Borrowing" shall mean, as to any Borrowing, the SOFR Loans comprising such Borrowing.

"SOFR Loan" shall mean a Loan that bears interest at a rate based on Adjusted Term SOFR, other than pursuant to clause (c)(ii) of the definition of "ABR".

# "Specified Collateral" means all Collateral consisting of the following:

- (i) all Accounts (as defined in the Uniform Commercial Code), other than Accounts which constitute identifiable proceeds of DIP Term Priority Collateral;
- (ii) all Chattel Paper (as defined in the Uniform Commercial Code), other than Chattel Paper which constitutes identifiable proceeds of DIP Term Priority Collateral;
- (iii) (iii)(x) all Deposit Accounts (as defined in the Uniform Commercial Code) (other than Deposit Accounts that are intended to solely contain identifiable proceeds of DIP Term Priority Collateral) and money and all cash, checks, other negotiable instruments, funds and other evidences of payments held therein, and (y) Securities Accounts (other than Securities Accounts (as defined in the Uniform Commercial Code) that are intended to solely contain identifiable proceeds of DIP Term Priority Collateral), Security Entitlements (as defined in the Uniform Commercial Code) and securities credited to such a Securities Account, and, in each case, all cash, checks and other property held therein or credited thereto; provided, however, that during the continuance of an Event of Default to the extent that identifiable proceeds of DIP Term Priority Collateral are deposited in any such Deposit Accounts or Securities Accounts, such identifiable proceeds shall be treated as Specified Collateral;
  - (iv) all Inventory (as defined in the Uniform Commercial Code);
- (v) to the extent relating to, evidencing or governing any of the items referred to in the preceding clauses (i) through (iv) constituting Specified Collateral, all Documents

(as defined in the Uniform Commercial Code), General Intangibles (as defined in the Uniform Commercial Code) (other than any Intellectual Property (as defined in the Security Agreement)), Instruments (as defined in the Uniform Commercial Code) (including promissory notes), and Commercial Tort Claims (as defined in the Uniform Commercial Code); provided, that to the extent any of the foregoing also relates to Collateral of a type not referred to in clauses (i) through (iv), only that portion relating to the items referred to in the preceding clauses (i) through (iv) shall be included in the Specified Collateral;

- (vi) to the extent relating to any of the items referred to in the preceding clauses (i) through (v) constituting Specified Collateral, all Supporting Obligations (as defined in the Uniform Commercial Code) and Letter-of-Credit Rights (as defined in the Uniform Commercial Code); provided, that to the extent any of the foregoing also relates to Collateral that does not otherwise constitute Specified Collateral, only that portion related to those items referred to in the preceding clauses (i) through (v) shall be included in the Specified Collateral;
- (vii) all books and Records (as defined in the Uniform Commercial Code) relating to the items referred to in the preceding clauses (i) through (vi) constituting Specified Collateral (including all books, databases, customer lists, engineer drawings and Records, whether tangible or electronic, which contain any information relating to any of the items referred to in the preceding clauses (i) through (vi)); and
- (viii) (viii) all collateral security and guarantees with respect to any of the foregoing and all cash, Money (as defined in the Uniform Commercial Code), insurance proceeds, Instruments, Securities (as defined in the Uniform Commercial Code), Financial Assets (as defined in the Uniform Commercial Code) and Deposit Accounts received as proceeds of any of the foregoing (such proceeds, "Specified Collateral Proceeds"); provided, however, that no proceeds of Specified Collateral Proceeds will constitute Specified Collateral unless such proceeds of Specified Collateral Proceeds would otherwise constitute Specified Collateral.

Notwithstanding the foregoing, "Specified Collateral" shall not include the DIP Proceeds Account, the Foreign Reserve Account, or any amounts deposited therein in accordance with the terms of this Agreement, or any proceeds of Avoidance Actions.

"SPV" shall have the meaning provided in Section 13.6(f).

"Stated Maturity" shall mean, with respect to any installment of principal on any series of Indebtedness, the date on which such payment of principal was scheduled to be paid in the original documentation governing such Indebtedness, and shall not include any contingent obligations to repay, redeem or repurchase any such principal prior to the date originally scheduled for payment thereof.

"Stock" shall mean shares of capital stock or shares in the capital, as the case may be (whether denominated as common stock or preferred stock or ordinary shares or preferred shares, as the case may be), beneficial, partnership or membership interests, participations or other equivalents (regardless of how designated) of or in a corporation, partnership, limited liability company or equivalent entity, whether voting or non-voting, *provided* that any instrument evidencing Indebtedness convertible or exchangeable for Stock shall not be deemed to be Stock

unless and until such instrument is so converted or exchanged; *provided*, *further* that, solely with respect to any CFC or CFC Holding Company, Stock shall also include any instrument or security treated as stock for U.S. federal income tax purposes.

"Stock Equivalents" shall mean all securities convertible into or exchangeable for Stock and all warrants, options or other rights to purchase or subscribe for any Stock, whether or not presently convertible, exchangeable or exercisable, *provided* that any instrument evidencing Indebtedness convertible or exchangeable for Stock Equivalents shall not be deemed to be Stock Equivalents unless and until such instrument is so converted or exchanged; *provided*, *further* that, solely with respect to any CFC or CFC Holding Company, Stock Equivalent shall also include any instrument or security treated as stock equivalent for U.S. federal income tax purposes.

"Subscription Contract" means each customer subscription contract entered into by the Borrower and its Subsidiaries.

"Subscription Contract ARR" means, with respect to each Subscription Contract, an amount equal to (i) the Monthly Booking Value thereof *times* (ii) twelve months.

"Subsequent Draw" shall have the meaning set forth in Section 2.1.

"Subsidiary" of any Person shall mean and include (a) any corporation more than 50% of whose Stock of any class or classes having by the terms thereof ordinary voting power to elect a majority of the directors of such corporation (irrespective of whether or not at the time Stock of any class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at the time owned by such Person directly or indirectly through Subsidiaries and (b) any limited liability company, unlimited company, partnership, association, joint venture or other entity of which such Person directly or indirectly through Subsidiaries has more than a 50% voting equity interest at the time or is a controlling general partner. Unless otherwise expressly provided, all references herein to a "Subsidiary" shall mean a Subsidiary of the Borrower.

"Subsidiary Guarantor" shall mean each Guarantor that is a Subsidiary of the Borrower.

"Superpriority Claim" shall mean a claim against any Debtor in any of the Chapter 11 Cases which is an administrative expense claim pursuant to section 364(c)(1) of the Bankruptcy Code, having priority over any and all administrative expenses of the kind specified in section 503(b) or 507(b) of the Bankruptcy Code.

"Survey" shall mean a survey of any Owned Real Property that constitutes Material Real Property (and all improvements thereon), including a survey based on aerial photography that is (a) (i) prepared by a licensed surveyor or engineer, (ii) certified by the surveyor (in a manner reasonable in light of the size, type and location of the Real Estate covered thereby) to the Administrative Agent and the Collateral Agent and (iii) sufficient, either alone or in connection with a survey (or "no change") affidavit in form and substance customary in the applicable jurisdiction, for the applicable title company to remove (to the extent permitted by Applicable Law) or amend all standard survey exceptions from the title insurance policy (or commitment) relating to such Owned Real Property that constitutes Material Real Property and issue such

endorsements or other survey coverage, to the extent available in the applicable jurisdiction, as the Collateral Agent may reasonably request or (b) otherwise reasonably acceptable to the Collateral Agent, taking into account the size, type and location of the Real Estate covered thereby.

"Swap Termination Value" shall mean, in respect of any one or more Hedging Agreements, after taking into account the effect of any legally enforceable netting agreement relating to such Hedging Agreements, (a) for any date on or after the date such Hedging Agreements have been closed out and termination value(s) determined in accordance therewith, such termination value(s), and (b) for any date prior to the date referenced in clause (a), the amount(s) determined as the mark-to-market value(s) for such Hedging Agreements, as determined based upon one or more mid-market or other readily available quotations provided by any recognized dealer in such Hedging Agreements (which may include a Lender or any Affiliate of a Lender).

"Taxes" shall mean any and all present or future taxes, duties, levies, imposts, assessments, deductions, withholdings or other similar charges imposed by any Governmental Authority whether computed on a separate, consolidated, unitary, combined or other basis and any interest, fines, penalties or additions to tax with respect to the foregoing.

"Term Commitment" shall mean, with respect to each Lender, (a) the amount of such Lender's commitment to make Term Loans in the amount set forth opposite such Lender's name on Schedule 1.1(a) as such Lender's "Term Commitment" or (b) if the Term Commitments (as determined without giving effect to any Borrowing) are adjusted effective or retroactively effective on the earlier of the occurrence of the Escrow Payment and the Final Order Entry Date to give effect to the B-3 Re-allocation, the amount of such Lender's commitment to make Term Loans in the amount set forth opposite such Lender's name on a commitment schedule provided and certified by the Borrower to the Administrative Agent upon the occurrence of the B-3 Re-allocation Triggering Event and acknowledged, by electronic email, by Akin (solely with respect to the Commitments of the Lenders in the Akin Ad Hoc Group (as defined in the Restructuring Support Agreement)) and Paul, Weiss (solely with respect to the Commitments of the Lenders in the PW Ad Hoc Group (as defined in the Restructuring Support Agreement)), consistent with Annex II to the Commitment Letter with allocations breakdowns among affiliated Lenders. The Administrative Agent may conclusively rely upon such adjusted commitment schedule for purposes of adjusting the Register, without independent inquiry or investigation.

"Term Facility" shall have the meaning specified in the Recitals to this Agreement.

"**Term Lender**" shall mean, at any time, a Lender with an outstanding Term Loan or a Term Commitment at such time.

"Term Loan" shall have the meaning set forth in Section 2.1(a).

"Term SOFR" shall mean,

(a) for any calculation with respect to a SOFR Loan, the Term SOFR Reference Rate for a tenor comparable to the applicable Interest Period on the day (such day, the "Periodic Term SOFR Determination Day") that is two (2) U.S. Government Securities Business Days prior to the first day of such Interest Period, as such rate

is published by the Term SOFR Administrator; provided, however, that if as of 5:00 p.m. (New York City time) on any Periodic Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as published by the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such Periodic Term SOFR Determination Day, and

for any calculation with respect to an ABR Loan on any day, the Term SOFR (b) Reference Rate for a tenor of one month on the day (such day, the "ABR Term **SOFR Determination Day**") that is two (2) U.S. Government Securities Business Days prior to such day, as such rate is published by the Term SOFR Administrator; provided, however, that if as of 5:00 p.m. (New York City time) on any ABR Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as published by the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such ABR Term SOFR Determination Day.

"Term SOFR Adjustment" shall mean a percentage equal to 0% per annum:

"Term SOFR Administrator" means CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the Term SOFR Reference Rate selected by the Administrative Agent in its reasonable discretion).

"Term SOFR Reference Rate" shall mean the forward-looking term rate based on SOFR.

"Transaction Expenses" shall mean any fees, costs, liabilities or expenses incurred or paid by Holdings, the Borrower or any of its respective Subsidiaries in connection with the Transactions, this Agreement and the other Credit Documents and the transactions contemplated hereby and thereby including in respect of the commitments, negotiation, syndication, documentation and closing (and post-closing actions in connection with the Collateral) of the Credit Facilities.

"Transactions" shall mean, collectively, (i) the execution of and funding under the Credit Documents and definitive documentation in respect of the DIP ABL Facility, if any, (ii) the Prepetition ABL Payoff, (iii) the other transactions contemplated by the Restructuring Support

Agreement and (iv) the payment of fees, costs, liabilities and expenses in connection with each of the foregoing and the consummation of any other transaction connected with the foregoing.

"Transferee" shall have the meaning provided in Section 13.6(e).

"Type" shall mean, as to any Term Loan, its nature as an ABR Loan or a SOFR Loan.

"UCC" shall mean the Uniform Commercial Code of the State of New York, or of any other state the laws of which are required to be applied in connection with the perfection of security interests in any Collateral.

"UK Financial Institution" means any BRRD Undertaking (as such term is defined under the PRA Rulebook (as amended from time to time) promulgated by the United Kingdom Prudential Regulation Authority) or any person falling within IFPRU 11.6 of the FCA Handbook (as amended from time to time) promulgated by the United Kingdom Financial Conduct Authority, which includes certain credit institutions and investment firms, and certain affiliates of such credit institutions or investment firms.

"UK Resolution Authority" means the Bank of England or any other public administrative authority having responsibility for the resolution of any UK Financial Institution.

"Unfunded Current Liability" of any Pension Plan shall mean the amount, if any, by which the Accumulated Benefit Obligation (as defined under Statement of Financial Accounting Standards No. 87 ("SFAS 87")) under the Pension Plan as of the close of its most recent plan year, determined in accordance with SFAS 87 as in effect on the Closing Date, exceeds the fair market value of the assets allocable thereto.

"U.S. Government Securities Business Day" shall mean any day except for (a) a Saturday, (b) a Sunday or (c) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities.

"U.S. Lender" shall have the meaning provided in Section 5.4(h).

"U.S. Trustee" shall have the meaning provided in the Restructuring Support Agreement.

"Variance Report" shall have the meaning set forth in Section 9.1(c).

"Variance Test Period" shall mean (i) initially, the first two weeks after the Petition Date, (ii) second, the first four-week period after the Petition Date and (iii) thereafter, each rolling four week period ending two weeks after the prior Variance Test Period.

"Voting Stock" shall mean, with respect to any Person, such Person's Stock or Stock Equivalents having the right to vote for the election of directors or other governing body of such Person under ordinary circumstances; *provided* that for the purpose of the definition of "Excluded Stock and Stock Equivalents" and in each reference to the Voting Stock of any CFC or

CFC Holding Company, Voting Stock shall also include any instrument treated as voting stock or stock equivalent for U.S. federal income tax purposes.

**"Wholly Owned"** shall mean, with respect to the ownership by a Person of a Subsidiary, that all of the Stock of such Subsidiary (other than directors' qualifying shares or nominee or other similar shares required pursuant to Applicable Law) are owned by such Person or another Wholly Owned Subsidiary of such Person.

"Withdrawal Liability" shall mean liability to a Multiemployer Plan as a result of a complete or partial withdrawal from such Multiemployer Plan, as such terms are defined in Title IV of ERISA.

"Write-Down and Conversion Powers" shall mean, (a) with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule and (b) with respect to the United Kingdom, any powers of the applicable Resolution Authority under the Bail-In Legislation to cancel, reduce, modify or change the form of a liability of any UK Financial Institution or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability or any of the powers under that Bail-In Legislation that are related to or ancillary to any of those powers.

## 1.2 Other Interpretive Provisions

With reference to this Agreement and each other Credit Document, unless otherwise specified herein or in such other Credit Document:

- (a) The meanings of defined terms are equally applicable to the singular and plural forms of the defined terms.
- (b) The words "herein", "hereto", "hereof" and "hereunder" and words of similar import when used in any Credit Document shall refer to such Credit Document as a whole and not to any particular provision thereof.
- (c) Article, Section, Exhibit and Schedule references are to the Credit Document in which such reference appears.
  - (d) The term "including" is by way of example and not limitation.
- (e) The term "documents" includes any and all instruments, documents, agreements, certificates, notices, reports, financial statements and other writings, however evidenced, whether in physical or electronic form.
- (f) The words "asset" and "property" shall be construed to have the same meaning and effect and refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract rights.

- (g) All references to "knowledge" or "awareness" of any Credit Party or a Subsidiary thereof means the actual knowledge of an Authorized Officer of a Credit Party or such Subsidiary.
- (h) In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including"; the words "to" and "until" each mean "to but excluding"; and the word "through" means "to and including".
- (i) Any reference herein to any Person shall be construed to include such Person's successors and permitted assigns and, in the case of any Governmental Authority, any other Governmental Authority that shall have succeeded to any or all of the functions thereof.
- (j) Section headings herein and in the other Credit Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Credit Document.
- For purposes of determining compliance with any one of Sections 10.1, 10.2, 10.3, 10.4, 10.5, 10.6, 10.7, 10.11 and 1.1, (i) in the event that any Lien, Investment, Indebtedness, merger, consolidation, amalgamation or similar fundamental change, Disposition, Restricted Payment, Affiliate transaction, contractual obligation or prepayment of Junior Indebtedness meets the criteria of more than one of the categories of transactions permitted pursuant to any clause of such Section, such transaction (or portion thereof) at any time and from time to time shall be permitted under one or more of such clauses as determined by the Borrower in its sole discretion at such time; provided that (x) all Indebtedness outstanding under the Credit Documents will be deemed at all times to have been incurred in reliance only on the exception in clause (a) of Section 10.1 and (ii) with respect to any Lien, Investment, Indebtedness, merger, consolidation, amalgamation or similar fundamental change, Disposition, Restricted Payment, Affiliate transaction, contractual obligation or prepayment of Junior Indebtedness or other applicable transaction in a currency other than Dollars, no Default or Event of Default shall be deemed to have occurred solely as a result of changes in rates of currency exchange occurring after the time such Lien, Investment, Indebtedness, merger, consolidation, amalgamation or similar fundamental change, Disposition, Restricted Payment, Affiliate transaction, contractual obligation or prepayment of Junior Indebtedness or other applicable transaction is made (so long as such Lien, Investment, Indebtedness, merger, consolidation, amalgamation or similar fundamental change, Disposition, Restricted Payment, Affiliate transaction, contractual obligation or prepayment of Junior Indebtedness or other applicable transaction at the time incurred or made was permitted hereunder).

# 1.3 Accounting Terms

All accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with, GAAP.

#### 1.4 Divisions

For all purposes under the Credit Documents, in connection with any division or plan of division under Delaware law (or any comparable event under a different jurisdiction's laws): (a) if any asset, right, obligation or liability of any Person becomes the asset, right, obligation or liability of a different Person, then it shall be deemed to have been transferred from the original Person to the subsequent Person, and (b) if any new Person comes into existence, such new Person shall be deemed to have been organized on the first date of its existence by the holders of its Equity Interests at such time.

## 1.5 References to Agreements, Laws, Etc

Unless otherwise expressly provided herein, (a) references to organizational documents, agreements (including the Credit Documents) and other Contractual Requirements shall be deemed to include all subsequent amendments, restatements, amendment and restatements, extensions, supplements and other modifications thereto, but only to the extent that such amendments, restatements, amendment and restatements, extensions, supplements and other modifications are permitted or not prohibited by any Credit Document and (b) references to any Applicable Law shall include all statutory and regulatory provisions consolidating, amending, replacing, supplementing or interpreting such Applicable Law.

## 1.6 Times of Day

Unless otherwise specified, all references herein to times of day shall be references to New York City time (daylight or standard, as applicable).

## 1.7 Timing of Payment or Performance

When the payment of any obligation or the performance of any covenant, duty or obligation is stated to be due or performance required on a day which is not a Business Day, the date of such payment (other than as described in the definition of Interest Period) or performance shall extend to the immediately succeeding Business Day.

## 1.8 Currency Equivalents Generally

In determining whether any Indebtedness, Investment, Lien, Disposition, Restricted Payment or any other amount under a "fixed amount" basket denominated in Dollars may be incurred in a currency other than Dollars, such amount shall be determined based on the currency exchange rate determined at the time of such incurrence (or, in the case of any revolving Indebtedness or any amount committed to be made, at the time it is first committed); provided that no Default or Event of Default shall be deemed to have occurred solely as a result of changes in rates of exchange occurring after the time such Indebtedness, Investment, Lien, Disposition, Restricted Payment or such other amount is incurred or made; *provided*, *further*, that for purpose of determining Consolidated Net Income, Consolidated EBITDA or any other amount determined based on Consolidated Net Income or Consolidated EBITDA, amounts in currencies other than Dollars shall be translated into Dollars at the currency exchange rates used in preparing the most recently delivered financial statements.

## 1.9 Classification of Loans and Borrowings

For purposes of this Agreement, Term Loans may be classified and referred to by Type (e.g., a "SOFR Loan"). Borrowings also may be classified and referred to by Type (e.g., a "SOFR Borrowing").

#### 1.10 Permitted Liens.

The permission of the existence of Liens set forth under Section 10.2 is not intended to subordinate or postpone, and shall not be interpreted as a subordinating or postponing, or as any agreement to subordinate or postpone, any Lien created under any of the Credit Documents to any such Lien; it being understood that the priority of any such Lien and the Lien created under any of the Credit Documents shall be determined by the Orders and by operation of the Applicable Law.

### **SECTION 2** Amount and Terms of Credit

### 2.1 The Loans.

Subject to the terms and conditions set forth herein and in the Orders, each Term Lender agrees, severally and not jointly, to make term loans (each a "Term Loan") in Dollars to the Borrower in two draws on or after the Closing Date, in an aggregate principal amount not to exceed its respective Term Commitment. For the avoidance of doubt, any Term Commitments (x) shall be reduced dollar for dollar at the time of funding of any Loans thereunder and shall terminate upon the earlier of (i) the funding of the Subsequent Draw and (ii) the Maturity Date. Each Borrowing shall consist of Term Loans of the same Type made on the same day by the Term Lenders ratably according to their respective Term Commitments; provided, that (x) the first Borrowing of Term Loans shall be in the principal amount equal to the lesser of (A) the full amount authorized by the Bankruptcy Court in the Interim Order and (B) \$400,000,000 (the "Initial Draw") and (y) the second Borrowing of Term Loans shall be in the principal amount equal to the difference between (i) the lesser of (A) the full amount of all Term Loans authorized by the Bankruptcy Court in the Final Order (inclusive of the Initial Draw) and (B) \$500,000,000 and (ii) the principal amount of the Initial Draw (the "Subsequent Draw"). Term Loans prepaid or repaid may not be reborrowed.

## 2.2 Maximum Number of Borrowings

After giving effect to all Borrowings, all conversions of Term Loans from one Type to the other, and all continuations of Term Loans as the same Type, there shall not be more than two (2) Interest Periods in effect unless otherwise agreed between the Borrower and the Administrative Agent.

## 2.3 Notice of Borrowing

(a) Each Borrowing shall be made upon the Borrower's irrevocable Notice of Borrowing to the Administrative Agent. Each such notice must be received by the Administrative Agent not later than (i) in the case of the Initial Draw, 1:00 p.m. one (1) U.S. Government Securities Business Day prior to the requested date of such Borrowing, and (ii) in the case of the Subsequent Draw, 12:00 p.m. ten (10) U.S. Government Securities Business Days prior to the

requested date of such Borrowing, provided that, in the case of this clause (ii), the Borrower may not submit such notice prior to the Final Order Entry Date. Each Notice of Borrowing shall be appropriately completed and signed by an Authorized Officer of the Borrower. Each Notice of Borrowing shall specify (i) whether the Borrower is requesting a Borrowing, (ii) the requested date of the Borrowing (which shall be a Business Day), (iii) the principal amount of Term Loans to be borrowed, (iv) the Type of Term Loans to be borrowed and (v) if applicable, the duration of the Interest Period with respect thereto. If the Borrower fails to specify a Type of Term Loan in a Notice of Borrowing, then the applicable Term Loans shall be made as ABR Loans. If the Borrower requests a Borrowing of SOFR Loans in any such Notice of Borrowing, but fails to specify an Interest Period, it will be deemed to have specified an Interest Period of one (1) month.

- (b) Following receipt of a Notice of Borrowing, the Administrative Agent shall promptly notify each Lender of the amount of its *pro rata* share of the Term Loans. In the case of each Borrowing, each Lender shall make the amount of its Term Loan available to the Administrative Agent in immediately available funds at the Administrative Agent's Office in Dollars not later than 1:00 p.m. on the Business Day specified in the applicable Notice of Borrowing. Upon satisfaction of the applicable conditions set forth in Section 6, the Administrative Agent shall make all funds so received available to the Borrower in like funds as received by the Administrative Agent either by (i) crediting the account of the Borrower on the books of the Administrative Agent with the amount of such funds or (ii) wire transfer of such funds, in each case in accordance with instructions provided to (and reasonably acceptable to) the Administrative Agent by the Borrower.
- (c) The Administrative Agent shall promptly notify the Borrower and the Lenders of the interest rate applicable to any Interest Period for SOFR Loans upon determination of such interest rate. The determination of the Adjusted Term SOFR by the Administrative Agent shall be conclusive in the absence of manifest error. At any time that ABR Loans are outstanding, the Administrative Agent shall notify the Borrower and the Lenders of any change in the prime rate used in determining the ABR promptly following the public announcement of such change.

### 2.4 Disbursement of Funds

- (a) No later than 12:00 p.m. on the date specified in each Notice of Borrowing, each Lender will make available its *pro rata* portion, if any, of each Borrowing requested to be made on such date in the manner provided below; *provided* that on the Closing Date, such funds may be made available at such earlier time as may be agreed among the Borrower, the Administrative Agent and the Lenders for the purpose of consummating the Transactions.
- (b) Each Lender shall make available all amounts required under any Borrowing for its applicable Commitments in immediately available funds to the Administrative Agent at the Administrative Agent's Office in Dollars, and the Administrative Agent will make available to the Borrower, by depositing to an account designated by the Borrower to the Administrative Agent the aggregate of the amounts so made available in Dollars. Unless the Administrative Agent shall have been notified by any Lender prior to the date of any such Borrowing that such Lender does not intend to make available to the Administrative Agent its portion of the Borrowing or Borrowings to be made on such date, the Administrative Agent may assume that such Lender has made such amount available to the Administrative Agent on such

date of Borrowing, and the Administrative Agent, in reliance upon such assumption, may (in its sole discretion and without any obligation to do so) make available to the Borrower a corresponding amount. If such corresponding amount is not in fact made available to the Administrative Agent by such Lender and the Administrative Agent has made available such amount to the Borrower, the Administrative Agent shall be entitled to recover such corresponding amount from such Lender. If such Lender does not pay such corresponding amount forthwith upon the Administrative Agent's demand therefor the Administrative Agent shall promptly notify the Borrower in writing and the Borrower shall immediately pay such corresponding amount to the Administrative Agent in Dollars. The Administrative Agent shall also be entitled to recover from such Lender or the Borrower interest on such corresponding amount in respect of each day from the date such corresponding amount was made available by the Administrative Agent to the Borrower to the date such corresponding amount is recovered by the Administrative Agent, at a rate *per annum* equal to (i) if paid by such Lender, the Overnight Rate or (ii) if paid by the Borrower, the then-applicable rate of interest or fees, calculated in accordance with Section 2.8, for the Term Loans.

- (c) Nothing in this Section 2.4 shall be deemed to relieve any Lender from its obligation to fulfill its commitments hereunder or to prejudice any rights that the Borrower may have against any Lender as a result of any default by such Lender hereunder (it being understood, however, that no Lender shall be responsible for the failure of any other Lender to fulfill its commitments hereunder).
  - 2.5 Repayment of Term Loans; Evidence of Debt
- (a) On the Maturity Date, either (i) the Term Loans shall be converted into Exit Term Loans in accordance with Section 2.15 and the Put Option Premium shall be paid in New Equity Interests pursuant to Section 4.1(c) and the Exit Fee shall be paid in cash in accordance with Section 4.1(c), or (ii) in all other instances, the Borrower shall repay to the Administrative Agent, for the benefit of the Lenders holding Term Loans, the then outstanding Term Loans and the Put Option Premium and the Exit Fee all in cash.
  - (b) The Term Loans shall not amortize.
  - (c) [Reserved].
- (d) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of the Borrower to the appropriate lending office of such Lender resulting from each Term Loan made by such lending office of such Lender from time to time, including the amounts of principal and interest payable and paid to such lending office of such Lender from time to time under this Agreement.
- (e) The Administrative Agent shall maintain the Register pursuant to Section 13.6(b), and a subaccount for each Lender, in which Register and subaccounts (taken together) shall be recorded (i) the amount of each Term Loan made hereunder and, if applicable, the relevant tranche thereof and the Type of each Term Loan made and the Interest Period applicable thereto, (ii) the amount of any principal or interest due and payable or to become due and payable from the Borrower to each Lender hereunder, (iii) the amount of any sum received by the Administrative

Agent hereunder from the Borrower and each Lender's share thereof, and (iv) any cancellation or retirement of Term Loans as contemplated by Section 13.6(g).

- (f) The entries made in the Register and accounts and subaccounts maintained pursuant to clauses (d) and (e) of this Section 2.5 shall, to the extent permitted by Applicable Law, be prima facie evidence of the existence and amounts of the obligations of the Borrower therein recorded; *provided*, *however*, that the failure of any Lender or the Administrative Agent to maintain such account, such Register or such subaccount, as applicable, or any error therein, shall not in any manner affect the obligation of the Borrower to repay (with applicable interest) the Term Loans made to the Borrower by such Lender in accordance with the terms of this Agreement.
- (g) The Borrower hereby agrees that, upon request of any Lender at any time and from time to time after the Borrower has made the Initial Draw, the Borrower shall provide to such Lender, at the Borrower's expense a promissory note substantially in the form of Exhibit B, evidencing the Term Loans owing to such Lender.
- (h) The Term Loans (together with the Put Option Premium and the Exit Fee) shall mature and become due and payable on the Maturity Date or shall be converted into Exit Term Loans in accordance with Section 2.15.

#### 2.6 SOFR/Base Rate Conversions and Continuations

Subject to the penultimate sentence of this clause (a), (x) the Borrower shall (a) have the option on any Business Day to convert all or a portion equal to at least the Minimum Borrowing Amount of the outstanding principal amount of any Term Loans of one Type into a Borrowing or Borrowings of another Type, and (y) the Borrower shall have the option on any Business Day to continue the outstanding principal amount of any SOFR Loans as SOFR Loans, for an additional Interest Period; provided that (i) no partial conversion of SOFR Loans shall reduce the outstanding principal amount of SOFR Loans made pursuant to a single Borrowing to less than the Minimum Borrowing Amount, (ii) ABR Loans may not be converted into SOFR Loans if an Event of Default is in existence on the date of the conversion and the Administrative Agent has or the Required Lenders have determined in its or their sole discretion not to permit such conversion, (iii) SOFR Loans may not be continued as SOFR Loans for an additional Interest Period if an Event of Default is in existence on the date of the proposed continuation and the Required Lenders have determined in their sole discretion not to permit such continuation, and (iv) Borrowings resulting from conversions pursuant to this Section 2.6 shall be limited in number as provided in Section 2.2. Each such conversion or continuation shall be effected by the Borrower by giving the Administrative Agent at the Administrative Agent's Office prior to 1:00 p.m. at least (i) one Business Day's in the case of a conversion into ABR Loans or (ii) three U.S. Government Securities Business Days', in the case of a continuation of, or conversion to, SOFR Loans, prior written notice, in each case substantially in the form of Exhibit A (each, a "Notice of Conversion or Continuation") specifying the Term Loans to be so converted or continued, the Type of Term Loans to be converted into or continued and, if such Term Loans are to be converted into, or continued as, SOFR Loans, the Interest Period to be initially applicable thereto (if no Interest Period is selected, the Borrower shall be deemed to have selected an Interest Period of one month's duration). The Administrative Agent shall give each applicable Lender notice as promptly as practicable of any such proposed conversion or continuation affecting any of its Term Loans.

- (b) If any Event of Default is in existence at the time of any proposed continuation of any SOFR Loans and the Required Lenders have determined in their sole discretion not to permit such continuation, such SOFR Loans shall be automatically converted on the last day of the current Interest Period into ABR Loans. If upon the expiration of any Interest Period in respect of SOFR Loans, the Borrower has failed to elect a new Interest Period to be applicable thereto as provided in clause (a) above, the Borrower shall be deemed to have elected to convert such Borrowing of SOFR Loans into a Borrowing of ABR Loans, effective as of the expiration date of such current Interest Period.
- (c) Notwithstanding anything to the contrary herein, the Borrower may deliver a Notice of Conversion or Continuation pursuant to which the Borrower elects to irrevocably continue the outstanding principal amount of any Term Loans subject to an interest rate Hedging Agreement as SOFR Loans for each Interest Period until the expiration of the term of such applicable Hedging Agreement.

## 2.7 Benchmark Replacement Setting

- (a) Benchmark Replacement. Notwithstanding anything to the contrary herein or in any other Credit Document, upon the occurrence of a Benchmark Transition Event, the Administrative Agent and the Borrower may amend this Agreement to replace the then-current Benchmark with a Benchmark Replacement. Any such amendment with respect to a Benchmark Transition Event will become effective at 5:00 p.m. (New York City time) on the fifth (5th) Business Day after the Administrative Agent has posted such proposed amendment to all affected Lenders and the Borrower so long as the Administrative Agent has not received, by such time, written notice of objection to such amendment from Lenders comprising the Required Lenders. No replacement of a Benchmark with a Benchmark Replacement pursuant to this Section 2.7(a)(i) will occur prior to the applicable Benchmark Transition Start Date.
- (b) Benchmark Replacement Conforming Changes. In connection with the use, administration, adoption or implementation of a Benchmark Replacement, the Administrative Agent will have the right to make Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Credit Document, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Credit Document.
- Agent will promptly notify the Borrower and the Lenders of (i) the implementation of any Benchmark Replacement and (ii) the effectiveness of any Conforming Changes in connection with the use, administration, adoption or implementation of a Benchmark Replacement. The Administrative Agent will notify the Borrower of (x) the removal or reinstatement of any tenor of a Benchmark pursuant to Section 2.7(d) and (y) the commencement of any Benchmark Unavailability Period. Any determination, decision or election that may be made by the Administrative Agent or, if applicable, any Lender (or group of Lenders) pursuant to this Section 2.7, including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made

in its or their sole discretion and without consent from any other party to this Agreement or any other Credit Document, except, in each case, as expressly required pursuant to this Section 2.7.

- (d) Unavailability of Tenor of Benchmark. Notwithstanding anything to the contrary herein or in any other Credit Document, at any time (including in connection with the implementation of a Benchmark Replacement), (i) if the then-current Benchmark is a term rate (including the Term SOFR Reference Rate) and either (A) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by the Administrative Agent in its reasonable discretion or (B) the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor for such Benchmark is not or will not be representative, then the Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for any Benchmark settings at or after such time to remove such unavailable or non-representative tenor and (ii) if a tenor that was removed pursuant to clause (i) above either (A) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (B) is not, or is no longer, subject to an announcement that it is not or will not be representative for a Benchmark (including a Benchmark Replacement), then the Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for all Benchmark settings at or after such time to reinstate such previously removed tenor.
- (e) Benchmark Unavailability Period. Upon the Borrower's receipt of notice of the commencement of a Benchmark Unavailability Period, the Borrower may revoke any pending request for a SOFR Borrowing of, conversion to or continuation of SOFR Loans to be made, converted or continued during any Benchmark Unavailability Period and, failing that, the Borrower will be deemed to have converted any such request into a request for a Borrowing of or conversion to ABR Loans. During a Benchmark Unavailability Period or at any time that a tenor for the then-current Benchmark is not an Available Tenor, the component of ABR based upon the then-current Benchmark or such tenor for such Benchmark, as applicable, will not be used in any determination of ABR.

### 2.8 Interest

- (a) The unpaid principal amount of each ABR Loan shall bear interest from the date of the Borrowing thereof until maturity (whether by acceleration or otherwise) at a rate *per annum* that shall at all times be the Applicable ABR Margin *plus* the ABR, in each case, in effect from time to time.
- (b) The unpaid principal amount of each SOFR Loan shall bear interest from the date of the Borrowing thereof until maturity thereof (whether by acceleration or otherwise) at a rate *per annum* that shall at all times be the Applicable SOFR Margin *plus* the relevant Adjusted Term SOFR, in each case in effect from time to time.
- (c) At all times during which an Event of Default is continuing, the Borrower shall pay interest on the principal amount of all Loans outstanding hereunder (and any past due amounts not constituting principal) at a rate *per annum* (the "**Default Rate**") that is equal to the rate that would otherwise be applicable thereto *plus* 2% to the fullest extent permitted by

Applicable Laws. Accrued and unpaid interest on past due amounts (including interest on past due interest) shall be due and payable upon demand.

- Borrowing to but excluding the date of any repayment thereof and shall be payable in Dollars; provided that any Term Loan that is repaid on the same date on which it is made shall bear interest for one day. Except as provided below, interest shall be payable (i) in respect of each ABR Loan, quarterly in arrears on the last Business Day of each March, June, September and December, (ii) in respect of each SOFR Loan, on the last day of each Interest Period applicable thereto and, in the case of an Interest Period in excess of three months, on each Business Day occurring at three-month intervals after the first day of such Interest Period, and (iii) in respect of each Term Loan, (A) on any prepayment; provided that interest on ABR Loans shall only become due pursuant to this clause (A) if the aggregate principal amount of the ABR Loans then-outstanding is repaid in full, (B) at maturity (whether by acceleration or otherwise) and (C) after such maturity, on demand.
- (e) All computations of interest hereunder shall be made in accordance with Section 5.5.

#### 2.9 Interest Periods

At the time the Borrower gives a Notice of Borrowing or Notice of Conversion or Continuation in respect of the making of, or conversion into or continuation as, a Borrowing of SOFR Loans in accordance with Section 2.6(a), the Borrower shall give the Administrative Agent written notice of the Interest Period applicable to such Borrowing, which Interest Period shall, at the option of the Borrower, be one, three or six months.

Notwithstanding anything to the contrary contained above:

- (a) the initial Interest Period for any Borrowing of SOFR Loans shall commence on the date of such Borrowing (including the date of any conversion from a Borrowing of ABR Loans) and each Interest Period occurring thereafter in respect of such Borrowing shall commence on the day on which the next preceding Interest Period expires;
- (b) if any Interest Period relating to a Borrowing of SOFR Loans begins on the last Business Day of a calendar month or begins on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period, such Interest Period shall end on the last Business Day of the calendar month at the end of such Interest Period;
- (c) if any Interest Period would otherwise expire on a day that is not a Business Day, such Interest Period shall expire on the next succeeding Business Day; *provided* that if any Interest Period in respect of a SOFR Loan would otherwise expire on a day that is not a Business Day but is a day of the month after which no further Business Day occurs in such month, such Interest Period shall expire on the next preceding Business Day; and
- (d) the Borrower shall not be entitled to elect any Interest Period in respect of any SOFR Loan if such Interest Period would extend beyond the applicable Maturity Date of such Term Loan.

The Administrative Agent will have the right to make Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Credit Document, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Credit Document. The Administrative Agent will promptly notify the Borrower and the Lenders of the effectiveness of any Conforming Changes in connection with the use or administration of Term SOFR.

- 2.10 Increased Costs, Illegality, Etc.
- (a) [Reserved].
- (b) [Reserved].
- If, after the Closing Date, any Change in Law relating to capital adequacy (c) or liquidity of any Lender or compliance by any Lender or its parent with any Change in Law relating to capital adequacy or liquidity occurring after the Closing Date, has or would have the effect of reducing the rate of return on such Lender's or its parent's or its Affiliates' capital or assets as a consequence of such Lender's commitments or obligations hereunder to a level below that which such Lender or its parent or any Affiliate thereof could have achieved but for such Change in Law (taking into consideration such Lender's or parent's policies with respect to capital adequacy or liquidity), then from time to time, promptly after written demand by such Lender (with a copy to the Administrative Agent), the Borrower shall pay to such Lender such additional amount or amounts as will compensate such Lender or its parent for such reduction, it being understood and agreed, however, that a Lender shall not be entitled to such compensation as a result of such Lender's compliance with, or pursuant to any request or directive to comply with, any Applicable Law as in effect on the Closing Date. Each Lender, upon determining in good faith that any additional amounts will be payable pursuant to this Section 2.10(c), will give prompt written notice thereof to the Borrower, which notice shall set forth in reasonable detail the basis of the calculation of such additional amounts, although the failure to give any such notice shall not, subject to Section 2.13, release or diminish the Borrower's obligations to pay additional amounts pursuant to this Section 2.10(c) upon receipt of such notice.
- (d) Notwithstanding the foregoing, no Lender shall demand compensation pursuant to this Section 2.10 if it shall not at the time be the general policy or practice of such Lender to demand such compensation in substantially the same manner as applied to other similarly situated borrowers under comparable syndicated credit facilities.
  - (e) [Reserved].
- (f) Subject to Section 2.7, if, on or prior to the first day of any Interest Period for any SOFR Loan:
  - (i) the Administrative Agent determines (which determination shall be conclusive and binding absent manifest error) that "Adjusted Term SOFR" cannot be determined pursuant to the definition thereof, or
  - (ii) the Required Lenders determine that for any reason in connection with any request for a SOFR Loan or a conversion thereto or a continuation thereof that

Adjusted Term SOFR for any requested Interest Period with respect to a proposed SOFR Loan does not adequately and fairly reflect the cost to such Lenders of making and maintaining such Loan, and the Required Lenders have provided notice of such determination to the Administrative Agent,

the Administrative Agent will promptly so notify the Borrower and each Lender.

Upon notice thereof by the Administrative Agent to the Borrower, any obligation of the Lenders to make SOFR Loans, and any right of the Borrower to continue SOFR Loans or to convert ABR Loans to SOFR Loans, shall be suspended (to the extent of the affected SOFR Loans or affected Interest Periods) until the Administrative Agent (with respect to clause (ii), at the instruction of the Required Lenders) revokes such notice. Upon receipt of such notice, (i) the Borrower may revoke any pending request for a borrowing of, conversion to or continuation of SOFR Loans (to the extent of the affected SOFR Loans or affected Interest Periods) or, failing that, the Borrower will be deemed to have converted any such request into a request for a Borrowing of or conversion to ABR Loans in the amount specified therein and (ii) any outstanding affected SOFR Loans will be deemed to have been converted into ABR Loans at the end of the applicable Interest Period. Upon any such conversion, the Borrower shall also pay accrued interest on the amount so converted, together with any additional amounts required pursuant to Section 2.11. Subject to Section 2.7, if the Administrative Agent determines (which determination shall be conclusive and binding absent manifest error) that "Adjusted Term SOFR" cannot be determined pursuant to the definition thereof on any given day, the interest rate on ABR Loans shall be determined by the Administrative Agent without reference to clause (c)(ii) of the definition of "ABR" until the Administrative Agent revokes such determination.

### 2.11 Compensation

If (i) any payment of principal of any SOFR Loan is made by the Borrower to or for the account of a Lender other than on the last day of the Interest Period for such SOFR Loan as a result of a payment or conversion pursuant to Section 2.5, 2.6, 2.10, 5.1, 5.2 or 13.7, as a result of acceleration of the maturity of the Term Loans pursuant to Section 11 or for any other reason, (ii) any Borrowing of SOFR Loans is not made as a result of a withdrawn Notice of Borrowing, (iii) any ABR Loan is not converted into a SOFR Loan as a result of a withdrawn Notice of Conversion or Continuation, (iv) any SOFR Loan is not continued as a SOFR Loan as a result of a withdrawn Notice of Conversion or Continuation or (v) any prepayment of principal of any SOFR Loan is not made as a result of a withdrawn notice of prepayment pursuant to Section 5.1 or 5.2, the Borrower shall, after receipt of a written request by such Lender (which request shall set forth in reasonable detail the basis for requesting such amount), pay to the Administrative Agent for the account of such Lender any amounts required to compensate such Lender for any additional losses, costs or expenses that such Lender may reasonably incur as a result of such payment, failure to convert, failure to continue or failure to prepay, including any loss, cost or expense (excluding loss of anticipated profits) actually incurred by reason of the liquidation or reemployment of deposits or other funds acquired by any Lender to fund or maintain such SOFR Loan. Notwithstanding the foregoing, no Lender shall demand compensation pursuant to this Section 2.11 if it shall not at the time be the general policy or practice of such Lender to demand such

compensation in substantially the same manner as applied to other similarly situated borrowers under comparable syndicated credit facilities.

## 2.12 Change of Lending Office

Each Lender agrees that, upon the occurrence of any event giving rise to the operation of Section 2.10(a)(ii), 2.10(a)(iii), 2.10(b) or 5.4 with respect to such Lender, it will, if requested by the Borrower, use reasonable efforts (subject to overall policy considerations of such Lender) to designate another lending office for any Term Loans affected by such event; *provided* that such designation is made on such terms that such Lender and its lending office suffer no economic, legal or regulatory disadvantage, with the object of avoiding the consequence of the event giving rise to the operation of any such Section. Nothing in this Section 2.12 shall affect or postpone any of the obligations of the Borrower or the right of any Lender provided in Section 2.10 or 5.4. The Borrower hereby agrees to pay all reasonable costs and expenses incurred by any Lender in connection with such designation.

#### 2.13 Notice of Certain Costs

Notwithstanding anything in this Agreement to the contrary, to the extent any notice required by Section 2.10, 2.11 or 5.4 is given by any Lender more than 180 days after such Lender has knowledge (or should have had knowledge) of the occurrence of the event giving rise to the additional cost, reduction in amounts, loss, tax or other additional amounts described in such Sections, such Lender shall not be entitled to compensation under Section 2.10, 2.11 or 5.4, as the case may be, for any such amounts incurred or accruing prior to the 181st day prior to the giving of such notice to the Borrower.

## 2.14 [Reserved]

### 2.15 Conversion to Exit Term Loans

On the Conversion Date, automatically and without any further consent or action required by any Agent or any Lender or notice by the Borrower, the aggregate principal amount of the Loans outstanding as of such date shall be automatically converted on a dollar-for-dollar basis for principal amount of the "Exit Term Loans" (as defined in the Exit Term Loan Term Sheet) in accordance with the Exit Term Loan Term Sheet and the Restructuring Support Agreement and deemed funded under the Exit Term Loan Facility; the Credit Parties (or the entities assuming the operations and assets of the Credit Parties in the Plan, to the extent such Person is required under the Exit Term Loan Term Sheet or the Plan to continue to be an obligor thereunder) shall assume all obligations in respect of the Loans hereunder that are converted into "Exit Term Loans" (as defined in the Exit Term Loan Term Sheet) and deemed funded under the Exit Term Loan Facility; each Lender shall be a lender under the Exit Term Loan Facility in accordance with the Restructuring Support Agreement and the Exit Term Loan Term Sheet; provided that all other Obligations (other than the principal amount of the Loans so converted), including, for the avoidance of doubt, accrued and unpaid interest and the Exit Fee, shall be paid in full in cash (other than the Put Option Premium, which shall be paid in New Equity Interests on the Conversion Date), upon which this Agreement shall terminate and be superseded and replaced in its entirety by the Exit Term Loan Facility. Notwithstanding the foregoing, all obligations of the Borrower and the other Credit Parties to the Agents and Lenders under this Agreement and any other Credit Document which are expressly stated in this Agreement or such other Credit Document as surviving such agreement's termination shall, as so specified, survive without prejudice and remain in full force and effect. Each of the Credit Parties, Agents and the Lenders shall take such actions and execute and deliver such agreements, instruments or other documents reasonably required to give effect to the provisions of this Section 2.15 and as are required to complete the schedules to the Exit Term Loan Facility or other agreements contemplated thereby.

## 2.16 Defaulting Lenders

Notwithstanding any provision of this Agreement to the contrary, if any Lender becomes a Defaulting Lender, then no Defaulting Lender shall be entitled to receive any fee payable under Section 4 or any interest at the Default Rate payable under Section 2.8(d) for any period during which that Lender is a Defaulting Lender (and the Borrower shall not be required to pay any such fee or interest that otherwise would have been required to have been paid to that Defaulting Lender).

## 2.17 Priority and Liens

Each of the Credit Parties hereby covenants and agrees that upon the entry of, and subject to, an Interim Order (and when applicable, the Final Order) and subject to the Carve Out in all respects and, subject to the DIP ABL Intercreditor Agreement, the Obligations: (i) pursuant to section 364(c)(1) of the Bankruptcy Code, shall at all times constitute allowed Superpriority Claims in the Chapter 11 Cases, which Superpriority Claims in respect of the Term Facility and the DIP ABL Facility, if any, shall rank pari passu with each other; (ii) pursuant to section 364(c)(2) of the Bankruptcy Code, shall at all times be secured by a valid, binding, continuing, enforceable perfected first priority Lien (subject to the terms of the Security Agreement and the Orders but which will, in any event, be subject to the Carve Out) on all of the Collateral, whether now existing or hereafter acquired, that is not subject to valid, perfected, nonvoidable liens in existence at the time of commencement of the Chapter 11 Cases or to valid, nonvoidable liens in existence at the time of such commencement that are perfected subsequent to such commencement as permitted by section 546(b) of the Bankruptcy Code; (iii) pursuant to section 364(c)(3) of the Bankruptcy Code, shall be secured by a valid, binding, continuing, enforceable perfected junior Lien upon all Collateral, whether now existing or hereafter acquired, that is subject to valid, perfected and nonvoidable Liens in favor of Persons that are not Affiliates of the Credit Parties and in existence at the time of the commencement of the Chapter 11 Cases or that is subject to valid Liens in existence at the time of the commencement of the Chapter 11 Cases that are perfected subsequent to such commencement as permitted by section 546(b) of the Bankruptcy Code (other than certain property that is subject to the existing Liens that secure the Prepetition First Lien Debt, which liens shall be primed by the liens described in the following clause (iv)); and (iv) pursuant to section 364(d)(l) of the Bankruptcy Code, shall be secured by a valid, binding, continuing, enforceable perfected senior priming Lien on all Collateral that is subject to the Liens which secure the Prepetition First Lien Debt (the "Primed Liens") other than with respect to Liens on Collateral that is Specified Collateral securing the DIP ABL Facility, if any, all of which Primed Liens shall be primed by and made subject and subordinate to the perfected first priority senior Liens to be granted to the Administrative Agent, which senior priming Liens in favor of the Administrative Agent shall also prime any Liens granted after the

commencement of the Chapter 11 Cases to provide adequate protection Liens to the extent of any diminution in the value of the collateral of the Primed Liens as provided in the Interim Order and the Final Order in respect of any of the Primed Liens, subject in each case to the Carve Out in all respects and as set forth in the Orders (the "**Priming Liens**") and with respect to perfection, solely to the extent it may be achieved by the entry of the Orders and the perfection steps required to be taken under the Security Agreement.

(b) The Priming Liens, (i) except as set forth in clause (c) below, shall be subject and junior to the Carve Out in all respects, (ii) shall be junior to Liens that are senior to the Primed Liens (unless such Liens are themselves Primed Liens), (iii) shall be senior in all respects to the interests of such property of the holders of the obligations in respect of the Primed Liens and (iv) shall also be senior to any Liens granted after the Petition Date to provide adequate protection in respect of the Primed Liens.

## (c) [Reserved].

- (d) The relative priorities of the Liens described in this Section 2.17 with respect to the Collateral shall be as set forth in the Interim Order (and, when entered, the Final Order). In accordance with the Interim Order (or, once entered, the Final Order), all of the Liens described in this Section 2.17 shall be effective and perfected upon entry of the Interim Order, without the necessity of the execution, recordation of filings by the Debtors of security agreements, control agreements, pledge agreements, financing statements or other similar documents, or the possession or control by the Administrative Agent of, or over, any Collateral, as set forth in the Interim Order.
- Further to Section 2.17(a) and the Interim Order (and, when entered, the Final Order), to secure the full and timely payment and performance of the Obligations, each Credit Party hereby MORTGAGES, GRANTS, BARGAINS, ASSIGNS, SELLS, CONVEYS and CONFIRMS, to the Administrative Agent, for the ratable benefit of the Secured Parties, the Real Estate and property and interests relating thereto (which, for the avoidance of doubt, shall include all of such Credit Party's right, title and interest now or hereafter acquired in and to (A) all improvements now owned or hereafter acquired by such Credit Party, (B) all materials, supplies, equipment, apparatus and other items of personal property now owned or hereafter acquired by such Credit Party and now or hereafter attached to, installed in or used in connection with the Real Estate, and all utilities whether or not situated in easements, and all equipment, inventory and other goods in which such Credit Party now has or hereafter acquires any rights or any power to transfer rights and that are or are to become fixtures (as defined in the Uniform Commercial Code) related to the Real Estate, (C) [reserved], (D) all reserves, escrows or impounds and all deposit accounts maintained by such Credit Party with respect to the Real Estate, (E) all leases, licenses, concessions, occupancy agreements or other agreements (written or oral, now or at any time in effect) which grant to any Person a possessory interest in, or the right to use, all or any part of the Real Estate, together with all related security and other deposits, (F) all of the rents, revenues, royalties, income, proceeds, profits, accounts receivable, security and other types of deposits, and other benefits paid or payable by parties to the leases for using, leasing, licensing possessing, operating from, residing in, selling or otherwise enjoying the Real Estate, (G) all other agreements, such as construction contracts, architects' agreements, engineers' contracts, utility contracts, maintenance agreements, management agreements, service contracts, listing agreements,

guaranties, warranties, permits, licenses, certificates and entitlements in any way relating to the construction, use, occupancy, operation, maintenance, enjoyment or ownership of the Real Estate, (H) all rights, privileges, tenements, hereditaments, rights-of-way, easements, appendages and appurtenances appertaining to the foregoing, (I) all property tax refunds payable with respect to the Real Estate, (J) all accessions, replacements and substitutions for any of the foregoing and all proceeds thereof, (K) all insurance policies, unearned premiums therefor and proceeds from such policies covering any of the above property now or hereafter acquired by such Credit Party as an insured party, and (L) all awards, damages, remunerations, reimbursements, settlements or compensation heretofore made or hereafter to be made to any Credit Party by any governmental authority pertaining to any condemnation or other taking (or any purchase in lieu thereof) of all or any Real Estate), TO HAVE AND TO HOLD to the Administrative Agent, and such Credit Party does hereby bind itself, its successors and assigns to WARRANT AND FOREVER DEFEND the title to such property, assets and interests unto the Administrative Agent. Notwithstanding the foregoing, excluded from the foregoing grant of Lien is any right, title and interest of any Credit Party in and to any Excluded Collateral.

## 2.18 No Discharge; Survival of Claims

Each of the Credit Parties agrees that prior to Payment in Full of the Obligations or deemed Payment in Full pursuant to Section 2.15, (a) its obligations under the Credit Documents shall not be discharged by the entry of an order confirming a Reorganization Plan (and each of the Credit Parties, pursuant to section 1141(d)(4) of the Bankruptcy Code, hereby waives any such discharge) and (b) the Superpriority Claim granted to the Agents and the Lenders pursuant to the Orders and the Liens granted to the Agents and the Lenders pursuant to the Orders shall not be affected in any manner by the entry of an order confirming a Reorganization Plan.

### 2.19 Payment of Obligations

Subject to the last paragraph of Section 11.20, upon the maturity (whether by acceleration or otherwise) of any of the Obligations of the Credit Parties under this Agreement or any of the other Credit Documents, the Administrative Agent and the Lenders shall be entitled to immediate payment of such Obligations without further application to or order of the Bankruptcy Court.

# **SECTION 3** [Reserved]

## **SECTION 4** Fees; Commitments

- 4.1 Fees. The Borrower shall pay to the Agents and the Lenders (as applicable), for their respective accounts, in Dollars:
- (a) all fees in the amounts and at the times specified in the Fee Letter due and payable to the Agents;
- (b) without duplication of amounts referred to in clause (a) above or (c), (d) and (e) below, to the Administrative Agent, for the account of each Lender, an upfront fee (the "Term Upfront Fee") (which, at the discretion of such Lender, may take the form of original issue discount) equal to 4.00% (the "Term Upfront Fee Percentage") of the aggregate principal

amount of the Loans funded hereunder in respect of each Borrowing referred to in Section 2.01(a), which Term Upfront Fee shall be earned, due and payable on the date of such Borrowing and calculated by multiplying the Term Upfront Fee Percentage by the aggregate principal amount of Loans funded by such Lender on the date of such Borrowing; provided that notwithstanding the foregoing, for the avoidance of doubt no applicable original issue discount shall reduce the amount of Obligations required to be prepaid or repaid with respect to the Loans owing hereunder;

- (c) without duplication of amounts referred to in clauses (a) and (b) above and (d) and (e) below, a fee (the "**Put Option Premium**") equal to 6.0% of the aggregate principal amount of the Commitments as of the date of the Commitment Letter without giving effect to any termination, reduction, or prepayment, which such Put Option Premium shall be earned on the date of the Commitment Letter and payable upon the repayment in full or deemed repayment in full pursuant to Section 2.15 or acceleration of the principal amount of the Term Loans, whether through optional prepayments, mandatory prepayments, at maturity, upon acceleration or otherwise, (x) to the extent the Plan shall have been consummated on or prior to the Maturity Date, to any Lender thereof or any Person designated by such Lender (pro rata in accordance with the outstanding principal amount of the Commitments and Loans held by such Lender on the date that the Put Option Premium is payable), in shares of New Equity Interests at a 37.5% discount to an implied equity value of \$538,812,500, or (y) otherwise, to the Administrative Agent, for the ratable account of each Lender thereof (pro rata in accordance with the outstanding principal amount of the Commitments and Loans held by such Lender on the date that the Put Option Premium is payable), in cash;
- (d) without duplication of amounts referred to in clauses (a), (b) and (c) above and (e) below, upon, and on the date of, any repayment or prepayment (whether pursuant to voluntary or mandatory prepayments provisions hereunder, and including on the Maturity Date) or acceleration of any Loans or termination of any Commitment, in full or in part (other than any termination of Commitments solely as a result of the Borrowing of any Term Loans hereunder), to the Administrative Agent, for the ratable account of each Lender thereof (pro rata in accordance with the outstanding principal amount of the Commitments and Loans held by such Lender on the date that the Exit Fee is payable), an exit fee (the "Exit Fee") in cash in an amount equal to 1.0% of the sum of the principal amount of such Loans repaid or prepaid or accelerated or Commitments terminated, which Exit Fee shall be earned on the Closing Date; and
- (e) without duplication of amounts referred to in clauses (a), (b), (c) and (d) above, to the Administrative Agent, for the account of each Lender, a fronting fee equal to 0.375% of the aggregate Commitments as of the Closing Date, which such fronting fee shall be earned, due and payable on the Closing Date;

Notwithstanding the foregoing, the Borrower shall not be obligated to pay any amounts to any Defaulting Lender pursuant to this Section 4.1 (subject to Section 2.17).

## 4.2 Mandatory Termination or Reduction of Commitments

The Commitments shall be automatically and promptly reduced on the date of any Borrowing of Term Loans pursuant to Section 2.1(a), by the amount of such Borrowing.

## **SECTION 5** Payments

## 5.1 Voluntary Prepayments

The Borrower shall have the right to prepay Term Loans, without premium or penalty (other than amounts, if any, required to be paid pursuant to Section 2.11 with respect to prepayments of SOFR Loans made on any date other than the last day of the applicable Interest Period, the Exit Fee and, solely in the case of the prepayment in full of all Term Loans then outstanding, the Put Option Premium), in whole or in part, from time to time on the following terms and conditions: (a) the Borrower shall give the Administrative Agent at the Administrative Agent's Office revocable written notice of its intent to make such prepayment, the amount of such prepayment and, in the case of SOFR Loans, the specific Borrowing(s) pursuant to which made, which notice shall be given by the Borrower no later than 1:00 p.m. three (3) U.S. Government Securities Business Days prior to such prepayment and shall be promptly transmitted by the Administrative Agent to each Lender, (b) each partial prepayment of Term Loans shall be in a minimum amount of \$1,000,000 and in multiples of \$500,000 in excess thereof; provided that no partial prepayment of SOFR Loans made pursuant to a single Borrowing shall reduce the outstanding SOFR Loans made pursuant to such Borrowing to an amount less than the Minimum Borrowing Amount for SOFR Loans and (c) any prepayment of SOFR Loans pursuant to this Section 5.1 on any day prior to the last day of an Interest Period applicable thereto shall be subject to compliance by the Borrower with the applicable provisions of Section 2.11. Each prepayment in respect of any Term Loans pursuant to this Section 5.1 shall be paid to the Term Lenders in accordance with their respective pro rata shares of the Term Loans. At the Borrower's election in connection with any prepayment pursuant to this Section 5.1, such prepayment shall not be applied to any Term Loan of a Defaulting Lender.

## 5.2 Mandatory Prepayments

### (a) Prepayment Events.

- (i) On each occasion that an Asset Sale Prepayment Event or a Recovery Prepayment Event occurs, the Borrower shall, within one Business Day after the receipt of Net Cash Proceeds of such Prepayment Event, in accordance with clauses (c) and (d) below, prepay (or cause to be prepaid) Term Loans with a principal amount equal to 100% of the Net Cash Proceeds from such Prepayment Event; *provided*, that no prepayment shall be required pursuant to this Section 5.2(a)(i) unless and until the amount at any time of Net Cash Proceeds from such Prepayment Events required to be applied at or prior to such time pursuant to this Section 5.2(a)(i) and not yet applied at or prior to such time to prepay Term Loans pursuant to this Section exceeds \$5,000,000 in the aggregate for all such Prepayment Events.
- (ii) On each occasion that a Debt Incurrence Prepayment Event occurs, the Borrower shall immediately, upon the receipt of the Net Cash Proceeds from the occurrence of such Debt Incurrence Prepayment Event, prepay (or cause to be prepaid), in accordance with clauses (c) and (d) below, Term Loans with a principal amount equal to 100% of the Net Cash Proceeds from such Debt Incurrence Prepayment Event.

- (iii) On each occasion that an Extraordinary Receipts Prepayment Event occurs, the Borrower shall, within one Business Day after the receipt of Net Cash Proceeds of such Prepayment Event, prepay (or cause to be prepaid), in accordance with clauses (c) and (d) below, Term Loans with a principal amount equal to 100% of the Net Cash Proceeds from such Extraordinary Receipts Prepayment Event.
- (b) Each such prepayment of Term Loans shall be subject to payment of the Exit Fee in accordance with Section 4.1, all accrued and unpaid interest to, but not including the date of repayment, and, solely in the case of the prepayment in full of all Term Loans then outstanding, payment of the Put Option Premium.
- (c) Application to Repayment Amounts. Each prepayment of Term Loans required by Section 5.2(a) (except as provided in Section 5.2(a)(iii)) or Section 5.2(b) shall be allocated to the Term Loans then outstanding (ratably to the Term Loans (or on a less than ratable basis, if agreed to by the Lenders providing such Term Loans) based on then remaining principal amounts of the Term Loans then outstanding) until paid in full.
- (d) Application to Term Loans. With respect to each prepayment of Term Loans required pursuant to Section 5.2(a) (other than Section 5.2(a)(iii)) or Section 5.2(b), subject to Section 11.11 in the case of Section 5.2(a)(ii) (when applicable), the Borrower may designate the Types of Term Loans that are to be prepaid and the specific Borrowing(s) pursuant to which made; *provided* that the Borrower pays any amounts, if any, required to be paid pursuant to Section 2.11 with respect to prepayments of SOFR Loans made on any date other than the last day of the applicable Interest Period. In the absence of a Rejection Notice or a designation by the Borrower as described in the preceding sentence, the Administrative Agent may, subject to the above, make such designation in its reasonable discretion with a view, but no obligation, to minimize breakage costs owing under Section 2.11.

### (e) [Reserved].

Rejection Right. The Borrower shall notify the Administrative Agent in writing of any mandatory prepayment of Term Loans required to be made pursuant to Section 5.2(a)(i) or Section 5.2(b), in each case at least three Business Days prior to the date such prepayment is required to be made (or such shorter period of time as agreed to by the Administrative Agent in its reasonable discretion). Each such notice shall be revocable and specify the anticipated date of such prepayment and provide a reasonably detailed estimated calculation of the amount of such prepayment. The Administrative Agent will promptly notify each Lender holding Term Loans to be prepaid in accordance with such prepayment notice of the contents of the Borrower's prepayment notice and of such Lender's *pro rata* share of the prepayment. Each Lender may reject all or a portion of its pro rata share of any such prepayment of Term Loans required to be made pursuant to Section 5.2(a)(i) or Section 5.2(b) (such declined amounts, the "Declined Proceeds") by providing written notice (each, a "Rejection Notice") to the Administrative Agent and the Borrower no later than 5:00 p.m. one Business Day after the date of such Lender's receipt of notice from the Administrative Agent regarding such prepayment. Each Rejection Notice shall specify the principal amount of the mandatory prepayment of Term Loans to be rejected by such Lender. If a Lender fails to deliver a Rejection Notice to the Administrative Agent within the time frame specified above or such Rejection Notice fails to specify the principal

amount of the Term Loans to be rejected, any such failure will be deemed an acceptance of the total amount of such prepayment of Term Loans. Any Declined Proceeds remaining thereafter shall be retained by the Borrower ("Retained Declined Proceeds").

Foreign Net Cash Proceeds. Notwithstanding any other provisions of this Section 5.2, to the extent that the repatriation of any or all of the Net Cash Proceeds from a Recovery Prepayment Event (a "Foreign Recovery Event") of, or any Disposition by, a Foreign Subsidiary giving rise to an Asset Sale Prepayment Event (a "Foreign Asset Sale"), are prohibited or delayed by applicable local law (including financial assistance, corporate benefit, restrictions on upstreaming of cash intra-group and fiduciary and statutory duties of the directors of the relevant subsidiaries) or material agreement (so long as not created in contemplation of such prepayment) or organizational document from being repatriated to the United States or, if the Borrower and the Required Lenders have determined in good faith that repatriation of any of or all the Net Cash Proceeds of any Foreign Recovery Event or any Foreign Asset Sale would have an material adverse tax consequence to the Borrower and its Subsidiaries (or any direct or indirect parent company of the Borrower), such portion of the Net Cash Proceeds so affected will not be required to be applied to repay Term Loans, at the times provided in this Section 5.2 but may be retained by the applicable Foreign Subsidiary for working capital purposes so long, but only so long, (i) as the applicable local law will not permit repatriation to the United States (the Borrower hereby agreeing to promptly take commercially reasonable actions reasonably required by the applicable local law or material agreement to permit such repatriation) or (ii) as the Borrower and Required Lenders determine in good faith such material adverse tax consequence to the Borrower and its Subsidiaries (or any direct or indirect parent company of the Borrower) would be incurred, and once such repatriation is permitted under the applicable local law or such material adverse tax consequence would no longer be incurred as determined by the Borrower in good faith (and in any event not later than ten (10) Business Days after such repatriation is permitted to occur or the Borrower and the Required Lenders determine in good faith that such material adverse tax consequence would no longer be incurred) applied (net of additional taxes payable or reserved against as a result thereof) apply an amount equal thereto to the repayment of the Term Loans as required pursuant to this Section 5.2.

## 5.3 Method and Place of Payment

(a) Except as otherwise specifically provided herein, all payments under this Agreement shall be made by the Borrower without set-off, counterclaim or deduction of any kind, to the Administrative Agent for the ratable account of the Lenders entitled thereto, as the case may be, not later than 2:00 p.m., in each case, on the date when due and shall be made in immediately available funds at the Administrative Agent's Office or at such other office as the Administrative Agent shall specify for such purpose by written notice to the Borrower, it being understood that written or facsimile notice by the Borrower to the Administrative Agent to make a payment from the funds in the Borrower's account at the Administrative Agent's Office shall constitute the making of such payment to the extent of such funds held in such account. All repayments or prepayments of any Term Loans (whether of principal, interest or otherwise) hereunder and all other payments under each Credit Document shall be made in Dollars. The Administrative Agent will thereafter cause to be distributed on the same day (if payment was actually received by the Administrative Agent prior to 2:00 p.m. or, otherwise, on the next Business Day) like funds relating to the payment of principal or interest or fees ratably to the Lenders entitled thereto.

(b) Any payments under this Agreement that are made later than 2:00 p.m. shall be deemed to have been made on the next succeeding Business Day. Whenever any payment to be made hereunder shall be stated to be due on a day that is not a Business Day, the due date thereof shall be extended to the next succeeding Business Day and, with respect to payments of principal, interest shall be payable during such extension at the applicable rate in effect immediately prior to such extension.

# 5.4 Net Payments

- Any and all payments made by or on behalf of the Borrower or any (a) Guarantor under this Agreement or any other Credit Document shall be made free and clear of, and without deduction or withholding for or on account of, any Taxes; provided that if the Borrower or any Guarantor or the Administrative Agent shall be required by Applicable Law (as determined in the good faith discretion of an applicable withholding agent) to deduct or withhold any Taxes from such payments, then (i) the Borrower or such Guarantor or the Administrative Agent shall make such deductions or withholdings and (ii) the Borrower or such Guarantor or the Administrative Agent shall timely pay the full amount deducted or withheld to the relevant Governmental Authority within the time allowed and in accordance with Applicable Law. If such a Tax is an Indemnified Tax, the sum payable by the Borrower or any Guarantor shall be increased as necessary so that after making all such required deductions and withholdings (including such deductions or withholdings applicable to additional sums payable under this Section 5.4), the Administrative Agent, the Collateral Agent or any Lender, as the case may be, receives an amount equal to the sum it would have received had no such deductions or withholdings been made. Whenever any Indemnified Taxes are payable by the Borrower or such Guarantor, as promptly as practicable thereafter, the Borrower or Guarantor shall send to the Administrative Agent for its own account or for the account of such Lender, as the case may be, a certified copy of an original official receipt (or other evidence acceptable to such Lender, acting reasonably) received by the Borrower or such Guarantor showing payment thereof.
- (b) The Borrower shall timely pay to the relevant Governmental Authority Other Taxes in accordance with Applicable Law, or at the option of the Administrative Agent, timely reimburse it for the payment of any Other Taxes that are paid by the Administrative Agent to the relevant Governmental Authority in accordance with Applicable Law.
- (c) The Borrower shall indemnify and hold harmless the Administrative Agent, the Collateral Agent and each Lender within fifteen Business Days after written demand therefor, for the full amount of any Indemnified Taxes imposed on the Administrative Agent, the Collateral Agent or such Lender as the case may be, on or with respect to any payment by or on account of any obligation of the Borrower or any Guarantor hereunder or under any other Credit Document (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section 5.4) payable or paid by such Agent or Lender or required to be withheld or deducted from a payment to such Agent or Lender and any reasonable out-of-pocket expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate setting forth reasonable detail as to the amount of such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), the Administrative Agent or the Collateral Agent (as applicable) on its own behalf or on behalf of a Lender shall be conclusive absent manifest error.

- (d) Each Lender shall severally indemnify the Administrative Agent, within 10 days after demand therefor, for (i) any Indemnified Taxes attributable to such Lender (but only to the extent that the Borrower has not already indemnified the Administrative Agent for such Indemnified Taxes and without limiting the obligation of the Borrower to do so), (ii) any Taxes attributable to such Lender's failure to comply with the provisions of Section 13.6 relating to the maintenance of a Register and (iii) any Excluded Taxes attributable to such Lender, in each case, that are payable or paid by the Administrative Agent in connection with this Agreement or any Credit Document, and any reasonable expenses arising therefrom or with respect thereof, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent shall be conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under this Agreement or any Credit Document or otherwise payable by the Administrative Agent to the Lender from any other source against any amount due to the Administrative Agent under this paragraph (d).
- Any Non-U.S. Lender claiming a basis for an exemption from or reduction of withholding Tax under the law of the jurisdiction in which the Borrower is resident for tax purposes, or under any treaty to which such jurisdiction is a party, with respect to payments hereunder or under any other Credit Document shall, to the extent it is legally able to do so, deliver to the Borrower (with a copy to the Administrative Agent), at the time or times prescribed by Applicable Law or reasonably requested by the Borrower or the Administrative Agent, such properly completed and executed documentation prescribed by Applicable Law as will permit such payments to be made without withholding or at a reduced rate of withholding or as will permit the Borrower or the Administrative Agent to determine the withholding or deduction required to be made. In addition, any Lender, if requested by the Borrower or the Administrative Agent, shall deliver such other documentation prescribed by Applicable Law or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in this Section 5.4(d), the completion, execution and submission of such documentation (other than such documentation set forth in Section 5.4(f), 5.4(i) and 5.4(j) below) shall not be required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.
- (f) Without limiting the generality of Section 5.4(e), each Non-U.S. Lender with respect to any amounts payable hereunder or under any other Credit Document shall, to the extent it is legally entitled to do so:
  - (i) deliver to the Borrower and the Administrative Agent, on or prior to the date on which such Non-U.S. Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), two executed copies of (x) in the case of a Non-U.S. Lender claiming exemption from U.S. federal withholding Tax under Section 871(h) or 881(c) of the Code with respect to payments of "portfolio interest", United States Internal Revenue Service Form W-8BEN or W-8BEN-E (together with a certificate substantially in the form

of Exhibit J-1 representing that such Non-U.S. Lender is not a bank within the meaning of Section 881(c)(3)(A) of the Code, is not a 10-percent shareholder (within the meaning of Section 871(h)(3)(B) of the Code) of the Borrower, any interest payment received by such Non-U.S. Lender under this Agreement or any other Credit Document is not effectively connected with the conduct of a trade or business in the United States and is not a controlled foreign corporation related to the Borrower (within the meaning of Section 864(d)(4) of the Code)), (y) Internal Revenue Service Form W-8BEN, Form W-8-BEN-E or Form W-8ECI, in each case properly completed and duly executed by such Non-U.S. Lender claiming complete exemption from, or reduced rate of, U.S. Federal withholding Tax on payments under any Credit Document or (z) to the extent a Non-U.S. Lender is not the beneficial owner with respect to any portion of any sums paid or payable to such Lender under any of the Credit Documents (for example, in the case of a typical participation or where Non-U.S. Lender is a pass through entity) Internal Revenue Service Form W-8IMY and all necessary attachments (including the forms described in clauses (x) and (y) above and in Section 5.4(i), Exhibit J-2, Exhibit J-3 and or other certification documents from each beneficial owner, as applicable); provided that if the Non-U.S. Lender is a partnership it may provide Exhibit J-4 on behalf of one or more of its direct or indirect partners that are claiming the portfolio interest exemption; and

(ii) deliver to the Borrower and the Administrative Agent two further copies of any such form or certification (or any applicable successor form) on or before the date that any such form or certification expires or becomes obsolete or inaccurate in any respect and after the occurrence of any event requiring a change in the most recent form previously delivered by it to the Borrower or the Administrative Agent.

If in any such case any Change in Law has occurred prior to the date on which any such delivery would otherwise be required that renders any such form inapplicable or would prevent such Non-U.S. Lender from duly completing and delivering any such form with respect to it, such Non-U.S. Lender shall promptly so advise the Borrower and the Administrative Agent.

If any Lender, the Administrative Agent or the Collateral Agent, as applicable, determines, in its sole discretion exercised in good faith, that it had received and retained a refund of an Indemnified Tax or additional sums payable under this Section 5.4 (including an Other Tax) for which a payment has been made by the Borrower pursuant to this Agreement, which refund in the good faith judgment of such Lender, the Administrative Agent or the Collateral Agent, as the case may be, is attributable to such payment made by the Borrower, then the Lender, the Administrative Agent or the Collateral Agent, as the case may be, shall reimburse the Borrower for such amount (net of all out-of-pocket expenses of such Lender, the Administrative Agent or the Collateral Agent, as the case may be, and without interest other than any interest received thereon from the relevant Governmental Authority with respect to such refund) as the Lender, the Administrative Agent or the Collateral Agent, as the case may be, determines in its sole discretion exercised in good faith to be the proportion of the refund as will leave it, after such reimbursement, in no better or worse position (taking into account expenses or any Taxes imposed on the refund) than it would have been in if the payment had not been required; provided that the Borrower, upon the request of the Lender, the Administrative Agent or the Collateral Agent, agrees to repay the amount paid over to the Borrower (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) to the Lender, the Administrative Agent or the Collateral Agent in the event the Lender, the Administrative Agent or the Collateral Agent is required to repay such refund to such Governmental Authority. A Lender shall claim any refund that it determines is available to it, unless it concludes in its sole discretion that it would be adversely affected by making such a claim. None of any Lender, the Administrative Agent or the Collateral Agent shall be obliged to disclose any information regarding its tax affairs that it deems confidential to any Credit Party in connection with this clause (g).

- (h) If the Borrower determines that a reasonable basis exists for contesting a Tax, each Lender or Agent, as the case may be, shall use reasonable efforts to cooperate with the Borrower as the Borrower may reasonably request in challenging such Tax. Subject to the provisions of Section 2.12, each Lender and Agent agrees to use reasonable efforts to cooperate with the Borrower as the Borrower may reasonably request to minimize any amount payable by the Borrower or any Guarantor pursuant to this Section 5.4. The Borrower shall indemnify and hold each Lender and Agent harmless against any out-of-pocket expenses incurred by such Person in connection with any request made by the Borrower pursuant to this Section 5.4(h). Nothing in this Section 5.4(h) shall obligate any Lender or Agent to take any action that such Person, in its sole judgment, determines may result in a material detriment to such Person.
- (i) Without limiting the generality of Section 5.4(d), with respect to any amounts payable hereunder or under any other Credit Document, each Lender or Agent that is a United States person under Section 7701(a)(30) of the Code (each, a "U.S. Lender") shall deliver to the Borrower and the Administrative Agent two United States Internal Revenue Service Forms W-9 (or substitute or successor form), properly completed and duly executed, certifying that such Lender or Agent is exempt from United States backup withholding (i) on or prior to the Closing Date (or on or prior to the date it becomes a party to this Agreement), (ii) on or before the date that such form expires or becomes obsolete or inaccurate in any respect, (iii) after the occurrence of a change in such Agent's or Lender's circumstances requiring a change in the most recent form previously delivered by it to the Borrower and the Administrative Agent and (iv) from time to time thereafter if reasonably requested by the Borrower or the Administrative Agent.
- If a payment made to any Agent or Lender would be subject to U.S. federal withholding Tax imposed under FATCA if such Agent or Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Sections 1471(b) or 1472(b) of the Code, as applicable), such Agent or Lender shall deliver to the Borrower and the Administrative Agent, at the time or times prescribed by law and at such time or times reasonably requested by the Borrower or the Agent, such documentation prescribed by Applicable Law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and such other documentation reasonably requested by the Administrative Agent and the Borrower as may be necessary for the Administrative Agent and the Borrower to comply with their obligations under FATCA, to determine whether such Agent or Lender has or has not complied with such Agent's or Lender's FATCA obligations and to determine the amount, if any, to deduct and withhold from such payment and deliver to the Borrower and the Administrative Agent two further copies of any such documentation on or before the date that any such documentation expires or becomes obsolete or inaccurate in any respect and after the occurrence of any event requiring a change in the documentation previously delivered by it to the Borrower or the Administrative Agent. Solely for purposes of this subsection (j), "FATCA" shall include any amendments made to FATCA after

the date of this Agreement and any current or future intergovernmental agreements and any Applicable Law implementing such agreement entered into in connection therewith.

(k) The agreements in this Section 5.4 shall survive the resignation or replacement of the Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of this Agreement and the payment of the Term Loans and all other amounts payable hereunder and under any other Credit Document.

## 5.5 Computations of Interest and Fees

Except as provided in the next succeeding sentence, interest on SOFR Loans and ABR Loans shall be calculated on the basis of a 360-day year for the actual days elapsed. Interest on ABR Loans in respect of which the rate of interest is calculated on the basis of the rate of interest in effect for such day as publicly announced from time to time by the Wall Street Journal as the "U.S. prime rate" and interest on overdue interest shall be calculated on the basis of a 365-(or 366-, as the case may be) day year for the actual days elapsed.

#### 5.6 Limit on Rate of Interest

- (a) No Payment Shall Exceed Lawful Rate. Notwithstanding any other term of this Agreement, the Borrower shall not be obligated to pay any interest or other amounts under or in connection with this Agreement or otherwise in respect of the Obligations in excess of the amount or rate permitted under or consistent with any Applicable Law.
- (b) Payment at Highest Lawful Rate. If the Borrower is not obliged to make a payment that it would otherwise be required to make, as a result of Section 5.6(a), the Borrower shall make such payment to the maximum extent permitted by or consistent with Applicable Laws.
- (c) Adjustment if Any Payment Exceeds Lawful Rate. If any provision of this Agreement or any of the other Credit Documents would obligate the Borrower to make any payment of interest or other amount payable to any Lender in an amount or calculated at a rate that would be prohibited by any Applicable Law, then notwithstanding such provision, such amount or rate shall be deemed to have been adjusted with retroactive effect to the maximum amount or rate of interest, as the case may be, as would not be so prohibited by Applicable Law, such adjustment to be effected, to the extent necessary, by reducing the amount or rate of interest required to be paid by the Borrower to the affected Lender under Section 2.8.
- (d) Spreading. In determining whether the interest hereunder is in excess of the amount or rate permitted under or consistent with any Applicable Law, the total amount of interest shall be spread throughout the entire term of this Agreement until its Payment in Full.
- (e) Notwithstanding the foregoing, and after giving effect to all adjustments contemplated thereby, if any Lender shall have received from the Borrower an amount in excess of the maximum permitted by any Applicable Law, then the Borrower shall be entitled, by notice in writing to the Administrative Agent to obtain reimbursement from that Lender in an amount equal to such excess, and pending such reimbursement, such amount shall be deemed to be an amount payable by that Lender to the Borrower.

## **SECTION 6** Conditions Precedent to the Closing Date

The Initial Draw on the Closing Date is subject to the satisfaction in all material respects of the conditions set forth below.

### 6.1 Credit Documents

The Administrative Agent and counsel to the Lenders shall have received (a) this Agreement, executed and delivered by an Authorized Officer of Holdings and the Borrower, (b) the Guarantee, executed and delivered by an Authorized Officer of each Guarantor as of the Closing Date, (c) the Security Agreement, executed and delivered by an Authorized Officer of each grantor party thereto as of the Closing Date and (d) a duly executed Notice of Borrowing delivered pursuant to Section 2.3(a). The Administrative Agent shall have received the Fee Letter executed and delivered by an Authorized Officer of the Borrower.

### 6.2 Collateral

- (a) All Non-Debtor Subsidiary Notes/Receivables in existence on the Closing Date shall be evidenced by the Foreign Intercompany Note and shall have been pledged pursuant to the Security Agreement and/or the Orders.
- (b) Proper Uniform Commercial Code financing statements covering the Collateral described therein shall be delivered to the Collateral Agent.
- (c) The Borrower shall deliver to the Collateral Agent a completed Perfection Certificate in form and substance reasonably acceptable to the Required Lenders, executed and delivered by an Authorized Officer of the Borrower, together with all attachments contemplated thereby.
- (d) With respect to all Owned Real Property that constitutes Collateral on which a "Building" or "Mobile Home" (as contemplated by the Flood Laws) is located, a completed "Life-of-Loan" Federal Emergency Management Agency Standard Flood Hazard Determination (together with, if such Real Estate is located in a special flood hazard area, (A) a notice about special flood hazard area status and flood disaster assistance duly executed by the Borrower and each Credit Party relating thereto, and (B) evidence of flood insurance in form and substance reasonably satisfactory to the Administrative Agent).

# 6.3 Legal Opinions

The Administrative Agent and the Lenders shall have received the executed customary legal opinions of Kirkland & Ellis LLP, special New York counsel to the Credit Parties, in form and substance reasonably satisfactory to the Administrative Agent and the Required Lenders. Holdings, the Borrower, the other Credit Parties and the Administrative Agent hereby instruct such counsel to deliver such legal opinions.

# 6.4 Closing Certificates

The Administrative Agent shall have received a certificate of the Credit Parties, dated the Closing Date, in respect of the conditions set forth in Sections 6.7, 6.8 and 6.16.

# 6.5 Secretary's Certificate; Authorization of Proceedings of Each Credit Party

The Administrative Agent and the Lenders shall have received, on or prior to the Closing Date, a certificate of each Credit Party, dated the Closing Date and executed by an Authorized Officer of such Credit Party, with appropriate insertions and attachments, including (a) a copy of the resolutions of the board of directors, other managers or general partner of each Credit Party (or a duly authorized committee thereof) authorizing (i) the execution, delivery and performance of the Credit Documents referred to in Section 6.1 (and any agreements relating thereto) to which it is a party and (ii) in the case of the Borrower, the extensions of credit contemplated hereunder, (b) true and complete copies of the Organizational Documents of each Credit Party as of the Closing Date, (c) good standing certificates (to the extent such concept exists in the relevant jurisdiction of organization) of the Borrower and the Guarantors and (d) the names, titles, incumbency and signature specimens of those representatives of such Credit Party who have been authorized by such resolutions and/or written consents to execute Credit Documents on behalf of such Credit Party.

#### 6.6 Fees

All fees required to be paid on the Closing Date pursuant to the Fee Letter and reasonable and documented out-of-pocket expenses required to be paid on the Closing Date pursuant to the Commitment Letter and this Agreement, in the case of expenses, to the extent invoiced at least one (1) Business Day prior to the Closing Date, shall have been paid, or shall be paid substantially concurrently with, the Initial Draw.

## 6.7 Representations and Warranties

The representations and warranties of the Borrower and each other Credit Party contained in Section 8 or any other Credit Document shall be true and correct in all material respects on and as of the Closing Date; provided that, to the extent that such representations and warranties specifically refer to an earlier date, they shall be true and correct in all material respects as of such earlier date; provided, further that, any representation and warranty that is qualified as to "materiality," "Material Adverse Effect" or similar language shall be true and correct (after giving effect to any qualification therein) in all respects on such respective dates.

### 6.8 Material Adverse Effect

Since September 30, 2022, there shall not have occurred any circumstance or condition, which individually or in the aggregate, constitutes or is reasonably expected to constitute, a Material Adverse Effect.

# 6.9 Approved Budget; DIP Projection

The Administrative Agent and the Lenders shall have received (a) the initial Approved Budget (dated as of a date not more than five (5) days prior to the Closing Date)) and (b) the DIP Projection, in each case, in form and substance acceptable to the Required Lenders.

## 6.10 Prepetition ABL Payoff Letter

The Administrative Agent and the Lenders shall have received an executed copy of the Prepetition ABL Payoff Letter.

# 6.11 Prepetition ABL Payoff

Substantially concurrently with the funding of the Initial Draw, the Prepetition ABL Payoff shall have occurred.

## 6.12 Governmental Approvals

All necessary governmental and third party consents and approvals necessary in connection with the Term Facility and the transactions contemplated hereby shall have been obtained on or prior to Closing Date (without the imposition of any adverse conditions that are not reasonably acceptable to the Required Lenders), after giving effect to the Interim Order and any other order of the Bankruptcy Court, and shall remain in effect; and, after giving effect to the Interim Order and any other order of the Bankruptcy Court entered on or prior to the Closing Date, no law or regulation shall be applicable that restrains, prevents or imposes materially adverse conditions upon the Term Facility or the transactions contemplated hereby.

### 6.13 Patriot Act

The Administrative Agent and each Lender who has requested the same shall have received (at least one (1) Business Day prior to the Closing Date) all documentation and other information about the Borrower and each Guarantor as has been reasonably requested in writing prior to the Closing Date by the Administrative Agent or the Lenders that is required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including without limitation the Patriot Act, to the extent such request has been delivered to the Borrower at least three (3) Business Days prior to the Closing Date.

### 6.14 The Chapter 11 Cases

- (a) The Petition Date shall have occurred.
- (b) The Interim Order Entry Date shall have occurred not later than three (3) Business Days following the Petition Date, and the Interim Order shall not have been vacated, reversed, modified, amended or stayed.
- (c) All First Day Orders shall have been entered by the Bankruptcy Court and all such entered First Day Orders shall be reasonably satisfactory in form and substance to the

Required Lenders, it being understood that drafts approved by both co-counsel to the Required Lenders, on or prior to the Petition Date are reasonably satisfactory.

- (d) No trustee under chapter 7 or chapter 11 of the Bankruptcy Code or examiner with enlarged powers beyond those set forth in section 1106(a)(3) and (4) of the Bankruptcy Code shall have been appointed in any of the Chapter 11 Cases.
- (e) The amount of Loans made on the Closing Date shall not exceed the amount authorized by the Interim Order.

# 6.15 Escrow Payment

The Escrow Payment shall have occurred or shall be occurring substantially contemporaneously with the Initial Draw, and none of the amounts so paid shall have been recovered from or required to be disgorged by any recipient thereof.

### 6.16 No Default

No Default or Event of Default shall exist at the time of, or would result from, the Initial Draw or the application of the proceeds therefrom.

## 6.17 No Litigation

There shall exist no unstayed action, suit, investigation, litigation, or proceeding pending in any court or before any arbitrator or governmental instrumentality (other than the Chapter 11 Cases) that would reasonably be expected to result in a Material Adverse Effect.

### 6.18 Restructuring Support Agreement

Holders holding not less than 66 2/3% of outstanding principal amount of the Prepetition First Lien Debt shall have entered into the Restructuring Support Agreement, in form and substance satisfactory to the Required Lenders. The Restructuring Support Agreement shall be in full force and effect as of the Closing Date and shall not have been amended or modified without the Required Lenders' prior written consent.

For purposes of determining compliance with the conditions specified in Section 6 on the Closing Date or Section 7 on the date of the Subsequent Draw, each Lender that has signed or authorized the signing of this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required under this Section 6 to be consented to or approved by or acceptable or satisfactory to a Lender unless the Administrative Agent shall have received notice from such Lender prior to such date specifying its objection thereto.

## **SECTION 7** Conditions Precedent to the Subsequent Draw.

The Subsequent Draw is subject to the satisfaction in all material respects of the conditions set forth below.

# 7.1 Closing Date

The Closing Date shall have occurred.

### 7.2 Interim Order

The Interim Order shall be in full force and effect and shall not have been vacated or reversed, shall not be subject to a stay, and shall not have been modified or amended without the prior written consent of the Required Lenders; provided that at the time of the Subsequent Draw, the Administrative Agent shall have received a final copy of an order of the Bankruptcy Court in substantially the form of the "form of" Interim Order filed with the DIP Motion on the Petition Date (with only such modifications thereto as are satisfactory in form and substance to the Borrower and the Required Lenders) and authorizing the Term Facility on a final basis (the "Final Order"), which, in any event, shall have been entered by the Bankruptcy Court no later than the date that is forty-five (45) days following the Petition Date, and at the time of the Subsequent Draw and the Borrower's delivery of the Notice of Borrowing with respect to the Subsequent Draw, in each case, the Final Order shall be in full force and effect, and shall not have been vacated or reversed, shall not be subject to a stay, and shall not have been modified or amended without the prior written consent of the Required Lenders; and if either the Interim Order or the Final Order is the subject of a pending appeal in any respect, neither the making of the Loans nor the performance by any Credit Party of any of their respective obligations under any of the Credit Documents shall be the subject of a presently effective stay pending appeal.

# 7.3 First Day Orders

All material First Day Orders shall have been entered by the Bankruptcy Court on a final basis and shall be reasonably satisfactory in form and substance to the Borrower and the Required Lenders.

# 7.4 Representations and Warranties

The representations and warranties of the Borrower and each other Credit Party contained in Section 8 or any other Credit Document shall be true and correct in all material respects on and as of the date of the Subsequent Draw; provided that, to the extent that such representations and warranties specifically refer to an earlier date, they shall be true and correct in all material respects as of such earlier date; provided, further that, any representation and warranty that is qualified as to "materiality," "Material Adverse Effect" or similar language shall be true and correct (after giving effect to any qualification therein) in all respects on such respective dates.

### 7.5 No Default or Event of Default

No Default or Event of Default shall exist at the time of, or would result from, the Subsequent Draw or the application of the proceeds therefrom.

## 7.6 Request for Borrowing

The Administrative Agent shall have received a duly executed Notice of Borrowing from the Borrower in accordance with the requirements thereof (together with, if applicable, the

certification and acknowledgment with respect to any adjustment to the Term Commitments as required in the definition thereof), no less than ten (10) Business Days prior to the date of the Subsequent Draw.

### 7.7 No Violation of Law

The making of such Loan shall not violate any requirement of Applicable Law applicable to the Credit Parties, after giving effect to the Orders and any other order of the Bankruptcy Court entered on or prior to the date of the applicable Borrowing, and shall not be enjoined, temporarily, preliminarily or permanently.

## 7.8 Payments of Fees and Expenses

The Administrative Agent and the Lenders shall have received the fees in accordance with the procedures set forth in the Interim Order and in the amounts previously agreed in writing by the Administrative Agent or otherwise pursuant to the Credit Documents, if any, on or prior to such date and all expenses (including the reasonable fees, disbursements and other charges of counsel) payable by the Credit Parties pursuant to the Commitment Letter, this Agreement, and the Interim Order for which, with respect to expenses, invoices have been presented at least one Business Day prior to such date shall have been paid; which amounts may be paid from the proceeds of the Term Loans funded on the date of the Subsequent Draw unless required to be paid by the Borrower prior to such date.

### 7.9 Material Adverse Effect

No Material Adverse Effect shall have occurred since the Petition Date.

## 7.10 The Chapter 11 Cases

- (a) None of the Chapter 11 Cases shall have been dismissed or converted to a chapter 7 case.
- (b) No trustee or examiner with enlarged powers beyond those set forth in section 1106(a)(3) and (4) of the Bankruptcy Code shall have been appointed in any of the Chapter 11 Cases.
- (c) The Escrow Payment shall have occurred or shall be occurring substantially contemporaneously with the Subsequent Draw, and none of the amounts so paid shall have been recovered from or required to be disgorged by any recipient thereof.

## 7.11 Milestones

All Milestones required to be satisfied by the Debtors as of the date of the proposed Borrowing shall have been satisfied or waived or extended in accordance with the Restructuring Support Agreement.

# 7.12 Approved Budget

The funding of such Term Loans, as applicable, shall be in accordance with the Approved Budget.

Each Notice of Borrowing (but not any Notice of Conversion or Continuation) submitted by a Borrower shall be deemed to be a representation and warranty that the conditions specified in this Section 7 have been satisfied on and as of the date of the applicable Borrowing.

## 7.13 No Litigation

There shall exist no unstayed action, suit, investigation, litigation, or proceeding pending in any court or before any arbitrator or governmental instrumentality (other than the Chapter 11 Cases) that would reasonably be expected to result in a Material Adverse Effect.

# **SECTION 8** Representations and Warranties.

In order to induce the Lenders to enter into this Agreement and to make the Term Loans as provided for herein, each of Holdings and the Borrower makes the following representations and warranties to the Agents and the Lenders, on behalf of itself and its Subsidiaries, all of which shall survive the execution and delivery of this Agreement and the making of the Term Loans:

## 8.1 Corporate Status; Compliance with Laws

Each of Holdings, the Borrower and each Subsidiary of the Borrower (a) is a duly organized and validly existing corporation or other entity in good standing (as applicable) under the laws of the jurisdiction of its organization and has the corporate or other organizational power and authority to own its property and assets and to transact the business in which it is engaged, except as would not reasonably be expected to result in a Material Adverse Effect, (b) has duly qualified and is authorized to do business and is in good standing (if applicable) in all jurisdictions where it is required to be so qualified, except where the failure to be so qualified would not reasonably be expected to result in a Material Adverse Effect and (c) is in compliance with all Applicable Laws, except to the extent that the failure to be in compliance would not reasonably be expected to result in a Material Adverse Effect.

### 8.2 Corporate Power and Authority

Each Credit Party has the corporate or other organizational power and authority to, subject to the entry of the Orders and subject to the terms thereof, execute, deliver and carry out the terms and provisions of the Credit Documents to which it is a party and has taken all necessary corporate or other organizational action to authorize the execution, delivery and performance of the Credit Documents to which it is a party. Subject to entry of the Orders and subject to the terms thereof, each Credit Party has duly executed and delivered each Credit Document to which it is a party and each such Credit Document constitutes the legal, valid and binding obligation of such Credit Party enforceable in accordance with its terms, subject to the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization and other similar laws relating to or affecting creditors' rights generally and general principles of equity (whether considered in a proceeding in

equity or law) (*provided* that, with respect to the creation and perfection of security interests with respect to Indebtedness, Stock and Stock Equivalents of Foreign Subsidiaries, only to the extent the creation and perfection of such obligation is governed by the UCC).

### 8.3 No Violation

Neither the execution, delivery or performance by any Credit Party of the Credit Documents to which it is a party nor the compliance with the terms and provisions thereof nor the consummation of the financing transactions contemplated hereby and thereby will (a) contravene any applicable provision of any Applicable Law (including material Environmental Laws) other than any contravention which would not reasonably be expected to result in a Material Adverse Effect, (b) except to the extent arising under the documents governing the Prepetition Debt, result in any breach of any of the terms, covenants, conditions or provisions of, or constitute a default under, or result in the creation or imposition of any Lien upon any of the property or assets of Holdings, the Borrower or any Subsidiary (other than Liens permitted hereunder) pursuant to the terms of any material indenture, loan agreement, lease agreement, mortgage, deed of trust or other material debt agreement or instrument to which Holdings, the Borrower or any Subsidiary is a party or by which it or any of its property or assets is bound (any such term, covenant, condition or provision, a "Contractual Requirement") other than any such breach, default or Lien that would not reasonably be expected to result in a Material Adverse Effect, or (c) violate any provision of the Organizational Documents of any Credit Party.

## 8.4 Litigation

Except for the Chapter 11 Cases and matters set forth on Schedule 8.4, there are no actions, suits or proceedings pending or, to the knowledge of the Borrower, threatened in writing with respect to Holdings, the Borrower or any of the Subsidiaries that have a reasonable likelihood of adverse determination and such determination would reasonably be expected to result in a Material Adverse Effect.

# 8.5 Margin Regulations

Neither the making of any Term Loan hereunder nor the use of the proceeds thereof will violate the provisions of Regulation T, U or X of the Board.

## 8.6 Governmental Approvals

Subject to the entry of the Orders and subject to the terms thereof, the execution, delivery and performance of the Credit Documents does not require any consent or approval of, registration or filing with, or other action by, any Governmental Authority, except for (i) such as have been obtained or made and are in full force and effect, (ii) filings and recordings in respect of the Liens created pursuant to the Security Documents and (iii) such licenses, authorizations, consents, approvals, registrations, filings or other actions the failure of which to obtain or make would not reasonably be expected to have a Material Adverse Effect.

# 8.7 Investment Company Act

None of the Credit Parties is an "investment company" within the meaning of, and subject to registration under, the Investment Company Act of 1940, as amended.

# 8.8 True and Complete Disclosure

- heretofore or contemporaneously furnished by or on behalf of Holdings, the Borrower, any of the Subsidiaries of the Borrower or any of their respective authorized representatives to the Administrative Agent and/or any Lender on or before the Closing Date (including all such information and data contained in the Credit Documents) regarding Holdings, the Borrower and its Subsidiaries in connection with the Transactions for purposes of or in connection with this Agreement or any transaction contemplated herein contained any untrue statement of any material fact or omitted to state any material fact necessary to make such information and data (taken as a whole) not materially misleading at such time in light of the circumstances under which such information or data was furnished, it being understood and agreed that for purposes of this Section 8.8(a), such factual information and data shall not include projections or estimates (including financial estimates, forecasts and other forward-looking information) and information of a general economic or general industry nature.
- (b) The projections provided to the Lenders who have elected to receive such projections are based upon good faith estimates and assumptions believed by the Borrower to be reasonable at the time made, it being recognized by the Agents and the Lenders that such projections, forward-looking statements, estimates and pro forma financial information are not to be viewed as facts or a guarantee of performance, and are subject to material contingencies and assumptions, many of which are beyond the control of the Credit Parties, and that actual results during the period or periods covered by any such projections, forward-looking statements, estimates and pro forma financial information may differ materially from the projected results.
  - 8.9 [Reserved]

#### 8.10 Tax Matters

Except where the failure of which would not be reasonably expected to have a Material Adverse Effect or is permitted or required by the Bankruptcy Code, (a) each of Holdings, the Borrower and each of the Subsidiaries has filed all federal income Tax returns and all other Tax returns, domestic and foreign, required to be filed by it (after giving effect to all applicable extensions) and has paid all material Taxes payable by it that have become due (whether or not shown on such Tax return), other than those (i) not yet delinquent or (ii) contested in good faith as to which adequate reserves have been provided to the extent required by law and in accordance with GAAP, (b) each of Holdings, the Borrower and each of the Subsidiaries has provided adequate reserves in accordance with GAAP for the payment of, all federal, state, provincial and foreign Taxes not yet due and payable, and (c) each of Holdings, the Borrower and each of the Subsidiaries has satisfied all of its Tax withholding obligations.

## 8.11 Compliance with ERISA

- (a) No ERISA Event has occurred or is reasonably expected to occur; and no Lien imposed under the Code or ERISA on the assets of the Borrower or any ERISA Affiliate exists (or is reasonably likely to exist) nor has the Borrower or any ERISA Affiliate been notified in writing that such a Lien will be imposed on the assets of Holdings, the Borrower or any ERISA Affiliate on account of any Pension Plan, except to the extent that a breach of any of the representations, warranties or agreements in this Section 8.11(a) would not result, individually or in the aggregate, in an amount of liability that would be reasonably likely to have a Material Adverse Effect. No Pension Plan has an Unfunded Current Liability that would, individually or when taken together with any other liabilities referenced in this Section 8.11(a), be reasonably likely to have a Material Adverse Effect.
- (b) All Foreign Plans are in compliance with, and have been established, administered and operated in accordance with, the terms of such Foreign Plans and Applicable Law, except for any failure to so comply, establish, administer or operate the Foreign Plans as would not reasonably be expected to have a Material Adverse Effect. All contributions or other payments which are due with respect to each Foreign Plan have been made in full and there are no funding deficiencies thereunder, except to the extent any such events would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

## 8.12 Subsidiaries

Schedule 8.12 lists each Subsidiary of Holdings (and the direct and indirect ownership interest of Holdings therein), in each case existing on the Closing Date.

# 8.13 Intellectual Property

Each of Holdings, the Borrower and the Subsidiaries has good and marketable title to, or a valid license or right to use, all patents, trademarks, servicemarks, trade names, copyrights and all applications therefor and licenses thereof, and all other intellectual property rights, free and clear of all Liens (other than Liens permitted hereunder), that are necessary for the operation of their respective businesses as currently conducted, except where the failure to have any such title, license or rights would not reasonably be expected to have a Material Adverse Effect.

#### 8.14 Environmental Laws

Except as would not reasonably be expected to have a Material Adverse Effect: (a) Holdings, the Borrower and the Subsidiaries and all Real Estate are in compliance with all Environmental Laws; (b) Holdings, the Borrower and the Subsidiaries have, and have timely applied for renewal of, all permits under Environmental Law to construct and operate their facilities as currently constructed; (c) except as set forth on Schedule 8.14, neither Holdings, the Borrower nor any Subsidiary is subject to any pending or, to the knowledge of the Borrower, threatened Environmental Claim or any other liability under any Environmental Law, including any such Environmental Claim, or, to the knowledge of the Borrower, any other liability under Environmental Law related to, or resulting from the business or operations of any predecessor in interest of any of them; (d) none of Holdings, the Borrower or any Subsidiary is conducting or financing or, to the knowledge of the Borrower, is required to conduct or finance, any

investigation, removal, remedial or other corrective action pursuant to any Environmental Law at any location; (e) to the knowledge of the Borrower, no Hazardous Materials have been released into the environment at, on or under any Real Estate currently owned or leased by Holdings, the Borrower or any Subsidiary, and (f) neither Holdings, the Borrower nor any Subsidiary has treated, stored, transported, released, disposed or arranged for disposal or transport for disposal of Hazardous Materials at, on, under or from any currently or, to the knowledge of the Borrower, formerly owned or leased Real Estate or facility. Except as provided in this Section 8.14, Holdings, the Borrower and the Subsidiaries make no other representations or warranties regarding Environmental Laws.

# 8.15 Properties

Except as set forth on Schedule 8.15, Holdings, the Borrower and the Subsidiaries have good title to or valid leasehold or easement interests or other license or use rights in all properties that are necessary for the operation of their respective businesses as currently conducted, free and clear of all Liens (other than any Liens permitted under this Agreement) and except where the failure to have such good title, leasehold or easement interests or other license or use rights would not reasonably be expected to have a Material Adverse Effect. Schedule 8.15 lists the street address for each Owned Real Property constituting Material Real Property (if any) by any Credit Party on the Closing Date.

#### 8.16 No Default

No Default or Event of Default has occurred, after giving effect to the funding of the Term Loans.

## 8.17 Security Interests

Subject to the qualifications set forth in Section 6.2 and the terms of the DIP ABL Intercreditor Agreement (if then in effect), with respect to each Credit Party, the Security Documents, taken as a whole, are effective to create in favor of the Collateral Agent, for the benefit of the Secured Parties, a legal, valid and enforceable first priority security interest (subject to Liens permitted hereunder) in the Collateral described therein, in each case, to the extent required under the Security Documents, the enforceability of which is subject to applicable bankruptcy, insolvency, reorganization, moratorium or other laws affecting creditors' rights generally and subject to general principles of equity, regardless of whether considered in a proceeding in equity or at law. In the case of (i) the Stock described in the Security Agreement that is in the form of securities represented by stock certificates or otherwise constituting certificated securities within the meaning of Section 8-102(a)(15) of the New York UCC ("Certificated Securities"), when certificates representing such Stock are delivered to the Collateral Agent along with instruments of transfer in blank or endorsed to the Collateral Agent, and (ii) all other Collateral constituting personal property described in the Security Agreement, when financing statements, intellectual property security agreements and other required filings, recordings, agreements and actions in appropriate form are executed and delivered, performed, recorded or filed in the appropriate offices, as the case may be, the Collateral Agent, for the benefit of the applicable Secured Parties, shall have a fully perfected Lien on, and security interest in, all right, title and interest of the Credit Parties in all Collateral that may be perfected by filing, recording or registering a financing

statement, an intellectual property security agreement or analogous document (to the extent such Liens may be perfected by possession of the Certificated Securities by the Collateral Agent or such filings, agreements or other actions or perfection is otherwise required by the terms of any Credit Document), in each case, to the extent required under the Security Documents, as security for the Obligations, in each case prior and superior in right to any other Lien (except, in the case of Liens permitted hereunder).

#### 8.18 Labor Matters

Except as, in the aggregate, would not reasonably be expected to have a Material Adverse Effect: (a) there are no strikes or other labor disputes against Holdings, the Borrower or any Subsidiary pending or, to the knowledge of the Borrower, threatened in writing; and (b) hours worked by and payment made for such work to employees of Holdings, the Borrower and each Subsidiary have not been in violation of the Fair Labor Standards Act or any other Applicable Law dealing with such matters.

### 8.19 Sanctioned Persons; Anti-Corruption Laws; Patriot Act

None of Holdings, the Borrower or any of its Subsidiaries or any of their respective directors or officers is subject to any economic embargoes or similar sanctions administered or enforced by the U.S. Department of State or the U.S. Department of the Treasury (including the Office of Foreign Assets Control) or any other applicable sanctions authority (collectively, "Sanctions", and the associated laws, rules, regulations and orders, collectively, "Sanctions Laws"). Each of Holdings, the Borrower and its Subsidiaries and their respective officers and directors is in compliance, in all material respects, with (i) all Sanctions Laws, (ii) the United States Foreign Corrupt Practices Act of 1977, as amended, and any other applicable anti-bribery or anticorruption laws, rules, regulations and orders (collectively, "Anti-Corruption Laws") and (iii) the Patriot Act and any other applicable anti-terrorism and anti-money laundering laws, rules, regulations and orders. No part of the proceeds of the Term Loans will be (i) used, directly or indirectly, in violation of the Patriot Act, the Anti-Corruption Laws, Sanctions Laws and/or any other anti-terrorism or anti-money laundering laws in any material respect or (ii) provided, given, used for the benefit of, or otherwise made available to (x) any individual or entity located or resident in or organized under the laws of any jurisdiction that is the target of comprehensive sanctions under Sanctions Laws (currently Cuba, Iran, North Korea, Syria, the Crimea region, or the so-called Donetsk or Luhansk People's Republic) or (y) any individual or entity that is the target of Sanctions Laws (including, without limitation, persons listed on the Specially Designated Nationals and Blocked Persons List maintained by the U.S. Department of the Treasury).

#### 8.20 Use of Proceeds

The Borrower will use the proceeds of the Term Loans in accordance with Section 9.13 of this Agreement.

### 8.21 Orders

The Interim Order is (and the Final Order when entered will be) effective to create in favor of the Administrative Agent, for the benefit of the Secured Parties, legal, valid, binding and enforceable perfected security interests in the Collateral without the necessity of the execution

of mortgages, security agreements, pledge agreements, financing statements or other agreements or documents.

# 8.22 Status of Obligations; Perfection and Priority of Security Interests

Subject to entry of the Interim Order (and the Final Order, as applicable) the Obligations shall have the status and priority set forth in Section 2.17 and, for the avoidance of doubt, are subject to the Carve Out in all respects.

# **SECTION 9** Affirmative Covenants

The Borrower hereby covenants and agrees that on the Closing Date and thereafter, until all Commitments have terminated and the Term Loans, together with interest, fees and all other Obligations, are Paid in Full:

#### 9.1 Information Covenants

The Borrower will furnish to the Administrative Agent (which shall promptly make such information available to the Lenders in accordance with its customary practice):

- (a) Monthly Operating Reports. Copies of all "Monthly Operating Reports" filed with the Bankruptcy Court for the applicable month within two (2) Business Days of such filing.
- (b) Approved Budget. On the last Business Day of each four-week period after the Closing Date, an updated Approved Budget in substance satisfactory to the Required Lenders; provided that the failure of such updated Approved Budget to be satisfactory to the Required Lender shall not be deemed to be a violation of this Section 9.1(b) and no Default or Event of Default shall result from the rejection by the Required Lenders of such updated Approved Budget; provided further that for the avoidance of doubt, the most recently effective Approved Budget shall continue to constitute the Approved Budget for purpose of this Agreement; provided further that such rejection by the Required Lenders shall result in an Event of Default two Business Days prior to the commencement of a weekly period that is not included in the most recently effective Approved Budget.
- (c) Variance Reports. On Wednesday following each two-week period (with the first such delivery on March 1, 2023 with respect to the first Variance Test Period, with the second such delivery of March 15, 2023 with respect to the second Variance Test Period and thereafter the Wednesday following each applicable Variance Test Period), a report in substantially the form of Exhibit M (the "Variance Report").
- (d) Within twenty days after the end of each fiscal month, a written report, in form and substance reasonably acceptable to the Required Lenders, containing the gross accounts receivable and inventory of the Credit Parties (as of the last day of the applicable month).

## (e) [Reserved].

- Officer of the Borrower or any Subsidiary obtains knowledge thereof, notice of (i) the occurrence of any event that constitutes a Default or Event of Default, which notice shall specify the nature thereof, the period of existence thereof and what action the Borrower proposes to take with respect thereto, (ii) any litigation, regulatory or governmental proceeding pending against the Borrower or any Subsidiary that has a reasonable likelihood of adverse determination and such determination would reasonably be expected to be determined adversely and, if so determined, to result in a Material Adverse Effect, (iii) the occurrence of any ERISA Event or any ERISA Event that is reasonably expected to occur, that would reasonably be expected to result in a Material Adverse Effect and (iv) any other event which would reasonably be expected to have a Material Adverse Effect.
- (g) Other Information. Promptly upon filing thereof, copies of any filings (including on Form 10-K, 10-Q or 8-K) or registration statements (other than drafts of pre-effective versions of registration statements) with, and reports to, the SEC or any analogous Governmental Authority in any relevant jurisdiction by Holdings, the Borrower or any Subsidiary (other than amendments to any registration statement (to the extent such registration statement, in the form it becomes effective, is delivered to the Administrative Agent), exhibits to any registration statement and, if applicable, any registration statements on Form S-8) and copies of all financial statements, proxy statements, notices and reports that Holdings, the Borrower or any Subsidiary shall send to the holders of any publicly issued debt with a principal amount in excess of \$30,000,000 of Holdings, the Borrower and/or any Subsidiary in their capacity as such holders (in each case to the extent not theretofore delivered to the Administrative Agent pursuant to this Agreement).
- (h) Requested Information. With reasonable promptness, following the reasonable request of the Administrative Agent, such other information (financial or otherwise) as the Administrative Agent on its own behalf or on behalf of any Lender (acting through the Administrative Agent) may reasonably request in writing from time to time; *provided* that, notwithstanding anything to the contrary in this Section 9.1(h), none of Holdings, the Borrower or any of its Subsidiaries will be required to provide any such other information pursuant to this Section 9.1(h) to the extent that (i) the provision thereof would violate any attorney client privilege (as reasonably determined by counsel (internal or external) to the Credit Parties), law, rule or regulation, or any contractual obligation of confidentiality binding on the Credit Parties or their respective Affiliates (so long as not entered into in contemplation hereof) or (ii) such information constitutes attorney work product (as reasonably determined by counsel (internal or external) to the Credit Parties).
- (i) Consolidated EBITDA. Within 45 days after the end of each fiscal quarter, commencing with the fiscal quarter ending March 31, 2023, a certificate of an Authorized Officer of the Borrower setting forth a calculation of the Consolidated EBITDA for the four fiscal quarters most recently ended.
- (j) (i) As soon as reasonably practicable in advance of filing with the Bankruptcy Court or delivering to the Creditors' Committee or the U.S. Trustee, as the case may be, the form of the Final Order and all other proposed orders and pleadings related to the Loans and the Credit Documents, any Reorganization Plan and/or any disclosure statement related thereto and (ii) by the earlier of (A) three (3) Business Days prior to being filed (and if impracticable, then

as soon as possible and in no event later than promptly after being filed) on behalf of any of the Debtors with the Bankruptcy Court or (B) at the same time as such documents are provided by any of the Debtors to any statutory committee appointed in the Chapter 11 Cases or the U.S. Trustee, all other notices, filings, motions, pleadings or other information concerning the financial condition of the Borrower or any of its Subsidiaries or other Indebtedness of the Credit Parties or any request for relief under section 363, 365, 1113 or 1114 of the Bankruptcy Code or Rule 9019 of the Federal Rules of Bankruptcy Procedure.

# 9.2 Books, Records and Inspections

- (a) The Borrower will, and will cause each Subsidiary to, permit officers and designated representatives of the Administrative Agent or the Required Lenders to visit and inspect any of the properties or assets of the Borrower or such Subsidiary in whomsoever's possession to the extent that it is within such party's control to permit such inspection (and shall use commercially reasonable efforts to cause such inspection to be permitted to the extent that it is not within such party's control to permit such inspection), and to examine the books and records of the Borrower and any such Subsidiary and discuss the affairs, finances and accounts of the Borrower and of any such Subsidiary with, and be advised as to the same by, its and their officers and independent accountants, all at such reasonable times and intervals and to such reasonable extent as the Required Lenders may desire (and subject, in the case of any such meetings or advice from such independent accountants, to such accountants' customary policies and procedures); provided that, excluding any such visits and inspections during the continuation of an Event of Default (i) the Required Lenders, may exercise rights of the Lenders under this Section 9.2 and (ii) the Required Lenders shall not exercise such rights more than two times in any calendar year; provided, further, that when an Event of Default exists, the Required Lenders may do any of the foregoing at the expense of the Borrower at any time during normal business hours and upon reasonable advance notice. The Required Lenders shall give the Borrower the opportunity to participate in any discussions with the Borrower's independent public accountants. Notwithstanding anything to the contrary in this Section 9.2, neither the Borrower nor any Subsidiary will be required under this Section 9.2 to disclose or permit the inspection or discussion of any document, information or other matter to the extent that such action would violate any attorney-client privilege (as reasonably determined by counsel (internal or external) to the Credit Parties), law, rule or regulation, or any contractual obligation of confidentiality (not created in contemplation thereof) binding on the Credit Parties or their respective Affiliates or constituting attorney work product (as reasonably determined by counsel (internal or external) to the Credit Parties).
- (b) The Borrower will, and will cause each Subsidiary to, maintain proper books of record and account, in which entries that are full, true and correct in all material respects and are in conformity, in all material respects, with GAAP shall be made of all material financial transactions and matters involving the assets of the business of the Borrower or such Subsidiary, as the case may be (it being understood and agreed that any Subsidiary may maintain its individual books and records in conformity with local standards or customs and that such maintenance shall not constitute a breach of the representations, warranties or covenants hereunder).

### 9.3 Maintenance of Insurance

The Borrower will, and will cause each Subsidiary to, at all times maintain in full force and effect, pursuant to self-insurance arrangements or with insurance companies that the Borrower believes (in the good faith judgment of the management of the Borrower, as applicable) are financially sound and responsible at the time the relevant coverage is placed or renewed, insurance in at least such amounts (after giving effect to any self-insurance which the Borrower believes (in the good faith judgment of management of the Borrower, as applicable) is reasonable and prudent in light of the size and nature of its business) and against at least such risks (and with such risk retentions) as the Borrower believes (in the good faith judgment of management of the Borrower, as applicable) is reasonable and prudent in light of the size and nature of its business and the availability of insurance on a cost-effective basis; and will furnish to the Administrative Agent, upon written reasonable request from the Administrative Agent, information presented in reasonable detail as to the insurance so carried, provided, however, that for so long as no Event of Default has occurred and is continuing, the Administrative Agent shall be entitled to make such request only once in any calendar year. With respect to each Owned Real Property that constitutes Material Real Property, obtain flood insurance in such total amount as the Administrative Agent may from time to time require, if at any time the area in which any improvements located on any Owned Real Property that constitutes Material Real Property is designated a "flood hazard area" in any Flood Insurance Rate Map published by the Federal Emergency Management Agency (or any successor agency), and otherwise comply with the National Flood Insurance Program as set forth in the Flood Disaster Protection Act of 1973, as amended from time to time.

## 9.4 Payment of Taxes

The Borrower will pay and discharge, and will cause each of the Subsidiaries to pay and discharge, all Taxes, assessments and governmental charges or levies imposed upon it or upon its income or profits, or upon any properties belonging to it, prior to the date on which penalties attach thereto, and all lawful claims in respect of any Taxes imposed, assessed or levied that, if unpaid, could reasonably be expected to become a material Lien upon any properties of the Borrower or any Subsidiary; *provided* that neither the Borrower nor any such Subsidiary shall be required to pay any such tax, assessment, charge, levy or claim (i) that is being contested in good faith and by proper proceedings if it has maintained adequate reserves (in the good faith judgment of management of the Borrower) with respect thereto in accordance with GAAP or (ii) with respect to which the failure to pay would not reasonably be expected to result in a Material Adverse Effect or would be permitted or required under the Bankruptcy Code.

## 9.5 Consolidated Corporate Franchises

The Borrower will do, and will cause each Subsidiary that is a Subsidiary to do, or cause to be done, all things necessary to preserve and keep in full force and effect its existence, corporate rights and authority, except to the extent that the failure to do so would not reasonably be expected to have a Material Adverse Effect; *provided*, *however*, that the Borrower and the Subsidiaries may consummate any transaction otherwise permitted hereby, including under Section 10.2, 10.3, 10.4 or 10.5.

# 9.6 Compliance with Statutes, Regulations, Etc

The Borrower will, and will cause each Subsidiary to, comply with all Applicable Laws applicable to it or its property, including all governmental approvals or authorizations required to conduct its business, and to maintain all such governmental approvals or authorizations in full force and effect, in each case except where the failure to do so would not reasonably be expected to have a Material Adverse Effect.

### 9.7 Lender Calls

Solely with respect to any commercial matters, the Credit Parties (including one or more other senior members of management) shall, and shall cause their financial advisors, if appropriate and applicable, at reasonable times to be mutually and reasonably agreed from time to time (and no more frequently than monthly) by Borrower and Administrative Agent, host telephonic conference calls with the Administrative Agent, the Lenders and their financial advisors.

## 9.8 Maintenance of Properties

The Borrower will, and will cause the Subsidiaries to, keep and maintain all property material to the conduct of its business in good working order and condition (ordinary wear and tear, casualty and condemnation excepted), except to the extent that the failure to do so would not reasonably be expected to have a Material Adverse Effect.

# 9.9 Cash Management Order

Each Credit Party shall maintain its cash management system in accordance with the Cash Management Order (subject to any approval or consent rights of the Lenders provided therein) in all material respects.

### 9.10 End of Fiscal Years

The Borrower will, for financial reporting purposes, cause its Fiscal Year to end on September 30 of each year (each a "Fiscal Year") and cause its Subsidiaries to maintain their fiscal years as in effect on the Closing Date; provided, however, that the Borrower may, upon written notice to the Administrative Agent change the Fiscal Year or the fiscal years of its Subsidiaries with the prior written consent of the Administrative Agent and the Required Lenders (not to be unreasonably withheld, conditioned, delayed or denied), in which case the Borrower and the Administrative Agent will, and are hereby authorized by the Lenders to, make any adjustments to this Agreement that are necessary in order to reflect such change in financial reporting.

## 9.11 Additional Guarantors and Grantors

Subject to any applicable limitations set forth in the Guarantee, the Security Documents, the DIP ABL Intercreditor Agreement (if in existence) and this Agreement (including Section 9.12), the Borrower will cause each direct or indirect Domestic Subsidiary of the Borrower (excluding any Excluded Subsidiary) formed or otherwise purchased or acquired after the Closing Date and each other Domestic Subsidiary of the Borrower that ceases to constitute an Excluded

Subsidiary to, upon the date of such formation, acquisition or cessation, as applicable (or such longer period as the Required Lenders may agree in its reasonable discretion), execute a supplement to each of the Guarantee and the Security Agreement in order to become a Guarantor under such Guarantee and a grantor/pledgor under the Security Agreement.

#### 9.12 Further Assurances

- (a) Subject to the applicable limitations set forth in this Agreement (including Section 9.11), the Security Documents and the DIP ABL Intercreditor Agreement (if in existence), the Borrower will, and will cause each other Credit Party to, execute any and all further documents, financing statements, agreements and instruments, and take all such further actions (including the filing and recording of financing statements, fixture filings, mortgages, deeds of trust and other documents) that may be required under any Applicable Law, or that the Collateral Agent may reasonably request, in order to grant, preserve, protect and perfect the validity and priority of the security interests created or intended to be created by the applicable Security Documents, all at the expense of the Borrower and the Subsidiaries.
- (b) Subject to any applicable limitations set forth in the Security Documents (including in any Mortgage), if any assets that are of the nature secured by any Security Documents (including any owned Real Estate or improvements thereto constituting Material Real Property) are acquired by the Borrower or any Subsidiary Guarantor after the Closing Date or are held by any Subsidiary on or after the time it becomes a Guarantor pursuant to Section 9.11 (other than assets constituting Collateral under the Security Documents that become subject to the Lien of any Security Document upon acquisition thereof or assets subject to a Lien granted pursuant to Section 10.2(d) or 10.2(g)), the Borrower will promptly notify the Collateral Agent thereof and, if requested by the Collateral Agent, will cause such assets to be subjected to a Lien securing the Obligations and will take, and cause the other Credit Parties to take, such actions as shall be necessary or reasonably requested by the Collateral Agent, as soon as commercially reasonable but in no event later than 30 days, unless extended by the Collateral Agent in its reasonable discretion, to grant and perfect such Liens consistent with the applicable requirements of the Security Documents, including actions described in paragraph (a) of this Section, all at the expense of the Credit Parties.
- (c) Any Mortgage delivered to the Collateral Agent in accordance with the preceding clause (b) shall be accompanied by those items set forth in clause (d) that are customary for the type of assets covered by such Mortgage. Any items that are customary for the type of assets covered by such Mortgage may be delivered within a commercially reasonable period of time after the delivery of a Mortgage if they are not reasonably available at the time the Mortgage is delivered.
- (d) With respect to any Owned Real Property that constitutes Material Real Property, to the extent requested by the Required Lenders, within 60 days, unless extended by the Required Lenders in their reasonable discretion, the Borrower will deliver, or cause to be delivered, to the Collateral Agent (i) a Mortgage with respect to each Owned Real Property that constitutes Material Real Property, executed by an Authorized Officer of each obligor party thereto, (ii) a policy or policies of title insurance insuring the Lien of each such Mortgage as a valid Lien on the Owned Real Property that constitutes Material Real Property described therein, free of any other

Liens except Permitted Encumbrances or consented to in writing (including via email) by the Collateral Agent, together with such endorsements and reinsurance as the Collateral Agent may reasonably request, together with evidence reasonably acceptable to the Collateral Agent of payment of all title insurance premiums, search and examination charges, escrow charges and related charges, fees, costs and expenses required for the issuance of the title insurance policies referred to above, (iii) a Survey, to the extent requested by the Required Lenders and reasonably necessary to satisfy the requirements of clause (ii) above, (iv) all other documents and instruments, including Uniform Commercial Code or other applicable fixture security financing statements, reasonably requested by the Collateral Agent to be filed, registered or recorded to create the Liens intended to be created by any such Mortgage and perfect such Liens to the extent required by, and with the priority required by, such Mortgage shall have been delivered to the Collateral Agent in proper form for filing, registration or recording and (v) written opinions of legal counsel in the states in which each such Owned Real Property that constitutes Material Real Property is located in customary form and substance. If any building or other improvement included in any Owned Real Property that constitutes Material Real Property is at any time located in an area identified by the Federal Emergency Management Agency (or any successor agency) as a special flood hazard area with respect to which flood insurance has been made available under the National Flood Insurance Act of 1968 (as now or as hereafter in effect or successor act thereto), then the Borrower shall, prior to delivery of the Mortgages, deliver or cause to be delivered, (i) a completed Federal Emergency Management Agency Standard Flood Determination with respect to each Owned Real Property that constitutes Material Real Property, in each case in form and substance reasonably satisfactory to the Collateral Agent and (ii) evidence of flood insurance with respect to each Owned Real Property that constitutes Material Real Property, to the extent and in amounts required by Applicable Laws, in each case in form and substance reasonably satisfactory to the Collateral Agent.

- (e) Notwithstanding anything herein to the contrary, if the Borrower and the Required Lenders mutually agree in their reasonable judgment (confirmed in writing to the Borrower and the Administrative Agent) that the cost or other consequences (including adverse tax and accounting consequences) of creating or perfecting any Lien on any property is excessive in relation to the benefits afforded to the Secured Parties thereby, then such property may be excluded from the Collateral for all purposes of the Credit Documents.
- (f) At the request of the Required Lenders, the Borrower and the Administrative Agent shall enter into a springing account control agreement with the account bank in form and substance satisfactory to the Required Lenders and the Borrower that is sufficient to give Administrative Agent "control" (for purposes of Articles 8 and 9 of the Uniform Commercial Code) over Foreign Reserve Account and/or the DIP Proceeds Account.

## 9.13 Use of Proceeds

(a) The Borrower will use the proceeds from the Loans (i) to consummate the Prepetition ABL Payoff, (ii) for working capital and general corporate purposes of the Debtors; (iii) to pay obligations arising from or related to the Carve Out; (iv) to pay professional fees in connection with the Chapter 11 Cases; (v) to make adequate protection payments; (vi) to pay fees and expenses incurred in connection with the transactions contemplated hereby and in the Restructuring Support Agreement; (vii) to fund an intercompany loan in an amount not to exceed

\$50,000,000 (the "Initial Intercompany Transaction") by Sierra to Avaya International Sales Ltd (the "Initial Intercompany Borrower") to be used for working capital and general corporate purposes of foreign Subsidiaries of the Borrower, including, but not limited to, to fund certain cash pooling accounts of certain foreign Subsidiaries of the Borrower and opened at Citibank, N.A. and its affiliates; and (viii) to fund a deposit of up to \$40,000,000 to a segregated account held by the Borrower (the "Foreign Reserve Account") for purposes of backstopping the liquidity of certain foreign non-Debtor affiliates to the extent necessary to preserve the value of the Debtors' international business; *provided* that the use of funds in the Foreign Reserve Account during the Chapter 11 Cases shall be subject to the Foreign Reserve Protocol.

Reserve Account Withdrawal") shall be subject to the following procedures (the "Foreign Reserve Protocol") in addition to the Intercompany Transfer Mechanic: (i) the Debtors shall submit in writing to the Administrative Agent, with a copy to the advisors to the Akin Ad Hoc Group (as defined in the Restructuring Support Agreement) and the PW Ad Hoc Group (as defined in the Restructuring Support Agreement) a withdrawal request (the "Foreign Reserve Account Withdrawal Notice"), which Foreign Reserve Account Withdrawal Notice shall include (a) the proposed foreign non-Debtor recipient of the transfer; (b) the proposed transfer amount; and (c) the proposed use of funds; (ii) the Administrative Agent shall, within three (3) Business Days from receipt of the Foreign Reserve Account Withdrawal Notice, advise the Debtors if the Required Lenders consent to any such Foreign Reserve Account Withdrawal, which consent shall not be unreasonably withheld. All Foreign Reserve Account Withdrawals shall be made in accordance with the Intercompany Transfer Mechanic.

# 9.14 Maintenance of Ratings

The Borrower will use commercially reasonable efforts to obtain and maintain (but not maintain any specific rating) a public corporate family and/or corporate credit rating, as applicable, and public ratings provided pursuant to this Agreement within 30 days of the Closing Date, in each case, from each of S&P and Moody's.

# 9.15 Changes in Business

The Borrower and the Subsidiaries, taken as a whole, will not fundamentally and substantively alter the character of their business, taken as a whole, from the business conducted by the Borrower and the Subsidiaries, taken as a whole, on the Closing Date and other business activities which are extensions thereof or otherwise similar, incidental, complementary, synergistic, reasonably related or ancillary to any of the foregoing, in each case as determined by the Borrower in good faith.

#### 9.16 Milestones

The Borrower shall comply with the following Milestones (as waived or extended in accordance with the Restructuring Support Agreement) (collectively, the "Milestones"):

(a) The Interim Order shall have been entered by no later than three (3) days after the Petition Date.

- (b) The Escrow Payment shall occur no later than the date when the Interim DIP Order is entered by the Bankruptcy Court; *provided* that, this milestone can only be extended or waived by the Required PW Ad Hoc Group Lenders without the need to obtain the consent of the other Lenders); *provided, further*, that the failure to satisfy this milestone shall not give rise to a Default or Event of Default, unless the Bankruptcy Court does not authorize the Escrow Payment pursuant to the Final DIP Order (unless previously authorized by the Bankruptcy Court under the Interim DIP Order or otherwise authorized by the Bankruptcy Court prior to or contemporaneously with entry of the Final DIP Order).
- (c) The Final Order shall have been entered by no later than forty-five (45) days after the Petition Date.
- (d) The Rights Offering (as defined in the Restructuring Support Agreement) shall have been commenced by no later than ten (10) days after the Petition Date.
- (e) The 2023 PBGC Settlement shall have been approved by the Bankruptcy Court no later than confirmation of the Plan.
- (f) The order provisionally approving the adequacy of the Disclosure Statement, in form and substance acceptable to the Debtors and the Required Lenders, shall have been entered no later than three (3) days after the Petition Date.
- (g) The Plan and related disclosure statement (the "**Disclosure Statement**") shall have been filed no later than one (1) day after the Petition Date.
- (h) The order approving the adequacy of the Disclosure Statement, in form and substance acceptable to the Debtors and the Required Lenders, shall have been entered no later than sixty (60) days after the Petition Date.
- (i) The order confirming the Plan, in form and substance acceptable to the Debtors and the Required Lenders, shall have been entered by no later than sixty (60) days after the Petition Date.
- (j) The substantial consummation of the Plan (as defined in the Restructuring Support Agreement) shall have occurred no later than ninety (90) days after the Petition Date.

### 9.17 Bankruptcy Matters

#### Each Debtor shall:

(a) cause all proposed (i) orders related to or affecting the Loans and other Obligations, the Prepetition Debt and the Credit Documents, any other financing or use of Cash Collateral, any sale or other disposition of Collateral outside the ordinary course, cash management, adequate protection, any plan of reorganization and/or any disclosure statement related thereto, (ii) orders concerning the financial condition of the Borrower or any of its Subsidiaries or other Indebtedness of the Debtors or seeking relief under section 363, 364 or 365 of the Bankruptcy Code or Rule 9019 of the Federal Rules of Bankruptcy Procedure, (iii) orders authorizing additional payments to critical vendors (outside of the relief approved in the First Day

Orders and "second day" orders) and (iv) orders establishing procedures for administration of the Chapter 11 Cases or approving significant transactions submitted to the Bankruptcy Court, in each case, proposed by the Debtors to be in accordance with and permitted by the terms of this Agreement;

(including in any Mortgage), including with respect to any asset subject to a separate notice period thereunder, if any assets that are of the nature secured by any Security Documents (including any owned Real Estate or improvements thereto) are acquired by the Borrower or any Subsidiary Guarantor after the Closing Date or are held by any Subsidiary on or after the time it becomes a Guarantor pursuant to Section 9.11, unless an effective lien has been created under the Security Agreement and perfected under an effective UCC financing statement, the Borrower will promptly notify the Collateral Agent thereof and, if requested by the Collateral Agent, will cause such assets to be subjected to a Lien securing the Obligations and will take, and cause the other Credit Parties to take, such actions as shall be necessary or reasonably requested by the Collateral Agent, as soon as commercially reasonable but in no event later than five (5) Business Days, unless extended by the Collateral Agent in its reasonable discretion, to grant and perfect such Liens consistent with the applicable requirements of the Security Documents, including actions described in paragraph (a) of this Section, all at the expense of the Credit Parties.

### 9.18 DIP Proceeds Account

The Borrower shall deposit or credit all net proceeds of the Term Loans, after giving effect to the Prepetition ABL Payoff and the funding of the Foreign Reserve Account, into a segregated deposit or money market account (the "DIP Proceeds Account"). All withdrawals from the DIP Proceeds Account shall be used solely for the permitted purposes described under Section 9.13 or to make payments on the Term Loans, including withdrawal of proceeds out of the DIP Proceeds Account in anticipation of upcoming payment obligations due within ten (10) Business Days after such withdrawal to another operating account of a Debtor, as determined by the Borrower in good faith. The Borrower shall provide accounting of the DIP Proceeds Account and proceeds thereof (without duplication of any such disclosure that would be included in the Variance Report) upon reasonable request by the Lenders; provided that such request shall not be made more than once every month for all Lenders and such information shall be delivered to the Administrative Agent to be available to all Lenders. The Borrower shall use reasonable efforts to ensure no amounts shall be deposited in the DIP Proceeds Account other than the proceeds of the Term Loans as described above and the proceeds of any investments held in or credited to such account and to the extent any such other amount is deposited in the DIP Proceeds Account, the Borrower shall promptly withdraw such amount from the DIP Proceeds Account.

# **SECTION 10 Negative Covenants**

The Borrower hereby covenants and agrees that on the Closing Date and thereafter, until all Commitments and all Term Loans, together with interest, fees and all other Obligations are Paid in Full:

#### 10.1 Limitation on Indebtedness

The Borrower will not, and will not permit the Subsidiaries to, create, incur, assume or suffer to exist any Indebtedness. Notwithstanding the foregoing, the limitations set forth in the immediately preceding sentence shall not apply to any of the following:

- (a) Indebtedness arising under the Credit Documents;
- (b) Indebtedness in respect of the Prepetition First Lien Debt;
- (c) Indebtedness of the Debtors under the DIP ABL Facility;
- (d) subject to compliance with Sections 10.5 and 10.13, Indebtedness of the Borrower or any Subsidiary owed to the Borrower or any Subsidiary; *provided* that all such Indebtedness of any Credit Party owed to any Person that is not a Credit Party shall be (x) unsecured and subordinated pursuant to an intercompany note reasonably satisfactory to the Required Lenders; *provided further* that such intercompany note shall not be required to be delivered by any Foreign Subsidiary until 30 days after the Closing Date;
- (e) subject to compliance with Section 10.5, Guarantee Obligations incurred by (i) Subsidiaries in respect of Indebtedness of the Borrower or any other Subsidiary that is permitted to be incurred under this Agreement and (ii) the Borrower in respect of Indebtedness of Subsidiaries that is permitted to be incurred under this Agreement; *provided* that (x) if the Indebtedness being guaranteed under this Section 10.1(e) is subordinated to the Obligations, such Guarantee Obligations shall be subordinated to the Guarantee of the Obligations on terms (taken as a whole) at least as favorable to the Lenders as those contained in the subordination of such Indebtedness and (y) a Subsidiary that is not a Credit Party may not, by virtue of this Section 10.1(e), guarantee Indebtedness that such Subsidiary could not otherwise incur under this Section 10.1;
- (f) Indebtedness in respect of any bankers' acceptance, bank guarantees, letter of credit, warehouse receipt or similar facilities entered into in the ordinary course of business (including in respect of workers compensation claims, health, disability or other employee benefits or property, casualty or liability insurance or self-insurance or other Indebtedness with respect to reimbursement-type obligations regarding workers compensation claims and similar obligations);
- (g) Guarantee Obligations (i) incurred in the ordinary course of business in respect of obligations of (or to) suppliers, customers, franchisees, lessors and licensees, or (ii) otherwise constituting Investments permitted by Section 10.5 (other than Investments permitted by Section 10.5(l) by reference to Section 10.1 and Section 10.5(q); *provided* that this clause (ii) shall not be construed to limit the requirements of Section 10.1(d) and (e);
- (h) Indebtedness (including Indebtedness arising under Capital Leases) incurred to finance the purchase price, cost of design, acquisition, construction, repair, restoration, replacement, expansion, installation or improvement of fixed or capital assets or otherwise in respect of Capital Expenditures, so long as such Indebtedness is incurred concurrently with or within 270 days of the acquisition, construction, repair, restoration, replacement, expansion, installation or improvement of such fixed or capital assets or incurrence of such Capital

Expenditure, and any refinancing indebtedness thereof, in an aggregate principal amount not to exceed \$20,000,000 at any time outstanding;

- (i) Indebtedness existing on the Petition Date;
- (j) Indebtedness in respect of Hedging Agreements; *provided* that such Hedging Agreements are not entered into for speculative purposes (as determined by the Borrower in good faith) and are entered into in the ordinary course of business;
- (k) (i) Indebtedness with respect to Existing Letters of Credit, Existing Cash Management Obligations and Existing Hedging Obligations and (ii) Indebtedness with respect to any other letter of credit existing on the Petition Date;
- (l) Indebtedness in respect of performance bonds, bid bonds, appeal bonds, surety bonds and completion guarantees and similar obligations not in connection with money borrowed, in each case provided in the ordinary course of business or consistent with past practice, including those incurred to secure health, safety and environmental obligations in the ordinary course of business (including in respect of construction or restoration activities) or consistent with past practice;
- (m) additional Indebtedness of the Credit Parties (other than Indebtedness for borrowed money) and Subsidiaries that are not Credit Parties; *provided* that the aggregate amount of Indebtedness incurred or incurred pursuant to this Section 10.1(m) shall not exceed \$30,000,000 at any time outstanding;
- (n) cash management services and other Indebtedness in respect of overdraft facilities, employee credit card programs, netting services, automatic clearinghouse arrangements and other cash management and similar arrangements in the ordinary course of business;
- (o) (i) Indebtedness incurred in the ordinary course of business in respect of obligations of the Borrower or any Subsidiary to pay the deferred purchase price of goods or services or progress payments in connection with such goods and services and (ii) Indebtedness in respect of intercompany obligations of the Borrower or any Subsidiary with the Borrower or any Subsidiary in respect of accounts payable incurred in connection with goods sold or services rendered in the ordinary course of business and consistent with past practice and not in connection with the borrowing of money;
- (p) Indebtedness arising from agreements of the Borrower or any Subsidiary providing for indemnification, adjustment of purchase price or similar obligations (including earnouts), in each case entered into in connection with Investments and the Disposition of any business, assets or Stock or Stock Equivalents permitted hereunder;
- (q) Indebtedness of the Borrower or any Subsidiary consisting of (i) financing of insurance premiums or (ii) take or pay obligations contained in supply agreements, in each case arising in the ordinary course of business;

- (r) Indebtedness representing deferred compensation, or similar arrangement, to employees, consultants or independent contractors of the Borrower and the Subsidiaries incurred in the ordinary course of business;
  - (s) [reserved];
- (t) Indebtedness consisting of obligations of the Borrower and the Subsidiaries under deferred compensation or other similar arrangements incurred by such Person in connection with the Transactions;
  - (u) [reserved];
  - (v) [reserved];
  - (w) [reserved];
  - (x) [reserved];
  - (y) [reserved];
  - (z) [reserved];
- (aa) to the extent constituting Indebtedness, customer deposits and advance payments (including progress payments) received in the ordinary course of business from customers for goods and services purchased in the ordinary course of business;
- (bb) (i) Indebtedness of the Borrower or any Subsidiary supported by a letter of credit, in a principal amount not in excess of the stated amount of such letter of credit so long as such letter of credit is otherwise permitted to be incurred pursuant to this Section 10.1 or (ii) obligations in respect of letters of support, guarantees or similar obligations issued, made or incurred for the benefit of the Borrower or any Subsidiary of the Borrower in connection with any statutory filing or the delivery of audit opinions performed in jurisdictions other than the United States;
  - (cc) [reserved];
  - (dd) [reserved];
  - (ee) [reserved]; and
- (ff) all premiums (if any), interest (including post-petition interest), fees, expenses, charges, and additional or contingent interest on obligations described in clauses (a) through (ee) above.

Accrual of interest or dividends, the accretion of accreted value, the accretion or amortization of original issue discount and the payment of interest or dividends in the form of additional Indebtedness or Disqualified Stock will not be deemed to be an incurrence or issuance of Indebtedness or Disqualified Stock for purposes of this covenant.

This Agreement will not treat (1) unsecured Indebtedness as subordinated or junior to secured Indebtedness merely because it is unsecured or (2) senior Indebtedness as subordinated or junior to any other senior Indebtedness merely because it has a junior lien priority with respect to the same collateral.

### 10.2 Limitation on Liens

The Borrower will not, and will not permit the Subsidiaries to, create, incur, assume or suffer to exist any Lien upon any property or assets of any kind (real or personal, tangible or intangible) of the Borrower or such Subsidiary, whether now owned or hereafter acquired, except:

- (a) Liens arising under the Credit Documents, including the Orders (which, for the avoidance of doubt, include Liens granted as adequate protection on account of the Primed Liens);
- (b) Liens securing Indebtedness permitted to be incurred pursuant to Section 10.1(c) and any cash management obligations (including the Existing Cash Management Obligations) and Hedging Obligations (including the Existing Hedging Obligations) secured under the same security documents in respect of the DIP ABL Facility; *provided* that such Liens shall not attach to any assets or property that is not Collateral and shall be subject to the DIP ABL Intercreditor Agreement;
  - (c) Liens securing the Prepetition First Lien Debt;
- (d) Liens securing Indebtedness permitted pursuant to Section 10.1(h); provided that except as otherwise permitted hereby, such Liens attach at all times only to the assets so financed except (1) for accessions to the property financed with the proceeds of such Indebtedness and the proceeds and the products thereof and (2) that individual financings of equipment provided by one lender may be cross collateralized to other financings of equipment provided by such lender;
  - (e) Liens existing on the Petition Date;
- (f) Liens on assets of the Debtors securing Indebtedness permitted under Section 10.1(c);
- (g) (i) prior to the incurrence of the DIP ABL Facility, Liens over cash and Cash Equivalents securing the Existing Letters of Credit, Existing Cash Management Obligations and Existing Hedging Obligations and (ii) on and after the incurrence of the DIP ABL Facility, Liens over cash and Cash Equivalents securing Existing Letters of Credit not deemed to be issued under the definitive documentation in respect of the DIP ABL Facility;
- (h) additional Liens on assets of any Subsidiary that is not a Credit Party securing Indebtedness of such Subsidiary permitted pursuant to Section 10.1 (or other obligations of such Subsidiary not constituting Indebtedness);
  - (i) [reserved];

- (j) [reserved];
- (k) additional Liens securing obligations of the Credit Parties (other than obligations for borrowed money) and Subsidiaries that are not Credit Parties, so long as the aggregate amount of obligations secured thereby at any time outstanding does not exceed \$20,000,000 at the time of incurrence or issuance;
  - (1) [reserved];
  - (m) Permitted Encumbrances; and
- (n) the supplement, amendment, amendment and restatement, modification, replacement, refinancing, refunding, restructuring, extension or renewal of any Lien permitted by clause (e) of this Section 10.2 upon or in the same assets theretofore subject to such Lien (or upon or in after-acquired property that is affixed or incorporated into the property covered by such Lien and accessions thereto or any proceeds or products thereof.

## 10.3 Limitation on Fundamental Changes

The Borrower will not, and will not permit the Subsidiaries to, consummate any merger, consolidation or amalgamation, or liquidate, wind up or dissolve itself (or suffer any liquidation or dissolution), or convey, sell, lease, assign, transfer or otherwise consummate the Disposition of, all or substantially all its business units, assets or other properties, except that:

- (a) [reserved];
- (b) so long as no Event of Default has occurred and is continuing or would result therefrom, any non-Debtor Subsidiary may be merged, amalgamated or consolidated with or into any one or more non-Debtor Subsidiaries;
  - (c) [reserved];
- (d) any Subsidiary that is not a Credit Party may sell, lease, transfer or otherwise Dispose of any or all of its assets (upon voluntary liquidation or otherwise) to the Borrower or any other Subsidiary;
- (e) the Borrower or any Subsidiary of the Borrower may sell, lease, transfer or otherwise Dispose of any or all of its assets (upon voluntary liquidation or otherwise) to any Credit Party; *provided* that the consideration for any such Disposition by any Person other than a Debtor shall not exceed the fair value of such assets;
- (f) any Subsidiary may liquidate or dissolve if (i) the Borrower determines in good faith that such liquidation or dissolution is in the best interests of the Borrower and is not materially disadvantageous to the Lenders and (ii) to the extent such Subsidiary is a Credit Party, any assets or business of such Subsidiary not otherwise disposed of or transferred in accordance with Section 10.4 or 10.5, or in the case of any such business, discontinued, shall be transferred to, or otherwise owned or conducted by, a Debtor after giving effect to such liquidation or dissolution;

- (g) any non-Debtor Subsidiary may change its legal form so long as no Event of Default has occurred and is continuing or would result therefrom;
  - (h) [reserved];
  - (i) the Transactions may be consummated; and
- (j) the Borrower and the Subsidiaries may consummate a merger, amalgamation dissolution, liquidation, windup, consolidation or Disposition, constituting, or otherwise resulting in, a transaction permitted by Section 10.4 (other than pursuant to (x) Section 10.4(d) and (y) the Disposition of all or substantially all of the assets of the Borrower and its Subsidiaries, taken as a whole, to any Person other than the Borrower or any Guarantor), an Investment permitted pursuant to Section 10.5 (other than Section 10.5(l)), and any Restricted Payments permitted pursuant to Section 10.6 (other than Section 10.6(f)).

## 10.4 Limitation on Disposition

The Borrower will not, and will not permit the Subsidiaries to make any Disposition, except that:

- (a) the Borrower and the Subsidiaries may sell, transfer or otherwise Dispose of (i) obsolete, negligible, immaterial, worn-out, uneconomical, scrap, used, or surplus or mothballed assets (including any such equipment that has been refurbished in contemplation of such Disposition) or assets no longer used or useful in the business or no longer commercially desirable to maintain, (ii) inventory or goods (or other assets) held for sale in the ordinary course of business, (iii) cash and Cash Equivalents, (iv) immaterial assets (including allowing any registrations or any applications for registration of any immaterial intellectual property rights to lapse or go abandoned in the ordinary course of business), and (v) assets for the purposes of charitable contributions or similar gifts to the extent such assets are not material to the ability of the Borrower and the Subsidiaries, taken as a whole, to conduct its business in the ordinary course;
- (b) the Borrower and the Subsidiaries may make Dispositions of assets with a fair market value not to exceed \$5,000,000 in the aggregate;
- (c) (i) the Borrower and the Subsidiaries may make Dispositions to the Borrower or any other Credit Party, (ii) any Subsidiary that is not a Credit Party may make Dispositions to the Borrower or any other Subsidiary of the Borrower and (iii) any Credit Party may make Dispositions to a non-Credit Party to the extent permitted under Section 10.13;
- (d) the Borrower and any Subsidiary may effect any transaction permitted by Sections 10.2, 10.3 (other than Section 10.3(j)), 10.5 (other than Section 10.5(l)) or 10.6 (other than Section 10.6(f));
- (e) the Borrower and any Subsidiary may lease, sublease, license (only on a non-exclusive basis, with respect to any intellectual property) or sublicense (only on a non-exclusive basis, with respect to any intellectual property) real, personal or intellectual property in the ordinary course of business;

- (f) Dispositions of property (including like-kind exchanges) to the extent that (i) such property is exchanged for credit against the purchase price of similar replacement property (excluding any boot thereon) or (ii) the proceeds of such Disposition are applied to the purchase price of such replacement property, in each case under Section 1031 of the Code or otherwise;
  - (g) [reserved];
- (h) Dispositions of (i) Investments in joint ventures (regardless of the form of legal entity) to the extent required by, or made pursuant to, customary buy/sell arrangements or put/call arrangements between the joint venture parties set forth in joint venture arrangements and similar binding arrangements or (ii) to joint ventures in connection with the dissolution or termination of a joint venture to the extent required pursuant to joint venture and similar arrangements;
  - (i) [reserved];
  - (j) Dispositions listed on Schedule 10.4;
- (k) transfers of property subject to a Recovery Event or in connection with any condemnation proceeding upon receipt of the Net Cash Proceeds of such Recovery Event or condemnation proceeding;
- (l) Dispositions or discounts of accounts receivable or notes receivable in connection with the collection or compromise thereof or the conversion of accounts receivable to notes receivable;
  - (m) [reserved];
- (n) the execution of (or amendment to), settlement of or unwinding of any Hedging Agreement;
  - (o) [reserved];
- (p) the surrender or waiver of contractual rights and settlement or waiver of contractual or litigation claims, in each case, in the ordinary course of business;
  - (q) [reserved];
  - (r) [reserved];
- (s) the Borrower and any Subsidiary may (i) terminate or otherwise collapse its cost sharing agreements with the Borrower or any Subsidiary and settle any crossing payments in connection therewith, (ii) convert any intercompany Indebtedness to Stock or any Stock to intercompany Indebtedness, (iii) settle, discount, write off, forgive or cancel any intercompany Indebtedness or other obligation owing by the Borrower or any Subsidiary or (iv) settle, discount, write off, forgive or cancel any Indebtedness owing by any present or former consultants, managers, directors, officers or employees of Holdings, the Borrower, any direct or indirect parent thereof, or any Subsidiary thereof or any of their successors or assigns;

- (t) any Disposition of property to the extent that (1) such property is exchanged for credit against the purchase price of similar replacement property that is purchased within 270 days thereof or (2) the proceeds of such Disposition are promptly applied to the purchase price of such replacement property (which replacement property is actually purchased within 270 days thereof);
  - (u) [reserved];
- (v) any swap of assets in exchange for services or other assets in the ordinary course of business of comparable or greater fair market value or usefulness to the business of the Borrower and the Subsidiaries, taken as a whole, as determined in good faith by the Borrower; and
- (w) Dispositions of any asset between or among the Borrower and/or any Subsidiary as a substantially concurrent interim Disposition in connection with a Disposition otherwise permitted pursuant to clauses (a) through (v) above; *provided* that after giving effect to any such Disposition, to the extent the assets subject to such Dispositions constituted Collateral, such assets shall remain subject to, or be rejoined to, the Lien of the Security Documents.

Notwithstanding the foregoing, no transfer or other Disposition of any intellectual property by a Credit Party to any Person that is not a Credit Party may be made except pursuant to Section 10.4(e).

### 10.5 Limitation on Investments

The Borrower will not, and will not permit the Subsidiaries, to make any Investment except:

- (a) extensions of trade credit, asset purchases (including purchases of inventory, supplies, materials and equipment) and the licensing or contribution of intellectual property pursuant to joint marketing arrangements, original equipment manufacturer arrangements or development agreements with other Persons, in each case in the ordinary course of business;
- (b) Investments in cash or Cash Equivalents when such Investments were made;
  - (c) [reserved];
- (d) Investments existing on, or made pursuant to legally binding written commitments in existence on, the Closing Date;
- (e) any Investment acquired by the Borrower or any Subsidiary (i) in exchange for any other Investment or accounts receivable held by the Borrower or any such Subsidiary in connection with or as a result of a bankruptcy, workout, reorganization, or recapitalization of, or settlement of delinquent accounts or disputes with or judgments against, the issuer, obligor or borrower of such original Investment or accounts receivable, (ii) as a result of a foreclosure by the Borrower or any Subsidiary with respect to any secured Investment or other transfer of title with respect to any secured Investment in default or (iii) as a result of the settlement, compromise or

resolution of litigation, arbitration or other disputes with Persons who are not Affiliates or in satisfaction or judgments against other Persons;

- (f) [reserved];
- (g) (i) Investments by the Borrower or any Subsidiary in the Borrower or any Subsidiary that is a Credit Party, (ii) subject to Section 10.13, Investments by any Subsidiary that is not a Credit Party in any other Subsidiary that is not a Credit Party, (iii) subject to Section 10.13, the advance of the Initial Intercompany Transaction to the Initial Intercompany Borrower and any intercompany Foreign Reserve Account Withdrawals, which shall be subject to the Intercompany Transfer Mechanic and, if applicable, the Foreign Reserve Protocol and (iv) the issuance of letters of credit under the DIP ABL Facility for the benefit of any Subsidiary that is not a Credit Party;
  - (h) [reserved];
  - (i) [reserved];
- (j) Investments constituting non-cash proceeds received from Disposition of assets permitted by Section 10.4;
  - (k) [reserved];
- (l) Investments consisting of or resulting from Indebtedness, Liens, Restricted Payments, fundamental changes and Dispositions permitted by Section 10.1 (other than Sections 10.1(d), 10.1(e) and 10.1(g)(ii)), 10.2, 10.3 (other than Section 10.3(j)), 10.4 (other than Section 10.4(d)), 10.6 (other than Section 10.6(f)), 10.7 or 10.8, as applicable;
- (m) loans and advances to any direct or indirect parent of the Borrower in lieu of, and not in excess of the amount of, Restricted Payments to the extent permitted to be made to such parent in accordance with Section 10.6; *provided* that the aggregate amount of such loans and advances shall reduce the ability of the Borrower and the Subsidiaries to make Restricted Payments under the applicable clauses of Section 10.6 by such amount;
- (n) Investments consisting of extensions of credit in the nature of accounts receivable or notes receivable arising from the grant of trade credit in the ordinary course of business, and Investments received in satisfaction or partial satisfaction thereof from financially troubled account debtors and other credits to suppliers in the ordinary course of business;
- (o) Investments in the ordinary course of business consisting of endorsements for collection or deposit and customary trade arrangements with customers consistent with past practices;
- (p) advances of payroll payments to employees, consultants or independent contractors or other advances of salaries or compensation to employees, consultants or independent contractors, in each case in the ordinary course of business;

	(q)	Guarantee	Obligations	of the Borro	wer or any	Subsidiary of	leases (	other
than	Capital Lease	s) or of other	· obligations	that do not	constitute	Indebtedness,	in each	case
enter	ed into in the o	ordinary cours	e of business	:				

- (r) [reserved];
- (s) Investments in Hedging Agreements permitted by Section 10.1;
- (t) [reserved];
- (u) Investments consisting of deposits of cash and Cash Equivalents as collateral support permitted under Section 10.2;
  - (v) [reserved];
  - (w) [reserved];
- (x) Investments consisting of purchases and acquisitions of assets and services in the ordinary course of business;
- (y) Investments in the ordinary course of business consisting of Article 3 endorsements for collection or deposit and Article 4 customary trade arrangements with customers consistent with past practice;
  - (z) [reserved];
- (aa) contributions in connection with compensation arrangements to a "rabbi" trust for the benefit of employees, directors, partners, members, consultants, independent contractors or other service providers or other grantor trust subject to claims of creditors in the case of a bankruptcy of the Borrower or any of its Subsidiaries;
  - (bb) Investments relating to pension trusts;
  - (cc) [reserved];
  - (dd) [reserved];
- (ee) Investments in deposit accounts, commodities and securities accounts opened in the ordinary course of business;
- (ff) Investments solely to the extent such Investments reflect an increase in the value of Investments otherwise permitted under this Agreement; and
- (gg) Investments in prepaid expenses, negotiable instruments held for collection and lease, utility and workers compensation, performance and similar deposits entered into as a result of the operations of the business in the ordinary course of business.

Notwithstanding the foregoing, no Investment consisting of or resulting from any transfer or other Disposition of any intellectual property by a Credit Party to a Subsidiary that is

not a Credit Party may be made except pursuant to (i) Section 10.5(l) (solely in respect of Dispositions permitted by Section 10.4(e) or (g)) or (ii) the proviso to Section 10.5(w).

# 10.6 Limitation on Restricted Payments

The Borrower will not, and will not permit the Subsidiaries to, declare or pay any Restricted Payments except that:

- (a) [reserved];
- (b) [reserved];
- (c) the Borrower may make Restricted Payments to Holdings in an amount required for Holdings to pay, in each case without duplication:
  - (i) [reserved];
  - (ii) (A) Holdings' general operating expenses incurred in the ordinary course of business and other corporate overhead costs and expenses (including administrative, legal, accounting and similar expenses provided by third parties) to the extent such costs and expenses are attributable to the ownership or operation of the Borrower and its Subsidiaries, or (B) any indemnification claims made by current directors or officers of the Borrower or Holdings (and not, for the avoidance of doubt, any former director or officer of the Borrower or Holdings) to the extent such claims are attributable to the ownership or operation of the Borrower or any Subsidiary;
  - (iii) franchise and excise Taxes and other fees, Taxes and expenses required to maintain the corporate existence of Holdings;
    - (iv) [reserved];
    - (v) [reserved]; and
  - (vi) customary salary, bonus, severance and other benefits payable to officers, employees or consultants of any direct or indirect parent company of the Borrower to the extent such salaries, bonuses and other benefits are attributable to the ownership or operation of the Borrower and its Subsidiaries;
    - (d) [reserved];
- (e) Restricted Payments consisting of or resulting from Liens, fundamental changes, Dispositions, Investments or other payments permitted by 10.2, 10.3 (other than Section 10.3(j)), 10.4 (other than Section 10.4(d)), 10.5 (other than Section 10.5(l)), 10.7 or 10.8, as applicable;
  - (f) [reserved];
  - (g) [reserved];

- (h) [reserved];
- (i) [reserved];
- (j) [reserved];
- (k) [reserved];
- (l) the Borrower may make payments described in Section 10.11 (other than Section 10.11(b));
  - (m) [reserved];
  - (n) [reserved];
  - (o) [reserved];
  - (p) [reserved];
  - (q) [reserved];
  - (r) [reserved];
- (s) each Subsidiary may make Restricted Payments to the Borrower and other Subsidiaries of the Borrower (and, in the case of a Restricted Payment by a non-Wholly Owned Subsidiary, to the Borrower and any other Subsidiary, as compared to the other owners of Stock in such Subsidiary, on a pro rata or more than pro rata basis based on their ownership interests of the relevant class of Stock).

# 10.7 Limitations on Debt Prepayments and Amendments

Except as expressly permitted by the terms and conditions set forth in the Orders, the Borrower will not, and will not permit the Subsidiaries to, mandatorily or voluntarily pay or prepay, repurchase or redeem or otherwise defease any Indebtedness of a Debtor that was incurred or arose prior to the filing of the Chapter 11 Cases.

# 10.8 Limitation on Subsidiary Distributions

The Borrower will not, and will not permit any Subsidiary that is not a Guarantor to, directly or indirectly, create or otherwise cause or suffer to exist or become effective any consensual encumbrance or consensual restriction on the ability of any such Subsidiary to (x) (i) pay dividends or make any other distributions to the Borrower or any Subsidiary that is a Guarantor on its Stock or Stock Equivalents or with respect to any other interest or participation in, or measured by, its profits or (ii) pay any Indebtedness owed to the Borrower or any Subsidiary that is a Guarantor, (y) make loans or advances to the Borrower or any Subsidiary that is Guarantor or (z) sell, lease or transfer any of its properties or assets to the Borrower or any Subsidiary that is a Guarantor, except (in each case) for such encumbrances or restrictions existing under or by reason of:

- (a) contractual encumbrances or restrictions in effect on the Closing Date, including pursuant to this Agreement, and documentation in respect of Existing Cash Management Obligations, Existing Hedging Agreements and contractual encumbrances or restrictions in respect of the DIP ABL Facility;
- (b) purchase money obligations and Capitalized Lease Obligations that impose restrictions of the nature discussed in clause (x), (y) or (z) above on the property so acquired, any replacements of such property or assets and additions and accessions thereto, after-acquired property subject to such arrangement, the proceeds and the products thereof and customary security deposits in respect thereof and in the case of multiple financings of equipment (or assets affixed or appurtenant thereto and additions and accessions) provided by any lender, other equipment (or assets affixed or appurtenant thereto and additions and accessions) financed by such lender (it being understood that such restriction shall not be permitted to apply to any property to which such restriction would not have applied but for such acquisition);
- (c) Applicable Laws or any applicable rule, regulation or order, or any request of any Governmental Authority having regulatory authority over the Borrower or any of its Subsidiaries;
  - (d) [reserved];
- (e) contracts for the sale of assets, including customary restrictions with respect to a Subsidiary of the Borrower pursuant to an agreement that has been entered into for the sale or Disposition of all or substantially all of the Stock or Stock Equivalents or assets of such Subsidiary and restrictions on transfer of assets subject to Liens permitted hereunder;
- (f) (x) secured Indebtedness otherwise permitted to be incurred pursuant to Sections 10.1 and 10.2 that limit the right of the debtor to Dispose of the assets securing such Indebtedness and (y) restrictions or encumbrances on transfers of assets subject to Liens permitted hereunder (but, with respect to any such Lien, only to the extent that such transfer restrictions apply solely to the assets that are the subject of such Lien);
- (g) restrictions or encumbrances on cash or other deposits or net worth imposed by customers under, or made necessary or advisable by, contracts entered into in the ordinary course of business:
- (h) restrictions or encumbrances imposed by other Indebtedness or Disqualified Stock of Subsidiaries permitted to be incurred subsequent to the Closing Date pursuant to the provisions of Section 10.1;
- (i) customary provisions in joint venture agreements or arrangements and other similar agreements or arrangements relating solely to such joint venture (including its assets and Subsidiaries) and the Stock or Stock Equivalents issued thereby;
- (j) customary provisions contained in leases, sub-leases, licenses, sub-licenses or similar agreements, in each case, entered into in the ordinary course of business;
  - (k) [reserved];

- (l) customary restrictions on leases, subleases, licenses, sublicenses or asset sale agreements otherwise permitted hereby so long as such restrictions relate to property interest, rights or the assets subject thereto;
- (m) customary provisions restricting assignment or transfer of any agreement entered into in the ordinary course of business;

# (n) [reserved]; or

(o) any encumbrances or restrictions of the type referred to in clauses (x), (y) and (z) above imposed by any amendments, modifications, restatements, renewals, increases, supplements, refundings, extensions, replacements or refinancings of the contracts, instruments or obligations referred to in clauses (a) through (n) above; *provided* that such amendments, modifications, restatements, renewals, increases, extensions, supplements, refundings, extensions, replacements, restructurings or refinancings (x) are, in the good faith judgment of the Borrower, not materially more restrictive with respect to such encumbrance and other restrictions taken as a whole than those prior to such amendment, modification, restatement, renewal, increase, extension, restructuring, supplement, refunding, replacement or refinancing or (y) do not materially impair the Borrower's ability to pay its obligations under the Credit Documents as and when due (as determined in good faith by the Borrower);

provided that (x) the priority of any preferred stock in receiving dividends or liquidating distributions prior to dividends or liquidating distributions being paid on common stock and (y) the subordination of (including the application of any standstill requirements to) loans or advances made to the Borrower or any Subsidiary that is a Guarantor to other Indebtedness incurred by the Borrower or any Subsidiary that is a Guarantor shall not be deemed to constitute such an encumbrance or restriction.

# 10.9 Amendment of Organizational Documents

The Borrower will not, nor will the Borrower permit any Credit Party to, amend or otherwise modify any of its Organizational Documents in a manner that is materially adverse to the Lenders, except as required by Applicable Laws.

# 10.10 Permitted Activities and Transfer to Parent Companies

Holdings will not engage in any material operating or business activities; provided that the following and any activities incidental thereto shall be permitted in any event: (i) its ownership of the Stock of the Borrower (but no other direct Subsidiaries), including receipt and payment of dividends and payments in respect of Indebtedness and other amounts in respect of Stock, (ii) the maintenance of its legal existence (including the ability to incur and pay, as applicable, fees, costs and expenses and taxes relating to such maintenance), (iii) the performance of its obligations with respect to the Transactions, the Credit Documents and any other documents governing Indebtedness permitted hereby, (iv) [reserved], (v) financing activities, including the issuance of securities, incurrence of debt, receipt and payment of dividends and distributions, making contributions to the capital of the Borrower and guaranteeing the obligations of the Borrower and the Subsidiaries, (vi) if applicable, participating in tax, accounting and other administrative matters as a member of the consolidated group and the provision of administrative

and advisory services (including treasury and insurance services) to its Subsidiaries of a type customarily provided by a holding company to its Subsidiaries, (vii) holding any cash or other property (but not operate any property), (viii) making and receiving of any dividends, payments in respect of Indebtedness or Investments permitted hereunder, (ix) providing indemnification to officers and directors, (x) [reserved], (xi) activities related to the consummation of the Transactions and activities contemplated thereby, (xii) [reserved], (xiii) [reserved], (xiv) [reserved], (xv) any transaction with the Borrower or any Subsidiary to the extent expressly permitted under this Section 10, (xvi) [reserved], (xvii) paying any Taxes it is obligated to pay, (xviii) [reserved] and (xix) any activities incidental or reasonably related to the foregoing.

## 10.11 Transactions with Affiliates

The Borrower will not conduct, nor will the Borrower permit any Subsidiary to, enter into any transaction of any kind with any Affiliate of the Borrower, whether or not in the ordinary course of business, other than:

- (a) transactions (i) between or among Credit Parties, (ii) between or among Subsidiaries that are not Credit Parties or (iii) between a Credit Party and a Subsidiary that is not a Credit Party in the ordinary course of business;
- (b) transactions permitted by Section 10.5 or Section 10.6 (other than Section 10.6(m));
- (c) transactions on terms substantially as favorable to the Borrower or such Subsidiary as would reasonably be obtainable by the Borrower or such Subsidiary at the time in a comparable arm's-length transaction with a Person other than an Affiliate;
  - (d) [reserved];
- (e) (i) employment, consulting and severance arrangements between the Borrower and the Subsidiaries (or any direct or indirect parent of the Borrower) and their respective officers, employees, directors or consultants in the ordinary course of business (including payments, loans and advances in connection therewith) and (ii) issuances of securities, or other payments, awards or grants in cash, securities or otherwise and other transactions pursuant to any equityholder, employee or director equity plan or stock or other equity option plan or any other management or employee benefit plan or agreement, other compensatory arrangement or any stock or other equity subscription, co-invest or equityholder agreement;
- (f) the payment of customary fees and reasonable out of pocket costs to, and indemnities provided on behalf of, directors, managers, consultants, officers and employees of the Borrower (or, to the extent attributable to the ownership of the Borrower and its Subsidiaries, any direct or indirect parent thereof) and the Subsidiaries of the Borrower;
  - (g) [reserved];
  - (h) [reserved];

- (i) transactions pursuant to permitted agreements in existence on the Closing Date and set forth on Schedule 10.11;
  - (j) [reserved];
  - (k) [reserved]; or
- (l) to the extent covered thereby, affiliate transactions authorized or approved pursuant to (i) the Interim Order, and, when applicable, the Final Order, or any First Day Order or (ii) any other order entered by the Bankruptcy Court without the objection of the Required Lenders; *provided*, for the avoidance of doubt, that such objection need not be filed with the Bankruptcy Court.

## 10.12 Permitted Variance

Subject to the Permitted Variance (as defined below), permit actual "total receipts" to exceed, or "operating disbursements" to be less than, for any Variance Test Period, total receipts or operating disbursements projected in the Approved Budget for such Variance Test Period to be determined on a bi-weekly basis.

# 10.13 Intercompany Limitations

- (a) Notwithstanding anything herein to the contrary and without limiting any other covenant contained herein, no credit support, guarantee, cash and/or Cash Equivalents (in each case, including proceeds of the Term Facility), Cash Collateral, may be transferred or otherwise provided by any Credit Party, directly or indirectly, to or for the benefit of any Subsidiary that is not a Credit Party (either in the form of Restricted Payments, Investments, intercompany advances, guarantee of obligations (including interest thereon) or otherwise (except for repayment of account payables for goods delivered or services rendered, including pursuant to shared services or other intercompany service agreements, in the ordinary course of business, consistent with past practice and subject to the Approved Budget)) other than (A) the issuance of letters of credit under the DIP ABL Facility for the benefit of any Subsidiary that is not a Credit Party or (B) if:
  - (i) any such amounts shall be made, (x) first, as an intercompany loan from the applicable Credit Party to Sierra, (y) second, as an intercompany loan from Sierra to the Initial Intercompany Borrower, and (z) third, except as prohibited by local law, as a substantially concurrent unsecured intercompany loan from the Initial Intercompany Borrower to the applicable non-Debtor Subsidiary which is the recipient or beneficiary of the proceeds, credit support or guarantee (the "Non-Debtor Subsidiary Transfer");
  - (ii) each Non-Debtor Subsidiary Transfer shall be evidenced by the Foreign Intercompany Note (the "Non-Debtor Subsidiary Notes/Receivables"). All such Non-Debtor Subsidiary Notes/Receivables payable to Sierra shall be pledged by Sierra as Collateral. To the extent the proceeds of such Investment is further transferred, in one or more steps, to another Subsidiary that is not a Debtor, each step of such transfer shall be evidenced by a Foreign Intercompany Note.

To the extent that local law limits the implementation of the foregoing and notwithstanding the fact that the Non-Debtor Subsidiary Notes/Receivables shall be unsecured obligations of the applicable non-Debtor Subsidiaries, the Orders shall provide that for all purposes in connection with the Cases (including, without limitation for purposes of creditor recoveries and value allocation in connection with these Cases (including in connection with any chapter 11 plan or otherwise)), all Non-Debtor Subsidiary Notes/Receivables shall be deemed to be secured obligations of the applicable non-Debtor Subsidiaries; the intent being to ensure that no creditor of the Debtors that asserts claims against Subsidiaries that are not Debtors on account of any claim against, obligation of, or transaction with, the Debtors (in respect of asserted joint and several liability or otherwise) unjustly benefits from the funding of Subsidiaries that are not Debtors through the use of proceeds of the Term Facility following the Petition Date. The provisions in this Section 10.13(a) are hereinafter referred to as the "Intercompany Transfer Mechanic."

# 10.14 Bankruptcy Matters

The Credit Parties will not permit, and will not permit any of their respective Subsidiaries to, without the Required Lenders' prior written consent (unless otherwise specified), do any of the following:

- (a) use any portion or proceeds of the Loans or the Collateral for payments or purposes that would violate the terms of the Orders;
- (b) incur, create, assume, suffer to exist or permit, except for the Carve Out or as otherwise expressly permitted by the Orders or any other order of the Bankruptcy Court reasonably acceptable to the Required Lenders, any other superpriority administrative claim which is *pari passu* with or senior to the claim of the Secured Parties against any Debtor;
- (c) subject to the terms of the Orders, assert, join, investigate, support or prosecute any claim or cause of action against any of the Secured Parties (in their capacities as such), unless such claim or cause of action is in connection with the enforcement of the Credit Documents against any of the Secured Parties;
- (d) other than as provided in any First Day Order or any Order, enter into any agreement to return any of its inventory to any of its creditors for application against any Prepetition Indebtedness, trade payables arising before the Petition Date or other claims arising before the Petition Date under section 546(c) of the Bankruptcy Code if, after giving effect to any such agreement, the aggregate amount applied to Prepetition Indebtedness, trade payables arising before the Petition Date and other claims arising before the Petition Date subject to all such agreements, setoffs and recoupments since the Petition Date would exceed \$500,000;
- (e) seek, consent to, or permit to resist any order granting authority to take any action that is prohibited by the terms of this Agreement, the Orders, the other Credit Documents or refrain from taking any action that is required to be taken by the terms of this Agreement, the Orders or any of the other Credit Documents;
- (f) subject to the terms of the Orders, object to, contest, delay, prevent or interfere with in any material manner the exercise of rights and remedies by the Agents, the Lenders or other Secured Parties with respect to the Collateral following the occurrence of an

Event of Default, including without limitation a motion or petition by any Secured Party to lift an applicable stay of proceedings to do the foregoing (*provided* that any Debtor may contest or dispute whether an Event of Default has occurred in accordance with the terms of the Orders and the Credit Documents);

(g) make or permit to be made any change to the Orders.

Notwithstanding anything set forth above, this Section 10 shall not prohibit (i) any transactions contemplated by the Restructuring Transactions Memorandum (as defined in the Plan), including any tax reorganization transaction or tax-related restructuring transaction or (ii) any order of the Bankruptcy Court, in each case of (i) and (ii), consented to by the "Required Consenting Stakeholders" (as defined in the Restructuring Support Agreement.

## **SECTION 11 Events of Default**

Upon the occurrence of any of the following specified events (each an "Event of Default"):

# 11.1 Payments

The Borrower shall (a) default in the payment when due of any principal of the Term Loans, (b) default, and such default shall continue for more than two Business Days, in the payment when due of any interest on the Term Loans or (c) default, and such default shall continue for more than five Business Days, in the payment when due of any Fees or any other amounts owing hereunder or under any other Credit Document; or

# 11.2 Representations, Etc.

Any representation, warranty or statement made or deemed made by any Credit Party herein or in any other Credit Document or any certificate delivered or required to be delivered pursuant hereto or thereto shall prove to be materially untrue on the date as of which made or deemed made; or

#### 11.3 Covenants

Any Credit Party shall:

- (a) default in the due performance or observance by it of any term, covenant or agreement contained in Section 9.5 (solely with respect to the Borrower) or Section 10; or
- (b) default in the due performance or observance by it of any term, covenant or agreement (other than those referred to in Section 11.1 or 11.2 or clause (a) of this Section 11.3) contained in this Agreement or any other Credit Document and such default shall continue unremedied for a period of at least 15 calendar days after receipt of written notice by the Borrower from the Administrative Agent; or

# 11.4 Default Under Other Agreements

The Borrower or any Subsidiary shall (i) default in any payment with respect to any Indebtedness (other than any Indebtedness described in Section 11.1 or Hedging Obligations) in excess of \$10,000,000 in the aggregate for the Borrower and such Subsidiaries beyond the period of grace or cure and following all required notices, if any, provided in the instrument or agreement under which such Indebtedness was created or (ii) default in the observance or performance of any agreement or condition relating to any such Indebtedness or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event shall occur or condition exist (other than any agreement or condition relating to, or provided in any instrument or agreement, under which such Hedging Obligations were created) beyond the period of grace or cure and following all required notices, if any, provided in the instrument or agreement under which such Indebtedness was created, if the effect of which default or other event or condition is to cause, or to permit the holder or holders of such Indebtedness (or a trustee or agent on behalf of such holder or holders) to cause, any such Indebtedness to become due or to be repurchased, prepaid, defeased or redeemed (automatically or otherwise), or an offer to repurchase, prepay, defease or redeem such Indebtedness to be made, prior to its Stated Maturity; or (b) without limiting the provisions of clause (a) above, any such Indebtedness shall be declared to be due and payable, or required to be prepaid other than by a regularly scheduled required prepayment (other than any Hedging Obligations) or as a mandatory prepayment, prior to the Stated Maturity thereof; provided that clauses (a) and (b) above shall not apply to secured Indebtedness that becomes due as a result of the voluntary sale or transfer of the property or assets securing such Indebtedness, if such sale or transfer is permitted hereunder and under the documents providing for such Indebtedness; provided, further, that this Section 11.4 shall not apply to (i) any Indebtedness if the sole remedy of the holder thereof following such event or condition is to elect to convert such Indebtedness into Stock or Stock Equivalents (other than Disqualified Stock) and cash in lieu of fractional shares, (ii) any such default that is remedied by or waived (including in the form of amendment) by the requisite holders of the applicable item of Indebtedness or contested in good faith by the Borrower or the applicable Subsidiary in either case, prior to acceleration of all the Term Loans pursuant to this Section 11, (iii) any Indebtedness outstanding hereunder or (iv) any Indebtedness of any Debtor that was incurred prior to the Petition Date unless such Indebtedness has been accelerated and the enforcement of remedies with respect to such Indebtedness shall not have been stayed by the commencement of the Chapter 11 Cases;; provided further that any failure to perform or observe any financial covenant in respect of the DIP ABL Facility shall not constitute a Default or Event of Default hereunder unless and until the lenders thereof have declared all obligations thereunder to be immediately due and payable and terminate the commitments thereunder and such declaration has not been rescinded; or

## 11.5 Bankruptcy

(a) An involuntary proceeding shall be commenced or a voluntary or involuntary petition shall be filed in a court of competent jurisdiction seeking (1) relief in respect of any Domestic Subsidiary that did not become a Debtor on the Petition Date, or of a substantial part of the assets of any Domestic Subsidiary that did not become a Debtor on the Petition Date, under the Bankruptcy Code (unless (A) prior to the commencement of such case such Domestic Subsidiary becomes a Credit Party hereunder, (B) within five (5) Business Days after the commencement of such case, such Subsidiary's case becomes jointly administered with the

Chapter 11 Cases and (C) each of the Interim Order and/or Final Order (as applicable) are made applicable to such Subsidiary as a post-petition Guarantor hereunder, (2) the appointment of a receiver, trustee, custodian, conservator or similar official for any material Foreign Subsidiary, or (3) the winding-up or liquidation of any material Foreign Subsidiary; provided that in the case of any of the foregoing with respect to any such Foreign Subsidiary, an Event of Default pursuant to this Section 11.5 shall only be deemed to occur if such occurrence shall result (or would reasonably be expected to result) in a Material Adverse Effect; or (b) any ancillary insolvency proceeding in any applicable foreign jurisdiction shall be commenced against any Debtor or any recognition, administrative, and substantive order shall be entered against any Debtor by the applicable court (other than a proceeding that would not reasonably be expected to result in a Material Adverse Effect or cause a material delay of the Chapter 11 Cases), in each case without prior consent of the Required Lenders or on terms not satisfactory to the Required Lenders; or

#### 11.6 ERISA

The occurrence of any ERISA Event, which would be reasonably likely to have a Material Adverse Effect; or

## 11.7 Guarantee

Any Guarantee provided by Holdings, the Borrower or any Subsidiary or any material provision thereof shall cease to be in full force or effect (other than pursuant to the terms hereof or thereof) or any such Guarantor thereunder or any other Credit Party shall deny or disaffirm in writing any such Guarantor's obligations under the Guarantee; or

# 11.8 Security Agreement

The Security Agreement or any other material Security Document pursuant to which the assets of any Credit Party are pledged as Collateral or any material provision thereof shall cease to be in full force or effect in respect of a material portion of the Collateral (other than pursuant to the terms hereof or thereof or any defect arising as a result of acts or omissions of the Collateral Agent or any Lender which do not result from a material breach by a Credit Party of its obligations under the Credit Documents) or any grantor thereunder or any other Credit Party shall deny or disaffirm in writing such grantor's obligations under the Security Agreement or any other such Security Document; or

# 11.9 Judgments

One or more final judgments or decrees shall be entered against the Borrower or any Subsidiary (excluding any First Day Order or any order fixing the amount of any claim in the Chapter 11 Cases) involving a liability requiring the payment of \$10,000,000 or more in the aggregate for all such final judgments and decrees for the Borrower and the Subsidiaries (to the extent not paid or covered by indemnity or insurance provided by a carrier that has not denied coverage) (and in the case of such a judgment against any of the Debtors, such judgment arose post-petition) and any such final judgments or decrees shall not have been satisfied, vacated, discharged or stayed (including pursuant to the Bankruptcy Code) or bonded pending appeal within 30 consecutive days after the entry thereof; or

# 11.10 Change of Control

A Change of Control shall occur; or

# 11.11 Dismissal; Conversion

(i) Any of the Chapter 11 Cases of the Debtors shall be dismissed or converted to a case under chapter 7 of the Bankruptcy Code or any Debtors shall file a motion or other pleading seeking the dismissal of any of Chapter 11 Case of any Debtor under section 1112 of the Bankruptcy Code or otherwise without causing Payment in Full of all Obligations hereunder or (ii) a trustee under chapter 11 of the Bankruptcy Code or an examiner with enlarged powers (powers beyond those set forth in sections 1106(a)(3) and (4) of the Bankruptcy Code) under section 1104(b) of the Bankruptcy Code shall be appointed in any of the Chapter 11 Cases of the Debtors and the order appointing such trustee or examiner shall not be reversed or vacated within thirty (30) days after the entry thereof (or the Credit Parties or their Affiliates shall have acquiesced to the entry of such order) unless consented to by the Required Lenders; or

# 11.12 Superpriority Claims

Other than as permitted by the Orders, (i) an application shall be filed by any Debtor for the approval of (i) any other Superpriority Claim or Lien, or an order of the Bankruptcy Court shall be entered granting any other than Superpriority Claim or Lien (other than the Carve Out), in any of the Chapter 11 Cases of the Debtors that is pari passu with or senior to the claims (as such word is defined in the Bankruptcy Code) or Liens of the Administrative Agent, the Lenders and the other Secured Parties against the Borrower or any other Credit Party hereunder or under any of the other Credit Documents (including the adequate protection Liens and claims provided for in the Orders) or (ii) any Liens senior or pari passu with (A) the Liens under the Prepetition First Lien Debt or (B) the adequate protection Liens granted on account of the Primed Liens; or

# 11.13 Stay Relief

The Bankruptcy Court shall enter an order or orders granting relief from the automatic stay applicable under section 362 of the Bankruptcy Code to the holder or holders of any security interest to (i) permit foreclosure (or the granting of a deed in lieu of foreclosure or the like) on any assets of any of the Debtors which have a value in excess of \$1,000,000 in the aggregate or (ii) permit other actions that would have a Material Adverse Effect on the Debtors or their estates (taken as a whole); or

# 11.14 Orders; Actions

Subject to the last paragraph of Section 11 in all respects, (i) the Final Order Entry Date shall not have occurred by the date that is forty-five (45) days following the Petition Date; (ii) an order of the Bankruptcy Court shall be entered reversing, amending, supplementing, staying, vacating or otherwise amending, supplementing or modifying the Interim Order or the Final Order or Holdings, the Borrower or any Subsidiary of the Borrower shall apply for the authority to do so, in each case in a manner that is adverse in any respect to the Administrative Agent or the Lenders, without the prior written consent of the Administrative Agent and the Required Lenders; (iii) an order of the Bankruptcy Court shall be entered denying or terminating use of Cash

Collateral by the Credit Parties or imposing any additional conditions on such use (and such order remains unstayed for more than three (3) Business Days) and the Credit Parties shall have not obtained use of Cash Collateral pursuant to an order consented to by, and in form and substance reasonably acceptable to, the Required Lenders; (iv) the Interim Order (prior to the entry of the Final Order) or Final Order (at all times thereafter) shall cease to create a valid and perfected Lien on the Collateral described therein or the Final Order shall cease to be in full force and effect; (v) any of the Credit Parties or any Subsidiary of the Borrower shall fail to comply with the Orders in any material respect; (vi) other than with respect to the Carve Out (as provided for in Section 2.17), an order in the Chapter 11 Cases shall be entered (without the consent of the Administrative Agent and the Required Lenders) charging any of the Collateral under section 506(c) of the Bankruptcy Code against the Lenders; (vii) the Final Order shall not authorize the borrowing by the Borrower of the full amount of the Commitments provided for hereunder; (viii) the entry of an order in the Chapter 11 Cases authorizing the use Cash Collateral or obtaining of financing pursuant to section 364 of the Bankruptcy Code (other than the Term Facility or the DIP ABL Facility, if any), unless such financing would (and actually does) provide for Payment in Full of all Obligations upon the consummation thereof; (ix) any order shall be entered in the Chapter 11 Cases providing adequate protection, other than the Interim Order or Final Order (as applicable) or pursuant to any First Day Order or "second day order" or any other order reasonably acceptable to the Administrative Agent or as otherwise permitted pursuant to Section 11.12; or (x) the Borrower or any of its Subsidiaries shall take any action in support of the items referred to in the foregoing clauses (i)-(ix); or

# 11.15 Prepetition Payments

Except as permitted by the Orders, any Debtor shall make any Prepetition Payment other than Prepetition Payments (i) authorized by the Bankruptcy Court in accordance with the First Day Orders or "second day orders" of the Bankruptcy Court reasonably satisfactory to the Required Lenders or (ii) as set forth under the Approved Budget; or

# 11.16 [Reserved]

# 11.17 Reorganization Plan

A Reorganization Plan that is not (i) the Plan or (ii) a plan of reorganization that is otherwise satisfactory to the Required Akin Ad Hoc Group Lenders and the Required PW Ad Hoc Group Lenders in their sole discretion shall be confirmed in any of the Chapter 11 Cases of the Debtors, or any of the Credit Parties or any of their Subsidiaries shall file, propose, support or fail to contest in good faith the filing or confirmation of any such plan or entry of any such order; or

# 11.18 Supporting Actions

Any Credit Party or any of their Subsidiaries shall file any pleading with the Bankruptcy Court in support of any matter set forth in Sections 11.1, 11.12, 11.13, 11.14, 11.5 or 11.17; or

## 11.19 Sale Motions

The Borrower shall file (or fail to oppose) any motion seeking an order authorizing the sale of all or substantially all of the assets of the Credit Parties; or

#### 11.20 Termination Date.

The "Termination Date" as defined in and under the Restructuring Support Agreement shall have occurred with respect to all Persons party thereto; or

## 11.21 Adverse Claims

Any Person shall assert any claim in the Chapter 11 Cases arising under section 506(c) of the Bankruptcy Code against any Agent, any Lender or the Collateral or any agent or trustee under, or the collateral securing, the Prepetition First Lien Debt, and, solely with respect to such claim asserted by any Person other than a Credit Party or any Affiliate thereof, either (i) the same shall remain unopposed by the Borrower for more than 5 Business Days, or (ii) in any event, any such claim shall not be disallowed, dismissed or withdrawn, with prejudice, within 30 days after the assertion thereof; or if any Agent, Lenders or the Collateral are surcharged (other than with respect to the Carve Out) pursuant to sections 105, 506(c) or any other section of the Bankruptcy Code without the prior written consent of the Required Lenders;

#### 11.22 Adverse Orders

Solely with respect to the Term Facility, any order is entered by the Bankruptcy Court sustaining any objection or challenge of any kind or nature to the validity, priority, or amount of the Liens with created pursuant to the Credit Documents in favor of or claims held by, or an action to recharacterize or subordinate any Agent or any Lender;

then, and in any such event, and at any time thereafter, if any Event of Default shall then be continuing, the Administrative Agent shall, at the written request of the Required Lenders, by written notice to the Borrower, take any or all of the following actions (subject to the terms of the Orders), without prejudice to the rights of the Administrative Agent or any Lender to enforce its claims against the Borrower, except as otherwise specifically provided for in this Agreement: (i) declare the principal of and any accrued interest and Fees in respect of any or all Term Loans and any or all Obligations owing hereunder and under any other Credit Document to be, whereupon the same shall become, forthwith due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby waived by the Borrower; (ii) direct the Collateral Agent to enforce any and all Liens and security interests created pursuant to the Security Documents and (iii) enforce any and all of the Administrative Agent's rights under the Guarantee.

Notwithstanding anything to the contrary contained herein, any Event of Default under this Agreement or similarly defined term under any other Credit Document, other than any Event of Default which cannot be waived without the written consent of each Lender directly and adversely affected thereby, shall be deemed not to be "continuing" if the events, act or condition that gave rise to such Event of Default have been remedied or cured (including by payment, notice, taking of any action or omitting to take any action) or have ceased to exist and the Borrower is in compliance with this Agreement and/or such other Credit Document.

Upon an acceleration of the Loans following an Event of Default, the amount of principal of, and premium on (if any), the Loans that becomes due and payable shall include the Put Option Premium and the Exit Fee, determined as of such date. Such Put Option Premium and Exit Fee shall become immediately due and payable in cash by the Credit Parties and shall

constitute part of the Obligations, whether due to acceleration pursuant to the terms of this Agreement (in which case it shall be due immediately, either upon the giving of notice to Borrower), by operation of law or otherwise (including on account of any bankruptcy filing) as if the Loans were being voluntarily prepaid or repaid as of such date, in view of the impracticability and extreme difficulty of ascertaining actual damages and by mutual agreement of the parties as to a reasonable calculation of each Lender's lost profits as a result thereof. Any Put Option Premium and Exit Fee payable pursuant to this Agreement shall be presumed to be the liquidated damages sustained by each Lender as the result of the early repayment or prepayment of the Loans and each of the Borrower and the other Credit Parties agrees that it is reasonable under the circumstances currently existing. The Put Option Premium shall also become due and payable under this Agreement in the event the Obligations (and/or this Agreement) are satisfied or released by foreclosure (whether by power of judicial proceeding, deed in lieu of foreclosure or by any other means) or the Obligations are reinstated pursuant to section 1124 of the Bankruptcy Code. In the event the Put Option Premium or Exit Fee is determined not to be due and payable by order of any court of competent jurisdiction, including by operation of the Bankruptcy Code, despite such a triggering event having occurred, such Put Option Premium and the Exit Fee shall nonetheless constitute Obligations under this Agreement for all purposes hereunder. EACH OF THE BORROWER AND THE OTHER CREDIT PARTIES EXPRESSLY WAIVE (TO THE FULLEST EXTENT THEY MAY LAWFULLY DO SO) THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE OR LAW THAT PROHIBITS OR MAY PROHIBIT THE COLLECTION OF THE FOREGOING PUT OPTION PREMIUM OR THE EXIT FEE IN CONNECTION WITH ANY SUCH ACCELERATION. Each of the Borrower and the other Credit Parties expressly agrees (to the fullest extent it may lawfully do so) that: (A) the Put Option Premium and the Exit Fee are reasonable and the product of an arm's-length transaction between sophisticated business people, ably represented by counsel; (B) the Put Option Premium and the Exit Fee shall be payable notwithstanding the then prevailing market rates at the time payment or redemption is made; (C) there has been a course of conduct between Lenders, the Borrower and the other Credit Parties giving specific consideration in this transaction for such agreement to pay the Put Option Premium and the Exit Fee and (D) the Borrower and the other Credit Parties shall be estopped hereafter from claiming differently than as agreed to in this paragraph. Each of the Borrower and the Guarantors expressly acknowledges that its agreement to pay or guarantee the payment of the Put Option Premium and the Exit Fee to the Lenders as herein described is a material inducement to Lenders to make the Loans.

# 11.23 Application of Proceeds

Subject to the Orders and the DIP ABL Intercreditor Agreement (if entered into), any amount received by the Administrative Agent or the Collateral Agent from any Credit Party (or from proceeds of any Collateral) following any acceleration of the Obligations under this Agreement or any Event of Default under Section 11.5 shall be applied in the following order, but subject and subordinate to the Carve Out in all respects, including with respect to amounts received to be funded into the Carve Out Reserve:

(a) *First*, to payment of that portion of the Obligations constituting fees, indemnities, expenses and other amounts (other than principal and interest, but including amounts payable under Section 2.10 and Section 13.5) payable to the Agents in their capacities as such;

- (b) Second, to payment of that portion of the Obligations constituting fees, indemnities and other amounts (other than amounts set forth below) payable to the Lenders (in their capacities as such) (including amounts payable under Section 2.10 and Section 13.5), ratably among them in proportion to the amounts described in this clause Second payable to them;
- (c) *Third*, to payment of that portion of the Obligations constituting accrued and unpaid interest on the Loans and other Obligations, ratably among the Administrative Agent and Lenders in proportion to the respective amounts described in this clause *Third* payable to them;
- (d) Fourth, to the payment of any remaining Obligations of the Credit Parties that are due and payable to the Administrative Agent and the other Secured Parties on such date, ratably based upon the respective aggregate amounts of all such Obligations owing to the Administrative Agent and such other Secured Parties on such date; and
- (e) *Fifth*, the balance, if any, after all of the Obligations have been indefeasibly Paid in Full, to the Borrower or as otherwise required by Applicable Law.

Notwithstanding the foregoing, the application to the Obligations of amounts received in respect of the Collateral is expressly subject to the priorities set forth in the Interim Order (and, when entered, the Final Order), and all such amounts shall first be allocated in accordance with such priorities.

For the avoidance of doubt and notwithstanding anything to the contrary herein or in any intercreditor agreement, the Credit Documents, or in any of the documents evidencing Prepetition Debt, the Carve Out shall be senior to all Liens and claims securing the Term Facility, and any and all other forms of adequate protection, liens or claims securing the Term Facility or the obligations under the Prepetition First Lien Debt, including the Primed Liens.

## **SECTION 12 The Agents**

## 12.1 Appointment

(a) Each Secured Party (other than the Administrative Agent) hereby irrevocably designates and appoints the Administrative Agent as the agent of such Secured Party under this Agreement and the other Credit Documents and irrevocably authorizes the Administrative Agent, in such capacity, to take such action on its behalf under the provisions of this Agreement and the other Credit Documents and to exercise such powers and perform such duties as are expressly delegated to the Administrative Agent by the terms of this Agreement and the other Credit Documents, together with such other powers as are reasonably incidental thereto. The provisions of this Section 12 (other than this Section 12.1 and Sections 12.2, 12.9, 12.12 and 12.13, in each case, with respect to the Borrower) are solely for the benefit of the Agents and the other Secured Parties, and the Credit Parties shall not have any rights as a third party beneficiary of such provision. Notwithstanding any provision to the contrary elsewhere in this Agreement, no Agent shall have any duties or responsibilities, except those expressly set forth herein or in any other Credit Document, any fiduciary relationship with any other Secured Party or any agency or trust obligations with respect to any Credit Party or any other Secured Party, and no implied

covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Credit Document or otherwise exist against such Agent.

(b) The Secured Parties hereby irrevocably designate and appoint the Collateral Agent as the agent with respect to the Collateral, and each of the Secured Parties hereby irrevocably authorizes the Collateral Agent, in such capacity, to take such action on its behalf under the provisions of this Agreement and the other Credit Documents and to exercise such powers and perform such duties as are expressly delegated to the Collateral Agent by the terms of this Agreement and the other Credit Documents, together with such other powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary elsewhere in this Agreement, the Collateral Agent shall have no duties or responsibilities except those expressly set forth herein or in any other Credit Document, any fiduciary relationship with any of the other Secured Parties or any agency or trust obligations with respect to any Credit Party or any other Secured Party, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Credit Document or otherwise exist against the Collateral Agent. For the avoidance of doubt, any references to the Administrative Agent herein in respect of the grant or holding of Liens in Collateral or in respect of indemnity, reimbursement, and set-off rights shall include the Collateral Agent.

# 12.2 Delegation of Duties

The Administrative Agent and the Collateral Agent may each execute any of its duties under this Agreement and the other Credit Documents by or through agents, sub-agents, employees or attorneys-in-fact (which shall be entitled to all rights, privileges, protections, immunities, and indemnities granted to the Agents under the Credit Documents) and shall be entitled to advice of counsel concerning all matters pertaining to such duties. Neither the Administrative Agent nor the Collateral Agent shall be responsible for the negligence or misconduct of any agents, sub-agents or attorneys-in-fact selected by it in the absence of gross negligence or willful misconduct by such Agent in selecting such agents, sub-agents or attorneys-in-fact (as determined in the final judgment of a court of competent jurisdiction).

# 12.3 Exculpatory Provisions

(a) No Agent nor any of its Related Parties shall be (a) liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Credit Document (except for its or such Person's own gross negligence or willful misconduct, as determined in the final judgment of a court of competent jurisdiction, in connection with its duties expressly set forth herein, (b) liable in any circumstance for any action taken or omitted to be taken by any of them with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as such Agent shall believe in good faith shall be necessary), (c) responsible in any manner to any of the Lenders or any participant for any recitals, statements, representations or warranties made by any of Holdings, the Borrower, any other Guarantor, any other Credit Party or any officer thereof contained in this Agreement or any other Credit Document or in any certificate, report, statement or other document referred to or provided for in, or received by such Agent under or in connection with, this Agreement or any other Credit Document or for the value, validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any other Credit Document, or the preservation,

perfection, maintenance or continuation of perfection, or priority of any Lien or security interest created or purported to be created under the Security Documents, or for any failure of Holdings, the Borrower, any other Guarantor or any other Credit Party to perform its obligations hereunder or thereunder. No Agent shall be under any obligation to any other Secured Party to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Credit Document, or to inspect the properties, books or records of any Credit Party or any Affiliate thereof.

- Nothing in this Agreement or any other Credit Document shall require any Agent or its Related Parties to expend or risk their own funds or otherwise incur any financial liability in the performance of any duties, obligations or responsibilities or in the exercise of any right, power, authority or discretion hereunder or under the other Credit Documents. No Agent shall be responsible or liable for any failure or delay in the performance of its obligations under this Agreement or any other Credit Document, in each case, arising out of or caused, directly or indirectly, by circumstances beyond its control. With respect to any approval, determination, designation, or judgment to be made by any Agent herein or in the other Credit Documents, such Agent shall be entitled to request that the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as such Agent shall believe in good faith shall be necessary) make or confirm such approval, determination, designation, or judgment. The Administrative Agent does not warrant or accept responsibility for, and shall not have any liability with respect to the continuation of, administration of, submission of, calculation of or any other matter related to the ABR, the Term SOFR Reference Rate, Adjusted Term SOFR or Term SOFR, or any component definition thereof or rates referred to in the definition thereof, any alternative, successor or replacement rate thereto (including any Benchmark Replacement) or any relevant adjustments thereto.
- (c) Each Lender confirms to the Administrative Agent, the Collateral Agent, each other Lender and each of their respective Related Parties that it (i) possesses (individually or through its Related Parties) such knowledge and experience in financial and business matters that it is capable, without reliance on the Administrative Agent, the Collateral Agent, any other Lender or any of their respective Related Parties, of evaluating the merits and risks (including tax, legal, regulatory, credit, accounting and other financial matters) of (x) entering into this Agreement, (y) making Term Loans and other extensions of credit hereunder and under the other Credit Documents and (z) in taking or not taking actions hereunder and thereunder, (ii) is financially able to bear such risks and (iii) has determined that entering into this Agreement and making Term Loans and other extensions of credit hereunder and under the other Credit Documents is suitable and appropriate for it.
- (d) Each Lender acknowledges that (i) it is solely responsible for making its own independent appraisal and investigation of all risks arising under or in connection with this Agreement and the other Credit Documents, (ii) that it has, independently and without reliance upon the Administrative Agent, the Collateral Agent, any other Lender or any of their respective Related Parties, made its own appraisal and investigation of all risks associated with, and its own credit analysis and decision to enter into, this Agreement based on such documents and information, as it has deemed appropriate and (iii) it will, independently and without reliance upon the Administrative Agent, the Collateral Agent, any other Lender or any of their respective Related Parties, continue to be solely responsible for making its own appraisal and investigation of all risks

124

arising under or in connection with, and its own credit analysis and decision to take or not take action under, this Agreement and the other Credit Documents based on such documents and information as it shall from time to time deem appropriate, which may include, in each case:

- (i) the financial condition, status and capitalization of the Borrower and each other Credit Party;
- (ii) the legality, validity, effectiveness, adequacy or enforceability of this Agreement and each other Credit Document and any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Credit Document;
- (iii) determining compliance or non-compliance with any condition hereunder to the making of a Term Loan and the form and substance of all evidence delivered in connection with establishing the satisfaction of each such condition; and
- (iv) the adequacy, accuracy and/or completeness of any information delivered by the Administrative Agent, the Collateral Agent, any other Lender or by any of their respective Related Parties under or in connection with this Agreement or any other Credit Document, the transactions contemplated hereby and thereby or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Credit Document.

# 12.4 Reliance by Agents

The Administrative Agent and the Collateral Agent shall be entitled to rely, and shall be fully protected in relying, upon any writing, resolution, notice, consent, certificate, affidavit, letter, telecopy, telex, electronic mail, or teletype message, statement, order or other document or instruction believed by it in good faith to be genuine and correct and to have been signed, sent or made by the proper Person or Persons and upon advice and statements of legal counsel (including counsel to Holdings and/or the Borrower), independent accountants and other experts selected by the Administrative Agent or the Collateral Agent. The Administrative Agent may deem and treat the Lender specified in the Register with respect to any amount owing hereunder as the owner thereof for all purposes unless a written notice of assignment, negotiation or transfer thereof shall have been filed with the Administrative Agent. The Administrative Agent and the Collateral Agent shall be fully justified in failing or refusing to take any action under this Agreement or any other Credit Document unless it shall first receive such advice or concurrence of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as such Agent shall believe in good faith shall be necessary) as it deems appropriate; provided, that it shall first be indemnified to its satisfaction by the Lenders against any and all liability and expense that may be incurred by it by reason of taking or continuing to take any such action. The Administrative Agent and the Collateral Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement and the other Credit Documents in accordance with a request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as such Agent shall believe in good faith shall be necessary), and such request and any action taken or failure to act pursuant thereto shall be binding upon all the Lenders and all future holders of the Term Loans; provided that none of the Administrative Agent or the Collateral

Agent shall be required to take any action that, in its opinion or in the opinion of its counsel, may expose it to liability or that is contrary to any Credit Document or Applicable Law. In addition, the Lenders hereby authorize Akin and Paul, Weiss to provide joint written directions (which may be made by electronic mail) to each Agent on behalf of the Required Lenders in regards to the Credit Documents and such other matters until such time as the Required Lenders or Akin and Paul, Weiss advise the Agents in writing that Akin and Paul, Weiss are no longer authorized to provide any joint written directions to the Agents on behalf of the Required Lenders. The giving of any such joint direction by Akin and Paul, Weiss shall be deemed a reaffirmation by Akin and Paul, Weiss that such authorization from the Required Lenders has been duly given. The Administrative Agent and Collateral Agent may rely on and act upon any such joint direction given by Akin and Paul, Weiss and need not inquire as to the due authorization thereof. Notwithstanding the foregoing, to the extent any Agent determines that any joint direction from Akin and Paul, Weiss requires any certification or supplementation, such Agent shall promptly inform the Required Lenders or Akin and Paul, Weiss, as applicable, of such determination and the Required Lenders or Akin and Paul, Weiss, as applicable, shall promptly provide to such Agent one or more additional clarifying or supplementing directions. Until such time as such Agent receives such additional direction or directions from the Required Lenders or Akin and Paul, Weiss, as applicable, such Agent shall be under no duty or obligation to take, and may refrain from taking, any course of action for which such Agent has requested additional direction.

#### 12.5 Notice of Default

Neither the Administrative Agent nor the Collateral Agent shall be deemed to have knowledge or notice of the occurrence of any Default or Event of Default hereunder unless the Administrative Agent or the Collateral Agent, as applicable, has received written notice from a Lender, Holdings or the Borrower referring to this Agreement, describing such Default or Event of Default and stating that such notice is a "notice of default". In the event that the Administrative Agent or the Collateral Agent receives such a notice, it shall give notice thereof to the Lenders, the Administrative Agent or the Collateral Agent, as applicable. The Administrative Agent and the Collateral Agent shall take such action with respect to such Default or Event of Default as shall be reasonably directed by the Required Lenders (subject to the protections of Section 12.3); provided that unless and until the Administrative Agent or the Collateral Agent, as applicable, shall have received such directions, the Administrative Agent or the Collateral Agent, as applicable, may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as is within its authority to take under this Agreement and otherwise as it shall deem advisable in the best interests of the Lenders except to the extent that this Agreement requires that such action be taken only with the approval of the Required Lenders or each of the Lenders, as applicable.

# 12.6 Non-Reliance on Administrative Agent, Collateral Agent and Other Lenders

Each Lender expressly acknowledges that none of the Administrative Agent, the Collateral Agent or any of their respective officers, directors, employees, agents, attorneys-in-fact or Affiliates has made any representations or warranties to it and that no act by the Administrative Agent or the Collateral Agent hereinafter taken, including any review of the affairs of Holdings, the Borrower, any other Guarantor or any other Credit Party, shall be deemed to constitute any

representation or warranty by the Administrative Agent or the Collateral Agent to any Lender. Each Lender represents to Administrative Agent and the Collateral Agent that it has, independently and without reliance upon the Administrative Agent, Collateral Agent or any other Lender, and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, operations, property, financial and other condition and creditworthiness of Holdings, the Borrower, each other Guarantor and each other Credit Party and made its own decision to make its Term Loans hereunder and enter into this Agreement. Each Lender also represents that it will, independently and without reliance upon the Administrative Agent, Collateral Agent or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Credit Documents, and to make such investigation as it deems necessary to inform itself as to the business, operations, property, financial and other condition and creditworthiness of Holdings, the Borrower, each other Guarantor and each other Credit Party. Except for notices, reports and other documents expressly required to be furnished to the Lenders by the Administrative Agent hereunder, none of the Administrative Agent or the Collateral Agent shall have any duty or responsibility to provide any Lender with any credit or other information concerning the business, assets, operations, properties, financial condition, prospects or creditworthiness of Holdings, the Borrower, any other Guarantor or any other Credit Party that may come into the possession of the Administrative Agent, the Collateral Agent or any of their respective officers, directors, employees, agents, attorneys-in-fact or Affiliates.

## 12.7 Indemnification

The Lenders agree to indemnify each Agent and its Related Parties, each in its capacity as such (to the extent not reimbursed by the Credit Parties and without limiting the obligation of the Credit Parties to do so), ratably according to their respective principal portions of the Term Loans in effect on the date on which indemnification is sought (or, if indemnification is sought after the date upon which the Term Loans shall have been paid in full, ratably in accordance with their respective principal portions of the Term Loans in effect immediately prior to such date), from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, fees, costs, expenses or disbursements of any kind whatsoever that may at any time occur (including at any time following the payment of the Term Loans) be imposed on, incurred by or asserted against such Agent or Related Party, including all fees, disbursements and other charges of counsel to the extent required to be reimbursed by the Credit Parties pursuant to Section 13.5, in any way relating to or arising out of the making of the Term Loans, this Agreement, any of the other Credit Documents or any documents contemplated by or referred to herein or therein or the transactions contemplated hereby or thereby or any action taken or omitted by such Agent or Related Party under or in connection with any of the foregoing (SUBJECT TO THE PROVISOS BELOW, WHETHER OR NOT CAUSED BY OR ARISING IN WHOLE OR IN PART, OUT OF THE COMPARATIVE, CONTRIBUTORY OR SOLE ORDINARY **NEGLIGENCE OF THE INDEMNIFIED PERSON**); provided that no Lender shall be liable to any Agent for the payment of any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, fees, costs, expenses or disbursements resulting from such Agent's gross negligence or willful misconduct as determined by a final and non-appealable judgment of a court of competent jurisdiction; provided, further, that no action taken in accordance with the directions of the Required Lenders (or such other number or percentage of the Lenders as

shall be required by the Credit Documents) shall be deemed to constitute gross negligence or willful misconduct for purposes of this Section 12.7. In the case of any investigation, litigation or proceeding giving rise to any liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind whatsoever that may at any time occur, be imposed upon, incurred by or asserted against any such indemnified Person in any way relating to or arising out of the making of the Term Loans, this Agreement, any of the other Credit Documents or any documents contemplated by or referred to herein or therein or the transactions contemplated hereby or thereby or any action taken or omitted by such indemnified Person under or in connection with any of the foregoing (including at any time following the payment of the Term Loans), this Section 12.7 applies whether any such investigation, litigation or proceeding is brought by any Lender or any other Person. Without limitation of the foregoing, each Lender shall reimburse such Agent upon demand for its ratable share of any costs or out-of-pocket expenses (including attorneys' fees) incurred by such Agent in connection with the preparation, execution, delivery, administration, modification, amendment or enforcement (whether through negotiations, legal proceedings or otherwise) of, or legal advice rendered in respect of rights or responsibilities under, this Agreement, any other Credit Document, or any document contemplated by or referred to herein, to the extent that such Agent is not reimbursed for such expenses by or on behalf of the Borrower; provided that such reimbursement by the Lenders shall not affect the Borrower's continuing reimbursement obligations with respect thereto. If any indemnity furnished to any Agent for any purpose shall, in the opinion of such Agent, be insufficient or become impaired, such Agent may call for additional indemnity and cease, or not commence, to do the acts indemnified against until such additional indemnity is furnished; provided that in no event shall this sentence require any Lender to indemnify any Agent against any liability, obligation, loss, damage, penalty, action, judgment, suit, cost, expense or disbursement in excess of such Lender's pro rata portion thereof; and provided, further, this sentence shall not be deemed to require any Lender to indemnify any Agent against any liability, obligation, loss, damage, penalty, action, judgment, suit, cost, expense or disbursement resulting from such Agent's gross negligence or willful misconduct (as determined by a final and non-appealable judgment of court of competent jurisdiction).

# 12.8 Agents in their Individual Capacities

Each Agent and its Affiliates may make loans to, accept deposits from and generally engage in any kind of business with Holdings, the Borrower, any other Guarantor, and any other Credit Party as though such Agent were not an Agent hereunder and under the other Credit Documents. With respect to the Term Loans made by it, each Agent shall have the same rights and powers under this Agreement and the other Credit Documents as any Lender and may exercise the same as though it were not an Agent, and the terms "Lender" and "Lenders" shall include each Agent in its individual capacity.

## 12.9 Successor Agents

Each of the Administrative Agent and Collateral Agent may resign at any time by notifying the other Agent, the Lenders and the Borrower. Upon receipt of any such notice of resignation, the Required Lenders shall have the right, subject to the consent of the Borrower (not to be unreasonably withheld or delayed) so long as no Event of Default has occurred and is continuing, to appoint a successor, which shall be a bank with an office in the United States, or an

Affiliate of any such bank with an office in the United States. If no such successor shall have been so appointed by the Required Lenders and shall have accepted such appointment within 30 days after the retiring Agent gives notice of its resignation, then the retiring Agent may on behalf of the Lenders appoint a successor Agent meeting the qualifications set forth above (including receipt of the Borrower's consent); provided that if such Agent shall notify the Borrower and the Lenders that no qualifying Person (including as a result of the absence of consent of the Borrower) has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (x) the retiring Agent shall be discharged from its duties and obligations hereunder and under the other Credit Documents (except that in the case of any collateral security held by the Collateral Agent on behalf of the Secured Parties under any of the Credit Documents, the retiring Collateral Agent shall continue to hold such collateral security until such time as a successor Collateral Agent is appointed) and (y) all payments, communications and determinations provided to be made by, to or through such Agent shall instead be made by or to each Lender directly, until such time as the Required Lenders with (except after the occurrence and during the continuation of an Event of Default) the consent of the Borrower (not to be unreasonably withheld) appoint successor Agents as provided for above in this paragraph. Upon the acceptance of a successor's appointment as the Administrative Agent or Collateral Agent, as the case may be, hereunder, and upon the execution and filing or recording of such financing statements, or amendments thereto, and such amendments or supplements to the Mortgages, and such other instruments or notices, as may be necessary or desirable, or as the Required Lenders may request, in order to continue the perfection of the Liens granted or purported to be granted by the Security Documents, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Agent, and the retiring Agent shall be discharged from all of its duties and obligations hereunder or under the other Credit Documents (if not already discharged therefrom as provided above in this Section). The fees payable by the Borrower (following the effectiveness of such appointment) to such Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Agent's resignation hereunder and under the other Credit Documents, the provisions of this Section 12 (including Section 12.7) and Section 13.5 shall continue in effect for the benefit of such retiring Agent, its sub-agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Agent was acting as an Agent.

# 12.10 Withholding Tax

To the extent required by any Applicable Law, the Administrative Agent may withhold from any interest payment to any Lender an amount equivalent to any applicable withholding Tax. If the Internal Revenue Service or any authority of the United States or other jurisdiction asserts a claim that the Administrative Agent did not properly withhold Tax from amounts paid to or for the account of any Lender (because the appropriate form was not delivered, was not properly executed, or because such Lender failed to notify the Administrative Agent or of a change in circumstances that rendered the exemption from, or reduction of, withholding Tax ineffective, or for any other reason), such Lender shall indemnify the Administrative Agent (to the extent that the Administrative Agent has not already been reimbursed by the Borrower (solely to the extent required by this Agreement) and without limiting the obligation of the Borrower to do so) fully for all amounts paid, directly or indirectly, by the Administrative Agent as Tax or otherwise, including penalties and interest, together with all expenses incurred, including legal

expenses, allocated staff costs and any out of pocket expenses. A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent shall be conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any amounts at any time owing to such Lender under this Agreement or any other Credit Document against any amount due the Administrative Agent under this Section 12.10.

# 12.11 Administrative Agent May File Proofs of Claim

In case of the pendency of any receivership, insolvency, liquidation, bankruptcy, reorganization, arrangement, adjustment, composition or other judicial proceeding relative to any Credit Party, the Administrative Agent (irrespective of whether the principal of any Term Loan shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand on the Borrower) shall be entitled and empowered, by intervention in such proceeding or otherwise:

- (a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Term Loans and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Secured Parties (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders and the Administrative Agent and their respective agents and counsel and all other amounts due the Lenders and the Administrative Agent hereunder) allowed in such judicial proceeding; and
- (b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same;

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Secured Party to make such payments to the Administrative Agent and, in the event that the Administrative Agent shall consent to the making of such payments directly to the Secured Parties, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Agents and their respective agents and counsel, and any other amounts due the Administrative Agent under Sections 4.1 and 13.5.

Nothing contained herein shall be deemed to authorize the Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Secured Party or to authorize the Administrative Agent to vote in respect of the claim of any Secured Party in any such proceeding.

# 12.12 Intercreditor Agreement

Each of the Collateral Agent and the Administrative Agent is hereby authorized to enter into the DIP ABL Intercreditor Agreement contemplated hereby at the direction of the Required Lenders, and the parties hereto acknowledge that the DIP ABL Intercreditor Agreement to which the Collateral Agent and/or the Administrative Agent is a party are each binding upon them. Each Secured Party (a) hereby agrees that it will be bound by and will take no actions contrary to the provisions of the DIP ABL Intercreditor Agreement and (b) hereby authorizes and

instructs the Collateral Agent and the Administrative Agent to enter into the DIP ABL Intercreditor Agreement and to subject the Liens on the Collateral securing the Obligations to the provisions thereof.

# 12.13 Security Documents and Guarantee; Agents under Security Documents and Guarantee

(a) Each Secured Party hereby further authorizes the Administrative Agent or the Collateral Agent, as applicable, on behalf of and for the benefit of the Secured Parties, to be the agent for and representative of the Secured Parties with respect to the Guarantee, the Collateral and the Security Documents, as applicable. Subject to Section 13.1, without further written consent or authorization from any Secured Party, the Administrative Agent or the Collateral Agent, as applicable, may execute any documents or instruments necessary to subordinate any Lien on any property granted to or held by the Administrative Agent or the Collateral Agent under any Credit Document to the holder of any Lien permitted under clause (d) Section 10.2. The Secured Parties hereby irrevocably agree that the Liens granted to the Collateral Agent by the Credit Parties on any Collateral shall be automatically released (i) upon the termination of this Agreement and the Payment in Full of all Obligations hereunder, (ii) upon the sale or other Disposition of such Collateral (including as part of or in connection with any other sale or other Disposition permitted hereunder) to any Person other than another Credit Party, to the extent such sale or other Disposition is made in compliance with the terms of this Agreement (and the Collateral Agent may rely conclusively on a certificate to that effect provided to it by any Credit Party upon its reasonable request without further inquiry), (iii) to the extent such Collateral is comprised of property leased to a Credit Party, upon termination or expiration of such lease, (iv) if the release of such Lien is approved, authorized or ratified in writing by the Required Lenders (or such other percentage of the Lenders whose consent may be required in accordance with this Section 13.1), (v) to the extent the property constituting such Collateral is owned by any Guarantor, upon the release of such Guarantor from its obligations under the Guarantee, (vi) as required to effect any sale or other Disposition of Collateral in connection with any exercise of remedies of the Collateral Agent pursuant to the Security Documents and (vii) if such assets constitute Excluded Collateral. Any such release shall not in any manner discharge, affect or impair the Obligations or any Liens (other than those being released) upon (or obligations (other than those being released) of the Credit Parties in respect of) all interests retained by the Credit Parties, including the proceeds of any sale, all of which shall continue to constitute part of the Collateral except to the extent otherwise released in accordance with the provisions of the Credit Documents. Additionally, the Secured Parties hereby irrevocably agree that the Subsidiary Guarantors shall be automatically released from the Guarantee upon consummation of any transaction resulting in such Subsidiary ceasing to constitute a Subsidiary. Upon request by any Agent at any time, the Required Lenders (or such other percentage of the Lenders whose consent may be required in accordance with Section 13.1) shall confirm in writing such Agent's authority to release any Collateral or Guarantee pursuant to this Agreement or any other Credit Document. The Lenders hereby authorize the Administrative Agent and the Collateral Agent, as applicable, and the Administrative Agent and the Collateral Agent agree to execute and deliver any instruments, documents, and agreements necessary or desirable or reasonably requested by the Borrower to evidence and confirm the release of any Guarantor or Collateral pursuant to the foregoing provisions of this paragraph, all without the further consent or joinder of any Lender; provided, that (1) no Agent shall be required to execute any document or take any action to evidence such release on terms that, in such Agent's opinion

or the opinion of its counsel, could expose such Agent to liability or create any obligation or entail any consequence other than the such release without recourse to, or representation, or warranty by such Agent, and (2) the Credit Parties shall have provided such Agent with such certifications or documents as such Agent shall reasonably request in order to demonstrate that the requested release is permitted under this Section 12.13(a).

Right to Realize on Collateral and Enforce Guarantee. Anything contained in any of the Credit Documents to the contrary notwithstanding, Holdings, the Borrower, the Agents and each Secured Party hereby agree that (i) no Secured Party shall have any right individually to realize upon any of the Collateral or to enforce the Guarantee, it being understood and agreed that all powers, rights and remedies hereunder and under the Guarantee may be exercised solely by the Administrative Agent, on behalf of the Secured Parties in accordance with the terms hereof and thereof and all powers, rights and remedies under the Security Documents may be exercised solely by the Collateral Agent on behalf of the Secured Parties, and (ii) in the event of a foreclosure by the Collateral Agent on any of the Collateral pursuant to a public or private sale or other Disposition, the Collateral Agent or any Secured Party may be the purchaser or licensor of any or all of such Collateral at any such sale or other Disposition and the Collateral Agent, as agent for and representative of the Secured Parties (but not any Lender or Lenders in its or their respective individual capacities unless Required Lenders shall otherwise agree in writing) shall be entitled, for the purpose of bidding and making settlement or payment of the purchase price for all or any portion of the Collateral sold at any such public sale, to use and apply any of the Obligations as a credit on account of the purchase price for any collateral payable by the Collateral Agent at such sale or other Disposition.

## 12.14 Survival

The agreements in this Section 12 shall survive the resignation or replacement of any Agent or any assignment of rights by, or the replacement of, a Lender, the termination of this Agreement and the Commitments and the payment, satisfaction, or discharge of the Term Loans and all other Obligations payable hereunder and under any other Credit Document.

#### **SECTION 13 Miscellaneous**

## 13.1 Amendments, Waivers and Releases

Except as otherwise expressly set forth in the Credit Documents (including Section 2.10(e)), neither this Agreement nor any other Credit Document (other than the Orders), nor any terms hereof or thereof, may be amended, supplemented or modified except in accordance with the provisions of this Section 13.1. The Required Lenders may, or, with the written consent of the Required Lenders, the Administrative Agent and/or the Collateral Agent may, from time to time, (a) enter into with the relevant Credit Party or Credit Parties written amendments, supplements or modifications hereto and to the other Credit Documents for the purpose of adding any provisions to this Agreement or the other Credit Documents or changing in any manner the rights of the Lenders or of the Credit Parties hereunder or thereunder or (b) waive in writing, on such terms and conditions as the Required Lenders or the Administrative Agent and/or Collateral Agent, as the case may be, may specify in such instrument, any of the requirements of this Agreement or the other Credit Documents or any Default or Event of Default and its consequences; *provided*,

however, that each such waiver and each such amendment, supplement or modification shall be effective only in the specific instance and for the specific purpose for which given; and provided, further, that no such waiver and no such amendment, supplement or modification shall:

- forgive or reduce any portion of any Term Loan or extend the final (i) scheduled maturity date of any Term Loan or reduce the stated rate, or forgive any portion thereof, or extend the date for the payment of any principal, any interest or Fee payable hereunder (other than as a result of waiving the applicability of any post-default increase in interest rates), or extend the final expiration date of any Lender's Commitment, or increase the aggregate amount of the Commitments of any Lender, in each case without the written consent of each Lender directly and adversely affected thereby; provided that, in each case for purposes of this clause (i), a waiver of any condition precedent in Section 6 of this Agreement, the waiver of any Default, Event of Default, default interest, mandatory prepayment or reductions, any modification, waiver or amendment to the financial definitions or financial ratios or any component thereof or the waiver of any other covenant shall not constitute an increase of any Commitment of a Lender, a reduction or forgiveness of any portion of any Term Loan or in the interest rates or the fees or premiums or a postponement of any date scheduled for the payment of principal or interest or an extension of the final maturity of any Term Loan, or the scheduled termination date of any Commitment; or
- (ii) (x) reduce the percentages specified in the definition of the term "Required Lenders" without the consent of each Lender, or (y) consent to the assignment or transfer by Holdings or the Borrower of their respective rights and obligations under any Credit Document to which it is a party (except as permitted pursuant to Section 10.3 or as contemplated by the definition of "Holdings"), alter the order of application set forth in Section 5.2(c) during the continuance of an Event of Default or Section 11.23 or change Section 13.8 or any other provision requiring pro rata sharing among the Lenders, in each case of this clause (y) without the written consent of each Lender directly and adversely affected thereby, or
- (iii) amend, modify or waive any provision of Section 12 or any other provision that affects the rights or duties of, or any fees or other amounts payable to, the Agents under this Agreement or any other Credit Document without the written consent of the then-current Administrative Agent and Collateral Agent in a manner that directly and adversely affects such Person; *provided* that only the consent of the parties to the Agent Fee Letter shall be required to amend, modify or supplement the terms thereof, or
- (iv) release all or substantially all of the value of the Guarantors under the Guarantee (except as expressly permitted by the Guarantee or this Agreement) or release all or substantially all of the Collateral under the Security Documents (except as expressly permitted by the Security Documents or this Agreement), in either case without the prior written consent of each Lender, or
- (v) modify or waive any provision of this Agreement in order to permit the incurrence of any financing pursuant to section 364 of the Bankruptcy Code (other than the Term Facility in the maximum amounts permitted after giving effect to the Final Order

and the DIP ABL Facility) that would be secured by the Collateral (or any portion thereof) on a pari passu or senior basis with the Obligations or that would benefit from any Superpriority Claim in the Chapter 11 Cases that is pari passu or senior to the Superpriority Claims with respect to the Obligations as provided in the Orders, without the written consent of each Lender; or

(vi) modify or waive any provision of Section 6.15, Section 7.10(c), or Section 9.16(b) without the prior written consent of the Required PW Ad Hoc Group Lenders; *provided* that any such modification or waiver shall not require the consent of any other Lender.

Any such waiver and any such amendment, supplement or modification shall apply equally to each of the affected Lenders and shall be binding upon Holdings, the Borrower, the applicable Credit Parties, such Lenders, the Administrative Agent and all future holders of the affected Term Loans. In the case of any waiver, Holdings, the Borrower, the applicable Credit Parties, the Lenders, the Administrative Agent shall be restored to their former positions and rights hereunder and under the other Credit Documents, and any Default or Event of Default waived shall be deemed to be cured and not continuing, it being understood that no such waiver shall extend to any subsequent or other Default or Event of Default or impair any right consequent thereon. In connection with the foregoing provisions, the Administrative Agent may, but shall have no obligations to, with the concurrence of any Lender, execute amendments, modifications, waivers or consents on behalf of such Lender.

Notwithstanding anything to the contrary herein, no Defaulting Lender shall have any right to approve or disapprove any amendment, modification, supplement, waiver or consent hereunder, except that the Commitment of such Lender may not be increased or extended without the consent of such Lender (it being understood that any Commitments or Term Loans held or deemed held by any Defaulting Lender shall be excluded for a vote of the Lenders hereunder requiring any consent of the Lenders, except as expressly provided for by this Agreement).

Notwithstanding the foregoing, this Agreement may be amended (or amended and restated) with the written consent of the Required Lenders, the Administrative Agent, Holdings and the Borrower (a) to add one or more additional credit facilities to this Agreement and to permit the extensions of credit from time to time outstanding thereunder and the accrued interest and fees in respect thereof to share ratably in the benefits of this Agreement and the other Credit Documents with the Term Loans and Commitments and the accrued interest and Fees in respect thereof and (b) to include appropriately the Lenders holding such credit facilities in any determination of the Required Lenders and other definitions related to such new Term Loans and Commitments.

Notwithstanding anything herein to the contrary, the Credit Documents may be amended to (i) add syndication or documentation agents and make customary changes and references related thereto and (ii) if applicable, add or modify "parallel debt" language in any jurisdiction in favor of the Collateral Agent or add sub-agents, in each case under clauses (i) and (ii), with the consent of only the Borrower and the Administrative Agent, and in the case of clause (ii), the Collateral Agent.

Notwithstanding anything in this Agreement (including, without limitation, this Section 13.1) or any other Credit Document to the contrary, (i) [reserved]; (ii) any provision of this Agreement or any other Credit Document (including, for the avoidance of doubt, any exhibit, schedule or other attachment to any Credit Document) may be amended by an agreement in writing entered into by the Borrower and the Administrative Agent (A) to cure any ambiguity, omission, mistake, defect or inconsistency (as reasonably determined by the Administrative Agent and the Borrower), (B) to effect administrative changes of a technical or immaterial nature (as reasonably determined by the Administrative Agent and the Borrower), (C) to correct incorrect crossreferences or similar inaccuracies, and (D) to add benefit to all of the Term Loans if adding such benefit is a condition to the incurrence of any Indebtedness permitted to be incurred under the Credit Documents; provided that in the case of clauses (A) and (B) above, the Lenders shall have received at least five Business Days' prior written notice thereof and the Administrative Agent shall not have received, within five Business Days of the date of such notice to the Lenders, a written notice from the Required Lenders stating that the Required Lenders object to such amendment; (iii) guarantees, collateral documents and related documents executed by the Credit Parties in connection with this Agreement may be in a form reasonably determined by the Administrative Agent and may be, together with any other Credit Document, entered into, amended, supplemented or waived, without the consent of any other Person, by the applicable Credit Party or Credit Parties and the Administrative Agent or the Collateral Agent in its or their respective sole discretion if applicable, (A) to effect the granting, perfection, protection, expansion or enhancement of any security interest in any Collateral or additional property to become Collateral for the benefit of the Secured Parties, (B) as required by local law or advice of counsel to give effect to, or protect any security interest for the benefit of the Secured Parties, in any property or so that the security interests therein comply with the Applicable Law or (C) to cure ambiguities, omissions, mistakes or defects (as reasonably determined by the Administrative Agent and the Borrower) or to cause such guarantee, collateral security document or other document to be consistent with this Agreement and the other Credit Documents; and (iv) the Credit Parties and the Collateral Agent, without the consent of any other Secured Party, shall be permitted to enter into amendments and/or supplements to any Security Documents in order to include customary provisions permitting the Collateral Agent to appoint sub-collateral agents or representatives to act with respect to Collateral matters thereunder in its stead.

Notwithstanding anything in this Agreement or any Security Document to the contrary, the Administrative Agent may, in its sole discretion, grant extensions of time (and direct the Collateral Agent to grant such extensions) for the satisfaction of any of the requirements under Sections 9.11, Section 9.12 or any Security Documents in respect of any particular Collateral or any particular Subsidiary if it determines that the satisfaction thereof with respect to such Collateral or such Subsidiary cannot be accomplished without undue expense or unreasonable effort or due to factors beyond the control of Holdings, the Borrower and the Subsidiaries by the time or times at which it would otherwise be required to be satisfied under this Agreement or any Security Document.

## 13.2 Notices

Unless otherwise expressly provided herein, all notices and other communications provided for hereunder or under any other Credit Document shall be in writing (including by facsimile or other electronic transmission). All such written notices shall be mailed, faxed or

delivered to the applicable address, facsimile number or e-mail address, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, as follows:

- (a) if to Holdings, the Borrower, the Administrative Agent or the Collateral Agent, to the address, facsimile number, e-mail address or telephone number specified for such Person on Schedule 13.2 or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the other parties; and
- (b) if to any other Lender, to the address, facsimile number, e-mail address or telephone number specified in its Administrative Questionnaire or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to Holdings, the Borrower, the Administrative Agent and the Collateral Agent.

All such notices and other communications shall be deemed to be given or made upon the earlier to occur of (i) actual receipt by the relevant party hereto and (ii) (A) if delivered by hand or by courier, when signed for by or on behalf of the relevant party hereto; (B) if delivered by mail, three Business Days after deposit in the mails, postage prepaid; (C) if delivered by facsimile, when sent and receipt has been confirmed by telephone; and (D) if delivered by e-mail, when delivered; provided that notices and other communications to the Administrative Agent or the Lenders shall not be effective until received during normal business hours on a Business Day for such recipient.

# 13.3 No Waiver; Cumulative Remedies

No failure to exercise and no delay in exercising, on the part of the Administrative Agent, the Collateral Agent or any Lender, any right, remedy, power or privilege hereunder or under the other Credit Documents shall operate as a waiver thereof, nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

# 13.4 Survival of Representations and Warranties

All representations and warranties made hereunder, in the other Credit Documents and in any document, certificate or statement delivered pursuant hereto or in connection herewith shall survive the execution and delivery of this Agreement and the making of the Term Loans hereunder.

# 13.5 Payment of Expenses; Indemnification

The Borrower agrees, within thirty (30) days after written demand therefor (including documentation reasonably supporting such request), or, in the case of expenses of the type described in clause (a) below incurred prior to the Closing Date, on the Closing Date, (a) to pay or reimburse the Agents and the Lenders for all their reasonable and documented out-of-pocket costs and expenses incurred in connection with the preparation, execution, delivery, negotiation and administration of this Agreement (including the B-3 Re-allocation) and the other Credit Documents and any other documents prepared in connection herewith or therewith (including any

amendment or waiver with respect thereto), and the consummation and administration of the transactions contemplated hereby and thereby, limited, in the case of legal expenses, to the reasonable and documented fees, disbursements and other expenses of Akin, Paul, Weiss, Ropes & Gray LLP, Debevoise & Plimpton LLP, and to the extent reasonably advisable, one local counsel for the applicable parties or groups in each relevant material jurisdiction and reasonable and documented fees and expenses of Centerview Partners, Alvarez & Marsal, Glenn Agre Bergman & Fuentes LLP, FTI Consulting, Inc., and (b) to pay or reimburse each Agent and Lender for all their documented out-of-pocket costs and expenses incurred in connection with the enforcement or preservation of any rights under this Agreement, the other Credit Documents and any such other documents, limited, in the case of legal expenses, to the reasonable and documented attorneys' fees and expenses of (x) the counsel named in clause (a) above for the applicable parties or groups and any local counsel for the applicable parties or groups for each jurisdiction, the extent advisable or (y) otherwise one primary counsel for the Agent, one primary counsel to the Lenders taken as a whole (and additional conflict counsel for similarly situated Lenders as a group) and one local counsel for each relevant jurisdiction), (c) to pay, indemnify, and hold harmless each Lender and each Agent from, any and all recording and filing fees and (d) to pay, indemnify, and hold harmless each Lender and each Agent and their respective Affiliates, and the directors, officers, partners, managers, employee, agents (including sub-agents and co-agents), advisors, attorneys-in-fact, trustees, controlling persons, and members of any of the foregoing, from and against any and all other liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever, including reasonable and documented out-of-pocket fees, disbursements and other charges (in the case of legal expenses, subject to the same limitation set forth in clause (b) above) or, with respect to the execution, delivery, enforcement, performance and administration of this Agreement, the other Credit Documents and any such other documents, including, any of the foregoing relating to the violation of, noncompliance with or liability under, any Environmental Law (other than by such indemnified person or any of its Related Parties (other than trustees and advisors)) or to any actual or alleged presence, release or threatened release into the environment of Hazardous Materials attributable to the operations of Holdings, the Borrower, any of the Borrower's Subsidiaries or any of the Real Estate (all the foregoing in this clause (c), collectively, the "indemnified liabilities") (SUBJECT TO THE PROVISO BELOW, WHETHER OR NOT CAUSED BY OR ARISING IN WHOLE OR IN PART, OUT OF THE COMPARATIVE, CONTRIBUTORY OR SOLE ORDINARY NEGLIGENCE OF THE INDEMNIFIED PERSON); provided that neither the Borrower nor any other Credit Party shall have any obligation hereunder to any Agent or any Lender or any of their respective Related Parties with respect to indemnified liabilities to the extent they result from (X) with respect to the Lenders and their Related Parties, (A) the gross negligence, bad faith or willful misconduct of such indemnified Person or any of its Related Parties as determined by a final non-appealable judgment of a court of competent jurisdiction, (B) a material breach of the obligations of such indemnified Person or any of its Related Parties under the Credit Documents as determined by a final non-appealable judgment of a court of competent jurisdiction, (C) disputes not involving an act or omission of Holdings, the Borrower or any other Credit Party and that is brought by an indemnified Person against any other indemnified Person, other than any claims against any indemnified Person in its capacity or in fulfilling its role as an Agent or any similar role under the Credit Facilities or (D) any settlement effected without the Borrower's prior written consent, but if settled with the Borrower's prior written consent (not to be unreasonably withheld, delayed, conditioned or denied) or if there is a final non-appealable judgment in any

such proceeding, the Borrower will indemnify and hold harmless such indemnified Person from and against any and all losses, claims, damages, liabilities and expenses by reason of such settlement or judgment in accordance with this Section 13.5, and (Y) with respect to the Agents and their Related Parties, the gross negligence or willful misconduct of such indemnified Person or any of its Related Parties as determined by a final non-appealable judgment of a court of competent jurisdiction. All amounts payable under this Section 13.5 shall be paid within 30 days of receipt by the Borrower of an invoice relating thereto setting forth such expense in reasonable detail. The agreements in this Section 13.5 shall survive repayment of the Term Loans and all other amounts payable hereunder.

No Credit Party nor any indemnified Person shall have any liability for any special, punitive, indirect or consequential damages resulting from this Agreement or any other Credit Document or arising out of its activities in connection herewith or therewith (whether before or after the Closing Date) (except, in the case of the Borrower's obligation hereunder to indemnify and hold harmless the indemnified Person, to the extent of any losses, claims, damages, liabilities and expenses incurred or paid by such indemnified Person to a third party unaffiliated with such indemnified Person). No indemnified Persons shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Credit Documents or the transactions contemplated hereby or thereby, except to the extent that such damages have resulted from the willful misconduct or gross negligence of any indemnified Person or any of its Related Parties (as determined by a final non-appealable judgment of a court of competent jurisdiction). This Section 13.5 shall not apply to Taxes.

Each indemnified Person, by its acceptance of the benefits of this Section 13.5, agrees to refund and return any and all amounts paid by the Borrower (or on its behalf) to it if, pursuant to limitations on indemnification set forth in this Section 13.5, such indemnified Person was not entitled to receipt of such amounts.

# 13.6 Successors and Assigns; Participations and Assignments

(a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that (i) neither Holdings nor the Borrower may assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Administrative Agent and each Lender (and any attempted assignment or transfer by Holdings or the Borrower without such consent shall be null and void) and (ii) no Lender may assign or otherwise transfer its rights or obligations hereunder except in accordance with this Section 13.6. Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants (to the extent provided in clause (c) of this Section 13.6), to the extent expressly contemplated hereby, the Related Parties of each of the Administrative Agent, the Collateral Agent and the Lenders and each other Person entitled to indemnification under Section 13.5) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) (i) Subject to the conditions set forth in clause (b)(ii) below, any Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitments and the Term Loans at the time owing to it) with the prior written consent (such consent not to be unreasonably withheld, delayed, conditioned or denied) of the Administrative Agent; *provided* that no consent of the Administrative Agent shall be required for any assignment of any Term Loan to a Lender, an Affiliate of a Lender or an Approved Fund. Notwithstanding the foregoing, any Lender that is a Lender on the Closing Date may assign its rights and obligations under this Agreement (including both its Loans and its unfunded Commitments) to any Commitment Party or any Affiliate or Approved Fund thereof without the consent of the Administrative Agent.

Notwithstanding the foregoing, no assignment described in this Section 13.6(b) shall be made to (w) Holdings or any of its Subsidiaries, (x) a natural person, (y) any investment vehicle established primarily for the benefit of a natural person or (z) any Person who is not or does not simultaneously become, or whose Affiliate or Approved Fund is not or does not simultaneously become, a "Consenting Stakeholder" under and as defined in the Restructuring Support Agreement, and any attempted assignment in violation of clauses (w) - (z) shall be null and void.

- (ii) Assignments shall be subject to the following additional conditions:
- (A) except in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund or an assignment of the entire remaining amount of the assigning Lender's Loans, the amount of Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent), shall not be less than \$1,000,000, unless the Administrative Agent otherwise consents (which consents shall not be unreasonably withheld, delayed, conditioned or denied); *provided*, that contemporaneous assignments to a single assignee made by Affiliates of Lenders and related Approved Funds shall be aggregated for purposes of meeting the minimum assignment amount requirements stated above:
- (B) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement;
- (C) The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption, together with a processing and recordation fee in the amount of \$3,500; provided that only one such processing and recordation fee shall be payable in connection with any series of related assignments by or among a Lender and its Affiliates occurring on or about the same date; provided, further, that the Administrative Agent may, in its sole discretion, elect to waive such processing and recordation fee in the case of any assignment;

- (D) the assignee, if it shall not be a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire and the applicable tax forms as required under Section 5.4; and
- (E) no assignments (other than by a Lender that was a Lender as of the Closing Date) shall be permitted prior to the earlier of the date of the Escrow Payment and the Final Order Entry Date.
- (iii) Subject to acceptance and recording thereof pursuant to clause (b)(iv) of this Section 13.6, from and after the effective date specified in each Assignment and Assumption, the assignee thereunder shall be a party hereto and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be entitled to the benefits of Sections 2.10, 2.11, 5.4 and 13.5). Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this Section 13.6 shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with clause (c) of this Section 13.6 (other than attempted assignments or transfers in violation of the last paragraph of Section 13.6(b)(i) above, which shall be null and void as provided above).
- (iv) The Administrative Agent, acting for this purpose as a non-fiduciary agent of the Borrower, shall maintain at the Administrative Agent's Office in the United States a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amount (and stated interest) of the Term Loans owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). Further, each Register shall contain the name and address of the Administrative Agent and the lending office through which each such Person acts under this Agreement. The entries in the Register shall be conclusive, absent manifest error, and the Borrower, the Administrative Agent, the Collateral Agent and the Lenders shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by Holdings, the Borrower, the Collateral Agent and any Lender (solely with respect to its own outstanding Term Loans and Commitments), at any reasonable time and from time to time upon reasonable prior notice.
- (v) Upon its receipt of a duly completed Assignment and Assumption executed by an assigning Lender and an assignee, the assignee's completed Administrative Questionnaire and all requested tax forms (unless the assignee shall already be a Lender hereunder), the processing and recordation fee referred to in clause (b) of this Section 13.6 (unless waived by the Administrative Agent) and any written consent to such assignment required by clause (b) of this Section 13.6, the Administrative Agent shall promptly accept

140

such Assignment and Assumption and record the information contained therein in the Register.

- (i) Any Lender may, without the consent of Holdings, the Borrower or the (c) Administrative Agent, sell participations to one or more banks or other entities any of which is not (w) a natural person, (x) any investment vehicle established primarily for the benefit of a natural person or (z) any Person who is not or does not simultaneously become, or whose Affiliate or Approved Fund is not or does not simultaneously become, a "Consenting Stakeholder" under and as defined in the Restructuring Support Agreement (each, a "Participant") (and any such attempted sales to the Persons identified in clauses (w) - (z) above shall be null and void) (provided that the last sentence of Section 13.6(b)(i) shall apply) in all or a portion of such Lender's rights and obligations under this Agreement (including all or a portion of its Commitments and the Term Loans owing to it); provided that (A) such Lender's obligations under this Agreement shall remain unchanged, (B) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, and (C) Holdings, the Borrower, the Administrative Agent and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement or any other Credit Document; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any consent, amendment, modification, supplement or waiver described in clause (i) or (iv) of the second proviso of the first paragraph of Section 13.1 that directly and adversely affects such Participant. Subject to clause (c)(ii) of this Section 13.6, the Borrower agrees that each Participant shall be entitled to the benefits of Sections 2.10, 2.11 and 5.4 to the same extent as if it were a Lender, and provided that such Participant agrees to be subject to the requirements and limitations of those Sections and Sections 2.12 and 13.7(a) as though it were a Lender and had acquired its interest by assignment pursuant to clause (b) of this Section 13.6. To the extent permitted by Applicable Law, each Participant also shall be entitled to the benefits of Section 13.8(b) as though it were a Lender; provided that such Participant agrees to be subject to Section 13.8(a) as though it were a Lender. Each Lender that sells a participation agrees, at the Borrower's request and expense, to use reasonable efforts to cooperate with the Borrower to effectuate the provisions of Section 13.7 with respect to any Participant.
  - (ii) A Participant shall not be entitled to receive any greater payment under Section 2.10, 2.11 or 5.4 than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent.
  - (iii) Each Lender that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of the Borrower, maintain a register on which it enters the name and address of each participant and the principal amounts (and stated interest) of each Participant's interest in the Term Loans (or other rights or obligations) held by it (the "Participant Register"). The entries in the Participant Register shall be conclusive, absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. No Lender shall have any obligation to disclose

all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any commitments, loans, letters of credit or its other obligations under any Credit Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan, letter of credit or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations.

- (d) Any Lender may, without the consent of Holdings, the Borrower or the Administrative Agent, at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank, and this Section 13.6 shall not apply to any such pledge or assignment of a security interest; *provided* that no such pledge or assignment of a security interest shall release a Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.
- (e) Subject to Section 13.16, the Borrower authorizes each Lender to disclose to any Participant, secured creditor of such Lender or assignee (each, a "Transferee"), any prospective Transferee and any prospective direct or indirect contractual counterparties to any swap or derivative transactions to be entered into in connection with or relating to Term Loans made hereunder any and all financial information in such Lender's possession concerning the Borrower and its Affiliates that has been delivered to such Lender by or on behalf of the Borrower and its Affiliates pursuant to this Agreement or that has been delivered to such Lender by or on behalf of the Borrower and its Affiliates in connection with such Lender's credit evaluation of the Borrower and its Affiliates prior to becoming a party to this Agreement.
- SPV Lender. Notwithstanding anything to the contrary contained herein, any Lender (a "Granting Lender") may grant to a special purpose funding vehicle (an "SPV"), identified as such in writing from time to time by the Granting Lender to the Administrative Agent and the Borrower, the option to provide to the Borrower all or any part of any Term Loan that such Granting Lender would otherwise be obligated to make the Borrower pursuant to this Agreement; provided that (i) nothing herein shall constitute a commitment by any SPV to make any Term Loan and (ii) if an SPV elects not to exercise such option or otherwise fails to provide all or any part of such Term Loan, the Granting Lender shall be obligated to make such Term Loan pursuant to the terms hereof. The making of a Term Loan by an SPV hereunder shall utilize the Commitment of the Granting Lender to the same extent, and as if, such Term Loan were made by such Granting Lender. Each party hereto hereby agrees that no SPV shall be liable for any indemnity or similar payment obligation under this Agreement (all liability for which shall remain with the Granting Lender). In furtherance of the foregoing, each party hereto hereby agrees (which agreement shall survive the termination of this Agreement) that, prior to the date that is one year and one day after the payment in full of all outstanding commercial paper or other senior indebtedness of any SPV, it shall not institute against, or join any other Person in instituting against, such SPV any bankruptcy, reorganization, arrangement, insolvency or liquidation proceedings under the laws of the United States or any State thereof. In addition, notwithstanding anything to the contrary contained in this Section 13.6, any SPV may (i) with notice to, but without the prior written consent of, the Borrower and the Administrative Agent and without paying any processing fee therefor, assign all or a portion of its interests in any Term Loans to the Granting Lender or to any financial institutions (consented to by the Borrower and the Administrative Agent) providing liquidity

and/or credit support to or for the account of such SPV to support the funding or maintenance of Term Loans and (ii) disclose on a confidential basis any non-public information relating to its Term Loans to any rating agency, commercial paper dealer or provider of any surety, guarantee or credit or liquidity enhancement to such SPV. This Section 13.6(f) may not be amended without the written consent of the SPV. Notwithstanding anything to the contrary in this Agreement, (x) no SPV shall be entitled to any greater rights under Sections 2.10, 2.11 and 5.4 than its Granting Lender would have been entitled to absent the use of such SPV and (y) each SPV agrees to be subject to the requirements of Sections 2.10, 2.11, and 5.4 as though it were a Lender and has acquired its interest by assignment pursuant to clause (b) of this Section 13.6.

# (g) [Reserved]

This Section 13.6 shall be construed so that all Term Loans are at all times maintained in "registered form" within the meaning of Section 5f.103-1(c) of the United States Treasury Regulations.

#### 13.7 Replacements of Lenders under Certain Circumstances

- The Borrower shall be permitted (x) to replace any Lender with a (a) replacement bank or institution or (y) terminate the Commitment of such Lender, as the case may be, and repay all Obligations of the Borrower due and owing to such Lender relating to the Term Loans and participations held by such Lender as of such termination date that (a) requests reimbursement for amounts owing pursuant to Section 2.10 or Section 5.4 (or the Borrower is required to pay any Indemnified Taxes or additional amounts to any Agent or Lender or to any Governmental Authority on account of any Agent or Lender pursuant to Section 5.4), (b) is affected in the manner described in Section 2.10(a)(iii) and as a result thereof any of the actions described in such Section is required to be taken or (c) becomes a Defaulting Lender; provided that, solely in the case of the foregoing clause (x), (i) no Event of Default shall have occurred and be continuing at the time of such replacement, (ii) the Borrower shall repay (or the replacement bank or institution shall purchase, at par) all Term Loans and other amounts (other than any disputed amounts), pursuant to Section 2.10, 2.11 or 5.4, as the case may be, owing to such replaced Lender prior to the date of replacement, (iii) the replacement bank or institution, if not already a Lender, an Affiliate of a Lender or an Approved Fund, and the terms and conditions of such replacement, shall be reasonably satisfactory to the Administrative Agent (solely to the extent such consent would be required under Section 13.6), (iv) the replaced Lender shall be obligated to make such replacement in accordance with the provisions of Section 13.6 (provided that the Borrower shall be obligated to pay the registration and processing fee referred to therein unless otherwise agreed) and (v) any such replacement shall not be deemed to be a waiver of any rights that the Borrower, the Administrative Agent or any other Lender shall have against the replaced Lender.
- (b) If any Lender (such Lender, a "Non-Consenting Lender") has failed to consent to a proposed amendment, modification, supplement, waiver, discharge or termination that pursuant to the terms of Section 13.1 requires the consent of either (i) all of the Lenders directly and adversely affected or (ii) all of the Lenders, and, in each case, with respect to which the Required Lenders or a majority (in principal amount) of the directly and adversely affected Lenders shall, in each such case, have granted their consent, then so long as no Event of Default then exists,

the Borrower shall have the right (unless such Non-Consenting Lender grants such consent) to (x) replace such Non-Consenting Lender by requiring such Non-Consenting Lender to assign its Term Loans and its Commitments hereunder to one or more assignees reasonably acceptable to the Administrative Agent (to the extent such consent would be required under Section 13.6) or (y) terminate the Commitment of such Lender, and in the case of a Lender, repay all Obligations of the Borrower due and owing to such Lender relating to the Term Loans and participations held by such Lender as of such termination date; *provided* that: (a) all Obligations of the Borrower hereunder owing to such Non-Consenting Lender being replaced shall be paid in full to such Non-Consenting Lender concurrently with such assignment, and (b) the replacement Lender shall purchase the foregoing by paying to such Non-Consenting Lender a price equal to the principal amount thereof *plus* accrued and unpaid interest thereon. In connection with any such assignment, the Borrower, Administrative Agent, such Non-Consenting Lender and the replacement Lender shall otherwise comply with Section 13.6.

(c) Nothing in this Section 13.7 shall be deemed to prejudice any right or remedy that Holdings or the Borrower may otherwise have at law or at equity.

#### 13.8 Adjustments; Set-off

- Credit Document, if any Lender (a "Benefited Lender") shall at any time receive any payment of all or part of its Term Loans, or interest thereon, or receive any collateral in respect thereof (whether voluntarily or involuntarily, by set-off, pursuant to events or proceedings of the nature referred to in Section 11.5, or otherwise), in a greater proportion than any such payment to or Collateral received by any other Lender, if any, in respect of such other Lender's Term Loans, or interest thereon, such Benefited Lender shall purchase for cash from the other Lenders a participating interest in such portion of each such other Lender's Term Loan, or shall provide such other Lenders with the benefits of any such collateral, or the proceeds thereof, as shall be necessary to cause such Benefited Lender to share the excess payment or benefits of such collateral or proceeds ratably with each of the Lenders; *provided, however*, that if all or any portion of such excess payment or benefits is thereafter recovered from such Benefited Lender, such purchase shall be rescinded, and the purchase price and benefits returned, to the extent of such recovery, but without interest.
- Event of Default, in addition to any rights and remedies of the Lenders provided by Applicable Law, each Lender shall have the right, without prior notice to Holdings, the Borrower, any such notice being expressly waived by Holdings, the Borrower to the extent permitted by Applicable Law but with the prior written consent of the Administrative Agent and the Required Lenders, upon any amount becoming due and payable by the Borrower hereunder (whether at the Stated Maturity, by acceleration or otherwise) to set-off and appropriate and apply against such amount any and all deposits (general or special, time or demand, provisional or final) (other than payroll, trust, tax, fiduciary, employee health and benefits, pension, 401(k) and petty cash accounts), in any currency, and any other credits, indebtedness or claims, in any currency, in each case whether direct or indirect, absolute or contingent, matured or unmatured, at any time held or owing by such Lender or any branch or agency thereof to or for the credit or the account of the Borrower. Each Lender agrees promptly to notify the Borrower and the Administrative Agent after any such set-

off and application made by such Lender; *provided* that the failure to give such notice shall not affect the validity of such set-off and application.

#### 13.9 Counterparts; Electronic Execution

This Agreement may be executed by one or more of the parties to this Agreement on any number of separate counterparts (including by facsimile or other electronic transmission), and all of said counterparts taken together shall be deemed to constitute one and the same instrument. A set of the copies of this Agreement signed by all the parties shall be lodged with the Borrower and the Administrative Agent. The words "execution," "signed," "signature," and words of like import in any Assignment and Assumption or any other Credit Document shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any Applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.

# 13.10 Severability

Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

# 13.11 INTEGRATION; ORDERS CONTROL

THIS WRITTEN AGREEMENT AND THE OTHER CREDIT DOCUMENTS REPRESENT THE FINAL AGREEMENT OF HOLDINGS, THE BORROWER, THE COLLATERAL AGENT, THE ADMINISTRATIVE AGENT AND THE LENDERS WITH RESPECT TO THE SUBJECT MATTER HEREOF, AND (1) THERE ARE NO PROMISES, UNDERTAKINGS, REPRESENTATIONS OR WARRANTIES BY HOLDINGS, THE BORROWER, THE ADMINISTRATIVE AGENT, THE COLLATERAL AGENT OR ANY LENDER RELATIVE TO SUBJECT MATTER HEREOF NOT EXPRESSLY SET FORTH OR REFERRED TO HEREIN OR IN THE OTHER CREDIT DOCUMENTS, (2) THIS AGREEMENT AND THE OTHER CREDIT DOCUMENTS MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES AND (3) THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES; PROVIDED THAT THE SYNDICATION PROVISIONS AND THE BORROWER'S AND HOLDINGS' CONFIDENTIALITY OBLIGATIONS IN THE COMMITMENT LETTER SHALL REMAIN IN FULL FORCE AND EFFECT PURSUANT TO THE TERMS THEREOF. TO THE EXTENT THAT ANY SPECIFIC PROVISION HEREOF IS INCONSISTENT WITH ANY OF THE ORDERS, THE INTERIM ORDER OR FINAL ORDER (AS APPLICABLE) SHALL CONTROL.

#### 13.12 GOVERNING LAW

THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK AND, TO THE EXTENT APPLICABLE, THE BANKRUPTCY CODE.

### 13.13 Submission to Jurisdiction; Waivers

Each party hereto irrevocably and unconditionally:

- (a) submits for itself and its property in any legal action or proceeding relating to this Agreement and the other Credit Documents to which it is a party, or for recognition and enforcement of any judgment in respect thereof, to the exclusive general jurisdiction of the Bankruptcy Court and, if the Bankruptcy Court does not have (or abstains from) jurisdiction, the courts of the State of New York, the courts of the United States of America for the Southern District of New York, in each case sitting in New York City in the Borough of Manhattan, and appellate courts from any thereof;
- (b) consents that any such action or proceeding shall be brought in such courts and waives (to the extent permitted by Applicable Law) any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same;
- (c) agrees that service of process in any such action or proceeding may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to such Person at its address set forth on Schedule 13.2 or at such other address of which the Administrative Agent shall have been notified pursuant to Section 13.2;
- (d) agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law or, in the case of the Administrative Agent, the Collateral Agent and the Lenders, shall limit the right to sue in any other jurisdiction;
- (e) subject to the last paragraph of Section 13.5, waives, to the maximum extent not prohibited by Applicable Law, any right it may have to claim or recover in any legal action or proceeding referred to in this Section 13.13 any special, exemplary, punitive or consequential damages; and
- (f) agrees that a final judgment in any action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Applicable Law.

# 13.14 Acknowledgments

Each of Holdings and the Borrower hereby acknowledges that:

(a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Credit Documents;

- (i) the credit facilities provided for hereunder and any related arranging or other services in connection therewith (including in connection with any amendment, waiver or other modification hereof or of any other Credit Document) are an arm's-length commercial transaction between Holdings and the Borrower, on the one hand, and the Administrative Agent, the Lenders and the other Agents on the other hand, and Holdings, the Borrower and the other Credit Parties are capable of evaluating and understanding and understand and accept the terms, risks and conditions of the transactions contemplated hereby and by the other Credit Documents (including any amendment, waiver or other modification hereof or thereof); (ii) in connection with the process leading to such transaction, each of the Administrative Agent and the other Agents, is and has been acting solely as a principal and is not the financial advisor, agent or fiduciary for any of Holdings, the Borrower, any other Credit Parties or any of their respective Affiliates, stockholders, creditors or employees or any other Person; (iii) neither the Administrative Agent nor any other Agent has assumed or will assume an advisory, agency or fiduciary responsibility in favor of Holdings, the Borrower or any other Credit Party with respect to any of the transactions contemplated hereby or the process leading thereto, including with respect to any amendment, waiver or other modification hereof or of any other Credit Document (irrespective of whether the Administrative Agent or any other Agent has advised or is currently advising Holdings, the Borrower, the other Credit Parties or their respective Affiliates on other matters) and neither the Administrative Agent or other Agent has any obligation to Holdings, the Borrower, the other Credit Parties or their respective Affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein and in the other Credit Documents; (iv) the Administrative Agent, each other Agent and each Affiliate of the foregoing may be engaged in a broad range of transactions that involve interests that differ from those of Holdings, the Borrower and their respective Affiliates, and neither the Administrative Agent nor any other Agent has any obligation to disclose any of such interests by virtue of any advisory, agency or fiduciary relationship; and (v) neither the Administrative Agent nor any other Agent has provided and none will provide any legal, accounting, regulatory or tax advice with respect to any of the transactions contemplated hereby (including any amendment, waiver or other modification hereof or of any other Credit Document) and Holdings and the Borrower has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate. Holdings and the Borrower agree not to claim that the Administrative Agent or any other Agent has rendered advisory services of any nature or respect, or owes a fiduciary or similar duty to Holdings, the Borrower or any other Affiliates, in connection with the transactions contemplated hereby or the process leading hereto.
- (c) no joint venture is created hereby or by the other Credit Documents or otherwise exists by virtue of the transactions contemplated hereby among the Lenders or among Holdings and the Borrower, on the one hand, and any Lender, on the other hand.

#### 13.15 WAIVERS OF JURY TRIAL

HOLDINGS, THE BORROWER, EACH AGENT AND EACH LENDER HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVE (TO THE EXTENT PERMITTED BY APPLICABLE LAW) TRIAL BY JURY IN ANY LEGAL ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT AND FOR ANY COUNTERCLAIM THEREIN.

# 13.16 Confidentiality

The Administrative Agent, each other Agent and each Lender shall hold all Confidential Information confidential; provided that the Administrative Agent, each other Agent and each Lender may make disclosure (a) as required by the order of any court or administrative agency or in any pending legal, judicial or administrative proceeding, or otherwise as required by Applicable Law, regulation or compulsory legal process (in which case such Lender, the Administrative Agent or such other Agent shall use commercially reasonable efforts to inform the Borrower promptly thereof to the extent lawfully permitted to do so (except with respect to any audit or examination conducted by bank accountants or any self-regulatory authority or governmental or regulatory authority exercising examination or regulatory authority), (b) to such Lender's or the Administrative Agent's or such other Agent's attorneys, professional advisors, independent auditors, trustees or Affiliates involved in the Transactions on a "need to know" basis and who are made aware of and agree to comply with the provisions of this Section 13.16, in each case on a confidential basis (with such Lender, the Administrative Agent or such other Agent responsible for such persons' compliance with this Section 13.16), (c) on a confidential basis to any bona fide prospective Lender, prospective participant or swap counterparty, (d) to the extent requested by any bank regulatory authority having jurisdiction over an Agent or a Lender or its Affiliates (including in any audit or examination conducted by bank accountants or any selfregulatory authority or governmental or regulatory authority exercising examination or regulatory authority), (e) to the extent such information: (i) becomes publicly available other than as a result of a breach of this Section 13.16 or other confidential or fiduciary obligation owed by the Administrative Agent, such other Agent or such Lender to the Borrower or its Affiliates or (ii) becomes available to the Administrative Agent, such other Agent or such Lender on a nonconfidential basis from a source other than Holdings, the Borrower or any Subsidiary or on behalf of Holdings, the Borrower or any Subsidiary that, to the knowledge (after due inquiry) of the Administrative Agent, such other Agent or such Lender, is not in violation of any confidentiality obligation owed to the Borrower or its Affiliates, (f) to the extent the Borrower shall have consented to such disclosure in writing (which may include through electronic means), (g) as is necessary in protecting and enforcing the rights of the Administrative Agent, such other Agent or such Lender with respect to this Agreement or any other Credit Document, (h) for purposes of establishing any defense available under Applicable Laws, including, without limitation, establishing a "due diligence" defense, (i) to the extent independently developed by the Administrative Agent, such other Agent or such Lender or any Affiliates thereof without reliance on confidential information, (i) on a confidential basis, to the rating agencies in consultation with the Borrower, (k) on a confidential basis, to the CUSIP Service Bureau or any similar agency in connection with the issuance and monitoring of CUSIP numbers with respect to the Term Loans or market data collectors, similar services providers to the lending industry and service providers to the Administrative Agent in connection with the administration and management of this Agreement and the Credit Documents and (1) to ClearPar® or any other pricing settlement provider. Each Lender, the Administrative Agent and each other Agent agrees that it will not provide to prospective Transferees or to any pledgee referred to in Section 13.6 or to prospective direct or indirect contractual counterparties to any swap or derivative transactions to be entered into in connection with or relating to Term Loans made hereunder any of the Confidential Information unless such Person is advised of and agrees to be bound by the provisions of this Section 13.16 or confidentiality provisions at least as restrictive as those set forth in this Section 13.16.

#### 13.17 Direct Website Communications

- Holdings and the Borrower may, at their option, provide to the Administrative Agent any information, documents and other materials that they are obligated to furnish to the Administrative Agent pursuant to the Credit Documents, including, all notices, requests, financial statements, financial and other reports, certificates and other information materials, but excluding any such communication (provided that such communications described in clauses (A) - (D) will be delivered pursuant to Section 13.2, including by e-mail) that (A) relates to a request for a new, or a conversion of an existing, Borrowing or other extension of credit (including any election of an interest rate or Interest Period relating thereto), (B) relates to the payment of any principal or other amount due under this Agreement prior to the scheduled date therefor, (C) provides notice of any Default or Event of Default under this Agreement, or (D) is required to be delivered to satisfy any condition precedent to the effectiveness of this Agreement and/or any Borrowing or other extension of credit thereunder (all such non-excluded communications being referred to herein collectively as "Communications"), by transmitting the Communications in an electronic/soft medium in a format reasonably acceptable to the Administrative Agent at an email address separately identified by the Administrative Agent; provided that: (i) upon written request by the Administrative Agent, Holdings or the Borrower shall deliver paper copies of such documents to the Administrative Agent for further distribution to each Lender until a written request to cease delivering paper copies is given by the Administrative Agent and (ii) Holdings or the Borrower shall notify (which may be by facsimile or electronic mail) the Administrative Agent of the posting of any such documents and provide to the Administrative Agent by electronic mail electronic versions (i.e., soft copies) of such documents. Each Lender shall be solely responsible for timely accessing posted documents or requesting delivery of paper copies of such documents from the Administrative Agent and maintaining its copies of such documents. Nothing in this Section 13.17 shall prejudice the right of Holdings, the Borrower, the Administrative Agent, any other Agent or any Lender to give any notice or other communication pursuant to any Credit Document in any other manner specified in such Credit Document.
- (b) The Administrative Agent agrees that the receipt of the Communications by the Administrative Agent at its e-mail address set forth above shall constitute effective delivery of the Communications to the Administrative Agent for purposes of the Credit Documents. Each Lender agrees that notice to it (as provided in the next sentence) specifying that the Communications have been posted to the Platform shall constitute effective delivery of the Communications to such Lender for purposes of the Credit Documents. Each Lender agrees (A) to notify the Administrative Agent in writing (including by electronic communication) from time to time of such Lender's e-mail address to which the foregoing notice may be sent by electronic transmission and (B) that the foregoing notice may be sent to such e-mail address.
- (c) Holdings and the Borrower further agree that the Agents may make the Communications available to the Lenders by posting the Communications on Intralinks or a substantially similar electronic transmission system (the "**Platform**"), so long as the access to such Platform is limited (i) to the Agents, the Lenders or any bona fide potential Transferee and (ii) remains subject the confidentiality requirements set forth in Section 13.16.

- THE PLATFORM AND THE COMMUNICATIONS ARE PROVIDED "AS IS" AND "AS AVAILABLE". THE AGENT PARTIES DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE COMMUNICATIONS OR THE ADEQUACY OF THE PLATFORM, AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS IN OR OMISSIONS IN THE PLATFORM FROM THE COMMUNICATIONS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY ANY AGENT PARTY IN CONNECTION WITH THE COMMUNICATIONS OR THE PLATFORM. In no event shall any Agent or their Related Parties (collectively, the "Agent Parties" and each an "Agent Party") have any liability to Holdings, the Borrower, any Lender or any other Person for losses, claims, damages, liabilities or expenses of any kind (whether in tort, contract or otherwise) arising out of Holdings', the Borrower's or any Agent's transmission of Communications through the internet or the Platform, except to the extent the liability of any Agent Party resulted from such Agent Party's (or any of its Related Parties' (other than trustees or advisors)) gross negligence or willful misconduct (as determined in a final non-appealable judgment of a court of competent jurisdiction).
- The Borrower and each Lender acknowledge that certain of the Lenders may be "public-side" Lenders (Lenders that do not wish to receive material non-public information with respect to Holdings, the Borrower, the Subsidiaries of the Borrower or their securities) and, if documents or notices required to be delivered pursuant to the Credit Documents or otherwise are being distributed through the Platform, any document or notice that Holdings or the Borrower has indicated contains only publicly available information with respect to Holdings, the Borrower and the Subsidiaries of the Borrower and their securities may be posted on that portion of the Platform designated for such public-side Lenders. If Holdings or the Borrower has not indicated whether a document or notice delivered contains only publicly available information, the Administrative Agent shall post such document or notice solely on that portion of the Platform designated for Lenders who wish to receive material nonpublic information with respect to Holdings, the Borrower, the Subsidiaries of the Borrower and their securities. Notwithstanding the foregoing, Holdings and the Borrower shall use commercially reasonable efforts to indicate whether any document or notice contains only publicly available information. Notwithstanding the foregoing, the following Communications may be posted on that portion of the Platform designated for such "public-side" Lenders unless the Borrower notifies the Administrative Agent promptly in writing that any such document contains material non-public information: (1) the Credit Documents, and (2) notification of changes in the terms of the Loans. In the event that any "public-side" Lender has elected for itself to not access any information disclosed through the Platform or otherwise, such "public-side" Lender acknowledges that no Agent shall have any responsibility for such Public Lender's decision to limit the scope of information it has obtained in connection with the Credit Documents, and shall not be liable for the failure to so disclose such information to such "public-side" Lender. Each of the Lenders, Holdings and the Borrower hereby approves distribution of the Communications through the Platform and understands and assumes the risks of such distribution.

#### 13.18 USA PATRIOT Act

Each Lender hereby notifies the Borrower that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "Patriot Act"), it is required to obtain, verify and record information that identifies each Credit Party, which information includes the name and address of each Credit Party and other information that will allow such Lender to identify each Credit Party in accordance with the Patriot Act.

# 13.19 Payments Set Aside

Agent or any Lender, or any Agent or any Lender exercises its right of setoff, and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by such Agent or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with the Chapter 11 Cases or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such setoff had not occurred, and (b) each Lender severally agrees to pay to the Administrative Agent upon demand its applicable share of any amount so recovered from or repaid by any Agent, *plus* interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the applicable Overnight Rate from time to time in effect.

# 13.20 Judgment Currency

If, for the purposes of obtaining judgment in any court, it is necessary to convert a sum due hereunder or any other Credit Document in one currency into another currency, the rate of exchange used shall be that at which in accordance with normal banking procedures the Administrative Agent could purchase the first currency with such other currency on the Business Day preceding that on which final judgment is given. The obligation of the Borrower in respect of any such sum due from it to the Administrative Agent or the Lenders hereunder or under the other Credit Documents shall, notwithstanding any judgment in a currency (the "Judgment Currency") other than that in which such sum is denominated in accordance with the applicable provisions of this Agreement (the "Agreement Currency"), be discharged only to the extent that on the Business Day following receipt by the Administrative Agent of any sum adjudged to be so due in the Judgment Currency, the Administrative Agent may in accordance with normal banking procedures purchase the Agreement Currency with the Judgment Currency. If the amount of the Agreement Currency so purchased is less than the sum originally due to the Administrative Agent from the Borrower in the Agreement Currency, the Borrower agrees, as a separate obligation and notwithstanding any such judgment, to indemnify the Administrative Agent or the Person to whom such obligation was owing against such loss. If the amount of the Agreement Currency so purchased is greater than the sum originally due to the Administrative Agent in such currency, the Administrative Agent agrees to return the amount of any excess to the Borrower (or to any other Person who may be entitled thereto under Applicable Law).

# 13.21 Acknowledgement and Consent to Bail-In of EEA Financial Institutions

Notwithstanding anything to the contrary in any Credit Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any Affected Financial Institution arising under any Credit Document, to the extent such liability is unsecured, may be subject to the Write-Down and Conversion Powers of the applicable Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:

- (a) the application of any Write-Down and Conversion Powers by the applicable Resolution Authority to any such liabilities arising hereunder which may be payable to it by any party hereto that is an Affected Financial Institution; and
- (b) the effects of any Bail-In Action on any such liability, including, if applicable:
  - (i) a reduction in full or in part or cancellation of any such liability;
  - (ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such Affected Financial Institution, its parent undertaking, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Credit Document; or
  - (iii) the variation of the terms of such liability in connection with the exercise of the write-down and conversion powers of the applicable Resolution Authority.

[Signature Page Follows]

# Exhibit B

**DIP-to-Exit ABL Commitment Letter** 

#### HIGHLY CONFIDENTIAL

#### **EXECUTION VERSION**

#### CITIGROUP GLOBAL MARKETS INC.

388 Greenwich Street New York, NY 10013

# BANK OF AMERICA, N.A.

One Bryant Park New York, NY 10036

#### **ROYAL BANK OF CANADA**

200 Vesey Street New York, NY 10281

# DEUTSCHE BANK AG NEW YORK BRANCH DEUTSCHE BANK SECURITIES INC.

1 Columbus Circle New York, NY 10019

February 13, 2023

Avaya Holdings Corp. Avaya Inc. 2605 Meridian Parkway, Suite 200 Durham, North Carolina 27713 Attention: John Sullivan, VP and Corporate Treasurer

# DIP-to-Exit ABL Facility Commitment Letter

Ladies and Gentlemen:

Avaya Holdings Corp., a Delaware corporation ("Holdings"), and Avaya Inc., a Delaware corporation (the "Company", and together with Holdings, "you"), have advised Citigroup Global Markets Inc. (collectively with certain of its affiliates as may be appropriate to perform the work or consummate the transactions contemplated herein, "CGMI"), Royal Bank of Canada (acting through such of its affiliates or branches as it deems appropriate, "Royal Bank"), RBC Capital Markets1 ("RBCCM" and, together with Royal Bank, "RBC"), Bank of America, N.A. ("Bank of America"), Deutsche Bank AG New York Branch ("DBNY") and Deutsche Bank Securities Inc. ("DBSI", and together with CGMI, Royal Bank, RBCCM, Bank of America and DBNY, "we", "us", "our" or the "Commitment Parties"), that each of Holdings, the Company and certain of its domestic subsidiaries, is considering filing voluntary petitions for relief under chapter 11 of the United States Bankruptcy Code (the "Bankruptcy Code"), and the Company desires to establish, for the Company as a debtor-in possession pursuant to the Bankruptcy Code, subject to the conditions set forth herein, a senior secured superpriority debtor-in-possession asset-based revolving credit facility in an aggregate principal amount of \$128,125,000 (the "DIP ABL Facility"), which DIP ABL Facility, upon the satisfaction of the conditions set forth herein, will be automatically converted into a senior secured asset-based revolving credit facility upon the exit from the bankruptcy proceedings by the Company and its affiliates (the "Exit ABL Facility", together with the DIP ABL Facility, collectively, the "DIP-to-Exit ABL Facility"), in each case, substantially as described in the Summary of Proposed Terms and Conditions attached as Exhibit A hereto (the "Term Sheet" and together with this commitment letter, the "Commitment Letter"). Each capitalized term used but not defined herein shall have the meaning assigned to such term in the Term Sheet.

<sup>&</sup>lt;sup>1</sup> RBC Capital Markets is a brand name for the capital markets activities of Royal Bank of Canada and its affiliates.

As used herein, the term "*Transactions*" means, collectively, the negotiation and entering into the DIP-to-Exit ABL Facility and all other transactions described in the Term Sheet and all transactions related thereto. The date on which the Company files a voluntary petition under chapter 11 of the Bankruptcy Code is referred to as the "*Petition Date*" and the date on which the closing of the DIP ABL Facility occurs is referred to herein as the "*Closing Date*". The definitive documentation in respect of the DIP-to-Exit ABL Facility is referred to herein as "*DIP ABL Loan Documents*".

#### 1. Commitments.

In connection with the foregoing, each of CGMI (on behalf of Citi (as defined below)), Royal Bank, Bank of America and Deutsche Bank (each in such capacity, a "Lender" and, collectively, the "Lenders"), is pleased to advise you of its several but not joint commitment to provide the aggregate principal amount of the DIP ABL Facility as set forth opposite its name on Schedule I hereto, solely upon the terms and subject to the conditions set forth in this Commitment Letter. In addition, each Lender severally but not jointly agrees that upon the substantial consummation of an acceptable plan of reorganization of the Company, such Lender's commitment under the DIP ABL Facility shall automatically convert into the Exit ABL Facility, which shall be subject only to (i) the occurrence of the Closing Date, (ii) the execution of documentation containing the terms set forth under the heading "VIII. DIP ABL Facility to Exit ABL Facility Terms and Conversion Conditions - Terms" in Exhibit A hereto and (ii) the satisfaction or waiver by all of the Commitment Parties of the applicable "Conversion Conditions" set forth under the heading "VIII. DIP ABL Facility to Exit ABL Facility Terms and Conversion Conditions - Terms" in Exhibit A hereto. For purposes of this Commitment Letter, "Citi" means CGMI, Citibank, N.A., Citicorp USA, Inc., Citicorp North America, Inc., and/or any of their affiliates as may be appropriate to consummate the transactions contemplated hereby. It is understood and agreed that CGMI is entering into this letter for and on behalf of Citi.

#### 2. Titles and Roles.

It is agreed that (a) each of Citi, RBCCM, BofA Securities and DBSI will act as a lead arranger and a bookrunner for the DIP-to-Exit ABL Facility (each in such capacity, an "Arranger" and, collectively, the "Arrangers"; together with the Lenders, collectively, the "Financial Institutions", "we" or "us"), (b) Citi will act as the "global coordinator" and (c) Citibank, N.A. will act as the sole administrative agent and sole collateral agent for the DIP-to-Exit ABL Facility (in such capacities, the "Administrative Agent"), in each case upon the terms and subject to the conditions set forth or referred to in this Commitment Letter and/or the Term Sheet. You agree that Citi will have "left" placement in any and all marketing materials or other documentation used in connection with the DIP-to-Exit ABL Facility. You further agree that no other titles will be awarded and no compensation will be paid in connection with the DIP-to-Exit ABL Facility (other than as expressly contemplated by this Commitment Letter and each fee letter dated on or about the date hereof and delivered herewith with respect to the Transactions (collectively, the "Fee Letters")) unless you and we shall so agree.

#### 3. [Reserved].

# 4. <u>Information</u>.

You hereby represent that (a) all written information concerning you or any of your subsidiaries or the Transactions other than the Projections and information of a general economic or industry specific nature (the "Information") that has been or will be made available to us by you or any of your representatives, when taken as a whole, is or will be, when furnished, correct in all material respects and does not or will not, when furnished, contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements contained therein not misleading in light

of the circumstances under which such statements are made (after giving effect to all supplements and updates thereto) and (b) the financial and/or business projections and other forward-looking information (the "*Projections*") that have been or will be made available to us by you or any of your representatives have been or will be prepared in good faith based upon assumptions that you believe to be reasonable at the time made and at the time such Projections are made available to us; it being recognized that such Projections are not to be viewed as facts and that actual results during the period or periods covered by any such Projections may differ significantly from the projected results, and that no assurance can be given that the projected results will be realized, that actual results may differ significantly from the projected results and that such differences may be material. You agree that if at any time prior to the Closing Date, any of the representations in the preceding sentence would be incorrect in any material respect if the Information and Projections were being furnished, and such representations were being made, at such time, then you will reasonably promptly supplement the Information and the Projections so that such representations will be correct under those circumstances. In arranging the DIP-to-Exit ABL Facility, we will be entitled to use and rely on the Information and the Projections without responsibility for independent verification thereof.

#### 5. Fees.

As consideration for the commitments of the Commitment Parties hereunder, and their respective agreements to perform the services described herein, you agree to pay to us the fees set forth in the Term Sheet and the Fee Letters subject to the conditions set forth therein, it being understood that, to the extent any such payments are to be made after the commencement of the chapter 11 cases under the Bankruptcy Code, such payments will also be subject to the entry of an order of the Bankruptcy Court authorizing Holdings, the Company and the other subsidiaries that are debtors under such chapter 11 cases to perform their obligations under this Commitment Letter and the Fee Letters and to pay the fees and expenses set forth herein; .

#### 6. Conditions Precedent.

The commitments of the Commitment Parties, and their respective agreements to perform the services described herein, are subject solely to the satisfaction or waiver by all of the Commitment Parties of the applicable conditions set forth in the Term Sheet section titled "Conditions to Effectiveness".

# 7. <u>Indemnification; Expenses</u>.

(A) You agree (a) to indemnify and hold harmless the Commitment Parties and each of their respective affiliates and controlling persons and each of their respective officers, directors, employees, agents, trustees, advisors, attorneys and other representatives (each, an "Indemnified Party") from and against any and all liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements (including reasonable and documented fees and out-of-pocket costs and expenses of one law firm retained by the Indemnified Parties, taken as a whole, plus one local counsel in each relevant jurisdiction (and, in the event of any conflict of interest among the Indemnified Parties, one additional counsel (and one local counsel in each relevant jurisdiction) for each group of Indemnified Parties similarly situated that is subject to such conflict)), joint or several, of any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against any such Indemnified Party (including without limitation, in connection with any investigation, litigation or proceeding or the preparation of any defense in connection therewith), in each case arising out of or in connection with or by reason of this Commitment Letter, the Fee Letters or the DIP ABL Loan Documents or any of the transactions contemplated hereby or thereby, any use made or proposed to be made with the proceeds of the DIP-to-Exit ABL Facility (any of the foregoing, an "Action"), regardless of

whether any such Indemnified Party is a party thereto (and regardless of whether such Action is initiated by your equity holders, creditors or any other third party or by Holdings or any of its respective subsidiaries or affiliates), except to the extent such liability, obligation, loss, damage, penalty, claim, demand, action, judgment, suit, cost, expense or disbursement (i) is found in a final non-appealable judgment by a court of competent jurisdiction to have resulted from the gross negligence, bad faith or willful misconduct of such Indemnified Party or any of its Related Parties (as defined below), (ii) arises from a material breach by such Indemnified Party (or its Related Parties) of its obligations under this Commitment Letter as found in a final nonappealable judgment of a court of competent jurisdiction or (iii) arises out of a dispute solely among Indemnified Parties and not arising out of any act or omission of the Company or any of its subsidiaries (other than any claims, damages, losses, liabilities and expenses against the administrative agent or any of the arrangers in their capacities as such) and (b) to pay or reimburse the Financial Institutions reasonably promptly from time to time, upon presentation of a reasonably detailed summary statement, for all reasonable documented out-of-pocket costs and expenses (including but not limited to expenses of our due diligence investigation, arranging expenses, travel expenses and fees, disbursements and other charges of counsel (except the allocated costs of in-house counsel but including the reasonable and documented fees and out-of-pocket costs and expenses of one primary outside counsel (which shall be Davis Polk & Wardwell LLP) and one local counsel in each other appropriate jurisdiction)), in each case, incurred in connection with the DIP-to-Exit ABL Facility and the preparation, negotiation and enforcement of this Commitment Letter, the Fee Letters, the DIP ABL Loan Documents and any ancillary documents or security arrangements in connection therewith. "Related Parties" means, with respect to each Indemnified Party or Protected Party (as defined below), such person's affiliates and such person's or affiliate's officers, directors, employees, advisors, agents or other representatives.

(B) None of the Commitment Parties, their respective affiliates nor controlling persons nor any of their respective officers, directors, employees, agents, trustees, advisors, attorneys and other representatives (each, a "*Protected Party*") shall be liable for any damages arising from the use by others of any information or other materials obtained through internet, electronic, telecommunications or other information transmission systems, except to the extent such damages have resulted from (in each case as finally determined by a court of competent jurisdiction in a final and non-appealable judgment) the willful misconduct, bad faith or gross negligence of such Protected Party (or its Related Parties). None of the Protected Parties or you or any of your affiliates or the respective directors, officers, employees, advisors, and agents of the foregoing shall be liable for any indirect, special, punitive or consequential damages in connection with this Commitment Letter, the Fee Letters, the DIP-to-Exit ABL Facility, the DIP ABL Loan Documents or the transactions contemplated hereby; *provided* that the foregoing shall not limit your indemnification obligations set forth herein to the extent any such indirect, special, punitive or consequential damages are included in any third-party claim with respect to which the applicable Protected Party is entitled to indemnification pursuant to this Section 7.

(C) You shall not be liable for any settlement, compromise or consent to the entry of any judgment in any Action effected without your prior written consent (which consent shall not be unreasonably withheld or delayed), but if settled with your written consent or if there is a final judgment in any such Action, you agree to indemnify and hold harmless each Indemnified Party from and against any and all losses, claims, damages, liabilities and expenses by reason of such settlement or judgment in accordance with, and to the extent required by, this Section 7. You shall not, without the prior written consent of the applicable Indemnified Party (which consent shall not be unreasonably withheld or delayed, it being understood that any consent withheld in connection with any settlement not effected in accordance with the succeeding clauses (a) and (b) shall be reasonable), effect any settlement of any pending or threatened Action in respect of which indemnity could have been sought hereunder by such Indemnified Party unless (a) such settlement includes an unconditional release of such Indemnified Party

in form and substance reasonably satisfactory to such Indemnified Party from all liability on claims that are the subject matter of such Action and (b) does not include any statement as to or any admission of fault, culpability or a failure to act by or on behalf of such Indemnified Party.

(D) The foregoing provisions shall be superseded in each case by the applicable provisions contained in the DIP ABL Loan Documents upon execution thereof to the extent covered thereby and thereafter shall have no further force and effect to such extent.

#### 8. Sharing Information; Absence of Fiduciary Relationship; Affiliate Activities.

You acknowledge that each Financial Institution may be providing debt financing, equity capital or other services (including financial advisory services) to other companies in respect of which you may have conflicting interests regarding the transactions described herein or otherwise. We will not furnish confidential information obtained from you by virtue of the transactions contemplated by this Commitment Letter or our other relationships with you to other companies. You also acknowledge that we do not have any obligation to use in connection with the transactions contemplated by this Commitment Letter, or to furnish to you, confidential information obtained by us from other companies.

You further acknowledge and agree that (a) no fiduciary, advisory or agency relationship between you and any Financial Institution is intended to be or has been created in respect of any of the transactions contemplated by this Commitment Letter, irrespective of whether any Financial Institution has advised or is advising you on other matters, (b) each Financial Institution, on the one hand, and you, on the other hand, have an arms-length business relationship that does not directly or indirectly give rise to, nor do you rely on, any fiduciary duty on the part of any Financial Institutions, (c) you are capable of evaluating and understanding, and you understand and accept, the terms, risks and conditions of the transactions contemplated by this Commitment Letter, (d) you have been advised that each Financial Institution is engaged in a broad range of transactions that may involve interests that differ from your interests and that Financial Institutions do not have any obligation to disclose such interests and transactions to you by virtue of any fiduciary, advisory or agency relationship, and (e) you waive, to the fullest extent permitted by law, any claims you may have against any Financial Institution for breach of fiduciary duty or alleged breach of fiduciary duty and agree that no Financial Institution shall have any liability (whether direct or indirect) to you in respect of such a fiduciary duty claim or to any person asserting a fiduciary duty claim on behalf of or in right of you, including your stockholders, employees or creditors.

You further acknowledge that each Financial Institution and its affiliates (collectively, the "Group") is a full service securities firms engaged in securities trading and brokerage activities as well as providing investment banking and other financial services. In the ordinary course of business, member of the Group may provide investment banking and other financial services to, and/or acquire, hold or sell, for its own accounts and the accounts of customers, equity, debt and other securities and financial instruments (including bank loans and other obligations) of, you and your subsidiaries and other companies with which you or your subsidiaries may have commercial or other relationships. With respect to any securities and/or financial instruments so held by any member of the Group, or any of its customers, all rights in respect of such securities and financial instruments, including any voting rights, will be exercised by the holder of the rights, in its sole discretion. Accordingly, there may be situations where parts of the Group and/or their clients either now have or may in the future have interests, or take actions, that may conflict with the Company's interests. For example, the Group may, in the ordinary course of business, engage in trading in financial products or undertake other investment businesses for their own account or on behalf of other clients, including without limitation, trading in or holding long,

short or derivative positions in securities, loans or other financial products of the Company or its affiliates or other entities connected with the DIP-to-Exit ABL Facility or the transactions contemplated hereby.

In recognition of the foregoing, the Company agrees that the Group is not required to restrict its activities as a result of this Commitment Letter and that the Group may undertake any business activity without further consultation with or notification to the Company. Neither this Commitment Letter nor the receipt by any Financial Institution of confidential information nor any other matter will give rise to any fiduciary, equitable or contractual duties (including without limitation, any duty of trust or confidence) that would prevent or restrict the Group from acting on behalf of other customers or for its own account. Furthermore, the Company agrees that neither the Group nor any member or business of the Group is under a duty to disclose to the Company or use on behalf of the Company any information whatsoever about or derived from those activities or to account for any revenue or profits obtained in connection with such activities. However, consistent with the Group's long-standing policy to hold in confidence the affairs of its customers, the Group will not use confidential information obtained from the Company except in connection with its services to, and its relationship with, the Company; provided, however, that the Group will be free to disclose information in any manner as required by law, regulation, regulatory authority or other applicable judicial or government order.

# 9. <u>Assignments; Amendments; Governing Law, Etc.</u>

This Commitment Letter shall not be assignable by (x) you without the prior written consent of the other parties hereto, or (y) any Commitment Party without the prior written consent of the Borrower (and, in the cases of clauses (x) and (y), any attempted assignment without such applicable consent shall be null and void), is intended to be solely for the benefit of the parties hereto (and Indemnified Parties and Protected Parties), and is not intended to confer any benefits upon, or create any rights in favor of, any person other than the parties hereto (and Indemnified Parties and Protected Parties). Any and all obligations of, and services to be provided by, any Financial Institution hereunder may be performed and any and all rights of such Financial Institution hereunder may be exercised by or through any of its respective affiliates or branches and the provisions of Section 7 shall apply with equal force and effect to any such entities so performing any such duties or activities, but such Financial Institution shall not be relieved of its obligations under this Commitment Letter. This Commitment Letter may not be amended or any provision hereof waived or modified except by an instrument in writing signed by each of us and you. This Commitment Letter may be executed in any number of counterparts, each of which shall be an original and all of which, when taken together, shall constitute one agreement. Delivery of an executed counterpart of a signature page of this Commitment Letter by facsimile or other electronic transmission shall be effective as delivery of a manually executed counterpart hereof. The words "execution," "signed," "signature," and words of like import in this Commitment Letter shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act. Section headings used herein are for convenience of reference only, are not part of this Commitment Letter and are not to affect the construction of, or to be taken into consideration in interpreting, this Commitment Letter. You acknowledge that information and documents relating to the Transactions may be transmitted through Syndtrak, Intralinks, the internet, e-mail or similar electronic transmission systems, and that none of us shall be liable for any damages arising from the unauthorized use by others of information or documents transmitted in such manner unless resulting from the gross negligence, bad faith or willful misconduct, as determined by a court of competent jurisdiction in a final and non-appealable judgment, of such Financial Institution or any of its Related Parties. Each Financial

Institution may, in consultation with you, place customary advertisements in financial and other newspapers and periodicals or on a home page or similar place for dissemination of customary information on the Internet or worldwide web as they may choose, and circulate similar promotional materials, after the closing of the Transactions in the form of a "tombstone" or otherwise describing the names of you and your affiliates (or any of them), and the amount, type and closing date of such Transactions, all at the expense of such Financial Institution. THIS COMMITMENT LETTER AND ANY CLAIM, CONTROVERSY OR DISPUTE ARISING UNDER OR RELATED TO THIS COMMITMENT LETTER SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.

#### 10. Jurisdiction.

Each party hereto hereby irrevocably and unconditionally (a) submits, for itself and its property, to the exclusive jurisdiction of the bankruptcy court having jurisdiction over the chapter 11 cases of Holdings, the Company and its subsidiaries (the "Bankruptcy Court") or, if such court denies jurisdiction or the Company elects not to file cases under the Bankruptcy Court, then any New York State court or Federal court of the United States of America sitting in the County of New York, and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Commitment Letter, the Fee Letters or the transactions contemplated hereby or thereby, and agrees that all claims in respect of any such action or proceeding may be heard and determined only in such court, (b) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any suit, action or proceeding arising out of or relating to this Commitment Letter, the Fee Letters or the transactions contemplated hereby or thereby in any such court, (c) waives, to the fullest extent permitted by law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such court, and (d) agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Each party hereto agrees that service of any process, summons, notice or document by registered mail addressed to you or us shall be effective service of process for any suit, action or proceeding brought in any such court.

#### 11. Waiver of Jury Trial.

EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES THE RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING, CLAIM OR COUNTERCLAIM BROUGHT BY OR ON BEHALF OF ANY PARTY RELATED TO OR ARISING OUT OF THIS COMMITMENT LETTER, THE FEE LETTERS OR THE PERFORMANCE OF SERVICES HEREUNDER OR THEREUNDER.

#### 12. Confidentiality.

This Commitment Letter is delivered to you on the understanding that neither this Commitment Letter nor the Fee Letters nor any of their terms or substance, shall be disclosed, directly or indirectly, by you to any other person except (a) to your equity holders, your affiliates, your equity holders' and your affiliates' officers, directors, employees, attorneys, accountants and advisors on a confidential and "need to know" basis, (b) as required by applicable law or regulation or legal, judicial or administrative proceedings or other compulsory process or as requested by any governmental authority (or necessary in connection with any of the foregoing) (in which case you agree, to the extent reasonably practicable and permitted by law, promptly to inform in writing in advance thereof), including as may be required to obtain court approval in connection with any acts or obligations to be taken pursuant to this Commitment Letter or the Fee Letters or the transactions contemplated hereby or thereby (but subject to

the provisions of clause (ii) of the following sentence), (c) you may disclose this Commitment Letter and the contents hereof (but not the Fee Letters or the contents thereof other than the existence thereof and the contents thereof as part of projections, pro forma information and a generic disclosure of aggregate sources and uses to the extent customary in marketing materials and other required filings) (i) in any marketing materials relating to the DIP-to-Exit ABL Facility, (ii) with respect to the Term Sheet, to any rating agencies and (iii) to potential debt providers to obtain commitments to the DIP-to-Exit ABL Facility from such potential debt providers, (d) in connection with any remedy or enforcement of any right under this Commitment Letter or the Fee Letters, and (e) to the extent such information becomes publicly available other than as a result of a breach of this paragraph. Notwithstanding anything to the contrary in the foregoing, you shall be permitted to (i) provide unredacted copies of the Commitment Letter and the Fee Letters to the Bankruptcy Court and the Office of the United State Trustee in connection with any motion seeking approval of the DIP-to-Exit ABL Facility, (ii) publicly disclose the Commitment Letter and the Fee Letters to the extent necessary to obtain approval of the Bankruptcy Court for the DIP-to-Exit ABL Facility, provided, that you agree to use commercially reasonable efforts to file and diligently pursue a motion or an ex parte request seeking an order authorizing you to file the Fee Letters under seal, (iii) provide unredacted copies of the Commitment Letter and the Fee Letters to the counsel and financial advisors to the official committee of unsecured creditors appointed in any of the chapter 11 cases of you or any of your affiliates, in each case so long as such disclosure is on a confidential "professionals eyes only" basis and (iv) publicly file the Commitment Letter (but not the Fee Letters) in order to comply with any public disclosure requirements under the applicable rules of the Securities Exchange Commission.

Each Financial Institution shall use all confidential information received by it in connection with this Commitment Letter and the transactions contemplated hereby solely for the purposes of providing the services and entering into the Transactions and shall treat confidentially, together with the terms and substance of this Commitment Letter and the Fee Letters, all such information; provided, however, that nothing herein shall prevent such Financial Institution from disclosing any such information (a) to rating agencies (provided that we will disclose such information only through you and with your prior consent), (b) to any Lenders or participants or prospective Lenders or participants or in respect of any swap or derivative transaction relating to the Company, Holdings or any of their respective subsidiaries or their respective obligations or any other transaction under which payments are to be made by reference to the Company, Holdings or any of their respective subsidiaries or their respective obligations; provided that the disclosure of any such information to any Lenders or prospective Lenders or participants or prospective participants or swap, derivative or other parties referred to above shall be made subject to the acknowledgment and acceptance by such Lender or prospective Lender or participant or prospective participant or swap, derivative or other parties that such information is being disseminated on a confidential basis for the benefit of the Company (on substantially the terms set forth in this paragraph or as is otherwise reasonably acceptable to you and each Financial Institution, including, without limitation, as agreed in any confidential information memorandum or other marketing materials or pursuant to customary "click-through" or similar electronic agreements) in accordance with the standard arranging processes of such Financial Institution or customary market standards for dissemination of such type of information, (c) in any legal, judicial, administrative proceeding or other compulsory process or otherwise as required by applicable law or regulations (in which case such Financial Institution shall promptly notify you, in advance, to the extent reasonably practicable and permitted by law), (d) upon the request or demand of any regulatory authority having jurisdiction over such Financial Institution or its affiliates (in which case such Financial Institution shall, except with respect to any audit or examination conducted by bank accountants or any governmental bank regulatory authority exercising examination or regulatory authority, promptly notify you, in advance, to the extent reasonably practicable and lawfully permitted to do so), (e) to the officers, directors, employees, legal counsel, independent auditors, professionals and other experts or agents of such Financial Institution

(collectively, "*Representatives*") who are informed of the confidential nature of such information and are or have been advised of their obligation to keep information of this type confidential and each Financial Institution shall be responsible for its Representatives' compliance with this paragraph, (f) to any of its respective affiliates (provided that any such affiliate is advised of its obligation to retain such information as confidential and each Financial Institution shall be responsible for its affiliates' compliance with this paragraph) solely in connection with the Transactions, (g) to the extent any such information becomes publicly available other than by reason of disclosure by such Financial Institution, its affiliates or Representatives in breach of this Commitment Letter, (h) to the extent that such information is received by such Financial Institution from a third party that is not, to such Financial Institution's knowledge, subject to confidentiality obligations owing to you or any of your respective affiliates or related parties, (i) to the extent that such information is independently developed by such Financial Institution or (j) for purposes of establishing a "due diligence" defense. The provisions of this paragraph shall automatically terminate two years following the date of this Commitment Letter.

In addition, any Commitment Party may disclose the existence of and information about the DIP ABL Facility to market data collectors or as part of a "case study" incorporated into promotional materials.

Please note that the Financial Institutions and their affiliates do not provide tax, accounting or legal advice. Notwithstanding any other provision herein, this Commitment Letter does not limit the disclosure of any tax strategies.

#### 13. Surviving Provisions.

The compensation, reimbursement, indemnification, confidentiality, jurisdiction, governing law and waiver of jury trial provisions contained herein and in the Fee Letters shall remain in full force and effect regardless of whether the DIP ABL Loan Documents shall be executed and delivered and notwithstanding the termination of this Commitment Letter or the Lenders' commitments hereunder and our agreements to perform the services described herein; provided that your obligations under this Commitment Letter and the Fee Letters, other than those provisions relating to confidentiality, compensation and information, shall automatically terminate and be superseded by the definitive documentation relating to the DIP-to-Exit ABL Facility to the extent covered thereby upon the occurrence of the Closing Date, and you shall automatically be released from all liability in connection therewith at such time. The termination of any commitments under this Commitment Letter for any reason shall not prejudice your rights and remedies in respect of any breach of this Commitment Letter.

#### 14. PATRIOT Act Notification.

We hereby notify you that pursuant to the requirements of the USA PATRIOT Act, Title III of Pub. L. 107-56 (signed into law October 26, 2001) (the "PATRIOT Act") and 31 C.F.R. § 1010.230 (the "Beneficial Ownership Regulation"), each Commitment Party and each Lender is required to (a) obtain, verify and record information that identifies the Company, Holdings and the other guarantors, which information includes the name, address, tax identification number and other information regarding the Company, Holdings and the other guarantors that will allow such Lender to identify such person in accordance with the PATRIOT Act and (b) obtain a certification regarding beneficial ownership required by the Beneficial Ownership Regulation (a "Beneficial Ownership Certification") from each Borrower (as defined in the DIP ABL Loan Documents). This notice is given in accordance with the requirements of the PATRIOT Act and the Beneficial Ownership Regulation and is effective as to each Commitment Party and each Lender.

# 15. Acceptance and Termination.

If the foregoing correctly sets forth our agreement with you, please indicate your acceptance of the terms of this Commitment Letter and of the Fee Letters by returning to us executed counterparts hereof and of the Fee Letters not later than 11:59 p.m., New York City time, on February 13, 2023. Each Commitment Party's commitments hereunder, and each Commitment Party's agreements to perform the services described herein, will expire automatically and without further action or notice and without further obligation to you at such time in the event that we have not received such executed counterparts in accordance with the immediately preceding sentence. In the event that (x) the Approval Order (as defined below) is not entered on or before 5:00 p.m., New York City time, on February 17, 2023 or (y) the Closing Date does not occur on or before 5:00 p.m., New York City time, on February 24, 2023, then this Commitment Letter and the commitments of each of the Commitment Parties hereunder, and our agreements to perform the services described herein, shall automatically terminate without further action or notice and without further obligation to you unless each of us shall, in our sole discretion, agree to an extension. You may terminate this Commitment Letter with respect to any or all of the Commitment Parties' commitments hereunder (on a ratable basis) at any time for any reason.

For purposes hereof, "Approval Order" means an order by the Bankruptcy Court in form and substance acceptable to the Commitment Parties approving the Company's entry into and performance under the Commitment Letter and the Fee Letters.

[Remainder of this page intentionally left blank]

# IN THE UNITED STATES BANKRUPTCY COURT FOR THE SOUTHERN DISTRICT OF TEXAS HOUSTON DIVISION

In re:		) Chapter 11
AVAYA INC., et al.,1		) Case No. 23-90088 (DRJ)
	Debtors.	) (Jointly Administered)

INTERIM ORDER
(I) AUTHORIZING THE DEBTORS
(A) TO OBTAIN POSTPETITION FINANCING
PURSUANT TO 11 U.S.C. §§ 105, 361, 362, 363(b),
364(c)(1), 364(c)(2), 364(c)(3), 364(d)(1) AND 364(e), AND (B) TO
UTILIZE CASH COLLATERAL PURSUANT TO 11 U.S.C. § 363,
(II) GRANTING ADEQUATE PROTECTION TO PREPETITION
SECURED PARTIES PURSUANT TO 11 U.S.C. §§ 361, 362, 363, 364,
503, 506(c) AND 507(b), (III) SCHEDULING FINAL HEARING PURSUANT TO
BANKRUPTCY RULES 4001(b) AND (c) and (IV) GRANTING RELATED RELIEF

Upon the motion (the "Motion")<sup>2</sup> of Avaya Inc. (the "Company" or the "Borrower"), Avaya Holdings Corp. ("Holdings"), and the subsidiaries of the Company that are debtors and debtors in possession (the "Subsidiary Guarantors," and, collectively, with the Company and Holdings, the "Loan Parties" or the "Debtors") in the above-captioned cases (the "Chapter 11 Cases"), pursuant to sections 105, 361, 362, 363(b), 363(c)(2), 364(c)(1), 364(c)(2), 364(c)(3), 364(d)(1), 364(e), 503, 506(c) and 507 of title 11 of the United States Code (the "Bankruptcy Code"), and Rules 2002, 4001, 6003, 6004 and 9014 of the Federal Rules of Bankruptcy Procedure

A complete list of each of the Debtors in these Chapter 11 Cases may be obtained on the website of the Debtors' proposed claims and noticing agent at <a href="http://www.kccllc.net/avaya">http://www.kccllc.net/avaya</a>. The location of Debtor Avaya Inc.'s principal place of business and the Debtors' service address in these Chapter 11 Cases is 350 Mount Kemble Avenue, Morristown, New Jersey 07960.

<sup>&</sup>lt;sup>2</sup> Capitalized terms not otherwise defined herein shall have the meanings ascribed to them in the Motion.

(the "Bankruptcy Rules"), and the local bankruptcy rules for the Southern District of Texas (the "Local Bankruptcy Rules"), seeking entry of this interim order (this "Interim Order"):

- a. authorizing the Borrower to obtain senior secured postpetition financing on a superpriority basis (the "DIP Financing" or the "DIP Facility") and for Holdings and the Subsidiary Guarantors (the "Guarantors") to guarantee the Borrower's obligations in connection with the DIP Facility, consisting of a non-amortizing term loan facility in an aggregate principal amount of up to \$500,000,000 (all amounts extended under the DIP Facility, the "DIP Loans"; and the commitments in respect thereof, the "DIP Commitments"); provided that (i) an initial draw in the principal amount of up to \$400,000,000 (the "Initial DIP Loans") shall be made available to the Debtors upon entry of this Interim Order, and (ii) the remaining undrawn portion of the DIP Facility shall be made available to the Debtors upon entry of the Final Order (as defined herein), all on the terms and conditions set forth in the DIP Documents (as defined herein);
- b. authorizing the Borrower and Holdings to enter into that certain DIP-to-Exit ABL Commitment Letter, and the fee letters executed in connection therewith (collectively, the "<u>DIP-to-Exit ABL Commitment Papers</u>") and to pay, on a final and irrevocable basis, the fees, expenses and other amounts payable under the DIP-to-Exit ABL Commitment Papers as such fees become earned, due and payable, including, without limitation, structuring, commitment, upfront and similar fees and legal fees and expenses;
- c. authorizing the Loan Parties to execute and enter into the *Superpriority Secured Debtor in Possession Credit* Agreement dated as of February [●], 2023, among the Borrower, the Guarantors, the lenders party thereto (the "<u>DIP Lenders</u>")<sup>3</sup> and Wilmington Savings Fund Society, FSB, as administrative agent and collateral agent (in such capacities, the "<u>DIP Agent</u>," and, collectively with the DIP Lenders, the "<u>DIP Secured Parties</u>"), substantially in the form attached to the Motion as <u>Exhibit A</u> (as amended, supplemented or otherwise modified from time to time in accordance with the terms hereof and thereof, the "<u>DIP Credit Agreement</u>" and, collectively with the schedules and exhibits attached thereto and all agreements, documents, instruments and/or amendments executed and delivered in connection therewith, the "<u>DIP Documents</u>") and to perform all such other and further acts as may be required in connection with the DIP Documents;
- d. authorizing the Loan Parties to use proceeds of the DIP Facility and Cash Collateral (as defined herein), in accordance with the terms of this Interim Order, the DIP Documents and the DIP Budget (as defined herein),
   (i) to indefeasibly repay in full in cash all of the indebtedness outstanding

2

The DIP Lenders include the lenders that are party to the DIP Commitment Letter, dated February 14, 2023.

under the Prepetition ABL Credit Agreement (as defined herein), including to cash collateralize all obligations in respect of letters of credit issued and secured hedging agreements, (ii) to fund certain intercompany loans to non-Debtor affiliates of the Debtors, (iii) for working capital and general corporate purposes of the Debtors, (iv) to pay obligations arising from or related to the Carve Out (as defined herein), (v) to pay Allowed Professional Fees (as defined herein), (vi) to pay Adequate Protection Obligations (as defined herein), and (vii) to pay fees and expenses incurred in connection with the transactions contemplated hereby;

- e. authorizing Debtor Sierra Communications International LLC ("Sierra Communications") to use proceeds of the Initial DIP Loans to fund an intercompany loan in an amount not to exceed \$50,000,000 to non-Debtor Avaya International Sales Ltd (the "Initial Intercompany Transaction") in accordance with the terms of this Interim Order, the DIP Documents and the Intercompany Transfer Mechanic (as defined herein);
- f. authorizing the Borrower to transfer up to \$40,000,000 of proceeds of the Initial DIP Loans to a segregated account held by Avaya, Inc. for purposes of backstopping the liquidity of certain foreign non-Debtor affiliates to the extent necessary to preserve the value of the Debtors' international business operations (the "Foreign Reserve Account");
- g. granting adequate protection, subject to the Carve Out, and to the extent set forth herein, to the Prepetition Secured Parties under the Existing Agreements on account of the Primed Liens and for the use of their Cash Collateral and the Prepetition Collateral (all as defined herein) to the extent of any diminution in value of their respective interests in the Prepetition Collateral;
- h. authorizing the Debtors to pay, on a final and irrevocable basis, the principal, interest, fees, expenses, and other amounts payable under the DIP Documents as such become earned, due and payable, including, without limitation, the Term Upfront Fee, the Put Option Premium, the Exit Fee (each as defined in the DIP Credit Agreement), and any agency, audit fees, appraisal fees, valuation fees, administrative and collateral agents' fees and expenses, and prepetition and postpetition reasonable fees and disbursements of each of the DIP Secured Parties' attorneys, advisors, accountants, appraisers, bankers and other consultants, all to the extent provided in, and in accordance with, this Interim Order and the DIP Documents;
- i. granting valid, enforceable, non-avoidable, and fully perfected liens and security interests pursuant to Bankruptcy Code section 364(c)(2) and priming liens pursuant to Bankruptcy Code section 364(d)(1) on the DIP Collateral and all proceeds thereof, including, any (subject to entry of the Final Order) Avoidance Proceeds, subject only to the Carve Out, and the

Permitted Liens (all as defined herein), if any, in each case on the terms and conditions set forth herein and in the DIP Documents, as applicable, to secure principal of, and accrued interest on, the DIP Loans, and all other fees, costs, expenses, indemnification obligations, reimbursement obligations, charges, premiums, if any, additional interest, and all other obligations of whatever nature owing, whether or not contingent, whenever arising, accrued, accruing, due, owing or chargeable under the DIP Facility (with respect to the aforementioned obligations the "DIP Obligations");

- j. granting superpriority administrative expense claims pursuant to Bankruptcy Code section 364(c)(1) against each of the Debtors' estates to the DIP Secured Parties, with respect to the DIP Obligations with priority over any and all administrative expenses of any kind or nature subject and subordinate only to the Carve Out on the terms and conditions set forth herein and in the DIP Documents:
- k. (i) the waiver of the Debtors' and the estates' ability to surcharge the DIP Collateral pursuant to Bankruptcy Code section 506(c) with respect to the DIP Secured Parties, and (ii) subject to and upon entry of a Final Order, the waiver of (x) the Debtors' and the estates' ability to surcharge against the Prepetition Collateral pursuant to Bankruptcy Code section 506(c) with respect to the Prepetition Secured Parties, effective as of the Petition Date, (y) the applicability of the "equities of the case" exception under Bankruptcy Code section 552(b) with respect to the proceeds, products, offspring or profits of the Prepetition Collateral, and (z) the doctrine of "marshaling" and any other similar equitable doctrine with respect to any of the Prepetition Collateral and the DIP Collateral;
- 1. subject to the Remedies Notice Period, authorization for the DIP Secured Parties to exercise remedies under the DIP Documents on the terms described herein and therein, upon the occurrence and during the continuation of a DIP Termination Date (as defined herein);
- m. waiving any applicable stay (including under Bankruptcy Rule 6004) and providing for immediate effectiveness of this Interim Order; and
- n. scheduling a final hearing (the "<u>Final Hearing</u>") to consider the relief requested in the Motion and entry of a final order authorizing and approving the relief requested in the Motion on a final basis (the "<u>Final Order</u>"), and approving the form of notice with respect to the Final Hearing.

Notice of the Motion and the Interim Hearing having been served by the Debtors as set forth in the affidavit of service filed at (Docket No. [●]), and it appearing that such notice was the best available under the circumstances; and the Court having reviewed the Motion; and the Interim

Hearing having been held by the Court on February [•], 2023; and the relief requested in the Motion being in the best interests of the Debtors, their creditors and their estates and all other parties in interest in these Chapter 11 Cases; and the Court having found and determined that the relief requested in the Motion is necessary to avoid immediate and irreparable loss and damage to the Debtors' estates; and the Court having determined that the legal and factual bases set forth in the Motion establish just cause for the relief granted herein; and upon the record made by the DIP Declarations and the First Day Declaration (collectively, the "Declarations"), and at the Interim Hearing and after due deliberation and sufficient cause appearing therefor;

#### IT IS FOUND, DETERMINED, ORDERED AND ADJUDGED, that:

- 1. *Disposition*. The relief requested in the Motion is granted on an interim basis in accordance with the terms of this Interim Order. Any objections to the Motion with respect to the entry of this Interim Order that have not been withdrawn, waived or settled, and all reservations of rights included therein, are hereby denied and overruled on the merits. This Interim Order shall become effective immediately upon its entry.
- 2. *Jurisdiction*. This Court has core jurisdiction over the Chapter 11 Cases, the relief requested in the Motion, and the parties and property affected hereby pursuant to 28 U.S.C. §§ 157(b) and 1334. Venue is proper before this Court pursuant to 28 U.S.C. §§ 1408 and 1409.
- 3. Relief Necessary to Avoid Immediate and Irreparable Harm. The interim relief granted herein is necessary to avoid immediate and irreparable harm to the Debtors and their estates pending the Final Hearing.
- 4. *Debtors' Stipulations*. Without prejudice to the rights of any party in interest (but subject in all respects to the limitations set forth in Paragraphs 29 and 31 herein) the Debtors admit, stipulate, acknowledge and agree that:

#### (a) Prepetition ABL Facility.

December 15, 2017, by and among the Company, certain other borrowers party thereto (together with the Company, the "Prepetition ABL Borrowers"), Citibank, N.A., as administrative agent and collateral agent (in such capacities, the "Prepetition ABL Agent"), and each lender from time to time party thereto (the "Prepetition ABL Lenders," and, together with the Prepetition ABL Agent, the "Prepetition ABL Secured Parties") (as amended, restated, amended and restated, supplemented or otherwise modified from time to time prior to the date hereof, the "Prepetition ABL Credit Agreement," and, together with the security agreements, pledge agreements, mortgages, deeds of trust and other security documents executed by any of the Prepetition ABL Borrowers in favor of the Prepetition ABL Secured Parties, the "Prepetition ABL Credit Documents," and such credit facility thereunder, the "Prepetition ABL Facility"), the Prepetition ABL Secured Parties provided the Prepetition ABL Borrowers with an asset based credit facility. Each of the Prepetition ABL Credit Documents is valid, binding, and enforceable in accordance with its terms.

Date"), the Prepetition ABL Borrowers and the Prepetition Guarantors (as defined herein) were justly and lawfully indebted and liable to the Prepetition ABL Secured Parties, without defense, counterclaim or offset of any kind, in the aggregate principal amount of approximately \$56 million in respect of loans made and approximately \$40 million in respect of letters of credit issued pursuant to, and in accordance with the terms of, the Prepetition ABL Credit Documents, plus accrued and unpaid interest thereon and fees, expenses (including any attorneys', accountants', appraisers' and financial advisors' fees, in each case, that are chargeable or reimbursable under

the Prepetition ABL Credit Documents), charges, indemnities and other obligations incurred in connection therewith (whether arising before, on or after the Petition Date) as provided in the Prepetition ABL Credit Agreement (collectively, the "Prepetition ABL Obligations"), which Prepetition ABL Obligations have been guaranteed on a joint and several basis by Holdings and all of the Subsidiary Guarantors (other than Sierra Communications and CTIntegrations, LLC) (the "Prepetition Guarantors").

(iii) The Prepetition ABL Obligations constitute the legal, valid and binding obligations of the Prepetition ABL Borrowers and the Prepetition Guarantors, enforceable in accordance with the terms of the Prepetition ABL Credit Documents (other than in respect of the stay of enforcement arising from Bankruptcy Code section 362); and no portion of the Prepetition ABL Obligations or any payments made to the Prepetition ABL Secured Parties or applied to or paid on account of the obligations owing under the Prepetition ABL Credit Documents, prior to the Petition Date is subject to any contest, attack, rejection, recovery, recoupment, reduction, defense, counterclaim, offset, subordination, recharacterization, avoidance or other claim, cause of action or other challenge of any nature under the Bankruptcy Code or applicable non-bankruptcy law.

(iv) The liens and security interests granted to the Prepetition ABL Secured Parties (the "Prepetition ABL Liens") pursuant to and in connection with the Prepetition ABL Credit Documents are: (i) valid, binding, perfected, enforceable, first-priority liens and security interests in the Prepetition ABL Priority Collateral; <sup>4</sup> (ii) valid, binding, perfected,

<sup>&</sup>lt;sup>4</sup> "<u>Prepetition ABL Priority Collateral</u>" has the meaning given to "<u>ABL Priority Collateral</u>" in the ABL Intercreditor Agreement (as defined herein).

enforceable, second-priority liens and security interests in the Prepetition Term Priority Collateral<sup>5</sup> (together with the Prepetition ABL Priority Collateral, the "<u>Prepetition Collateral</u>"); (iii) not subject to avoidance, recharacterization, subordination, recovery, attack, effect, counterclaim, defense or Claim under the Bankruptcy Code or applicable non-bankruptcy law; and (iv) as of the Petition Date, with respect to Prepetition Collateral that is Prepetition Term Priority Collateral, subject and subordinate to the Prepetition First-Priority Liens (as defined herein).

- (v) The aggregate value of the Prepetition ABL Priority Collateral substantially exceeds the aggregate amount of the Prepetition ABL Obligations.
- (vi) None of the Debtors has any claims or causes of action against, or with respect to, the Prepetition ABL Secured Parties as of the Petition Date.

# (b) Prepetition Term Loan Facility.

December 15, 2017, by and among the Company, Holdings, Goldman Sachs Bank USA, as administrative agent and collateral agent ("Goldman" and, in such capacities, the "Prepetition Term Loan Agent"), the other parties thereto and each lender from time to time party thereto (the "Prepetition Term Loan Lenders," and, together with the Prepetition Term Loan Agent, the "Prepetition Term Loan Secured Parties") (as amended, restated, amended and restated, supplemented or otherwise modified from time to time prior to the date hereof, the "Prepetition Term Loan Credit Agreement," and together with any "Credit Documents," as defined in the Prepetition Term Loan Credit Agreement, and any guarantees, security agreements, pledge agreements, mortgages, deeds of trust and other security documents executed by any of the Debtors

8

<sup>&</sup>lt;sup>5</sup> "<u>Prepetition Term Priority Collateral</u>" has the meaning given to "<u>Term Priority Collateral</u>" in the ABL Intercreditor Agreement.

in favor of the Prepetition Term Loan Agent, for its benefit and for the benefit of the Prepetition Term Loan Lenders, the "Prepetition Term Loan Credit Documents"), which provided a term loan facility pursuant to which the Prepetition Term Loan Secured Parties made loans to the Borrower comprising the Tranche B-1 Term Loans, the Tranche B-2 Term Loans and the Tranche B-3 Term Loans (each as defined in the Prepetition Term Loan Credit Agreement) (collectively, the "Prepetition Term Loans"). Each of the Prepetition Term Loan Credit Documents is valid, binding, and enforceable in accordance with its terms. No less than \$220,606,000 of the proceeds from the issuance of the Tranche B-3 Term Loans is held in escrow pursuant to that certain Escrow Agreement (as defined in the RSA).

(ii) As of the Petition Date, the Company and the Prepetition Guarantors were justly and lawfully indebted and liable to the Prepetition Term Loan Secured Parties in respect of the (A) Legacy Term Loans in the aggregate principal amount of no less than \$1,543,000,000, plus accrued and unpaid interest thereon and fees, expenses (including any attorneys', accountants', appraisers' and financial advisors' fees, in each case, that are chargeable or reimbursable under the Prepetition Term Loan Credit Documents), charges, indemnities and other obligations incurred in connection therewith as provided in the Prepetition Term Loan Credit Documents (collectively, the "Prepetition Legacy Term Loan Obligations") and (B) Tranche B-3 Term Loans in the aggregate principal amount of no less than \$350,000,000, plus accrued and unpaid interest thereon and fees, expenses (including any attorneys', accountants', appraisers' and financial advisors' fees, in each case, that are chargeable or reimbursable under the Prepetition Term Loan Credit Documents), charges, indemnities and other obligations incurred in connection therewith as provided in the Prepetition Term Loan Credit Documents (the "Prepetition Tranche

<sup>&</sup>lt;sup>6</sup> The Tranche B-1 Term Loans and the Tranche B-2 Term Loans are, together, the "Legacy Term Loans."

<u>B-3 Term Loan Obligations</u>," and together with the Prepetition Legacy Term Loan Obligations, the "Prepetition Term Loan Obligations").

# (c) <u>Prepetition Legacy Notes.</u>

- (i) Pursuant to that certain Indenture, dated as of September 25, 2020, by and among the Company, the guarantors party thereto and Wilmington Trust, National Association, as trustee and as notes collateral agent (the "Prepetition Legacy Notes Trustee") (as amended, restated, amended and restated, supplemented or otherwise modified from time to time prior to the date hereof, the "Prepetition Legacy Notes Indenture," and, together with any supplements, security agreements, pledge agreements, mortgages, deeds of trust and other security documents executed by any of the Debtors in favor of the Prepetition Legacy Notes Trustee, for its benefit and for the benefit of the noteholders (in such capacities, the "Prepetition Legacy Notes Parties"), the "Prepetition Legacy Notes Documents"), the Company issued those certain 6.125% senior secured first lien notes (the "Prepetition Legacy Notes"). Each of the Prepetition Legacy Notes Documents is valid, binding, and enforceable in accordance with its terms.
- (ii) <u>Prepetition Legacy Notes Obligations</u>. As of the Petition Date, the Company and the Prepetition Guarantors were justly and lawfully indebted and liable to the Prepetition Legacy Notes Parties in the aggregate principal amount of no less than \$1,000,000,000, plus accrued and unpaid interest thereon and fees, expenses (that are chargeable or reimbursable under the Prepetition Legacy Notes Documents) and the Applicable Premium (as defined in the Prepetition Legacy Notes Indenture) (collectively, the "<u>Prepetition Legacy Notes Obligations</u>").

# (d) <u>Prepetition Secured Exchangeable Notes.</u>

(i) Pursuant to that certain Indenture, dated as of July 12, 2022, by and among the Company, the Prepetition Guarantors, and Wilmington Trust, National Association as

Trustee, exchange agent and notes collateral agent (the "Prepetition Secured Exchangeable Notes Trustee," and, together with the Prepetition Legacy Notes Trustee, the Prepetition Term Loan Agent, the "Prepetition First Lien Agents") (as amended, restated, amended and restated, supplemented or otherwise modified from time to time prior to the date hereof, the "Prepetition Secured Exchangeable Notes Indenture," and, together with any supplements, security agreements, pledge agreements, mortgages, deeds of trust and other security documents executed by any of the Debtors in favor of the Prepetition Secured Exchangeable Notes Trustee, for its benefit and for the benefit of the noteholders (in such capacities, the "Prepetition Secured Exchangeable Notes Parties," and, together with the Prepetition Term Loan Secured Parties and the Prepetition Legacy Notes Parties, the "Prepetition Secured Parties"), the "Prepetition Secured Exchangeable Notes Documents"), the Company issued those certain 8.00% senior secured first lien notes (the "Prepetition Secured Exchangeable Notes"). Each of the Prepetition Secured Exchangeable Notes Documents is valid, binding and enforceable in accordance with its terms.

- (ii) As of the Petition Date, the Company and the Prepetition Guarantors were justly and lawfully indebted and liable to the Prepetition Secured Exchangeable Notes Parties in the aggregate principal amount of no less than \$250,000,000 plus accrued and unpaid interest thereon and fees, expenses (that are chargeable or reimbursable under the Prepetition Secured Exchangeable Notes Documents) and the Notes Special Premium (as defined in the Prepetition Secured Exchangeable Notes Indenture) (collectively, the "Prepetition Secured Exchangeable Notes Obligations," and, together with the Prepetition Term Loan Obligations and the Prepetition Legacy Notes Obligations, the "Prepetition First Lien Obligations").
- (e) The Prepetition First Lien Obligations constitute the legal, valid and binding obligations of the Company and the Prepetition Guarantors, as applicable, enforceable in

accordance with their terms (other than in respect of the stay of enforcement arising from Bankruptcy Code section 362); and, other than as set forth in this Interim Order, no portion of the Prepetition First Lien Obligations or any payments made to the Prepetition Secured Parties or applied to or paid on account of the obligations owing under the Prepetition Term Loan Credit Documents, Prepetition Legacy Notes Documents and the Prepetition Secured Exchangeable Notes Documents (collectively, the "Existing Agreements") prior to the Petition Date is subject to any contest, attack, rejection, recovery, recoupment, reduction, defense, counterclaim, offset, subordination, recharacterization, avoidance or other claim, cause of action or other challenge of any nature under the Bankruptcy Code or applicable non-bankruptcy law.

- (f) The liens and security interests granted to the Prepetition Secured Parties pursuant to and in connection with the Existing Agreements (the "Prepetition First-Priority Liens" or the "Primed Liens") are: (i) valid, binding, perfected, enforceable, first-priority liens and security interests in the Prepetition Term Priority Collateral; (ii) valid, binding, perfected, enforceable, second-priority liens, and security interests in the Prepetition ABL Priority Collateral; (iii) not subject to avoidance, recharacterization, subordination, recovery, attack, effect, counterclaim, defense, or Claim under the Bankruptcy Code or applicable non-bankruptcy law; and (iv) as of the Petition Date, with respect to Prepetition Collateral that is Prepetition ABL Priority Collateral, subject and subordinate to the liens and security interests in favor of the Prepetition ABL Secured Parties.
- (g) As of the Petition Date, the Primed Liens were valid, binding, enforceable, non-avoidance, and properly perfected, and were senior in priority over any and all other liens on the Prepetition Collateral, subject only to certain senior liens senior by operation of law or as permitted by the Prepetition ABL Credit Documents and the Existing Agreements (solely to the

extent any such permitted liens were valid, properly perfected, non-avoidable, and senior in priority to the Primed Liens as of the Petition Date or were in existence immediately prior to the Petition Date that were perfected subsequent to the Petition Date as permitted by Bankruptcy Code section 546(b)) (such liens, "Permitted Liens").

- (h) [Reserved].
- (i) Each of (i) that certain ABL Intercreditor Agreement, dated as of December 15, 2017, among the Company, the Prepetition ABL Agent, the Prepetition Term Loan Agent and the Prepetition Legacy Notes Trustee, and certain other parties signatory thereto (as amended, supplemented or otherwise modified prior to the date hereof, the "ABL Intercreditor Agreement") and (ii) the First Lien Pari Intercreditor Agreement, dated as of September 25, 2020, by and among the Company, the Prepetition Term Loan Agent, the Prepetition Legacy Notes Trustee, the Prepetition Secured Exchangeable Notes Trustee and certain other parties signatory thereto (as amended, supplemented or otherwise modified prior to the date hereof, the "First Lien Pari Intercreditor Agreement," and, together with the ABL Intercreditor Agreement, the "ICAs") are binding and enforceable against the Company and the applicable Prepetition Guarantors in accordance with their terms, and the Company and the applicable Prepetition Guarantors are not entitled to take any action that would be contrary to the provisions thereof.
- (j) All cash, securities, cash equivalents and other property of the Company and the Prepetition Guarantors (and the proceeds therefrom) as of the Petition Date, including, without limitation, all cash proceeds of the Prepetition Collateral (including cash on deposit at the Depository Institutions as of the Petition Date, securities or other property, whether subject to control agreements or otherwise, in each case that constitutes Prepetition Collateral) and all cash securities or other property (and the proceeds therefrom) and other amounts on deposit or

maintained by such parties in any account or accounts with any depository institution (collectively, the "Depository Institutions"), were subject to rights of set-off and (i) valid, perfected, enforceable, first priority liens under the Prepetition ABL Credit Documents and applicable law and (ii) valid, perfected, enforceable, second-priority liens under the Existing Agreements and applicable law, as applicable, for the benefit of the Prepetition ABL Secured Parties and the Prepetition Secured Parties, respectively, and are "cash collateral" of the Prepetition ABL Secured Parties and the Prepetition Secured Parties within the meaning of Bankruptcy Code section 363(a) (the "Cash Collateral").

- 5. Findings Regarding the DIP Financing and Cash Collateral.
- (a) Good and sufficient cause has been shown for the entry of this Interim Order and to avoid immediate and irreparable loss or damage to the Debtors' estates.
- (b) The Debtors have an immediate need to obtain the DIP Financing and continue to use the Prepetition Collateral (including Cash Collateral) in order to, among other things, avoid the liquidation of their estates, and to permit the orderly continuation of the operation of their businesses, to maintain business relationships with vendors, suppliers and customers, to make payroll, to make capital expenditures, to pay adequate protection, to indefeasibly repay the Prepetition ABL Obligations and to satisfy other working capital and operational needs.
- (c) The Debtors are unable to obtain financing on more favorable terms from sources other than the DIP Lenders under the DIP Documents and are unable to obtain adequate unsecured credit allowable under Bankruptcy Code section 503(b)(1) as an administrative expense. The Debtors are also unable to obtain secured credit allowable under Bankruptcy Code sections 364(c)(1), 364(c)(2) and 364(c)(3) without granting to the DIP Agent and the DIP Lenders, subject to the Carve Out and the Permitted Liens, the DIP Liens (as defined herein) and the DIP

Superpriority Claims (as defined herein) and incurring the Adequate Protection Obligations (as defined herein), in each case, under the terms and conditions set forth in this Interim Order and in the DIP Documents.

- (d) The Debtors have an immediate need to fund the Initial Intercompany Transaction and to establish the Foreign Reserve Account in order to ensure the continuing operation of certain foreign non-Debtor subsidiaries in the ordinary course and thereby preserving the value of the Debtors' interests in the foreign non-Debtor subsidiaries for the overall benefit of the Debtors' estates and creditors.
- (e) Based on the Motion, the Declarations filed in support of the Motion, and the record presented to the Court at the Interim Hearing (including the First Day Declaration), the terms of the DIP Financing and the terms on which the Debtors may continue to use the Prepetition Collateral (including Cash Collateral) pursuant to this Interim Order and the DIP Documents are fair and reasonable, reflect the Debtors' exercise of prudent business judgment consistent with their fiduciary duties and constitute reasonably equivalent value and fair consideration.
- (f) The Prepetition ABL Repayment (as defined below) reflects the Debtors' exercise of prudent business judgment consistent with their fiduciary duties.
- (g) The Prepetition Secured Parties have consented or are deemed to have consented to the Debtors' use of Cash Collateral and the other Prepetition Collateral (solely in accordance with the terms of this Interim Order and the DIP Documents), and the Loan Parties' entry into the DIP Documents in accordance with and subject to the terms and conditions set forth in this Interim Order and the DIP Documents.
- (h) The DIP Financing (including the Prepetition ABL Repayment) and the use of the Prepetition Collateral (including Cash Collateral) have been negotiated in good faith and at

arm's-length among the Debtors, the DIP Agent, the DIP Lenders, the Prepetition ABL Secured Parties, and the Prepetition Secured Parties and all of the Loan Parties' obligations and indebtedness arising under, in respect of, or in connection with, the DIP Financing and the DIP Documents, including, without limitation: (i) all DIP Loans made to and guarantees issued by the Debtors pursuant to the DIP Documents; (ii) any Cash Management Obligations (as defined in the DIP Documents); (iii) any Hedging Obligations (as defined in the DIP Documents); and (iv) any other DIP Obligations, shall be deemed to have been extended by the DIP Agent and the DIP Lenders in good faith, as that term is used in Bankruptcy Code section 364(e) and in express reliance upon the protections offered by Bankruptcy Code section 364(e), and the DIP Agent and the DIP Lenders (and the successors and assigns thereof) shall be entitled to the full protection of Bankruptcy Code section 364(e) in the event that this Interim Order or any provision hereof is vacated, reversed or modified, on appeal or otherwise.

- (i) The Prepetition ABL Secured Parties and the Prepetition Secured Parties have acted in good faith regarding the DIP Financing (including the Prepetition ABL Repayment) and the Debtors' use of the Prepetition Collateral (including the Cash Collateral) to fund the administration of the Debtors' estates and continued operation of their businesses (including the incurrence and payment of the Adequate Protection Obligations), in accordance with the terms hereof, and the Prepetition Secured Parties (and the successors and assigns thereof), shall be entitled to the full protection of Bankruptcy Code section 363(m) in the event that this Interim Order or any provision hereof is vacated, reversed or modified, on appeal or otherwise.
- (j) The Prepetition Secured Parties are entitled to the adequate protection provided in this Interim Order as and to the extent set forth herein pursuant to Bankruptcy Code sections 361, 362, 363 and 364. Based on the Motion, the Declarations and on the record presented

to the Court, the terms of the proposed adequate protection arrangements for the use of and diminution of value of the Prepetition Collateral (including the Cash Collateral), if any, are fair and reasonable, and reflect the Debtors' prudent exercise of business judgment; *provided* that, nothing in this Interim Order or the other DIP Documents shall (w) be construed as the affirmative consent by any of the Prepetition Secured Parties for the use of Cash Collateral, other than on the terms set forth in this Interim Order and in the context of the DIP Financing authorized by this Interim Order, (x) be construed as a consent by any party to the terms of any other financing or any other lien encumbering the Prepetition Collateral (whether senior or junior), (y) prejudice, limit or otherwise impair the rights of any of the Prepetition Secured Parties, subject to any applicable provisions of the ICAs, to seek new, different, or additional adequate protection or assert the interests of any of the Prepetition Secured Parties, or (z) in the event of any such request for new, different, or additional relief per clause (y) all parties' rights (including the Debtors') to oppose such relief are fully reserved.

- 6. Authorization of the DIP Financing, the DIP Documents and the DIP-to-Exit ABL Commitment Papers.
- (a) The Loan Parties are hereby authorized to execute, enter into and perform all obligations under the DIP Documents and the DIP-to-Exit ABL Commitment Papers. The Borrower is hereby authorized pursuant to this Interim Order to forthwith borrow money pursuant to the DIP Credit Agreement, and the Guarantors are hereby authorized to guarantee the Borrower's obligations with respect to such borrowings as described herein, subject to any conditions and limitations under the DIP Documents, which shall be used for all purposes as outlined in Paragraphs d through f herein and under the DIP Documents, including, without limitation and as applicable, to indefeasibly repay in full, in cash all Prepetition ABL Obligations,

to fund the Foreign Reserve Account, to fund the Initial Intercompany Transaction, to provide working capital for the Debtors, to pay Adequate Protection Obligations, for general corporate purposes, to pay interest, fees and expenses in accordance with this Interim Order and the DIP Documents, to fund the Carve Out and to pay Allowed Professional Fees.

- (b) In furtherance of the foregoing and without further approval of this Court, each Loan Party is authorized to perform all acts, to make, execute and deliver all instruments and documents (including, without limitation, the execution or recordation of security agreements, mortgages and financing statements), and to pay all fees required under the DIP Documents and the DIP-to-Exit ABL Commitment Papers or that may otherwise be reasonably necessary for or in connection with the Loan Parties' performance of their obligations under the DIP Documents and the DIP-to-Exit ABL Commitment Papers, including as applicable and, without limitation:
- (i) the execution and delivery of, and performance under, each of the DIP Documents and the DIP-to-Exit ABL Commitment Papers;
- (ii) the execution and delivery of, and performance under, one or more amendments, waivers, consents or other modifications to and under the DIP Documents, in each case, in such form as the requisite parties under the DIP Documents may agree, it being understood that no further approval of the Court shall be required for authorizations, amendments, waivers, consents, or other modifications to and under the DIP Documents that are either non-material or not adverse to the Debtors (and any fees and other expenses (including any attorneys', accountants', appraisers', and financial advisors' fees), amounts, charges, costs, indemnities, and other obligations paid in accordance and connection therewith, but excluding, for the avoidance of doubt, any amendment, consent, or waiver fee). In the case of material amendments, waivers, consents or other modifications to the DIP Documents, the Debtors shall provide notice (which may be

provided through electronic mail or facsimile) to the lead counsel to the Creditors' Committee, respective counsel to the Akin Ad Hoc Group (as defined in the RSA) and the PW Ad Hoc Group (as defined in the RSA), and the U.S. Trustee, which parties shall have ten (10) Business Days from the date of such notice within which to object, in writing, to such material amendment, waiver, consent or other modification. If any such party timely objects to such material amendment, waiver, consent or other modification to the DIP Documents, such material amendment, waiver, consent or other modification shall only be permitted pursuant to an order of the Court. The foregoing shall be without prejudice to the Debtors' right to seek approval from the Court of a material amendment, waiver, consent or other modification on an expedited basis. For the avoidance of doubt, the extension of a Milestone (as defined in the DIP Credit Agreement) or the delivery of an updated DIP Budget shall not constitute a material amendment, modification, waiver, or supplement to the DIP Documents;

(iii) the non-refundable payment to the DIP Agent or the DIP Lenders, as the case may be, of all fees (which fees shall be, and shall be deemed to have been, approved upon entry of this Interim Order and, upon payment thereof, in accordance with the terms of the DIP Documents and this Interim Order, shall not be subject to any contest, attack, rejection, recoupment, reduction, defense, counterclaim, offset, subordination, recharacterization, avoidance or other claim, cause of action or other challenge of any nature under the Bankruptcy Code, under applicable non-bankruptcy law or otherwise), and any amounts due (or that may become due) in respect of the indemnification obligations, in each case referred to in the DIP Documents (and in any separate letter agreements between any or all Debtors, on the one hand, and the DIP Agent and/or DIP Lenders, on the other, in connection with the DIP Financing) and the DIP-to-Exit ABL Commitment Papers and the costs and expenses as may be due from time to time, including,

without limitation, fees and expenses of the professionals retained by the DIP Agent, the DIP Lenders and the Commitment Parties (as defined in the DIP-to-Exit ABL Commitment Papers), in each case, as provided for in this Interim Order, the DIP Documents and the DIP-to-Exit ABL Commitment Papers, without the need to file retention motions or fee applications or to provide notice to any party;

- (iv) the creation of intercompany loans in accordance with the Intercompany Transfer Mechanic; and
- (v) the performance of all other acts required under or in connection with the DIP Documents and the DIP-to-Exit ABL Commitment Papers.
- (c) Upon execution and delivery thereof, the DIP Documents and the DIP-to-Exit ABL Commitment Papers shall constitute valid, binding and unavoidable obligations of the Loan Parties, enforceable against each Loan Party thereto in accordance with the terms of the DIP Documents, the DIP-to-Exit ABL Commitment Papers and this Interim Order. No obligation, payment, transfer or grant of security under the DIP Documents or this Interim Order to the DIP Agent and/or the DIP Lenders shall be stayed, restrained, voidable or recoverable under the Bankruptcy Code or under any applicable law (including, without limitation, under Bankruptcy Code sections 502(d), 548 or 549), or subject to any defense, reduction, setoff, recoupment, claim or counterclaim.
  - 7. *Indefeasible Payment of the Prepetition ABL Obligations.*
- (a) Upon entry of this Interim Order, the Debtors shall use the proceeds of the Initial DIP Loans to (i) indefeasibly pay in full in cash Prepetition ABL Obligations other than in respect of letters of credit, secured hedging obligations and secured cash management obligations and (ii) cash collateralize all (A) letters of credit issued pursuant to the Prepetition ABL Credit

Documents; (B) obligations in respect of outstanding Secured Hedging Agreements (as defined in the Prepetition ABL Credit Agreement); and (C) obligations in respect of outstanding Secured Cash Management Agreements (as defined in the Prepetition ABL Credit Agreement), in each case on terms satisfactory to the applicable L/C Issuer, Hedge Bank and Cash Management Bank (each as defined in the Prepetition ABL Credit Agreement) (the "Prepetition ABL Repayment").

- (b) In furtherance of the Prepetition ABL Repayment, the Debtors are hereby authorized to (i) establish cash collateral accounts under the respective control of each L/C Issuer, Hedge Bank and Cash Management Bank (the "Cash Collateral Accounts"), (ii) deposit proceeds of the Initial DIP Loans into the Cash Collateral Accounts, (iii) enter into cash collateral agreements with each L/C Issuer, Hedge Bank and Cash Management Bank governing the terms pursuant to which the applicable Prepetition ABL Obligations will be cash collateralized (the "Cash Collateral Agreements") and (iv) perform all acts and pay all fees that may be reasonably necessary for or in connection with the Debtors' performance of their obligations under the Cash Collateral Agreements.
  - (c) Upon the occurrence of the Prepetition ABL Repayment,
- (i) the liens and security interests granted to the Prepetition ABL Secured Parties pursuant to and in connection with the Prepetition ABL Security Documents shall be automatically released and terminated;
- (ii) the Cash Collateral Agreements shall constitute valid, binding and enforceable postpetition agreements of the Debtors party thereto;
- (iii) pursuant to Bankruptcy Code section 364(c)(2), each of the L/C Issuers, Hedge Banks and Cash Management Banks shall have an exclusive, valid, binding, continuing, enforceable, fully-perfected, first priority security interest in and lien upon each Cash

Collateral Account subject to its control, including: (A) all funds held in each Cash Collateral Account or credited thereto, all rights to renew or withdraw the same, and all certificates and instruments, if any, from time to time representing or evidencing the Cash Collateral Account; (B) any notes, certificates of deposit, instruments, financial assets (as defined in Section 8-102(9) of the Uniform Commercial Code) or investment property evidencing or arising out of investment of any funds held in or credited to the Cash Collateral Account pursuant to the applicable Cash Collateral Agreement or otherwise held in the Cash Collateral Account; (C) any interest, dividends, cash, instruments and other property from time to time received, receivable or otherwise distributed in respect of or in exchange for any or all of the then existing collateral described in the foregoing clauses (A) and (B); and (D) all proceeds of any and all of the foregoing, which liens shall secure the relevant L/C Obligations (as defined in the Prepetition ABL Credit Agreement) and Secured Hedging Agreements (such liens, the "Cash Collateral Liens").

the extent necessary to permit each Hedge Bank, L/C Issuer and Cash Management Bank to direct the transfer, redemption, disposition or setoff of deposits or other assets in the Cash Collateral Accounts to the extent permitted by the applicable Cash Collateral Agreement, including to (i) reimburse the L/C Issuer for amounts drawn pursuant to any of the letters of credit secured by such Cash Collateral Agreement, (ii) pay amounts due and owing to any Hedge Bank in connection with any foreign exchange transaction, (iii) pay amounts due and owing to any Cash Management Bank in respect of Cash Management Obligations (as defined in the Prepetition ABL Credit Agreement) under a Secured Cash Management Agreement and (iv) pay fees or other amounts owed to the L/C Issuer, Hedge Bank or Cash Management Bank from time to time.

- (e) The Prepetition ABL Repayment shall be indefeasible upon entry of this Interim Order and not subject to the Challenge Period (as defined herein). Except as set forth in each applicable Cash Collateral Agreement, the Debtors shall not be authorized to use Cash Collateral in the Cash Collateral Accounts. The Cash Collateral Accounts shall constitute Excluded Assets and shall not be subject to the DIP Liens; *provided* that any proceeds released from the Cash Collateral Accounts shall be subject to the DIP Liens. Notwithstanding anything to the contrary in this Interim Order, the Cash Collateral Accounts shall not be subject to the Carve Out and shall not be used to fund the Carve Out Reserves.
- 8. B-3 Escrow Claims. Upon the entry of this Interim Order, the Debtors shall be authorized to execute and deliver an escrow release agreement providing for the termination of the Escrow Agreement, and, upon such termination of the Escrow Agreement, the Escrow Agent (as defined in the RSA) shall be authorized to release all of the Escrow Cash (as defined in the RSA) to the Prepetition Term Loan Agent on account of the Escrow B-3 Claims, and the Prepetition Term Loan Agent shall be authorized to forthwith remit the Escrow Cash without any holdback to the holders of the Tranche B-3 Term Loans, in accordance with the terms and conditions set forth in Paragraph 4.04 of the RSA (the "Escrow Release"); provided, however, that to the extent there is a successful Challenge to the validity of the Escrow Release, the Court shall fashion an appropriate equitable remedy with respect to such Escrow Release. Notwithstanding anything to the contrary in this Interim Order (including paragraph 30), (i) the Escrow Agent and the Prepetition Term Loan Agent, solely in their capacities as such, shall not incur any liability in connection with the Escrow Release; (ii) the indemnification, expense reimbursement, and other similar provisions in the Escrow Agreement shall be fully enforceable by the Escrow Agent against each of the Debtors and survive termination of the Escrow Agreement; (iii) any claims of the

Escrow Agent or the Prepetition Term Loan Agent against the Debtors under the indemnification, expense reimbursement, and other similar provisions in the Escrow Agreement or the other Prepetition Term Loan Credit Documents, in each case solely in connection with the Escrow Release, shall constitute allowed administrative expense claims against each of the Debtors under sections 503(b) and 507 of the Bankruptcy Code; and (iv) the indemnification, expense reimbursement, and other similar provisions in the Escrow Agreement and the other Prepetition Term Loan Credit Documents shall apply to any action taken by the Escrow Agent or the Prepetition Term Loan Agent in connection with the Escrow Release.

9. DIP Superpriority Claims. Subject in all respects to the Carve Out, pursuant to Bankruptcy Code section 364(c)(1), all of the DIP Obligations shall constitute allowed superpriority administrative expense claims against the Loan Parties in each of the Chapter 11 Cases (without the need to file any proof of claim) with priority over any and all claims against the Loan Parties, now existing or hereafter arising, of any kind whatsoever, including, without limitation, all administrative expenses of the kind specified in Bankruptcy Code sections 503(b) and 507(b) and any and all administrative expenses or other claims arising under Bankruptcy Code sections 105, 326, 328, 330, 331, 365, 503(b), 506(c), 507(a), 507(b), 726, 1113 or 1114 (including the Adequate Protection Obligations), whether or not such expenses or claims may become secured by a judgment lien or other non-consensual lien, levy or attachment, which allowed claims (the "DIP Superpriority Claims") shall for purposes of Bankruptcy Code section 1129(a)(9)(A) be considered administrative expenses allowed under Bankruptcy Code section 503(b), and which DIP Superpriority Claims shall be payable from and have recourse to all pre- and postpetition property of the Loan Parties and all proceeds thereof (excluding Avoidance Actions but including, in respect of DIP Superpriority Claims on account DIP Obligations only, effective upon entry of the Final Order, Avoidance Proceeds), subject only to the Carve Out. The DIP Superpriority Claims shall be entitled to the full protection of Bankruptcy Code section 364(e) in the event that this Interim Order or any provision hereof is vacated, reversed or modified, on appeal or otherwise.

- 10. *DIP Liens*. As security for the DIP Obligations, subject and subordinate in all respects to the Carve Out, effective and perfected upon the entry of this Interim Order and without the necessity of the execution, recordation or filing by the Loan Parties, the DIP Agent or any DIP Lender of mortgages, security agreements, control agreements, pledge agreements, financing statements or other similar documents, or the possession or control by the DIP Agent of, or over, any DIP Collateral, the following security interests and liens are hereby granted to the DIP Agent for their own respective benefit and the benefit of the DIP Lenders, subject only to the Carve Out and the Permitted Liens (all such liens and security interests granted to the DIP Agent, for their respective benefit and for the benefit of the respective DIP Lenders, pursuant to this Interim Order and the DIP Documents, the "DIP Liens"):
- (a) <u>First Lien on Unencumbered Property</u>. Pursuant to Bankruptcy Code section 364(c)(2), and subject and subordinate in all respects to the Carve Out, valid, binding, continuing, enforceable, fully-perfected first priority senior security interests in and liens upon all DIP Collateral, to the extent such DIP Collateral is not subject to valid, perfected and non-avoidable liens as of the Petition Date or valid and non-avoidable liens in favor of third parties that were in existence immediately prior to the Petition Date that are perfected as permitted by Bankruptcy Code section 546(b) ("<u>Unencumbered Property</u>"), with the relative priorities among the DIP Obligations as set forth on **Exhibit A** attached hereto;
- (b) <u>Liens Junior to Certain Other Liens</u>. Pursuant to Bankruptcy Code section 364(c)(3), and subject and subordinate in all respects to the Carve Out, valid, binding, continuing,

enforceable, fully-perfected junior security interests in and liens on the DIP Collateral, to the extent such DIP Collateral is subject to (i) valid, perfected and non-avoidable liens as of the Petition Date or (ii) valid and unavoidable liens in favor of third parties that were in existence immediately prior to the Petition Date that are perfected as permitted by Bankruptcy Code section 546(b), in each case other than the Primed Liens, with the relative priorities among the DIP Liens as set forth on **Exhibit A** attached hereto; and

- (c) <u>Priming Liens</u>. Pursuant to Bankruptcy Code section 364(d)(1), and subject and subordinate in all respects to the Carve Out, a valid, binding, continuing, enforceable, fully-perfected priming senior security interests in and liens upon the Prepetition Collateral, which security interests and liens shall prime the Primed Liens in accordance with the priorities shown on **Exhibit A** attached hereto.
- 11. Relative Priority of DIP Liens. The DIP Liens securing the DIP Obligations are valid, automatically perfected, non-avoidable, senior in priority, and superior to any security, mortgage, collateral interest, lien, or claim to any of the DIP Collateral, except that the DIP Liens shall be subject to the Carve Out in all respects and shall otherwise be junior only to the Permitted Liens. In the event of an enforcement of remedies in respect of the DIP Facility and the application of the DIP Collateral, such DIP Collateral shall be applied as specified on **Exhibit A** attached hereto.
- 12. *DIP Collateral*. For purposes of this Interim Order, "<u>DIP Collateral</u>" shall mean all owned or hereafter acquired, whether first arising prior to, on, or following the Petition Date, assets and property of the Loan Parties (including, without limitation, inventory, accounts receivable, equipment, property, plant, equipment, owned real property, investment property, insurance proceeds, deposit accounts (other than payroll, trust, tax accounts and accounts holding

cash collateral backstopping letters of credit, cash management obligations and hedging obligations in connection with the Prepetition ABL Repayment), rights under leases and other contracts, patents, copyrights, trademarks, tradenames, and other intellectual property, and capital stock of subsidiaries) and the proceeds thereof, including, subject to entry of the Final DIP Order, the Avoidance Action Proceeds (as defined herein), but not including the Excluded Assets (as defined herein). For the avoidance of doubt, DIP Collateral shall include a pledge of 100% of equity interests of the direct foreign subsidiaries held by any Loan Party; provided that, for the avoidance of doubt, no foreign subsidiaries shall provide a guarantee or pledge collateral with respect to the DIP Facility.

the DIP Documents, the DIP Collateral shall not include (the "Excluded Assets") (a) any claims and causes of action under Bankruptcy Code sections 502(d), 544, 545, 547, 548, 549, and 550 (collectively, the "Avoidance Actions") and, prior to entry of the Final Order, the proceeds of Avoidance Actions (it being understood that subject only to and effective upon entry of the Final Order, with respect to the DIP Obligations only, the DIP Collateral shall include any proceeds or property recovered, unencumbered or otherwise from successful Avoidance Actions, whether by judgment, settlement or otherwise ("Avoidance Proceeds"); (b) leased (not owned) real property; (c) intent-to-use trademarks, assets subject to enforceable contractual restrictions or statutory prohibitions (subject to customary UCC overrides) and certain other exceptions to be agreed; (d) any Excluded Stock and Stock Equivalents (each defined in the DIP Credit Agreement); (e) any margin stock; (f) any assets with respect to which granting a security interest in such assets is prohibited by or would violate law, treaty, rule, or regulation or determination of an arbitrator or a court or other Governmental Authority or which would require obtaining the consent, approval,

license or authorization of any Governmental Authority (as defined in the DIP Credit Agreement) (unless such consent, approval, license or authorization has been received; *provided* that there shall be no obligation to obtain such consent) or create a right of termination in favor of any governmental or regulatory third party, in each case after giving effect to Sections 9-406, 9-407, 9-408 or 9-409 of the Uniform Commercial Code of any relevant jurisdiction or other Applicable Law (as defined in the DIP Credit Agreement), and (g) any payroll, tax and trust account and any Cash Collateral Account; excluding, in each case of clauses (b) through (g), the proceeds and receivables thereof (to the extent not otherwise constituting Excluded Assets); *provided* that with respect to clause (f), such property shall be Excluded Assets only to the extent and for so long as such prohibition, violation, invalidation or consent right, as applicable, is in effect and in the case of any such agreement or consent, was not created in contemplation thereof or of the creation of a security interest therein.

## 14. Carve Out.

(a) Carve Out. As used in this Interim Order, the "Carve Out" means the sum of (i) all fees required to be paid to the Clerk of the Court and to the Office of the United States Trustee under section 1930(a) of title 28 of the United States Code plus interest at the statutory rate (without regard to the notice set forth in (iii) below); (ii) all reasonable fees and expenses up to \$100,000 incurred by a trustee under section 726(b) of the Bankruptcy Code (without regard to the notice set forth in (iii) below); (iii) to the extent allowed at any time, whether by interim order, procedural order, or otherwise, all unpaid fees and expenses (the "Allowed Professional Fees") incurred by persons or firms retained by the Debtors pursuant to section 327, 328, or 363 of the Bankruptcy Code (the "Debtor Professionals") and the Creditors' Committee pursuant to section 328 or 1103 of the Bankruptcy Code (the "Committee Professionals" and, together with the Debtor

Professionals, the "<u>Professional Persons</u>") at any time before or on the first business day following delivery by the DIP Agent of a Carve Out Trigger Notice (as defined below), whether allowed by the Court prior to or after delivery of a Carve Out Trigger Notice; and (iv) Allowed Professional Fees of Professional Persons in an aggregate amount not to exceed \$10,000,000 incurred after the first business day following delivery by the DIP Agent of the Carve Out Trigger Notice, to the extent allowed at any time, whether by interim order, procedural order, or otherwise (the amounts set forth in this clause (iv) being the "<u>Post-Carve Out Trigger Notice Cap</u>"). For purposes of the foregoing, "<u>Carve Out Trigger Notice</u>" shall mean a written notice delivered by email (or other electronic means) by the DIP Agent to the Debtors, their lead restructuring counsel, the U.S. Trustee, respective counsel to the Akin Ad Hoc Group (as defined in the RSA) and the PW Ad Hoc Group (as defined in the RSA), and counsel to the Creditors' Committee, which notice may be delivered following the occurrence and during the continuation of an Event of Default (as defined in the DIP Credit Agreement) and acceleration of the DIP Obligations under the DIP Facility, stating that the Post-Carve Out Trigger Notice Cap has been invoked.

(b) Carve Out Reserves. On the day on which a Carve Out Trigger Notice is given by the DIP Agent to the Debtors with a copy to counsel to the Creditors' Committee and the respective counsel to the Akin Ad Hoc Group and the PW Ad Hoc Group (the "Termination Declaration Date"), the Carve Out Trigger Notice shall (i) be deemed a draw request and notice of borrowing by the Debtors for DIP Loans under the DIP Facility (each, as defined in the DIP Credit Agreement) (on a pro rata basis based on the then-outstanding DIP Commitments), in an amount equal to the then unpaid amounts of the Allowed Professional Fees (any such amounts actually advanced shall constitute DIP Loans) and (ii) also constitute a demand to the Debtors to utilize all cash on hand as of such date and any available cash thereafter held by any Debtor to fund a reserve

in an amount equal to the then unpaid amounts of the Allowed Professional Fees. The Debtors shall deposit and hold such amounts in a segregated account at the DIP Agent in trust to pay such then unpaid Allowed Professional Fees (the "Pre-Carve Out Trigger Notice Reserve") prior to any and all other claims. On the Termination Declaration Date, the Carve Out Trigger Notice shall also (i) be deemed a request by the Debtors for DIP Loans under the DIP Facility (on a pro rata basis based on the then-outstanding DIP Commitments), in an amount equal to the Post-Carve Out Trigger Notice Cap (any such amounts actually advanced shall constitute DIP Loans) and (ii) constitute a demand to the Debtors to utilize all cash on hand as of such date and any available cash thereafter held by any Debtor, after funding the Pre-Carve Out Trigger Notice Reserve, to fund a reserve in an amount equal to the Post-Carve Out Trigger Notice Cap. The Debtors shall deposit and hold such amounts in a segregated account at the DIP Agent in trust to pay such Allowed Professional Fees benefiting from the Post-Carve Out Trigger Notice Cap (the "Post-Carve Out Trigger Notice Reserve" and, together with the Pre-Carve Out Trigger Notice Reserve, the "Carve Out Reserves") prior to any and all other claims. On the first business day after the DIP Agent gives such notice to such DIP Lenders (as defined in the DIP Credit Agreement), notwithstanding anything in the DIP Credit Agreement to the contrary, including with respect to the existence of a Default (as defined in the DIP Credit Agreement) or Event of Default, the failure of the Debtors to satisfy any or all of the conditions precedent for DIP Loans under the DIP Facility, any termination of the DIP Commitments following an Event of Default, or the occurrence of the DIP Termination Date, each DIP Lender with outstanding DIP Commitments (on a pro rata basis based on the then-outstanding DIP Commitments) shall make available to the DIP Agent such DIP Lender's pro rata share with respect to such borrowing in accordance with the DIP Facility. All funds in the Pre-Carve Out Trigger Notice Reserve shall be used first to pay

the obligations set forth in clauses (i) through (iii) of the definition of Carve Out set forth above (the "Pre-Carve Out Amounts"), but not, for the avoidance of doubt, the Post-Carve Out Trigger Notice Cap, until paid in full, and then, to the extent the Pre-Carve Out Trigger Notice Reserve has not been reduced to zero, to pay the DIP Agent for the benefit of the DIP Lenders, unless the DIP Obligations have been indefeasibly paid in full, in cash, and all DIP Commitments have been terminated, in which case any such excess shall be paid to the Prepetition Secured Parties in accordance with their rights and priorities as set forth on Exhibit A hereto. All funds in the Post-Carve Out Trigger Notice Reserve shall be used first to pay the obligations set forth in clause (iv) of the definition of Carve Out set forth above (the "Post-Carve Out Amounts"), and then, to the extent the Post-Carve Out Trigger Notice Reserve has not been reduced to zero, to pay the DIP Agent for the benefit of the DIP Lenders, unless the DIP Obligations have been indefeasibly paid in full, in cash, and all DIP Commitments have been terminated, in which case any such excess shall be paid to the Prepetition Secured Parties in accordance with their rights and priorities as set forth on **Exhibit A** hereto. Notwithstanding anything to the contrary in the DIP Documents, or this Interim Order, if either of the Carve Out Reserves is not funded in full in the amounts set forth in this Paragraph 14, then, any excess funds in one of the Carve Out Reserves following the payment of the Pre-Carve Out Amounts and Post-Carve Out Amounts, respectively, shall be used to fund the other Carve Out Reserve, up to the applicable amount set forth in this Paragraph 14, prior to making any payments to the DIP Agent or the Prepetition Secured Parties, as applicable. Notwithstanding anything to the contrary in the DIP Documents or this Interim Order, following delivery of a Carve Out Trigger Notice, the DIP Agent and the Prepetition First Lien Agents shall not sweep or foreclose on cash (including cash received as a result of the sale or other disposition of any assets) of the Debtors until the Carve Out Reserves have been fully funded, but shall have a security interest in any residual interest in the Carve Out Reserves, with any excess paid to the DIP Agent for application in accordance with the DIP Documents. Further, notwithstanding anything to the contrary in this Interim Order, (i) disbursements by the Debtors from the Carve Out Reserves shall not constitute DIP Loans or increase or reduce the DIP Obligations, (ii) the failure of the Carve Out Reserves to satisfy in full the Allowed Professional Fees shall not affect the priority of the Carve Out, and (iii) in no way shall the Initial Budget, Budget, Carve Out, Post-Carve Out Trigger Notice Cap, Carve Out Reserves, or any of the foregoing be construed as a cap or limitation on the amount of the Allowed Professional Fees due and payable by the Debtors. For the avoidance of doubt and notwithstanding anything to the contrary in this Interim Order, the DIP Facility, or in any Existing Agreements, the Carve Out shall be senior to all liens and claims securing the DIP Facility, the Adequate Protection Liens, and the 507(b) Claim, and any and all other forms of adequate protection, liens, or claims securing the DIP Obligations or the Prepetition First Lien Obligations.

- (c) Payment of Allowed Professional Fees Prior to the Termination

  Declaration Date. Any payment or reimbursement made prior to the occurrence of the

  Termination Declaration Date in respect of any Allowed Professional Fees shall not reduce the

  Carve Out.
- (d) No Direct Obligation To Pay Allowed Professional Fees. None of the DIP Agent, DIP Lenders, or the Prepetition Secured Parties shall be responsible for the payment or reimbursement of any fees or disbursements of any Professional Person incurred in connection with the Chapter 11 Cases or any successor cases under any chapter of the Bankruptcy Code. Nothing in this Interim Order or otherwise shall be construed to obligate the DIP Agent, the DIP Lenders, or the Prepetition Secured Parties, in any way, to pay compensation to, or to

reimburse expenses of, any Professional Person or to guarantee that the Debtors have sufficient funds to pay such compensation or reimbursement.

- payment or reimbursement made on or after the occurrence of the Termination Declaration Date in respect of any Allowed Professional Fees shall permanently reduce the Carve Out on a dollar-for-dollar basis. Any funding of the Carve Out shall be added to, and made a part of, the DIP Obligations secured by the DIP Collateral and shall be otherwise entitled to the protections granted under this Interim Order, the DIP Documents, the Bankruptcy Code, and applicable law.
- obligations to the DIP Secured Parties and Prepetition Secured Parties and the liens, security interests, and superpriority claims granted herein, under the DIP Documents, and/or under the Existing Agreements, including, without limitation, the DIP Liens, the DIP Superpriority Claims, the Prepetition First Lien Adequate Protection Liens, the Primed Liens, the Adequate Protection Claims, and the Prepetition First Lien Obligations, shall be subject in all respects and subordinate to the Carve Out.
  - 15. Protection of DIP Lenders' Rights.
- (a) Until the indefeasible Payment in Full (as defined in the DIP Documents) of all DIP Obligations and the termination of all remaining DIP Commitments under the DIP Facility, the Prepetition Secured Parties shall: (i) have no right to and shall take no action to foreclose upon, or recover in connection with, the liens granted thereto pursuant to the Existing Agreements or this Interim Order, or otherwise seek to exercise or enforce any rights or remedies against DIP Collateral, including in connection with the Prepetition First Lien Adequate Protection Liens except to the extent authorized by an order of this Court; (ii) be deemed to have consented

to any transfer, disposition or sale of, or release of liens on, any DIP Collateral, to the extent such transfer, disposition, sale or release is authorized under the DIP Documents; and (iii) deliver or cause to be delivered, at the Loan Parties' cost and expense, any termination statements, releases and/or assignments in favor of the DIP Agent or the DIP Lenders or other documents necessary to effectuate and/or evidence the release, termination and/or assignment of liens on any portion of such Collateral subject to any sale or disposition.

- (b) To the extent the Prepetition First Lien Agents or any other Prepetition Secured Parties have possession of any Prepetition Collateral or DIP Collateral or have control with respect to any Prepetition Collateral or DIP Collateral, then such Prepetition Secured Party shall be deemed to maintain such possession or exercise such control as gratuitous bailee and/or gratuitous agent for perfection for the benefit of the DIP Agent and the DIP Lenders and shall comply with the instructions of the DIP Agent with respect to the exercise of such control, and the DIP Agent agrees, that such Prepetition Secured Parties shall be deemed, without incurring any liability or duty to any party, to maintain possession or control of any Prepetition Collateral or DIP Collateral in its possession or control as gratuitous bailee and/or gratuitous agent for perfection for the benefit of the Prepetition Secured Parties with respect to bank accounts.
- (c) No rights, protections or remedies of the DIP Agent or the DIP Lenders granted by the provisions of this Interim Order or the DIP Documents shall be limited, modified or impaired in any way by: (i) any actual or purported withdrawal of the consent of any party to the Debtors' authority to continue to use Cash Collateral; (ii) any actual or purported termination of the Debtors' authority to continue to use Cash Collateral; or (iii) the terms of any other order or stipulation related to the Debtors' continued use of Cash Collateral or the provision of adequate protection to any party.

- of the DIP Collateral, the DIP Lenders, the DIP Agent, the Prepetition Collateral, the Prepetition First Lien Adequate Protection Liens or the Prepetition Secured Parties shall be subject to the equitable doctrine of "marshaling" or any other similar doctrine, and all proceeds thereof shall be received and used in accordance with this Interim Order. Further, subject only to and effective upon entry of the Final Order, in no event shall the "equities of the case" exception in Bankruptcy Code section 552(b) apply to the secured claims of the Prepetition Secured Parties.
- 17. Limitation on Charging Expenses. Except to the extent of the Carve Out, no costs or expenses of administration of the Chapter 11 Cases or any future proceeding that may result therefrom, including liquidation in bankruptcy or other proceedings under the Bankruptcy Code, shall be charged against or recovered from the DIP Collateral pursuant to Bankruptcy Code section 506(c) or any similar principle of law, as such pertains to the DIP Secured Parties or the DIP Obligations without the prior written consent of the DIP Agent and no such consent shall be implied from any other action, inaction, or acquiescence by the DIP Agent or the DIP Lenders, and nothing contained in this Interim Order shall be deemed to be a consent by the DIP Agent or the DIP Lenders or the Prepetition Secured Parties to any charge, lien, assessment or claim against the DIP Collateral under Bankruptcy Code section 506(c) or otherwise. Subject only to and effective upon entry of the Final Order, except to the extent of the Carve Out, no costs or expenses of administration of the Chapter 11 Cases or any future proceeding that may result therefrom, including liquidation in bankruptcy or other proceedings under the Bankruptcy Code, shall be charged against or recovered from the Prepetition Collateral (including Cash Collateral) pursuant to Bankruptcy Code section 506(c) or any similar principle of law, without the prior written consent of the Prepetition First Lien Agents, as the case may be, and no such consent shall be

implied from any other action, inaction, or acquiescence by the Prepetition Secured Parties, and nothing contained in this Interim Order shall be deemed to be a consent by the Prepetition Secured Parties to any charge, lien, assessment or claim against the Prepetition Collateral under Bankruptcy Code section 506(c) or otherwise.

- 18. Payments Free and Clear. Subject in all respects to the Carve Out, any and all payments or proceeds remitted to the DIP Agent on behalf of the DIP Lenders, the Prepetition ABL Agent on behalf of the Prepetition ABL Secured Parties or Prepetition First Lien Agents on behalf of the Prepetition Secured Parties pursuant to the provisions of this Interim Order or the DIP Documents shall be received free and clear of any claim, charge, assessment or other liability.
- 19. *Use of Cash Collateral*. Subject to the terms and conditions of this Interim Order and the DIP Documents, and in accordance with the DIP Budget (subject to the Permitted Variances (as defined in the DIP Documents)), the Debtors are authorized to use Cash Collateral (excluding the Cash Collateral Accounts) until the expiration of the Remedies Notice Period (as defined herein) following the DIP Termination Date (as defined herein); *provided* that the Prepetition Secured Parties are granted adequate protection as hereinafter set forth. Nothing in this Interim Order shall authorize the disposition of any assets of the Debtors outside the ordinary course of business, or any Debtor's use of any Cash Collateral or other proceeds resulting therefrom, except as permitted by this Interim Order and the DIP Documents, and in accordance with the DIP Budget (subject to the Permitted Variances), as applicable.
- 20. Adequate Protection for the Prepetition Secured Parties. The Prepetition Secured Parties are entitled, pursuant to Bankruptcy Code sections 361, 362, 363(e) and 364(d)(1), to adequate protection against the diminution in value, if any, of their interests in the Prepetition Collateral, including the Cash Collateral, in an amount equal to the aggregate diminution in the

value of the Prepetition Secured Parties' respective interests in the Prepetition Collateral from and after the Petition Date, if any, resulting from the sale, lease or use by the Debtors of the Prepetition Collateral, the priming of the Primed Liens by the DIP Liens pursuant to the DIP Documents and this Interim Order and the imposition of the automatic stay pursuant to Bankruptcy Code section 362 (the "<u>First Lien Adequate Protection Claim</u>"). In consideration of the foregoing, the Prepetition Secured Parties are hereby granted the following (collectively, the "<u>Adequate Protection Obligations</u>"), in each case, subject in all respects to the Carve Out:

- (a) Prepetition First Lien 507(b) Claims. Each of the Prepetition Term Loan Secured Parties, the Prepetition Legacy Notes Parties and the Prepetition Secured Exchangeable Notes Parties are hereby granted allowed superpriority administrative expense claims as provided for in Bankruptcy Code section 507(b) in the amount of the First Lien Adequate Protection Claim (the "Prepetition First Lien 507(b) Claims"), which Prepetition First Lien 507(b) Claims shall have recourse to and be payable from all of the DIP Collateral, including, without limitation, subject to entry of the Final Order, the Avoidance Proceeds. The Prepetition First Lien 507(b) Claims shall be subject and subordinate to the Carve Out and the DIP Superpriority Claims;
- (b) <u>Prepetition First Lien Adequate Protection Liens</u>. The Prepetition First Lien Agents (for the benefit of the applicable Prepetition Secured Parties) are hereby granted valid, perfected replacement security interests in and liens (the "<u>Prepetition First Lien Adequate Protection Liens</u>") upon all of the DIP Collateral including, without limitation, subject to entry of the Final Order, the Avoidance Proceeds, in each case, junior to the Carve Out, the DIP Liens and any other liens that are senior to DIP Liens;

- (c) <u>Reporting</u>. The Prepetition Secured Parties shall be entitled to delivery of all reports and notices deliverable to the DIP Secured Parties pursuant to section 9.1 of the DIP Credit Agreement; and
- (d) Adequate Protection Fees and Expenses. As further adequate protection, the Debtors are authorized and directed to pay, in accordance with Paragraph 33 of this Interim Order, the reasonable and documented prepetition and postpetition fees and expenses (the "Adequate Protection Fees and Expenses") of the Prepetition Secured Parties as follows: (1) the Prepetition Term Loan Agent and the Escrow Agent, including the reasonable and documented fees and disbursements of Davis Polk & Wardwell LLP and Porter Hedges LLP; (2) the Prepetition Legacy Notes Trustee, including the reasonable and documented fees and disbursements of (x) one primary counsel and (y) one local counsel; (3) the Prepetition Secured Exchangeable Notes Trustee, including the reasonable and documented fees and disbursements of (x) one primary counsel and (y) one local counsel; (4) the Akin Ad Hoc Group (including, without limitation, Akin Gump Strauss Hauer & Feld LLP, Centerview Partners LP and Alvarez & Marsal North America, LLC, plus one local counsel in each applicable jurisdiction and, in the event of any actual conflict of interest, one additional counsel to the affected parties) (in each case in accordance with the terms of the engagement letters between such professionals and the Company); (5) the PW Ad Hoc Group (including, without limitation, Paul, Weiss, Rifkind, Wharton & Garrison LLP, Glenn Agre Bergman & Fuentes LLP, FTI Consulting, Inc., plus one local counsel in each applicable jurisdiction and, in the event of any actual conflict of interest, one additional counsel to the affected parties) (in each case in accordance with the terms of the engagement letters between such professionals and the Company, if applicable, or the terms and conditions of the DIP Credit Agreement); (6) Debevoise & Plimpton LLP and one local bankruptcy counsel (if necessary), as

counsel to certain holders of the Prepetition Secured Exchangeable Notes; and (7) a board search firm jointly retained by the Akin Ad Hoc Group and the PW Ad Hoc Group.

- 21. Reservation of Rights of Prepetition Secured Parties. Under the circumstances and given that the above-described adequate protection is consistent with the Bankruptcy Code, including section 506(b) thereof, the Court finds that the adequate protection provided herein is reasonable and sufficient to protect the interests of the Prepetition Secured Parties; provided that, the Prepetition Secured Parties may request further or different adequate protection; provided further, that all parties' rights (including the Debtors') to contest such adequate protection requests are fully preserved.
  - 22. *Perfection of DIP Liens and the Prepetition First Lien Adequate Protection Liens.*
- (a) This Interim Order shall be sufficient and conclusive evidence of the creation, validity, perfection, and priority of all liens granted herein, including the DIP Liens and the Adequate Protection Liens, without the necessity of filing or recording any financing statement, mortgage, notice, or other instrument or document which may otherwise be required under the law or regulation of any jurisdiction or the taking of any other action (including, for the avoidance of doubt, entering into any deposit account control agreement or mortgage with respect to any ship or vessel) to validate or perfect (in accordance with applicable non-bankruptcy law) the DIP Liens or the Adequate Protection Liens or to entitle the DIP Agent, the other DIP Secured Parties, and the Prepetition Secured Parties to the priorities granted herein.
- (b) The DIP Agent, on behalf of the DIP Lenders, and the Prepetition First Lien Agents, on behalf of, or at the direction of, the applicable Prepetition Secured Parties are hereby authorized (unless otherwise agreed between the Debtors and the DIP Lenders or the Prepetition Secured Parties in the applicable DIP Documents or Existing Agreements), but not required, to file

or record (and to execute in the name of the Debtors, as their true and lawful attorneys, with full power of substitution, to the maximum extent permitted by law) financing statements, trademark filings, copyright filings, notices of lien or similar instruments in any jurisdiction, or take possession of or control over cash or securities, or take any other action in order to validate and perfect the liens and security interests granted to them hereunder. Whether or not the DIP Agent, on behalf of the DIP Lenders, or the Prepetition First Lien Agents, on behalf of, or at the direction of, the Prepetition Secured Parties, shall, in their sole discretion, choose to file such financing statements, trademark filings, copyright filings, notices of lien or similar instruments, or take possession of or control over any cash or securities, or otherwise confirm perfection of the liens and security interests granted to them hereunder, such liens and security interests shall be deemed valid, perfected, allowed, enforceable, non-avoidable and not subject to challenge, dispute or subordination, at the time and on the date of entry of this Interim Order. Upon the request of the DIP Agent, each of the Prepetition Secured Parties and the Loan Parties, without any further consent of any party, is authorized (in the case of the Prepetition Secured Parties) (unless otherwise agreed between the Debtors and the DIP Lenders or the Prepetition Secured Parties in the applicable DIP Documents or Existing Agreements) and directed (in the case of the Loan Parties) to take, execute, deliver and file such instruments (in each case, without representation or warranty of any kind) to enable the DIP Agent to further validate, perfect, preserve and enforce the DIP Liens. All such documents will be deemed to have been recorded and filed as of the Petition Date.

(c) A certified copy of this Interim Order may, in the discretion of the DIP Agent or the Prepetition First Lien Agents, be filed with or recorded in filing or recording offices in addition to or in lieu of such financing statements, notices of lien or similar instruments, and all filing offices are hereby authorized and directed to accept such certified copy of this Interim Order

for filing and/or recording, as applicable. The automatic stay of Bankruptcy Code section 362(a) shall be modified to the extent necessary to permit the DIP Agent or the Prepetition First Lien Agents to take all actions, as applicable, referenced in this subparagraph (c) and the immediately preceding subparagraph (b).

23. DIP Budget. Attached to this Interim Order as **Exhibit B** is an initial 13-week budget approved by the DIP Agent and the Required DIP Lenders in accordance with the DIP Documents (the "Initial DIP Budget"). The Initial DIP Budget reflects, among other things, the Debtors' anticipated operating receipts, anticipated operating disbursements, anticipated nonoperating disbursements, net operating cash flow and liquidity for each calendar week covered thereby. The Initial DIP Budget may be modified, amended, extended, and updated from time to time in accordance with the DIP Documents and once approved by the Required DIP Lenders in accordance with the DIP Documents, shall modify, replace, supplement or supersede, as applicable, the Initial DIP Budget for the periods covered thereby (the Initial DIP Budget, and each subsequent approved budget shall constitute, without duplication, a "DIP Budget"). The Initial DIP Budget has been thoroughly reviewed by the Debtors, their management, and their advisors. The Debtors, their management, and their advisors believe the Initial DIP Budget and the estimate of administrative expenses due or accruing during the period covered by the Initial DIP Budget were developed using reasonable assumptions, and based on those assumptions, the Debtors believe there should be sufficient available assets to pay all administrative expenses due or accruing during the period covered by the Initial DIP Budget. The Initial DIP Budget is an integral part of this Interim Order, and the DIP Secured Parties and the Prepetition Secured Parties are relying, in part, upon the Debtors' agreement to comply with the Initial DIP Budget (subject to permitted

variances), in determining to enter into the DIP Facility and to allow the Debtors' use of Cash Collateral in accordance with the terms of this Interim Order and the DIP Documents.

- 24. *DIP Termination Date*. For the purposes of this Interim Order, the "<u>DIP Termination Date</u>" shall be earliest to occur of (a) the date that is six (6) months after the Closing Date; (b) the date that is 45 days after the Petition Date if the Final Order has not been entered prior to the expiration of such 45-day period, unless otherwise extended by the Required DIP Lenders; (c) the substantial consummation (as defined in section 1101 of the Bankruptcy Code and which for purposes hereof shall be no later than the "effective date" thereof) of a plan of reorganization filed in the Chapter 11 Cases that is confirmed pursuant to an order entered by the Bankruptcy Court; (d) the acceleration of the Loans and the termination of the DIP Commitments with respect to the DIP Facility in accordance with the DIP Credit Agreement; (e) the consummation of a sale of all or substantially all of the assets of the Borrower (or the Borrower and the Guarantors) pursuant to section 363 of the Bankruptcy Code; and (f) the termination of the Restructuring Support Agreement.
- 25. Remedies upon Event of Default. The Debtors shall promptly provide notice to the DIP Agent and the Prepetition First Lien Agents (with a copy to counsel of the Akin Ad Hoc Group and counsel to the PW Ad Hoc Group) of the occurrence of any DIP Termination Date. Upon the occurrence and during the continuation of any DIP Termination Date, the DIP Agent (at the direction of the Required DIP Lenders in accordance with the applicable DIP Credit Agreement) shall be required to provide five (5) business days' written notice (such period, the "Remedies Notice Period" and such notice, a "Termination Declaration") to the Debtors, counsel to any Creditors' Committee (if appointed), counsel to the Prepetition First Lien Agents, counsel of the Akin Ad Hoc Group, counsel to the PW Ad Hoc Group and the U.S. Trustee of the applicable DIP

Agent's intent to exercise its rights and remedies, prior to the exercise of any of the following rights: (1) declaring all DIP Obligations, including any and all accrued interest, premiums, fees and expenses constituting the DIP Obligations owing under the DIP Documents, to be immediately due and payable; (2) declaring the commitment of each DIP Lender to make DIP Loans to be terminated, whereupon such commitments and obligation shall be terminated to the extent any such commitment remains under the DIP Facility; (3) the termination of the DIP Facility and the DIP Documents as to any future liability or obligation of the DIP Secured Parties, but without affecting any of the DIP Liens or the DIP Obligations; (4) termination and/or revocation of the Debtors' right, if any, under this Interim Order and the DIP Documents to use any Cash Collateral of the DIP Secured Parties; (5) charging of interest at the default rate under the DIP Facility; (6) freezing of monies or balances in the DIP Proceeds Account;<sup>7</sup> (7) enforcing any and all rights against the DIP Collateral in possession of the applicable DIP Agent, including, without limitation, disposition of the DIP Collateral, solely for the application towards the Carve Out and the DIP Obligations in accordance with their respective priorities; and (8) taking any other actions or exercise any other rights or remedies permitted under this Interim Order, the DIP Documents, or applicable law; provided that the DIP Lenders shall not be obligated to make any DIP Loans or advances under the DIP Facility during any Remedies Notice Period. The DIP Agent may provide

<sup>&</sup>quot;DIP Proceeds Account" has the meaning ascribed to it in the Debtors' Emergency Motion for Entry of Interim and Final Orders (I) Authorizing the Debtors to (A) Continue Using the Cash Management System, (B) Maintain Existing Bank Accounts and Business Forms and Books and Records, and (C) Continue Using the Investment Account and the Investment Policy, (II) Authorizing Continued Intercompany Transactions, (III) Granting Administrative Expenses Status to Postpetition Intercompany Transactions, and (IV) Granting Related Relief (the "Cash Management Motion") filed contemporaneously herewith.

- a Termination Declaration, notwithstanding the provisions of Bankruptcy Code section 362, without any application, motion or notice to, hearing before, or order from the Court.
- Agent, the DIP Lenders, the Debtors, the Creditors' Committee, and the applicable Prepetition Secured Parties consents to a hearing on an expedited basis to consider (a) whether a DIP Termination Date has occurred and (b) any appropriate relief (including, without limitation, the Debtors' non-consensual use of Cash Collateral). During the Remedies Notice Period, notwithstanding anything to the contrary set forth in Paragraph 25, the Debtors shall continue to have the right to use Cash Collateral in accordance with the terms of this Interim Order, solely to pay necessary expenses set forth in the DIP Budget to avoid immediate and irreparable harm to the Debtors' estates. At the end of the Remedies Notice Period, unless the Court has entered an order to the contrary or otherwise fashioned an appropriate remedy, including whether DIP Termination Date has occurred, the Debtors' right to use Cash Collateral shall immediately cease, unless otherwise provided herein, and the DIP Agent and DIP Lenders shall have the rights set forth immediately below.
- 27. Certain Rights and Remedies Following DIP Termination Date. During the Remedies Notice Period, prior to the exercise or enforcement of any rights against the DIP Collateral (other than as set forth in Paragraph 26 hereof), the DIP Agent (at the direction of the Required DIP Lenders in accordance with the DIP Documents) shall be required to file an emergency motion with the Court or file the appropriate written notice in accordance with the applicable Court procedures on five (5) Business Days' notice (the "Stay Relief Hearing") to determine whether a DIP Termination Event has occurred (and the Loan Parties and the Creditors' Committee, if any, shall not object to the shortened notice with respect to such Stay Relief Hearing).

In the event the Court determines during a Stay Relief Hearing that a DIP Termination Date has occurred, the Court may fashion an appropriate remedy, which may include, inter alia, the exercise of any and all rights or remedies available to the DIP Secured Parties under this Interim Order, the DIP Documents or applicable law against the DIP Collateral; *provided* that the rights of the Debtors to contest such relief are expressly preserved; *provided*, *further*, that in the event that a party challenges the applicable DIP Agent's assertion that a DIP Termination Event has occurred or has occurred and is continuing and the Court is unavailable for a hearing during the Remedies Notice Period, the automatic stay pursuant to Bankruptcy Code section 362 shall remain in effect as to all actions other than those expressly identified in Paragraph 26 until the Court has an opportunity to rule on such challenge.

- 28. Preservation of Rights Granted Under This Interim Order. Subject in all respects to Paragraph 28(d) hereof:
- (a) Other than the Carve Out, the Permitted Liens, and other claims and liens expressly granted by this Interim Order or as permitted pursuant to the DIP Documents, no claim or lien having a priority superior to or *pari passu* with those granted by this Interim Order to the DIP Agent and the DIP Lenders, or the Prepetition Secured Parties, respectively, shall be granted or allowed while any of the DIP Obligations or the Adequate Protection Obligations remain outstanding. Except as otherwise expressly provided in the DIP Documents or this Interim Order and set forth in **Exhibit A** hereto, and subject to the Carve Out in all respects, the DIP Liens and the Prepetition First Lien Adequate Protection Liens shall not be: (i) subject or subordinate to any lien or security interest that is avoided and preserved for the benefit of the Debtors' estates under Bankruptcy Code section 551; (ii) subordinated to or made *pari passu* with any other lien or security interest, whether under Bankruptcy Code section 364(d) or otherwise; (iii) subordinated

to or made *pari passu* with any liens arising after the Petition Date including, without limitation, any liens or security interests granted in favor of any federal, state, municipal or other domestic or foreign governmental unit (including any regulatory body), commission, board or court for any liability of the Debtors; and (iv) subject or subordinate to any intercompany or affiliate liens or security interests against the Debtors.

- (b) Notwithstanding any order that may be entered dismissing any of the Chapter 11 Cases under Bankruptcy Code section 1112 or otherwise is at any time entered: (i) the DIP Superpriority Claims, the Prepetition First Lien 507(b) Claims, the DIP Liens, the Prepetition First Lien Adequate Protection Liens and the Cash Collateral Liens shall continue in full force and effect and shall maintain their priorities as provided in this Interim Order until all DIP Obligations, Adequate Protection Obligations, L/C Obligations and Hedging Obligations shall have been indefeasibly paid in full in cash (and that such DIP Superpriority Claims, Prepetition First Lien 507(b) Claims, DIP Liens, Prepetition First Lien Adequate Protection Liens and Cash Collateral Liens shall, notwithstanding such dismissal, remain binding on all parties in interest); (ii) the other rights granted by this Interim Order shall not be affected; and (iii) this Court shall retain jurisdiction, notwithstanding such dismissal, for the purposes of enforcing the claims, liens and security interests referred to in this Paragraph 28 and otherwise in this Interim Order.
- (c) If any or all of the provisions of this Interim Order are hereafter reversed, modified, vacated or stayed, such reversal, modification, vacation or stay shall not affect: (i) the validity, priority or enforceability of the Carve Out, any DIP Obligations, or Adequate Protection Obligations incurred prior to the actual receipt of written notice by the DIP Agent, the Prepetition First Lien Agents, the Akin Ad Hoc Group and the PW Ad Hoc Group, as applicable, of the effective date of such reversal, modification, vacation or stay; (ii) the validity, priority or

enforceability of the DIP Liens, the Prepetition First Lien Adequate Protection Liens or the Cash Collateral Liens; or (iii) the Prepetition ABL Repayment. Notwithstanding any such reversal, modification, vacation or stay of any use of Cash Collateral, any DIP Obligations or any Adequate Protection Obligations incurred by the Debtors to the DIP Agent, the DIP Lenders, and the Prepetition Secured Parties, as the case may be, prior to the actual receipt of written notice by the DIP Agent, the Hedge Banks, the L/C Issuers, the Cash Management Banks, the Prepetition First Lien Agents, the Akin Ad Hoc Group and the PW Ad Hoc Group, as applicable, of the effective date of such reversal, modification, vacation or stay shall be governed in all respects by the original provisions of this Interim Order, and the DIP Agent, the DIP Lenders, the Prepetition ABL Secured Parties and the Prepetition Secured Parties shall be entitled to all the rights, remedies, privileges and benefits granted in Bankruptcy Code section 364(e), this Interim Order and the DIP Documents.

(d) Except as expressly provided in this Interim Order or in the DIP Documents, the Carve Out, the DIP Liens, the DIP Superpriority Claims, the Cash Collateral Liens, the Prepetition First Lien Adequate Protection Liens, the Prepetition First Lien 507(b) Claims and the other Adequate Protection Obligations and all other rights and remedies of the DIP Agent, the DIP Lenders, the Prepetition ABL Secured Parties and the Prepetition Secured Parties granted by the provisions of this Interim Order and the DIP Documents shall survive, and shall not be modified, impaired or discharged by: (i) the entry of an order converting any of the Chapter 11 Cases to a case under chapter 7, dismissing any of the Chapter 11 Cases, terminating the joint administration of these Chapter 11 Cases or by any other act or omission; (ii) the entry of an order approving the sale of any DIP Collateral pursuant to Bankruptcy Code section 363(b) (except to the extent permitted by the DIP Documents); or (iii) the entry of an order confirming a chapter 11 plan in

any of the Chapter 11 Cases and, pursuant to Bankruptcy Code section 1141(d)(4), the Debtors have waived any discharge as to any remaining DIP Obligations or Adequate Protection Obligations. The terms and provisions of this Interim Order and the DIP Documents shall continue in these Chapter 11 Cases, in any successor cases if these Chapter 11 Cases cease to be jointly administered and in any superseding chapter 7 cases under the Bankruptcy Code, and the Carve Out, the DIP Liens, the DIP Superpriority Claims, the Prepetition First Lien Adequate Protection Liens, the Prepetition First Lien 507(b) Claims, all of the other Adequate Protection Obligations, the Cash Collateral Liens and all other rights and remedies of the DIP Agent, the DIP Lenders and the Prepetition Secured Parties granted by the provisions of this Interim Order and the DIP Documents shall continue in full force and effect until the DIP Obligations are indefeasibly paid in full in cash, as set forth herein and in the DIP Documents, and the commitments under the DIP Facility have been terminated. Any successor to the Debtors (including, without limitation, any chapter 7 or chapter 11 trustee appointed or elected for any of the Debtors or any other estate representative appointed in the Chapter 11 Cases or any successor cases) shall be bound by the terms of the Interim Order and the Final Order to the same extent as the Debtors, including with respect to the Stipulations.

29. *Releases*. Subject to the rights and limitations set forth in Paragraph 31, effective upon entry of the Interim Order, each of the Debtors, and the Debtors' estates, on their own behalf and on behalf of each of their predecessors, their successors, and assigns shall, to the maximum extent permitted by applicable law, unconditionally, irrevocably and fully forever release, remise, acquit, relinquish, irrevocably waive and discharge each of the DIP Lenders, the DIP Agent, the Prepetition ABL Secured Parties, the Prepetition Secured Parties (with the exception of the Preserved Claims), the Escrow Agent, and each of their respective former, current, or future

officers, employees, directors, agents, representatives, owners, members, partners, financial advisors, legal advisors, shareholders, managers, consultants, accountants, attorneys, affiliates, predecessors and predecessors in interest, each solely in their capacities as such, of and from any and all claims, demands, liabilities, responsibilities, disputes, remedies, causes of action, indebtedness and obligations, rights, assertions, allegations, actions, suits, controversies, proceedings, losses, damages, injuries, attorneys' fees, costs, expenses, or judgments of every type, whether known, unknown, asserted, unasserted, suspected, unsuspected, accrued, unaccrued, fixed, contingent, pending, or threatened including, without limitation, all legal and equitable theories of recovery, arising under common law, statute or regulation or by contract (under U.S. laws), of every nature and description that exist on the date hereof arising out of, relating to, or in connection with any of the (a) the Prepetition ABL Credit Documents, the Existing Agreements or the transactions contemplated under such documents, and (b) the DIP Documents or the transactions contemplated under such documents, including, without limitation, (i) any so-called "lender liability" or equitable subordination claims or defenses, (ii) any and all claims and causes of action arising under the Bankruptcy Code, and (iii) any and all claims and causes of action regarding the validity, priority, perfection, or availability of the liens of the Prepetition Secured Parties (including Avoidance Actions).

30. *Preserved Claims*. Notwithstanding anything herein to the contrary, including the Stipulations (as defined herein), the Debtors reserve all rights with respect to (i) any potential claims and causes of action with respect to: (i) the Tranche B-3 Term Loans, including but not limited to, the obligations, guarantees, and security interests granted in connection with the same

(the "Preserved Tranche B-3 Claims"), which are proposed to be settled under the RSA Plan8 including, but not limited to, in respect of the Escrow Release provided for in Paragraph 8 hereof, in the event the RSA Plan (subject to any amendments thereto) is not confirmed or substantially consummated; *provided*, that any such claim, challenge, or cause of action with respect to any Preserved Tranche B-3 Claims shall be tolled while (a) the RSA remains in effect or (b) the Debtors are prosecuting any other chapter 11 plan that provides for the unimpairment of all general unsecured claims at the Debtors (other than at Holdings) (the "Unimpairing Plan") (together, the "Tolling Conditions"), *provided further* that the Preserved Tranche B-3 Claims shall be released pursuant to Paragraph 29 hereof upon the substantial consummation of an RSA Plan or an Unimpairing Plan; and (ii) claims against any entity (including any Prepetition Secured Party) that was the beneficiary of the repurchase, redemption or other satisfaction by any Debtor entity of HoldCo Convertible Notes<sup>9</sup> prior to the Petition Date (together with the Preserved Tranche B-3 Claims, the "Preserved Claims").

- 31. *Effect of Stipulations on Third Parties.*
- (a) Except as set forth in Paragraph 30 of this Interim Order, the Debtors' acknowledgments, stipulations, and releases set forth in Paragraphs 4 and 29 of this Interim Order (collectively, the "<u>Stipulations</u>") shall be binding on the Debtors, the Debtors' estates, and their respective representatives, successors, and assigns in all circumstances. The Stipulations contained in this Interim Order, shall be binding upon all other parties in interest and all of their

The "RSA Plan" shall mean the plan filed pursuant to the Restructuring Support Agreement, dated as of February 14, 2023, by and among the Company and its direct and indirect subsidiaries and the other parties signatory thereto (the "RSA").

<sup>&</sup>lt;sup>9</sup> "<u>HoldCo Convertible Notes</u>" means the unsecured notes issued under the indenture dated as of June 11, 2018, between Holdings and the Bank of New York Mellon Trust Company, N.A., as trustee, as amended, restated amended and restated, supplemented, or otherwise modified from time to time.

respective successors and assigns, including any chapter 7 or chapter 11 trustee (a "Trustee") and any statutory or non-statutory committees appointed or formed in the Chapter 11 Cases, including the Creditors' Committee (if any) and any other person or entity acting or seeking to act on behalf of the Debtors' estates in all circumstances and for all purposes, unless (a) the Creditors' Committee, if any, or any other party in interest (including any Trustee), in each case, with requisite standing (in each case to the extent requisite standing is obtained pursuant to an order of this Court entered prior to the Challenge Deadline and subject in all respects to any agreement or applicable law that may limit or affect such entity's right to ability to commence such proceeding), has duly and timely filed an adversary proceeding or contested matter (each, a "Challenge") challenging the validity, perfection, enforceability, allowability, priority or extent of the obligations in respect of the Prepetition First Lien Obligations or the Prepetition First-Priority Liens or otherwise asserting or prosecuting any Avoidance Actions or any other claims, counterclaims or causes of action, objections, contests or defenses against the Prepetition Secured Parties in connection with any matter related to the Prepetition First Lien Obligations or the Existing Agreements (collectively, the "Claims and Defenses") by no later than the earlier of (w) the commencement of a hearing to consider confirmation of a chapter 11 plan and (x) sixty (60) days after the Petition Date (the "Challenge Period"); provided that any Trustee appointed prior to the expiration of the Challenge Period will have the longer of (y) the remaining Challenge Period or (z) forty-five (45) days from the date of Trustee's appointment to commence a Challenge; provide, further that so long as the Tolling Conditions are met, the Challenge Period shall be tolled with respect to the Creditors' Committee and, upon the failure of the Tolling Conditions, the Creditors' Committee will have sixty (60) days from the date of such failure to bring any Challenge, and (b) there is entered a final, non-appealable order in favor of the plaintiff in any such timely

filed Challenge sustaining such Challenge. Any pleadings filed in any Challenge Proceeding shall set forth with specificity the basis for such Challenge (and any Challenge not so specified prior to the Challenge Deadline shall be deemed forever waived, released and barred). The Court may fashion any appropriate remedy following a successful Challenge.

- (b) If no Challenge is timely and properly filed prior to the expiration of the Challenge Period or the Court does not rule in favor of the plaintiff in any such proceeding, then without further order of this Court (x) the obligations in respect of the Prepetition First Lien Obligations shall constitute allowed claims, not subject to any Claims and Defenses (whether characterized as a counterclaim, setoff, subordination, recharacterization, defense, avoidance, contest, attack, objection, recoupment, reclassification, reduction, disallowance, recovery, disgorgement, attachment, "claim" (as defined by Bankruptcy Code section 101(5)), impairment, subordination (whether equitable, contractual or otherwise), or other challenge of any kind pursuant to the Bankruptcy Code or applicable nonbankruptcy law), for all purposes in these Chapter 11 Cases and any subsequent chapter 7 case; (y) the Prepetition First-Priority Liens shall not be subject to any other or further Challenge, including, without limitation, any Claims and Defenses, which shall be deemed to be forever waived and barred, and all parties in interest shall be enjoined from seeking to exercise the rights of the Debtors' estates, including any successor thereto (including any estate representative or a Trustee, whether such Trustee is appointed or elected prior to or following the expiration of the Challenge Period); and (z) the Stipulations shall be of full force and effect and forever binding upon the applicable Debtor's estate and all creditors, interest holders, and other parties in interest in these Chapter 11 Cases and any successor cases.
- (c) If any Challenge is timely filed prior to the expiration of the Challenge Period, (i) the Stipulations contained in this Interim Order shall nonetheless remain binding and

preclusive (as provided in the second sentence of this paragraph) on the Creditors' Committee, if any, any other statutory or non-statutory committees appointed or formed in the Chapter 11 Cases, any other person or party in these cases, including any Trustee and any other person or entity acting or seeking to act on behalf of the Debtors' estates, except as to any such findings and admissions that were expressly and successfully challenged in such Challenge as set forth in a final, non-appealable order of a court of competent jurisdiction, and (ii) any Claims and Defenses not brought in a timely filed Challenge shall be forever barred; *provided* that, if and to the extent any Challenges to a particular Stipulation or admission are withdrawn, denied or overruled by a final non-appealable order, such Stipulation also shall be binding on the Debtors' estates and all parties in interest. Nothing in this Interim Order vests or confers on any person, including a Creditors' Committee (if any), standing or authority to pursue any cause of action belonging to the Debtors or their estates, including, without limitation, Challenges with respect to the Stipulations, and all rights to object to such standing are expressly reserved.

- 32. Expenses and Indemnification of DIP Agent and the DIP Lenders.
- (a) All reasonable and documented out-of-pocket expenses and administrative fees and "seasoning fees," to the extent applicable, for each of the DIP Agent and the DIP Lenders (as set forth below), in connection with (i) the preparation, negotiation, and execution of the DIP Documents, whether or not the DIP Facility is successfully consummated; (ii) the syndication and funding of the DIP Loans; (iii) the creation, perfection or protection of the liens under the DIP Documents (including all search, filing, and recording fees), if any; and (iv) the on-going administration of the DIP Documents (including the preparation, negotiation, and execution of any amendments, consents, waivers, assignments, restatements or supplements thereto), are to be paid by the Loan Parties in accordance with Paragraph 33, including, for the avoidance of doubt, all

reasonable documented fees, costs and expenses of (1) counsel to the DIP Agent, Ropes & Gray LLP and one local counsel, (2) counsel to the DIP Lenders, Akin Gump Strauss Hauer & Feld LLP and Paul, Weiss, Rifkind, Wharton & Garrison LLP, and their respective local counsel, (3) the financial advisors to the DIP Lenders, Centerview Partners LP, Alvarez & Marsal North America, LLC and FTI Consulting, Inc.

- (b) In addition, the Loan Parties will indemnify the DIP Lenders, the DIP Agent and their respective affiliates, and hold them harmless from and against all reasonable and documented out-of-pocket costs, expenses (with respect to legal fees and expenses, limited to the reasonable and documented out-of-pocket legal fees and expenses of one primary counsel and local counsel for the DIP Agent) and liabilities arising out of or relating to the transactions contemplated hereby and any actual or proposed use of the proceeds of any loans made under the DIP Facility in accordance with, and subject to the limitations of, the DIP Documents.
- disbursements pursuant to this Interim Order (to the extent incurred after the Petition Date) shall be made within ten (10) business days (the "Review Period") (which time period may be extended by the applicable professional) after the receipt by: (i) the Debtors, (ii) counsel for the Debtors, (iii) counsel for the Akin Ad Hoc Group, (iv) counsel to the PW Ad Hoc Group, (v) the Creditors' Committee, if any, (vi) the U.S. Trustee, (vii) counsel for the DIP Agent, and (viii) counsel to the Prepetition Term Loan Agent and Escrow Agent (collectively, the "Fee Notice Parties") of invoices therefor (the "Invoiced Fees") and without the necessity of filing formal fee applications with the Court, including such amounts arising before, on or after the Petition Date. The invoices for such Invoiced Fees shall include the number of hours billed (except for financial advisors compensated on other than an hourly basis) and the expenses incurred by the applicable

professional; provided, however, that any such invoice: (i) may be redacted to protect privileged, confidential or proprietary information and (ii) shall not be required to contain individual time detail (provided, that such invoice shall contain (except for financial advisors compensated on other than an hourly basis), summary data regarding hours worked by each timekeeper for the applicable professional and such timekeepers' hourly rates). The Fee Notice Parties may object to any portion of the Invoiced Fees (the "Disputed Invoiced Fees") within the Review Period by filing with the Court a motion or other pleading, on at least ten (10) days' prior written notice of any hearing on such motion or other pleading, setting forth the specific objections to the Disputed Invoiced Fees in reasonable narrative detail and the bases for such objections; provided that only the Disputed Invoiced Fees shall not be paid until the objection is resolved by the applicable parties in good faith or by order of the Court; provided, further, that payment of any undisputed portion of Invoiced Fees shall be promptly paid within five (5) business days following the expiration of the Review Period. If no objection is filed to the Invoiced Fees is filed with the Review period, then such Invoiced Fees shall be promptly paid, without further of, or application to, the Court or notice to any other party, and, in any case, within five (5) business days following the expiration of the Review Period and shall not be subject any further review, challenge, or disgorgement. Any and all fees, costs, and expenses paid prior to the Petition Date by any of the Debtors (i) to the DIP Agent or the other DIP Secured Parties, (ii) to the Prepetition Secured Parties, or (iii) to the Prepetition ABL Secured Parties, in each case, in connection with the Chapter 11 Cases are hereby approved in full.

34. Limitation on Use of the DIP Facility, the DIP Collateral, and the Prepetition Collateral (Including the Cash Collateral).

(a) Notwithstanding anything herein or in any other order of this Court to the contrary, none of the DIP Facility, the Prepetition Collateral, any Cash Collateral or the Carve Out (other than the Investigation Budget (as defined herein)) may be used to (a) investigate, analyze, commence, prosecute, threaten, litigate, object to, contest, or challenge in any manner or raise any defenses to, the validity, perfection, priority, extent or enforceability of any amount due under the DIP Documents, the Existing Agreements or the liens or claims granted under this Interim Order, the DIP Documents or the Existing Agreements, including the Primed Liens, the Cash Collateral Liens and the DIP Liens, or any mortgage, security interest, or lien with respect thereto, or any other rights or interests or replacement liens with respect thereto or any other rights or interests of any of the DIP Agent, the other DIP Secured Parties, the Prepetition Term Loan Agent, the Prepetition ABL Secured Parties, or the Prepetition Secured Parties, (b) assert any Claims and Defenses, including any Avoidance Actions, or any other causes of action against the DIP Agent, the DIP Lenders, the Prepetition Secured Parties or their respective agents, affiliates, subsidiaries, directors, officers, representatives, attorneys or advisors, (c) prevent, hinder, or otherwise delay the DIP Agent's or the Prepetition First Lien Agents' assertion, enforcement, or realization on the Prepetition Collateral or the DIP Collateral, in accordance with the DIP Documents, the Existing Agreements or this Interim Order, the exercise of rights by the DIP Agent or the Prepetition Secured Parties once an Event of Default has occurred and is continuing, or any other rights or interest of any of the DIP Agent, the DIP Lenders, the Prepetition ABL Secured Parties or the Prepetition Secured Parties following the occurrence of a DIP Termination Date and after the Remedies Notice Period, (d) seek to subordinate (other than to the Carve-Out or as set forth in this Interim Order) or recharacterize the DIP Obligations or any of the Prepetition First Lien Obligations, or to disallow or avoid any claim, mortgage, security interest, lien, or replacement

lien or payment thereunder, (e) seek to modify any of the rights granted to the DIP Agent, the DIP Lenders, or any of the Prepetition First Lien Agents hereunder or under the DIP Documents or the Existing Agreements, in the case of each of the foregoing clauses (a) through (d), without such party's prior written consent, (e) pay any amount on account of any claims arising prior to the Petition Date unless such payments are approved by an order of this Court or otherwise permitted under the DIP Documents, (f) file any motion seeking approval of a sale of any DIP Collateral without the consent of the Required DIP Lenders, other than a sale that indefeasibly satisfies the DIP Obligations in full in cash, (g) challenge the Escrow Release or the Escrow Payment or seek to transfer any portion of the Escrow Cash to the Debtors' estates, or (h) pay Allowed Professional Fees, disbursements, costs or expenses incurred by any person, including, without limitation, the Creditors' Committee (if any), in connection with any of the foregoing; provided, that this paragraph (including, for the avoidance of doubt, the Investigation Budget) shall not limit (or be deemed to limit) the Loan Parties' rights to challenge the claims of any creditor who received payment pursuant to such repurchase, redemption or other satisfaction by any Debtor entity of the HoldCo Convertible Notes with the proceeds of the Tranche B-3 Term Loans or otherwise prior to the Petition Date. The "Investigation Budget" means a cap of \$125,000 with respect to Allowed Professional Fees to be incurred by the Creditors' Committee under the investigation budget.

(b) For the avoidance of doubt and notwithstanding anything to the contrary herein or in the DIP Documents, the Debtors shall not be authorized to use the DIP Facility or the DIP Collateral to pay fees or expenses for the Creditors' Committee, if any, in excess of the Investigation Budget to investigate Claims and Defenses against the Prepetition Secured Parties or to initiate or prosecute proceedings or actions on account of any Claims and Defenses against the Prepetition Secured Parties, including for the avoidance of doubt, the Preserved Claims; *provided* 

further that nothing contained in this Paragraph 34 shall prohibit the Debtors from responding or objecting to or complying with discovery requests of any Creditors' Committee, in whatever form, made in connection with such investigation or the payment from the DIP Collateral of professional fees of Debtor Professionals related thereto or from contesting or challenging whether a DIP Termination Event has in fact occurred. Except to the extent expressly permitted by the terms of the DIP Documents and this Interim Order, none of the Debtors, any Creditors' Committee (if appointed), or any trustee or other estate representative appointed in the Chapter 11 Cases or in any other proceedings superseding or relating to any of the foregoing and/or upon the dismissal of any of the Chapter 11 Cases or any such successor cases (collectively, the "Successor Cases") or any other person or entity may use or seek to use Cash Collateral or, to sell, or otherwise dispose of DIP Collateral or Prepetition Collateral, in each case, without the consent of the Required Lenders (as defined in the DIP Credit Agreement) (collectively the, "Required DIP Lenders").

Reserve Account (each, a "Foreign Reserve Account Withdrawal") in accordance with the DIP Documents shall be subject to the following procedures (the "Foreign Reserve Protocol") in addition to the Intercompany Transfer Mechanic: (i) the Debtors shall submit in writing to the DIP Agent, with a copy to the advisors to the Akin Ad Hoc Group and the PW Ad Hoc Group a withdrawal request (the "Foreign Reserve Account Withdrawal Notice"), which Foreign Reserve Account Withdrawal Notice shall specify (a) the proposed foreign non-Debtor recipient of the transfer; (b) the proposed transfer amount; and (c) the proposed use of funds; and (ii) the DIP Agent shall, within three (3) business days from receipt of the Foreign Reserve Account Withdrawal Notice, advise the Debtors if the Required DIP Lenders consent to any such Foreign Reserve Account Withdrawal, which consent shall not be unreasonably withheld. All Foreign

Reserve Account Withdrawals must be made in accordance with the Intercompany Transfer Mechanic (as defined herein).

36. Intercompany Transfer Mechanic. No DIP Proceeds, DIP Collateral or any Cash Collateral may be transferred or otherwise provided by any Debtor, directly or indirectly, to or for the benefit of any non-Debtor subsidiary or subsequently by any non-Debtor subsidiary to any other non-Debtor subsidiary (either in the form of restricted payments, investments, intercompany advances, guarantee of obligations or otherwise (except for repayment of account payables for goods delivered or services rendered postpetition in accordance with the order approving the relief sought in the Cash Management Motion, including pursuant to shared services or other intercompany service agreements, in the ordinary course of business, consistent with past practice and subject to the DIP Budget)), other than the Initial Intercompany Transaction and, subject to the Foreign Reserve Protocol, any Foreign Reserve Account Withdrawal ("Non-Debtor Subsidiary Transfer"), which, in each case, shall be, (i) first, lent to Sierra Communications; (ii) second lent by Sierra Communications to Avaya International Sales Ltd (the "Initial Intercompany Borrower"), evidenced by an intercompany unsecured note or intercompany ledger entry, in each case as is reasonably satisfactory to the Required DIP Lenders and the Debtors and (iii) further lent by the Initial Intercompany Borrower, in one or more steps, to the applicable non-Debtor subsidiary; each step of such transfer set forth in clause (iii) shall be evidenced by an intercompany unsecured note or intercompany ledger entry in each case as is reasonably satisfactory to the Required DIP Lenders and the Debtors (the transfers, intercompany notes or intercompany ledger entry referenced in clauses (ii) and (iii), collectively, the "Non-Debtor Subsidiary Notes/Receivable"). All Non-Debtor Subsidiary Notes/Receivables held by Sierra Communications shall accrue interest at the same rate as the DIP Term Loan Obligations and be pledged to secure the DIP Obligations.

Notwithstanding the fact that the Non-Debtor Subsidiary Notes/Receivables shall be unsecured obligations of the applicable non-Debtor subsidiaries, for all purposes in connection with the Chapter 11 Cases, the Non-Debtor Subsidiary Notes/Receivables shall be deemed to be secured obligations of the applicable non-Debtor subsidiaries for purposes of allocation of value in the Chapter 11 Cases. The provisions in this paragraph are hereinafter referred to as the "Intercompany Transfer Mechanic."

- 37. Loss or Damage to Collateral. Nothing in this Interim Order, the DIP Documents, or any other documents related to these transactions shall in any way be construed or interpreted to impose or allow the imposition upon the DIP Agent, any DIP Lender, the Prepetition ABL Secured Parties, or Prepetition Secured Parties of any liability for any claims arising from the prepetition or postpetition activities of the Debtors in the operation of their business, or in connection with their restructuring efforts. So long as the DIP Agent, the DIP Lenders and the Prepetition Secured Parties comply with their obligations under the DIP Documents and this Interim Order and their obligations, if any, under applicable law (including the Bankruptcy Code), (a) the DIP Agent, the DIP Lenders and the Prepetition Secured Parties shall not, in any way or manner, be liable or responsible for (i) the safekeeping of the DIP Collateral, (ii) any loss or damage thereto occurring or arising in any manner or fashion from any cause, (iii) any diminution in the value thereof or (iv) any act or default of any carrier, servicer, bailee, custodian, forwarding agency or other person and (b) all risk of loss, damage or destruction of the DIP Collateral shall be borne by the Debtors.
- 38. Reservation of Rights Under the First Lien Pari Intercreditor Agreement. Subject to the Prepetition ABL Repayment, pursuant to Bankruptcy Code section 510, the First Lien Pari Intercreditor Agreement and any other applicable intercreditor or subordination provisions

contained in any of the Existing Agreements (i) shall remain in full force and effect, (ii) shall continue to govern the relative priorities, rights, and remedies of the Prepetition Secured Parties (including the relative priorities, rights and remedies of such parties with respect to the replacement liens and administrative expense claims and superpriority administrative expense claims granted, or amounts payable, by the Debtors under this Interim Order or otherwise and the modifications of the automatic stay), and (iii) shall not be deemed to be amended, altered, or modified by the terms of this Interim Order or the DIP Documents, unless expressly set forth herein or therein.

- 39. *Interim Order Governs*. In the event of any inconsistency between the provisions of this Interim Order and the DIP Documents, the provisions of this Interim Order shall govern.
- 40. Binding Effect; Successors and Assigns. The DIP Documents and the provisions of this Interim Order, including all findings herein, shall be binding upon all parties in interest in these Chapter 11 Cases, including, without limitation, the DIP Agent, the DIP Lenders, the Prepetition Secured Parties, the Creditors' Committee (if any), any non-statutory committees appointed or formed in these Chapter 11 Cases, the Debtors and their respective successors and assigns (including any chapter 7 or chapter 11 trustee hereinafter appointed or elected for the estate of any of the Debtors, an examiner appointed pursuant to Bankruptcy Code section 1104, or any other fiduciary appointed as a legal representative of any of the Debtors or with respect to the property of the estate of any of the Debtors) and shall inure to the benefit of the DIP Agent, the DIP Lenders, the Prepetition ABL Secured Parties, the Prepetition Secured Parties and the Debtors and their respective successors and assigns; provided, however, that the DIP Agent, the DIP Lenders and the Prepetition Secured Parties shall have no obligation to permit the use of the Prepetition Collateral (including Cash Collateral) or to extend any financing to any chapter 7 trustee, chapter 11 trustee or similar responsible person appointed for the estates of the Debtors).

41. *Limitation of Liability*. In determining to make any loan or other extension of credit under the DIP Documents, to permit the use of Cash Collateral or in exercising any rights or remedies as and when permitted pursuant to this Interim Order or the DIP Documents, none of the DIP Agent, the DIP Lenders and the Prepetition Secured Parties shall (i) be deemed to be in "control" of the operations or participating in the management of the Debtors; (ii) owe any fiduciary duty to the Debtors, their respective creditors, shareholders or estates; and (iii) be deemed to be acting as a "Responsible Person" or "Owner" or "Operator" with respect to the operation or management of the Debtors (as such terms or similar terms are used in the United States Comprehensive Environmental Response, Compensation and Liability Act, 29 U.S.C. §§ 9601, et seq., as amended, or any similar federal or state statute).

#### 42. Master Proof of Claim.

(a) Each of the Prepetition Term Loan Agent, Prepetition Legacy Notes Trustee and Prepetition Secured Exchangeable Notes Trustee will not be required to file proofs of claim in any of the Chapter 11 Cases or any successor cases for any claim allowed herein, including claims arising under the Existing Agreements. The Stipulations shall be deemed to constitute a timely filed proof of claim for each of the Prepetition Term Loan Agent, Prepetition Legacy Notes Trustee and Prepetition Secured Exchangeable Notes Trustee, and such parties shall be treated under section 502(a) of the Bankruptcy Code as if they had timely filed a proof of claim. Any order entered by the Court in relation to the establishment of a bar date for any claim (including without limitation administrative claims) in any of the Chapter 11 Cases or any successor cases shall not apply to (i) the DIP Agent or the DIP Secured Parties, or (ii) the Prepetition Secured Parties with respect to the Prepetition First Lien Obligations or any claims arising under the Existing Agreements.

(b) In order to facilitate the processing of claims, to ease the burden upon the Court and to reduce an unnecessary expense to the Debtors' estates, each of the Prepetition Term Loan Agent, Prepetition Legacy Notes Trustee and Prepetition Secured Exchangeable Notes Trustee is authorized, but not directed, in their sole discretion, to file in the Debtors' lead chapter 11 case In re Avaya Inc., et al., Case No. 23-90088 (DRJ), a single, master proof of claim on behalf of the Prepetition Term Loan Secured Parties, Prepetition Legacy Notes Parties and Prepetition Secured Exchangeable Notes Parties, as applicable, on account of any and all of their respective claims arising under the applicable Existing Agreements and hereunder (each, a "Master Proof of Claim") against each applicable Debtor. Upon the filing of a Master Proof of Claim against each of the Debtors, the (i) Prepetition Secured Term Loan Agent and Prepetition Term Loan Secured Parties, (ii) Prepetition Legacy Notes Trustee and Prepetition Legacy Notes Parties, and (iii) Prepetition Secured Exchangeable Notes Trustee and Prepetition Secured Exchangeable Notes Secured Parties, as applicable, and each of their respective successors and assigns, shall be deemed to have filed a proof of claim in the amount set forth opposite its name therein in respect of its claims against each of the Debtors of any type or nature whatsoever with respect to the applicable Existing Agreements, and the claim of each Prepetition Secured Party (and each of its respective successors and assigns), named in a Master Proof of Claim shall be treated as if such entity had filed a separate proof of claim in each of these Chapter 11 Cases. The Master Proofs of Claim shall not be required to identify whether any Prepetition Secured Party acquired its claim from another party and the identity of any such party or to be amend to reflect a change in the holders of the claims set forth therein or a reallocation among such holders of the claims asserted therein resulting from the transfer of all or any portion of such claims. The provisions of this Paragraph 42 and each Master Proof of Claim are intended solely for the purpose of administrative

convenience and shall not affect the right of each Prepetition Secured Party (or its successors in interest) to vote separately on any plan proposed in these Chapter 11 Cases. The Master Proofs of Claim shall not be required to attach any instruments, agreements or other documents evidencing the obligations owing by each of the Debtors to the applicable Prepetition Secured Parties, which instruments, agreements or other documents will be provided upon written request to counsel to the Prepetition First Lien Agents, as applicable.

- 43. *Information and Other Covenants*. The Loan Parties shall comply in all material respects with the reporting requirements set forth in the DIP Documents. The Debtors shall maintain their cash management arrangements in a manner consistent with that described in the Cash Management Motion and any successor or final orders with respect thereto.
- 44. *Insurance*. To the extent that any of the Prepetition ABL Agent, the Prepetition Term Loan Agent, the Prepetition Legacy Notes Trustee and the Prepetition Secured Exchangeable Notes Trustee is listed as loss payee under the Borrower's or Guarantors' insurance policies, the DIP Agent are also deemed to be the loss payee under such insurance policies and shall act in that capacity and distribute any proceeds recovered or received in respect of any such insurance policies, first, to the payment in full of the DIP Obligations subject to the priority set forth on **Exhibit A** hereto (other than contingent indemnification obligations as to which no claim has been asserted), and second, to the payment of the Prepetition First Lien Obligations.
- 45. Effectiveness. This Interim Order shall constitute findings of fact and conclusions of law and shall take effect and be fully enforceable <u>nunc pro tunc</u> to the Petition Date immediately upon entry hereof. Notwithstanding Bankruptcy Rules 4001(a)(3), 6004(h), 6006(d), 7062, or 9014 or any Local Bankruptcy Rule, or Rule 62(a) of the Federal Rules of Civil Procedure, this

Interim Order shall be immediately effective and enforceable upon its entry and there shall be no stay of execution or effectiveness of this Interim Order.

- 46. *Headings*. Section headings used herein are for convenience only and are not to affect the construction of or to be taken into consideration in interpreting this Interim Order.
- 47. Payments Held in Trust. Except as expressly permitted in this Interim Order or the DIP Documents and subject to the Carve Out in all respects, in the event that any person or entity receives any payment on account of a security interest in DIP Collateral, receives any DIP Collateral or any proceeds of DIP Collateral or receives any other payment with respect thereto from any other source prior to indefeasible payment in full in cash of all DIP Obligations under the DIP Documents, and termination of the commitments in accordance with the DIP Documents, such person or entity shall be deemed to have received, and shall hold, any such payment or proceeds of DIP Collateral in trust for the benefit of the DIP Agent and the DIP Lenders based on the priorities set forth on Exhibit A hereto and shall immediately turn over such proceeds to the DIP Agent, or as otherwise instructed by this Court, for application in accordance with the DIP Documents and this Interim Order based on the priorities set forth on Exhibit A hereto; provided that, prior to the Prepetition ABL Repayment, the foregoing shall not apply to the receipt of ABL Priority Collateral (or any proceeds thereof to the extent not constituting Term Loan Priority Collateral) by any Prepetition ABL Secured Party on account of its security interest in ABL Priority Collateral.
- 48. *Credit Bidding*. Upon entry of this Interim Order and subject to the terms of the DIP Documents and the lien priorities set forth herein: (i) the DIP Agent and the DIP Lenders shall have the right to credit bid as part of any asset sale process and shall have the right to credit bid up to the full amount of the DIP Obligations during any sale of the Loan Parties' assets (in whole or

in part), including without limitation, sales occurring pursuant to Bankruptcy Code section 363 or included as part of any restructuring plan subject to confirmation under Bankruptcy Code section 1129(b)(2)(A)(ii)-(iii); and (ii) the Prepetition Secured Parties shall have the right to credit bid as part of any asset sale process and shall have the right to credit bid the full amount of their respective claims, including, for the avoidance of doubt, Adequate Protection Claims, if any, during any sale of the Debtors' assets (in whole or in part) with respect to any asset subject to a duly perfected lien in favor of the Prepetition Secured Parties as of the Petition Date, including without limitation, sales occurring pursuant to Bankruptcy Code section 363 or included as part of any restructuring plan subject to confirmation under Bankruptcy Code section 1129(b)(2)(A)(ii), and shall each automatically be deemed a "qualified bidder" with respect to any disposition of DIP Collateral or Prepetition Collateral (as applicable) under or pursuant to (a) Bankruptcy Code section 363, (b) a plan of reorganization or plan of liquidation under Bankruptcy Code section 1129, or (c) a sale or disposition by a chapter 7 trustee for any of the Debtors under Bankruptcy Code section 725. The DIP Agent (at the direction of the Required DIP Lenders) and the Prepetition First Lien Agents (at the direction of the requisite lenders or holders under the applicable Existing Agreements), shall each have the absolute right to assign, transfer, sell, or otherwise dispose of its rights to credit bid, except as may be set forth in the DIP Documents.

49. *No Waiver by Failure to Seek Relief*. The failure of the DIP Agent, DIP Lenders, or Prepetition Secured Parties to seek relief or otherwise exercise their rights and remedies under this Interim Order, the DIP Documents, the Existing Agreements, or applicable law, as the case may be, shall not constitute a waiver of any of the rights hereunder, thereunder, or otherwise of the DIP Agent, DIP Lenders or Prepetition Secured Parties.

- 50. *Necessary Action*. The Debtors are authorized to take all such actions as are necessary or appropriate to implement the terms of this Interim Order.
- 51. Retention of Jurisdiction. The Court shall retain jurisdiction to enforce the provisions of this Interim Order, and this retention of jurisdiction shall survive the confirmation and consummation of any chapter 11 plan for any one or more of the Debtors notwithstanding the terms or provisions of any such chapter 11 plan or any order confirming any such chapter 11 plan.
- 52. Final Hearing. The Final Hearing is scheduled for \_\_\_\_\_\_\_, 2023 at \_\_\_\_\_\_.m. before this Court.
- 53. Objections. Any party in interest objecting to the relief sought at the Final Hearing shall file and serve written objections, which objections shall be served upon (a) co-counsel to the Debtors, (i) Kirkland & Ellis LLP, 601 Lexington Avenue, New York, New York 10022, Attn: Joshua A. Sussberg, P.C., (joshua.sussberg@kirkland.com), Aparna Yenamandra, P.C., (aparna.yenamandra@ kirkland.com), Rachael M. Bentley (rachael.bentley@kirkland.com), and Andrew Townsell (andrew.townsell@kirkland.com); Kirkland & Ellis LLP, 300 North LaSalle, Chicago, Illinois 60654 Attn: Patrick J. Nash, Jr., P.C. (patrick.nash@kirkland.com), and (ii) Jackson Walker LLP, 1401 McKinney Street, Suite 1900, Houston, Texas 77010, Attn: Matthew D. Cavenaugh (mcavenaugh@jw.com), Rebecca Blake Chaikin (rchaikin@jw.com), and Genevieve M. Graham (ggraham@jw.com); (b) counsel to the DIP Agent, Ropes & Gray LLP, 1211 Avenue of the Americas, New York, New York 10036-8704, Attn: Mark Somerstein and Patricia Chen (mark.somerstein@ropesgray.com and patriica.chen@ropesgray.com); (c) counsel to the Akin Ad Hoc Group, Akin Gump Strauss Hauer & Feld LLP, One Bryant Park, Bank of America Tower, New York, New York 10036, Attn: Ira S. Dizengoff (idizengoff@akingump.com), Philip C. Dublin (pdublin@akingump.com), and

Naomi Moss (nmoss@akingump.com); (d) counsel to the PW Ad Hoc Group, Paul, Weiss, Rifkind, Wharton & Garrison LLP, Attn: Andrew N. Rosenberg (arosenberg@paulweiss.com), Brian S. Hermann (bhermann@paulweiss.com), Brian Bolin (bbolin@paulweiss.com), Joseph M. Graham (jgraham@paulweiss.com), and Xu Pang (xpang@paulweiss.com); (e) counsel to the Prepetition Term Loan Agent and Escrow Agent, Davis Polk & Wardwell LLP, 450 Lexington Avenue, New York, New York 10017, Attn: Brian M. Resnick (brian.resnick@davispolk.com) and Michael Pera (michael.pera@davispolk.com) and (f) any other party that has filed a request for notices with this Court, in each case to allow actual receipt by the foregoing no later than

54. *Notice of Entry of Interim Order*. The Debtors shall promptly serve copies of this Interim Order to the parties having been given notice of the Interim Hearing, to any party that has filed a request for notices with this Court and to the Creditors' Committee after the same has been appointed, or such Creditors' Committee's counsel, if the same shall have been appointed.

Dated:		, 2023
	Houston, Texas	

\_\_\_\_\_, 2023 at [\_\_\_\_] p.m. (EST).

UNITED STATES BANKRUPTCY JUDGE

# Exhibit A

### **Lien Priorities on DIP Collateral**

1 <sup>st</sup>	Carve Out
2 <sup>nd</sup>	Permitted Liens
3 <sup>rd</sup>	• DIP Liens
4 <sup>th</sup>	Prepetition First Lien Adequate Protection Liens
5 <sup>th</sup>	Prepetition First-Priority Liens

# Exhibit B

# **Initial DIP Budget**

**DIP Budget** 

(\$ in thousands)

		Filing												
		Week 1 2/11	Week 2 2/18	Week 3 2/25	Week 4 3/4	Week 5 3/11	Week 6 3/18	Week 7 3/25	Week 8 4/1	Week 9 4/8	Week 10 4/15	Week 11 4/22	Week 12 4/29	Week 13 5/6
		2/17	2/24	3/3	3/10	3/17	3/24	3/31	4/7	4/14	4/21	4/28	5/5	5/12
		Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast
Cash	Total Receipts	18,000	10,067	12,173	18,660	24,460	28,000	40,840	18,719	17,367	23,709	18,924	15,564	13,592
S	Payroll	5,088	563	26,775	513	2,041	477	27,422	824	2,257	986	27,400	435	1,737
Operating Plow	Operating Expenses	33,150	12,175	9,854	30,272	5,668	6,142	12,100	23,947	6,236	8,503	8,866	24,296	4,646
	Net I/C Activity	1,000	800		1,000	3,000		1,800		3,000	-		1,800	-
	Total Operating Disbursements	39,238	13,538	36,629	31,786	10,709	6,619	41,322	24,771	11,492	9,489	36,266	26,531	6,383
	OPERATING CASH FLOW	(21,238)	(3,471)	(24,456)	(13,126)	13,751	21,381	(482)	(6,052)	5,875	14,220	(17,343)	(10,967)	7,209
	Professional Fees	-	-	-	-	10,220	-		-	-	11,177	-		
an the	Cash Collateralization of L/C	40,000	-	-	(40,000)	-	-	-	-	-	-			
E E	Cash Collateralization of Customer / Tax / Customs Sureties	-	-	-	33,000	-	-	-	-	-	-	-	-	-
urs	Initial Intercompany Transfer	50,000	-	-	-	-	-	-	-	-	-	-	-	-
Disbu	Foreign Reserve Account	40,000	-	-	-	-	-	-	-	-	-	-	-	-
	Cure Costs	-	-	-	-	-	-	-	-	-	-	-	-	-
Other	Utility Deposit	500	-	-	-	-	-	-	-	-	_		-	-
· ·	Total	130,500	-	-	(7,000)	10,220	-	-	-	-	11,177	-	-	-
Ħ	ABL Paydown	56,000	-	-	-	-	-	-	-	-	-	-	-	-
ing me	Net Debt Activity	(400,000)	-	-	-	-	-	(100,000)	-	-	-	-	-	-
Financing Disbursement	Financing Fees	24,947	-	-	-	-	-	-	-	-	-	-	-	-
Fin	Interest	135		<u> </u>	<u> </u>	5,121					6,163			
۵	Total	(318,918)	-	-	-	5,121	-	(100,000)	-	-	6,163	-	-	-
	NET CASH FLOW	167,180	(3,471)	(24,456)	(6,126)	(1,590)	21,381	99,518	(6,052)	5,875	(3,120)	(17,343)	(10,967)	7,209
	Beginning Cash	22,017	189,197	185,727	161,270	155,145	153,555	174,937	274,455	268,403	274,278	271,157	253,815	242,847
	Change in Cash	167,180	(3,471)	(24,456)	(6,126)	(1,590)	21,381	99,518	(6,052)	5,875	(3,120)	(17,343)	(10,967)	7,209
	ENDING DOMESTIC CASH	189,197	185,727	161,270	155,145	153,555	174,937	274,455	268,403	274,278	271,157	253,815	242,847	250,057
	FOREIGN RESERVE ACCOUNT	40,000	40,000	40,000	40,000	40,000	40,000	40,000	40,000	40,000	40,000	40,000	40,000	40,000
.₹	L/C CASH COLLATERALIZATION ACCOUNT	40,000	40,000	40,000			_		· -		· _	· _		· _
Liquidity	TOTAL DOMESTIC CASH	269,197	265,727	241,270	195,145	193,555	214,937	314,455	308,403	314,278	311,157	293,815	282,847	290,057
글		203,137	203,727	241,270	155,145	155,555	214,337	314,433	300,403	314,270	311,137	255,015	202,047	250,037
	Memo:	260 107	265 727	241 270	105 145	102 555	214 027	214 455	200 402	214 270	211 157	202.015	202.047	200.057
	Domestic Cash	269,197	265,727	241,270	195,145	193,555	214,937	314,455	308,403	314,278	311,157	293,815	282,847	290,057
	International Cash TOTAL CONSOLIDATED CASH	135,965	129,493	125,475	125,425	129,433	141,755	141,098	140,734	146,003	146,833	138,076	134,728	132,710
		405,162	395,220	366,746	320,570	322,988	356,692	455,552	449,137	460,280	457,990	431,891	417,575	422,767
	Additional DIP TL Funds Available	100,000	100,000	100,000	100,000	100,000	100,000	•	-	-	-	-	-	-